THE STATE OF NEVADA

2	IN THE SUPREME COURT OF T	HE STATE OF NEVADA	
3	FRONT SIGHT MANAGEMENT LLC, a		
4		No.: <u>Electron</u> ically File Sep 11 2020 04:3	9 p.m.
5		Dist. Ct. Case No: 知道為both 1984 Brow Clerk of Supreme	n Court
6	VS.	'	
7	THE EIGHTH JUDICIAL DISTRICT		
8	COURT OF THE STATE OF NEVADA, IN AND FOR THE COUNTY OF CLARK;		
9	and THE HONORABLE TIMOTHY C.		
10	WILLIAMS, DISTRICT COURT JUDGE,		
11	Respondents,		
12	and		
13	anu		
14	LAS VEGAS DEVELOPMENT FUND		
15	LLC, a Nevada Limited Liability Company; EB5 IMPACT CAPITAL REGIONAL		
16	CENTER LLC, a Nevada Limited Liability		
17	Company; EB5 IMPACT ADVISORS		
18	LLC, a Nevada Limited Liability Company; ROBERT W. DZIUBLA, individually and		
19	as President and CEO of LAS VEGAS		
20	DEVELOPMENT FUND LLC and EB5 IMPACT ADVISORS LLC; JON		
21	FLEMING, individually and as an agent of		
22	LAS VEGAS DEVELOPMENT FUND		
23	LLC and EB5 IMPACT ADVISORS LLC; LINDA STANWOOD, individually and as		
24	Senior Vice President of LAS VEGAS		
25	DEVELOPMENT FUND LLC and EB5 IMPACT ADVISORS LLC,		
26	INITACT ADVISORS LLC,		
27	Real Parties in Interest.		

27

1	
2	PETITION FOR WRIT OF MANDAMUS, OR ALTERNATIVELY,
3	PROHIBITION
4	
5	
6	PETITIONER'S APPENDIX
7	VOLUME XVII
8 9 10 11 12 13 14 15 16 17 18	John P. Aldrich, Esq. Nevada Bar No. 6877 Jamie S. Hendrickson, Esq. Nevada Bar No. 12770 ALDRICH LAW FIRM, LTD. 7866 West Sahara Avenue Las Vegas, Nevada 89117 702-853-5490 jaldrich@johnaldrichlawfirm.com jamie@johnaldrichlawfirm.com
20	
21	
22	
23	
24	
25	
26	
27	
28	

CHRONOLOGICAL INDEX

<u>VOLUME I</u>	PAGES
Complaint (09/14/2018)	0001-0028
Amended Complaint (10/04/2018)	0029-0057
Affidavit of Service on Robert W. Dziubla (10/17/2018)	0058
Affidavit of Service on Linda Stanwood (10/17/2018)	0059
Affidavit of Service on EB5 Impact Advisors LLC (10/17/2018)	0060
Affidavit of Service on EB5 Impact Capital Regional Center LLC (10/18/2018)	0061
Affidavit of Service on Las Vegas Development Fund LLC (10/18/2018)	0062
Affidavit of Service on Chicago Title Company (10/22/2018)	0063
Notice of Entry of Order Admitting to Practice (11/15/2018)	0064-0068
Notice of Entry of Order on Plaintiff's Petition for Appointment of Receiver and for an Accounting (11/27/2018)	0069-0074
Notice of Entry of Order Granting Plaintiff's Motion for Protective Order (11/27/2018)	0075-0079
Notice of Entry of Protective Order (11/27/2018)	0080-0098
Notice of Entry of Order Granting Temporary Restraining Order and Expunging Notice of Default (11/27/2018)	0099-0104
Order Setting Settlement Conference (12/06/2018)	0105-0106
Second Amended Complaint (01/04/2019)	0107-0250

1 2	VOLUME II	<u>PAGES</u>
3	Second Amended Complaint (01/04/2019) (cont'd)	0251-0322
5	Notice of Entry of Order on Plaintiff's Motion for Preliminary Injunction (01/17/2019)	0323-0327
6 7 8 9	Notice of Entry of Order on Plaintiff's Renewed Motion for an Accounting Related to Defendants Las Vegas Development Fund LLC and Robert Dziubla and for Release of Funds (01/17/2019)	0328-0332
10 11	Notice of Entry of Order on Defendants' Motion to Dismiss Plaintiff's First Amended Complaint (01/17/2019)	0333-0337
12 13	Notice of Entry of Order on Plaintiff's Motion to Disqualify C. Keith Greer as Attorney of Record for Defendants (01/25/2019)	0338-0343
14 15	Notice of Entry of Disclaimer of Interest of Chicago Title Company and Stipulation and Order for Dismissal (02/05/2019)	0344-0350
16 17 18	Defendant Las Vegas Development Fund LLC's Motion for Appointment of Receiver and Request for Order Shortening Time (02/06/2019)	0351-0378
19 20 21	Declaration of Robert Dziubla in Support of Defendant Las Vegas Development Fund LLC's Motion for Appointment of Receiver [redacted in district court filing] (02/06/2019)	0379-0500
22	VOLUME III	PAGES
23 24 25	Declaration of Robert Dziubla in Support of Defendant Las Vegas Development Fund LLC's Motion for Appointment of Receiver [redacted in district court filing] (02/06/2019) (cont'd)	0501-0558
26 27	Declaration of C. Keith Greer in Support of Defendant's Motion for Receivership (02/06/2019)	0559-0601

1 2 3 4	Motion to Seal and/or Redact Pleadings and Exhibits to Protect Confidential Information, Motion to Amend Paragraph 2.3 of Protective Order, Motion for Order Shortening Time and Order Shortening Time (02/15/2019)	0602-0628
5	Notice of Entry of Order Shortening Time (02/15/2019)	0629-0658
6 7 8	Opposition Memorandum of Defendant Las Vegas Development Fund, LLC to Plaintiff's Motion to Seal and/or Redact Pleadings and Exhibits (02/19/2019)	0659-0669
9 10	Opposition to Defendant Las Vegas Development Fund LLC's Motion for Appointment of Receiver (02/22/2019)	0670-0730
11 12	Errata to Opposition to Defendant Las Vegas Development Fund LLC's Motion for Appointment of Receiver (02/22/2019)	0731-0740
13 14	Defendant Las Vegas Development Fund LLC's Reply to Plaintiff's Opposition to Defendant's Motion for Appointment of	0741-0750
15	Receiver (02/26/2019)	
15 16	1	<u>PAGES</u>
15	Receiver (02/26/2019)	PAGES 0751-0755
15 16 17 18	Receiver (02/26/2019) VOLUME IV Defendant Las Vegas Development Fund LLC's Reply to Plaintiff's Opposition to Defendant's Motion for Appointment of	
15 16 17 18 19 20 21	Receiver (02/26/2019) VOLUME IV Defendant Las Vegas Development Fund LLC's Reply to Plaintiff's Opposition to Defendant's Motion for Appointment of Receiver (02/26/2019) (cont'd) Supplemental Declaration of Robert W. Dziubla in Support of Defendant LVD Fund's Reply to Plaintiff's Opposition to	0751-0755

1 2 3	Defendant Las Vegas Development Fund, LLC's Opposition to Plaintiff's Second Motion for Temporary Restraining Order and Preliminary Injunction (03/19/2019)	0837-0860
4 5 6 7	Supplemental Declaration of Defendant Robert Dziubla in Support of Defendant Las Vegas Development Fund, LLC's Opposition to Plaintiff's Second Motion for Temporary Restraining Order and Preliminary Injunction (03/19/2019)	0861-0875
8	Notice of Entry of Order (03/19/2019)	0876-0881
9 10 11	Errata to Supplemental Declaration of Robert Dziubla in Support of Defendants' Opposition to Plaintiff's Second Motion for Temporary Restraining Order and Preliminary Injunction (03/20/2019)	0882-0892
12 13	Notice of Entry of Order (04/10/2019)	0893-0897
14	Notice of Entry of Order (04/10/2019)	0898-0903
15 16	Notice of Entry of Order (04/10/2019)	0904-0909
17	Notice of Entry of Order (04/10/2019)	0910-0916
18 19	Defendants' Answer to Plaintiff's Second Amended Complaint and Counterclaim (04/23/2019)	0917-1000
20 21	VOLUME V	PAGES
22 23	Defendants' Answer to Plaintiff's Second Amended Complaint and Counterclaim (04/23/2019) (cont'd)	1001-1083
24	Notice of Entry of Order (05/16/2019)	1084-1089
252627	Reporter's Transcript of Motion (Preliminary Injunction Hearing) (06/03/2019)	1090-1250

1 2	<u>VOLUME VI</u>	<u>PAGES</u>
3 4	Reporter's Transcript of Motion (Preliminary Injunction Hearing) (06/03/2019) (cont'd)	1251-1313
5	Order Setting Settlement Conference (06/04/2019)	1314-1315
6 7 8	Acceptance of Service of Counterclaim on Counterdefendants Front Sight Management, LLC, Ignatius Piazza, Jennifer Piazza, VNV Dynasty Trust I and VNV Dynasty Trust II (06/14/2019)	1316-1317
9	Notice of Entry of Order (06/25/2019)	1318-1324
10	Notice of Entry of Stipulation and Order Regarding Defendants' Judicial Foreclosure Cause of Action (06/25/2019)	1325-1330
12 13 14	Reporter's Transcript of Preliminary Injunction Hearing (07/22/2019)	1331-1500
15	VOLUME VII	PAGES
16 17	Reporter's Transcript of Preliminary Injunction Hearing (07/22/2019) (cont'd)	1501-1513
18	Reporter's Transcript of Preliminary Injunction (07/23/2019)	1514-1565
19 20	Business Court Order (07/23/2019)	1566-1572
21 22 23	Order Re Rule 16 Conference, Setting Civil Jury Trial, Pre- Trial/Calendar Call and Deadlines for Motions; Discovery Scheduling Order (08/20/2019)	1573-1577
24 25	Notice of Entry of Order Granting in Part and Denying in Part Counterdefendants' Motions to Dismiss Counterclaim (09/13/2019)	1578-1584
26 27 28	Notice of Entry of Order Denying Plaintiff's Motion for Temporary Restraining Order and Preliminary Injunction related to Investor Funds and Interest Payments (09/13/2019)	1585-1591

1 2 3	Notice of Entry of Order Staying All Subpoenas For Documents and Depositions which were Served on Non-Parties by Plaintiff (09/13/2019)	1592-1599
5	Plaintiff's Motion for Sanctions (09/17/2019)	1600-1643
6	Reporter's Transcript of Hearing (Preliminary Injunction Hearing) (09/20/2019)	1644-1750
8	VOLUME VIII	PAGES
9 10	Reporter's Transcript of Hearing (Preliminary Injunction Hearing) (09/20/2019) (cont'd)	1751-1930
11	Order Scheduling Hearing (09/27/2019)	1931-1932
12 13 14	Counterdefendants VNV Dynasty Trust I and VNV Dynasty Trust II's Answer to Counterclaim (09/30/2019)	1933-1957
15	Counterdefendant Dr. Ignatius Piazza's Answer to Counterclaim (09/30/2019)	1958-1981
17 18	Counterdefendant Front Sight Management LLC's Answer to Counterclaim (09/30/2019)	1982-2000
19	VOLUME IX	PAGES
20 21	Counterdefendant Front Sight Management LLC's Answer to Counterclaim (09/30/2019) (cont'd)	2001-2005
22 23	Counterdefendant Jennifer Piazza's Answer to Counterclaim (09/30/2019)	2006-2029
24 25	Defendant EB5 Impact Advisors LLC's Opposition to Plaintiff's Motion for Sanctions (09/30/2019)	2030-2040
26 27 28	Declaration of Robert Dziubla in Opposition to Plaintiff's Motion for Sanctions (09/30/2019)	2041-2044

1 2 3	Reporter's Transcript of Motions (Defendants' Motions to Quash Subpoena to Wells Fargo Bank, Signature Bank, Open Bank and Bank of Hope) (10/09/2019)	2045-2232
5	Reply to Opposition to Plaintiff's Motion for Sanctions (10/18/2019)	2233-2250
6 7	VOLUME X	PAGES
8 9	Reply to Opposition to Plaintiff's Motion for Sanctions (10/18/2019) (cont'd)	2251-2297
10	Notice of Intent to Issue Subpoena to Lucas Horsfall, LLP (10/22/2019)	2298-2378
12	Notice of Intent to Issue Subpoena to Bank of America, N.A. (10/22/2019)	2379-2459
14	Plaintiff's Motion to Quash Subpoenas (10/29/2019)	2460-2478
15 16 17	Defendants' Opposition to Plaintiff's Motion to Quash Subpoenas to Third Parties Bank of America and Lucas Horsfall, Murphy & Pindroh, LLP (11/6/2019)	2479-2500
18 19	VOLUME XI	PAGES
20 21	Defendants' Opposition to Plaintiff's Motion to Quash Subpoenas to Third Parties Bank of America and Lucas Horsfall, Murphy & Pindroh, LLP (11/6/2019) (cont'd)	2501-2655
222324	Notice of Entry of Order Granting Defendants' Motion to Advance Hearing regarding Plaintiff's Motion to Quash Subpoenas (11/08/2019)	2656-2660
252627	Reply to Opposition to Motion to Quash Subpoenas (11/15/2019)	2661-2750

1 2	VOLUME XII	<u>PAGES</u>
3 4	Reply to Opposition to Motion to Quash Subpoenas (11/15/2019) (cont'd)	2751-2776
5	Notice of Entry of Order Shortening Time (11/15/2019)	2777-2785
6 7 8 9	Notice of Entry of Order Granting in Part and Denying in Part Defendants' Motions to Quash Plaintiff's Subpoenas to Non-Parties Empyrean West, Jay Carter and David Keller (12/6/2019)	2786-2793
10 11	Notice of Entry of Order Granting Defendant's Motions to Quash Plaintiff's Subpoenas to Non-Party Banks (12/6/2019)	2794-2800
12 13	Notice of Entry of Stipulation and Order Regarding Exhibit (12/6/2019)	2801-2816
14 15	Notice of Entry of Order Denying Plaintiff's Motion to Quash Subpoenas to Plaintiff's Bank and Accountant (12/6/2019)	2817-2822
16	Notice of Entry of Order Shortening Time (12/11/2019)	2823-2836
17 18	Notice of Entry of Order (12/18/2019)	2837-2840
19	Notice of Entry of Stipulation and Order (12/18/2019)	2841-2846
20 21 22	Notice of Entry of Order Denying Plaintiff's Motion to Quash Subpoenas to Morales Construction, Top Rank Builders and All American Concrete and Masonry (12/19/2019)	2847-2853
232425	Notice of Entry of Order Denying Plaintiff's Motion for Sanctions Related to Defendant EB5IA's Accounting Records (12/19/2019)	2854-2860
26 27 28	Notice of Entry of Order Denying Plaintiff's Motion to Stay Enforcement of Order Denying Plaintiff's Motion to Quash Subpoenas to Bank of America and Lucas Horsfall (01/02/2020)	2861-2866

1		
2	Notice of Entry of Order (01/17/2020)	2867-2874
3	Statement of Undisputed Facts (01/17/2020)	2875-3000
4	VOLUME XIII	PAGES
5 6	Statement of Undisputed Facts (01/17/2020) (cont'd)	3001-3080
7 8 9	Notice of Entry of Findings of Fact, Conclusions of Law, and Order Denying Defendant Las Vegas Development Fund LLC's Motion to Dissolve Temporary Restraining Order and to Appoint a Receiver (01/23/2020)	3081-3091
10 11 12	Notice of Entry of Order on Status Check Regarding Discovery Responses/Plaintiff's Motion to Compel (01/23/2020)	3092-3095
13	Motion for Summary Judgment as to the Counterclaims Against VNV Dynasty Trust I and VNV Dynasty Trust II (01/23/2020)	3096-3143
15	Motion for Summary Judgment as to the Counterclaims Against Jennifer Piazza (01/23/2020)	3144-3166
17 18	Defendant and Counter Claimant LVDF's Objections to Plaintiff and Counter Defendant's Statement of Undisputed Facts (02/03/2020)	3167-3222
19 20 21	Defendant and Counterclaimant LVD Fund's Opposition to Counterdefendant Jennifer Piazza's Motion for Summary Judgment [redacted in district court filing] (02/03/2020)	3223-3239
22 23 24 25	Defendant and Counterclaimant LVD Fund's Opposition to VNV Dynasty Trust I and VNV Dynasty Trust II's Motion for Summary Judgment [redacted in district court filing] (02/03/2020)	3240-3250
26		

1 2	VOLUME XIV	<u>PAGES</u>
3 4 5	Defendant and Counterclaimant LVD Fund's Opposition to VNV Dynasty Trust I and VNV Dynasty Trust II's Motion for Summary Judgment [redacted in district court filing] (02/03/2020) (cont'd)	3251-3256
6 7 8 9	Declaration of C. Keith Greer in Support of Defendant and Counterclaimants' Oppositions to Jennifer Piazza and the VNV Dynasty Trust I and II Motions for Summary Judgment (02/03/2020)	3257-3326
10	Notice of Entry of Order (02/07/2020)	3327-3330
11 12 13 14	Motion to Seal and/or Redact Portions of Defendants' Oppositions to Jennifer Piazza and the VNV Trusts' Motions for Summary Judgment to Protect Confidential Financial Information, Motion for Order Shortening Time and Order Shortening Time (02/11/2020)	3331-3348
15 16	Notice of Entry of Order Shortening Time (02/11/2020)	3349-3368
17 18 19 20	Defendant Las Vegas Development Fund LLC's Opposition to Motion to Seal and/or Redact portions of Defendants' Oppositions to Jennifer Piazza and the NVN Trusts' Motions for Summary Judgment to Protect Confidential Financial Information (02/14/2020)	3369-3380
21 22	Notice of Entry of Order Regarding February 5, 2020 Status Check (02/19/2020)	3381-3385
23 24	Notice of Entry of Stipulation and Order Resetting Hearings and Briefing Schedule (02/25/2020)	3386-3391
25 26	Response to Defendant LVDF's Objections to Statement of Undisputed Facts and Countermotion to Strike (02/28/2020)	3392-3411
27 28	Notice of Entry of Order (03/02/2020)	3412-3416

2	Notice of Entry of Order (03/03/2020)	3417-3421
3	Notice of Entry of Order (03/12/2020)	3422-3429
4	Notice of Entry of Order (04/01/2020)	3430-3436
5	Notice of Entry of Order (04/01/2020)	3437-3441
7 8 9	Defendant and Counterclaimant Las Vegas Development Fund, LLC's Notice of Motion and Motion for Leave to Amend the Countercomplaint [redacted in district court filing] (04/03/2020)	3442-3500
10	VOLUME XV	PAGES
12 13 14	Defendant and Counterclaimant Las Vegas Development Fund, LLC's Notice of Motion and Motion for Leave to Amend the Countercomplaint [redacted in district court filing] (04/03/2020) (cont'd)	3501-3640
15 16 17	Declaration of C. Keith Greer in Support of Las Vegas Development Fund, LLC's Motion for Leave to Amend the Countercomplaint (04/04/2020)	3641-3645
18 19	Opposition to Motion for Leave to Amend Counterclaim (04/17/2020)	3646-3692
20212223	Notice of Entry of Stipulation and Order to Replace Exhibit "A" to Defendant's Motion for Leave to Amend the Countercomplaint [redacted in district court filing] (04/20/2020)	3693-3750
24	<u>VOLUME XVI</u>	PAGES
25262728	Notice of Entry of Stipulation and Order to Replace Exhibit "A" to Defendant's Motion for Leave to Amend the Countercomplaint [redacted in district court filing] (04/20/2020) (cont'd)	3751-3891

1	Nation of Entry of Order (04/28/2020)	2002 2006
2	Notice of Entry of Order (04/28/2020)	3892-3896
3	Reply in Support of Defendant and Counterclaimant Las Vegas	3897-4000
4	Development Fund, LLC's Motion for Leave to Amend the Counterclaim [redacted in district court filing] (04/29/2020)	
5	VOLUME XVII	PAGES
7	Reply in Support of Defendant and Counterclaimant Las Vegas Development Fund, LLC's Motion for Leave to Amend the	4001-4006
8	Counterclaim [redacted in district court filing] (04/29/2020)	
9	(cont'd)	
10	Defendant Las Vegas Development Fund, LLC's Motion for	4007-4016
11	Clarification on Order Shortening Time (05/01/2020)	4007-4010
12		
13	Opposition to Defendant Las Vegas Development Fund LLC's Motion for Clarification on Order Shortening Time	4017-4045
14	(05/11/2020)	
15		1016 10 - 6
16	Notice of Entry of Stipulation and Order to Extend Discovery Deadlines and Continue Trial (Second Request) (05/13/2020)	4046-4056
17	Deadinies and Continue That (Second Request) (03/13/2020)	
18	Amended Order Setting Jury Trial (05/13/2020)	4057-4061
19	Notice of Entry of Order Granting Las Vegas Development	4062-4067
20	Fund, LLC's Motion to Compel Production of Documents or, in	
21	the Alternative, Motion for Preliminary Injunction to Address Front Sight's Continuing Violation of Section 5.10 of the	
22	Construction Loan Agreement and Request for Limited Relief	
23	From the Protective Order (05/18/2020)	
24	Notice of Entry of Order Granting Defendant and	4068-4072
25	Counterclaimant Las Vegas Development Fund, LLC's Notice	
26	of Motion and Motion for Leave to Amend the Countercomplaint (06/04/2020)	
27		

1 2 3	Defendants' Answer to Plaintiff's Second Amended Complaint; and First Amended Counterclaim [redacted in district court filing] (06/04/2020)	4073-4250
5	VOLUME XVIII	PAGES
6 7	Defendants' Answer to Plaintiff's Second Amended Complaint; and First Amended Counterclaim [redacted in district court filing] (06/04/2020) (cont'd)	4251-4262
8 9 10	Notice of Entry of Order Granting Defendant Las Vegas Development Fund, LLC's Motion for Clarification on Order Shortening Time (06/05/2020)	4263-4268
11 12 13 14	Notice of Entry of Findings of Fact, Conclusions of Law and Order Denying Plaintiff Front Sight Management, LLC's Motion to Extinguish LVDF's Deed of Trust, or Alternatively to Grant Senior Debt Lender Romspen a First Lien Position, and Motion to Deposit Funds Pursuant to NRCP 67 (06/08/2020)	4269-4275
15 16 17	Notice of Entry of Order Denying Plaintiff's Motion to Quash Subpoenas to Summit Financial Group and US Capital Partners, Inc. (06/08/2020)	4276-4281
18 19 20	Notice of Entry of Order Denying Counter Defendants VNV Dynasty Trust I and VNV Dynasty Trust II's Motion for Summary Judgment (06/08/2020)	4282-4287
21 22	Notice of Entry of Order Denying Counter Defendant Jennifer Piazza's Motion for Summary Judgment (06/08/2020)	4288-4293
23	Notice of Entry of Order Shortening Time (06/12/2020)	4294-4305
2425	Affidavit of Service – Michael G. Meacher (06/16/2020)	4306-4308
26	Affidavit of Service – Top Rank Builders Inc. (06/16/2020)	4309-4311
27 28	Affidavit of Service – All American Concrete & Masonry Inc. (06/16/2020)	4312-4314

2	Affidavit of Service – Morales Construction, Inc. (06/16/2020)	4315-4317
3	Notice of Entry of Order Denying Front Sight Management	4318-4327
4	LLC's Motion for Partial Summary Judgment With Findings of	
5	Fact and Conclusions of Law (06/22/2020)	
6	Notice of Entry of Order Granting in Part Motion for Sanctions	4328-4333
7	and/or to Compel Actual Responses to Plaintiff's First Sets of Interrogatories to Defendants (06/22/2020)	
8	Notice of Entry of Findings of Fact and Conclusions of Law and	4334-4342
9	Order Granting In Part and Denying In Part Defendants' Motion	7337-7372
10	for Protective Order Regarding Discovery of Consultants and	
11	Individual Investors Confidential Information (07/06/2020)	
12	Notice of Entry of Order Denying Without Prejudice Plaintiff's	4343-4349
13	Motion for Sanctions for Violation of Court Orders Related to Defendants Responses to Plaintiffs Requests for Production of	
14	Documents to Defendants (07/06/2020)	
15	Nation of Francis of Onlan Counting Default and Mation for	1250 1250
16	Notice of Entry of Order Granting Defendants' Motion for Protective Order Regarding the Defendants' Private Financial	4350-4356
17	Information (07/10/2020)	
18	Acceptance of Service on Behalf of Efrain Rene Morales-	4357-4359
19	Moreno (07/23/2020)	1337 1337
20	Counterdefendant Jennifer Piazza's Answer to First Amended	4360-4386
21	Counterclaim (08/21/2020)	4300-4360
22	M: (00/26/2020)	4207 4200
23	Minutes of the Court (08/26/2020)	4387-4389
24	Notice of Entry of Stipulation and Order to Extend Discovery	4390-4403
25	Deadlines (09/02/2020)	
26		
27		

ALPHABETICAL INDEX

1

3		Volume(s)	Pages
4 5 6 7	Acceptance of Service of Counterclaim on Counterdefendants Front Sight Management, LLC, Ignatius Piazza, Jennifer Piazza, VNV Dynasty Trust I and VNV Dynasty Trust II (06/14/2019)	VI	1316-1317
8 9	Acceptance of Service on Behalf of Efrain Rene Morales-Moreno (07/23/2020)	XVIII	4357-4359
10 11	Affidavit of Service on Chicago Title Company (10/22/2018)	I	0063
12 13	Affidavit of Service on EB5 Impact Advisors LLC (10/17/2018)	I	0060
14 15	Affidavit of Service on EB5 Impact Capital Regional Center LLC (10/18/2018)	I	0061
16 17	Affidavit of Service on Las Vegas Development Fund LLC (10/18/2018)	I	0062
18 19	Affidavit of Service on Linda Stanwood (10/17/2018)	I	0059
20 21	Affidavit of Service on Robert W. Dziubla (10/17/2018)	I	0058
22 23	Affidavit of Service – All American Concrete & Masonry Inc. (06/16/2020)	XVIII	4312-4314
24 25	Affidavit of Service – Michael G. Meacher (06/16/2020)	XVIII	4306-4308
26 27 28	Affidavit of Service – Morales Construction, Inc. (06/16/2020)	XVIII	4315-4317

1 2	Affidavit of Service – Top Rank Builders Inc. (06/16/2020)	XVIII	4309-4311
3 4	Amended Complaint (10/04/2018)	I	0029-0057
5	Amended Order Setting Jury Trial (05/13/2020)	XVII	4057-4061
6 7	Business Court Order (07/23/2019)	VII	1566-1572
8	Complaint (09/14/2018)	I	0001-0028
9 10	Counterdefendant Dr. Ignatius Piazza's Answer to Counterclaim (09/30/2019)	VIII	1958-1981
11 12	Counterdefendant Front Sight Management LLC's Answer to Counterclaim (09/30/2019)	VIII / IX	1982-2005
13 14	Counterdefendant Jennifer Piazza's Answer to Counterclaim (09/30/2019)	IX	2006-2029
15 16	Counterdefendant Jennifer Piazza's Answer to First Amended Counterclaim (08/21/2020)	XVIII	4360-4386
17 18 19	Counterdefendants VNV Dynasty Trust I and VNV Dynasty Trust II's Answer to Counterclaim (09/30/2019)	VIII	1933-1957
20212223	Declaration of C. Keith Greer in Support of Defendant and Counterclaimants' Oppositions to Jennifer Piazza and the VNV Dynasty Trust I and II Motions for Summary Judgment (02/03/2020)	XIV	3257-3326
24 25 26	Declaration of C. Keith Greer in Support of Defendant LVD Fund's Reply to Plaintiff's Opposition to Defendant's Motion to Appoint Receiver (02/26/2019)	IV	0762-0769
27 28	Declaration of C. Keith Greer in Support of Defendant's Motion for Receivership (02/06/2019)	III	0559-0601
	· ·		

1	Declaration of C. Keith Greer in Support of Las	XV	3641-3645
2	Vegas Development Fund, LLC's Motion for Leave		
3	to Amend the Countercomplaint (04/04/2020)		
4	Declaration of Robert Dziubla in Opposition to	IX	2041-2044
5	Plaintiff's Motion for Sanctions (09/30/2019)		
6	Declaration of Robert Dziubla in Support of	II / III	0379-0558
7	Defendant Las Vegas Development Fund LLC's	117 111	0317 0330
8	Motion for Appointment of Receiver [redacted in		
9	district court filing] (02/06/2019)		
10	Defendant and Counter Claimant LVDF's	XIII	3167-3222
11	Objections to Plaintiff and Counter Defendant's Statement of Undisputed Facts (02/03/2020)		
12	Statement of Chaispated Lacts (02/03/2020)		
13	Defendant and Counterclaimant Las Vegas	XIV / XV	3442-3640
14	Development Fund, LLC's Notice of Motion and Motion for Leave to Amend the Countercomplaint		
15	[redacted in district court filing] (04/03/2020)		
16	Defendant and Counterclaimant LVD Fund's	XIII	3223-3239
17	Opposition to Counterdefendant Jennifer Piazza's		0220 0209
18	Motion for Summary Judgment [redacted in district court filing] (02/03/2020)		
19	court juingj (02/03/2020)		
20	Defendant and Counterclaimant LVD Fund's	XIII / XIV	3240-3256
21	Opposition to VNV Dynasty Trust I and VNV Dynasty Trust II's Motion for Summary Judgment		
22	[redacted in district court filing] (02/03/2020)		
23	Defendant EB5 Impact Advisors LLC's Opposition	IX	2030-2040
24	to Plaintiff's Motion for Sanctions (09/30/2019)		2020 2010
25	Defendant Las Vegas Develonment Fund LLC's	II	0351-0378
26	Defendant Las Vegas Development Fund LLC's Motion for Appointment of Receiver and Request for	11	0331-0378
27	Order Shortening Time (02/06/2019)		

1 2	Defendant Las Vegas Development Fund, LLC's Motion for Clarification on Order Shortening Time	XVII	4007-4016
3	(05/01/2020)		
4	Defendant Las Vegas Development Fund LLC's	XIV	3369-3380
5	Opposition to Motion to Seal and/or Redact portions	711 (3307 3300
6	of Defendants' Oppositions to Jennifer Piazza and		
7	the NVN Trusts' Motions for Summary Judgment to Protect Confidential Financial Information		
8	(02/14/2020)		
9	Defendant Las Vegas Development Fund, LLC's	IV	0837-0860
10	Opposition to Plaintiff's Second Motion for	1 V	0037-0000
11	Temporary Restraining Order and Preliminary		
12	Injunction (03/19/2019)		
13	Defendant Las Vegas Development Fund LLC's	III / IV	0741-0755
14	Reply to Plaintiff's Opposition to Defendant's Motion for Appointment of Receiver (02/26/2019)		
15	Defendants' Answer to Plaintiff's Second Amended	IV / V	0917-1083
16	Complaint and Counterclaim (04/23/2019)	1, ,	0717 1005
17	D-C14-2 A4- D1-:4:502- C1 A1-1	3/3/11 /	4072 4262
18	Defendants' Answer to Plaintiff's Second Amended Complaint and First Amended Counterclaim	XVII / XVIII	4073-4262
19	[redacted in district court filing] (06/04/2020)		
20	Defendants' Opposition to Plaintiff's Motion to	X / XI	2479-2655
21	Quash Subpoenas to Third Parties Bank of America	$\mathbf{A} / \mathbf{A} \mathbf{I}$	2419-2033
22	and Lucas Horsfall, Murphy & Pindroh, LLP		
23	(11/6/2019)		
24	Errata to Opposition to Defendant Las Vegas	III	0731-0740
25	Development Fund LLC's Motion for Appointment		
26	of Receiver (02/22/2019)		
27			

1 2 3 4	Errata to Supplemental Declaration of Robert Dziubla in Support of Defendants' Opposition to Plaintiff's Second Motion for Temporary Restraining Order and Preliminary Injunction (03/20/2019)	IV	0882-0892
5	Minutes of the Court (08/26/2020)	XVIII	4387-4389
6 7	Motion for Summary Judgment as to the Counterclaims Against Jennifer Piazza (01/23/2020)	XIII	3144-3166
8 9 10	Motion for Summary Judgment as to the Counterclaims Against VNV Dynasty Trust I and VNV Dynasty Trust II (01/23/2020)	XIII	3096-3143
11 12 13 14	Motion to Seal and/or Redact Pleadings and Exhibits to Protect Confidential Information, Motion to Amend Paragraph 2.3 of Protective Order, Motion for Order Shortening Time and Order Shortening Time (02/15/2019)	III	0602-0628
15 16 17 18 19	Motion to Seal and/or Redact Portions of Defendants' Oppositions to Jennifer Piazza and the VNV Trusts' Motions for Summary Judgment to Protect Confidential Financial Information, Motion for Order Shortening Time and Order Shortening Time (02/11/2020)	XIV	3331-3348
202122	Notice of Entry of Disclaimer of Interest of Chicago Title Company and Stipulation and Order for Dismissal (02/05/2019)	II	0344-0350
23 24 25 26 27	Notice of Entry of Findings of Fact and Conclusions of Law and Order Granting In Part and Denying In Part Defendants' Motion for Protective Order Regarding Discovery of Consultants and Individual Investors Confidential Information (07/06/2020)	XVIII	4334-4342

1 2 3 4	Notice of Entry of Findings of Fact, Conclusions of Law, and Order Denying Defendant Las Vegas Development Fund LLC's Motion to Dissolve Temporary Restraining Order and to Appoint a Receiver (01/23/2020)	XIII	3081-3091
5 6 7 8 9	Notice of Entry of Findings of Fact, Conclusions of Law and Order Denying Plaintiff Front Sight Management, LLC's Motion to Extinguish LVDF's Deed of Trust, or Alternatively to Grant Senior Debt Lender Romspen a First Lien Position, and Motion to Deposit Funds Pursuant to NRCP 67 (06/08/2020)	XVIII	4269-4275
10	Notice of Entry of Order (03/19/2019)	IV	0876-0881
12	Notice of Entry of Order (04/10/2019)	IV	0893-0897
13 14	Notice of Entry of Order (04/10/2019)	IV	0898-0903
15	Notice of Entry of Order (04/10/2019)	IV	0904-0909
16	Notice of Entry of Order (04/10/2019)	IV	0910-0916
17 18	Notice of Entry of Order (05/16/2019)	V	1084-1089
19	Notice of Entry of Order (06/25/2019)	VI	1318-1324
20	Notice of Entry of Order (12/18/2019)	XII	2837-2840
21 22	Notice of Entry of Order (01/17/2020)	XII	2867-2874
23	Notice of Entry of Order (02/07/2020)	XIV	3327-3330
24 25	Notice of Entry of Order (03/02/2020)	XIV	3412-3416
26	Notice of Entry of Order (03/03/2020)	XIV	3417-3421
27 28	Notice of Entry of Order (03/12/2020)	XIV	3422-3429

1 2	Notice of Entry of Order (04/01/2020)	XIV	3430-3436
3	Notice of Entry of Order (04/01/2020)	XIV	3437-3441
4	Notice of Entry of Order (04/28/2020)	XVI	3892-3896
5 6	Notice of Entry of Order Admitting to Practice (11/15/2018)	I	0064-0068
7	Notice of Entry of Order Denying Counter	XVIII	4288-4293
8 9	Defendant Jennifer Piazza's Motion for Summary Judgment (06/08/2020)	21 111	1200 1255
10			
11	Notice of Entry of Order Denying Counter Defendants VNV Dynasty Trust I and VNV Dynasty	XVIII	4282-4287
12	Trust II's Motion for Summary Judgment (06/08/2020)		
14	Notice of Entry of Order Denying Front Sight Management LLC's Motion for Partial Summary	XVIII	4318-4327
15 16	Judgment With Findings of Fact and Conclusions of Law (06/22/2020)		
17			
18	Notice of Entry of Order Denying Plaintiff's Motion for Sanctions Related to Defendant EB5IA's	XII	2854-2860
19	Accounting Records (12/19/2019)		
20	Notice of Entry of Order Denying Plaintiff's Motion	VII	1585-1591
21	for Temporary Restraining Order and Preliminary	V 11	1303-1391
22	Injunction related to Investor Funds and Interest		
23	Payments (09/13/2019)		
24	Notice of Entry of Order Denying Plaintiff's Motion	XII	2847-2853
25	to Quash Subpoenas to Morales Construction, Top Rank Builders and All American Concrete and		
26	Masonry (12/19/2019)		
27			

1			
2	Notice of Entry of Order Denying Plaintiff's Motion	XII	2817-2822
	to Quash Subpoenas to Plaintiff's Bank and Accountant (12/6/2019)		
3	Accountant (12/0/2019)		
4	Notice of Entry of Order Denying Plaintiff's Motion	XVIII	4276-4281
5 6	to Quash Subpoenas to Summit Financial Group and US Capital Partners, Inc. (06/08/2020)		
7	Notice of Entry of Order Denying Plaintiff's Motion	XII	2861-2866
8	to Stay Enforcement of Order Denying Plaintiff's		
9	Motion to Quash Subpoenas to Bank of America and Lucas Horsfall (01/02/2020)		
10			
11	Notice of Entry of Order Denying Without Prejudice Plaintiff's Motion for Sanctions for Violation of	XVIII	4343-4349
12	Court Orders Related to Defendants Responses to		
13	Plaintiffs Requests for Production of Documents to		
14	Defendants (07/06/2020)		
15	Notice of Entry of Order Granting Defendant and	XVII	4068-4072
16	Counterclaimant Las Vegas Development Fund,		
17	LLC's Notice of Motion and Motion for Leave to Amend the Countercomplaint (06/04/2020)		
18	- · · · · · · · · · · · · · · · · · · ·		
19	Notice of Entry of Order Granting Defendant Las Vegas Development Fund, LLC's Motion for	XVIII	4263-4268
20	Clarification on Order Shortening Time (06/05/2020)		
21	Notice of Entry of Order Granting Defendant's	XII	2794-2800
22	Motions to Quash Plaintiff's Subpoenas to Non-	AII	219 1 -2000
23	Party Banks (12/6/2019)		
24	Notice of Entry of Order Granting Defendants'	XVIII	4350-4356
25	Motion for Protective Order Regarding the Defendants' Private Financial Information		
26	(07/10/2020)		
27			

1 2	Notice of Entry of Order Granting Defendants'	XI	2656-2660
3	Motion to Advance Hearing regarding Plaintiff's Motion to Quash Subpoenas (11/08/2019)		
4	Notice of Entry of Order Granting in Part and	VII	1578-1584
5	Denying in Part Counterdefendants' Motions to Dismiss Counterclaim (09/13/2019)		
6	Distinss Counterclaim (07/13/2017)		
7	Notice of Entry of Order Granting in Part and	XII	2786-2793
8	Denying in Part Defendants' Motions to Quash Plaintiff's Subpoenas to Non-Parties Empyrean		
9	West, Jay Carter and David Keller (12/6/2019)		
10	Notice of Entry of Order Granting in Part Motion for	XVIII	4328-4333
12	Sanctions and/or to Compel Actual Responses to Plaintiff's First Sets of Interrogatories to Defendants		
13	(06/22/2020)		
14	Nation of Entry of Order Counting Los Vages	XVII	4062-4067
15	Notice of Entry of Order Granting Las Vegas Development Fund, LLC's Motion to Compel	AVII	4002-4007
16	Production of Documents or, in the Alternative, Motion for Preliminary Injunction to Address Front		
17	Sight's Continuing Violation of Section 5.10 of the		
18	Construction Loan Agreement and Request for Limited Relief From the Protective Order		
19	(05/18/2020)		
20	Notice of Entry of Order Granting Plaintiff's Motion	I	0075-0079
21	for Protective Order (11/27/2018)	1	0075-0077
22	Notice of Feture of Onder Counting Towns one	T	0000 0104
23	Notice of Entry of Order Granting Temporary Restraining Order and Expunging Notice of Default	I	0099-0104
24	(11/27/2018)		
25	Notice of Entry of Order on Defendants' Motion to	II	0333-0337
26	Dismiss Plaintiff's First Amended Complaint		
27	(01/17/2019)		
28			

1 2	Notice of Entry of Order on Plaintiff's Motion for	II	0323-0327
	Preliminary Injunction (01/17/2019)		
3 4	Notice of Entry of Order on Plaintiff's Motion to	II	0338-0343
5	Disqualify C. Keith Greer as Attorney of Record for Defendants (01/25/2019)		
6		_	0000
7	Notice of Entry of Order on Plaintiff's Petition for I Appointment of Receiver and for an Accounting		0069-0074
8	(11/27/2018)		
9	Notice of Entry of Order on Plaintiff's Renewed	II	0328-0332
10	Motion for an Accounting Related to Defendants Las	11	0320-0332
11	Vegas Development Fund LLC and Robert Dziubla and for Release of Funds (01/17/2019)		
12	and for Release of Fands (01/17/2017)		
13	Notice of Entry of Order on Status Check Regarding Discovery Responses/Plaintiff's Motion to Compel	XIII	3092-3095
14	(01/23/2020)		
15	N. CE. CO. I. D. T. E. I. 5	3/13 /	2201 2205
16	Notice of Entry of Order Regarding February 5, 2020 Status Check (02/19/2020)	XIV	3381-3385
17	Nation of Entry of Order Chartening Time	111	0629-0658
18	Notice of Entry of Order Shortening Time (02/15/2019)	III	0029-0038
19	Nation of Fortune of Outland the ortenian a Time	XII	2777 2795
20	Notice of Entry of Order Shortening Time (11/15/2019)		2777-2785
21			
22	Notice of Entry of Order Shortening Time (12/11/2019)	XII	2823-2836
23			
24	Notice of Entry of Order Shortening Time	XIV	3349-3368
25	(02/11/2020)		
26	Notice of Entry of Order Shortening Time	XVIII	4294-4305
27	(06/12/2020)		
28			

2	Notice of Entry of Order Staying All Subpoenas For Documents and Depositions which were Served on Non-Parties by Plaintiff (09/13/2019)	VII	1592-1599
3	Non-Fattles by Flamuii (09/13/2019)		
5	Notice of Entry of Protective Order (11/27/2018)	I	0080-0098
6	Notice of Entry of Stipulation and Order (12/18/2019)	XII	2841-2846
7			
8	Notice of Entry of Stipulation and Order Regarding Defendants' Judicial Foreclosure Cause of Action	VI	1325-1330
9	(06/25/2019)		
10			
11	Notice of Entry of Stipulation and Order Regarding Exhibit (12/6/2019)	XII	2801-2816
12			
13	Notice of Entry of Stipulation and Order Resetting Hearings and Briefing Schedule (02/25/2020)	XIV	3386-3391
14			
15	Notice of Entry of Stipulation and Order to Extend Discovery Deadlines (09/02/2020)	XVIII	4390-4403
16	Biscovery Beatimes (05/102/2020)		
17	Notice of Entry of Stipulation and Order to Extend Discovery Deadlines and Continue Trial (Second	XVII	4046-4056
18 19	Request) (05/13/2020)		
19	Notice of Entry of Stipulation and Order to Replace	VV / VVI	3693-3891
20	Exhibit "A" to Defendant's Motion for Leave to	XV / XVI	3093-3091
21 22	Amend the Countercomplaint [redacted in district court filing] (04/20/2020)		
23 24	Notice of Intent to Issue Subpoena to Bank of America, N.A. (10/22/2019)	X	2379-2459
۷٦	111101100, 11.11. (10/22/2017)		
25	Notice of Intent to Issue Subpoena to Lucas Horsfall,	X	2298-2378
26	LLP (10/22/2019)		
27			

1 2 3 4	Opposition Memorandum of Defendant Las Vegas Development Fund, LLC to Plaintiff's Motion to Seal and/or Redact Pleadings and Exhibits (02/19/2019)	III	0659-0669
5 6 7	Opposition to Defendant Las Vegas Development Fund LLC's Motion for Appointment of Receiver (02/22/2019)	III	0670-0730
8 9	Opposition to Defendant Las Vegas Development Fund LLC's Motion for Clarification on Order Shortening Time (05/11/2020)	XVII	4017-4045
10 11 12	Order Re Rule 16 Conference, Setting Civil Jury Trial, Pre-Trial/Calendar Call and Deadlines for Motions; Discovery Scheduling Order (08/20/2019)	VII	1573-1577
13 14	Order Scheduling Hearing (09/27/2019)	VIII	1931-1932
15	Order Setting Settlement Conference (12/06/2018)	I	0105-0106
16	Order Setting Settlement Conference (06/04/2019)	VI	1314-1315
17 18	Plaintiff's Motion for Sanctions (09/17/2019)	VII	1600-1643
19	Plaintiff's Motion to Quash Subpoenas (10/29/2019)	X	2460-2478
20212223	Plaintiff's Second Motion for Temporary Restraining Order and Preliminary Injunction, Motion for Order Shortening Time, and Order Shortening Time (03/01/19)	IV	0770-0836
242526	Reply in Support of Defendant and Counterclaimant Las Vegas Development Fund, LLC's Motion for Leave to Amend the Counterclaim [redacted in district court filing] (04/29/2020)	XVI / XVII	3897-4006
27 28	Reply to Opposition to Motion to Quash Subpoenas (11/15/2019)	XI / XII	2661-2776

1 2	Reply to Opposition to Plaintiff's Motion for Sanctions (10/18/2019)	IV / X	2233-2297
3 4 5	Reporter's Transcript of Hearing (Preliminary Injunction Hearing) (09/20/2019)	VII / VIII	1644-1930
6	Reporter's Transcript of Motion (Preliminary Injunction Hearing) (06/03/2019)	V / VI	1090-1313
8 9 10	Reporter's Transcript of Motions (Defendants' Motions to Quash Subpoena to Wells Fargo Bank, Signature Bank, Open Bank and Bank of Hope) (10/09/2019)	IX	2045-2232
11 12	Reporter's Transcript of Preliminary Injunction Hearing (07/22/2019)	VI / VII	1331-1513
13 14	Reporter's Transcript of Preliminary Injunction (07/23/2019)	VII	1514-1565
15 16 17	Response to Defendant LVDF's Objections to Statement of Undisputed Facts and Countermotion to Strike (02/28/2020)	XIV	3392-3411
18 19	Second Amended Complaint (01/04/2019)	I / II	0107-0322
20	Statement of Undisputed Facts (01/17/2020)	XII / XIII	2875-3080
21222324	Supplemental Declaration of Defendant Robert Dziubla in Support of Defendant Las Vegas Development Fund, LLC's Opposition to Plaintiff's Second Motion for Temporary Restraining Order and Preliminary Injunction (03/19/2019)	IV	0861-0875
25262728	Supplemental Declaration of Robert W. Dziubla in Support of Defendant LVD Fund's Reply to Plaintiff's Opposition to Defendant's Motion to Appointment of Receiver (02/26/2019)	IV	0756-0761

limited to, inducing Front Sight to improperly use funds for the personal benefit of Counter Defendants Ignatius Piazza, Jennifer Piazza, and VNV Trust Defendants.

- 93. Front Sight did in fact breach the contract as stated specifically above.
- 94. The breach was caused by the wrongful and unjustified conduct;
- 95. As a direct and proximate result of Counter Defendants' intentional acts to induce Front Sight to breach the CLA, Counter Claimant sustained damages in the amount to be proven at trial.
- 96. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.

FOURTH CAUSE OF ACTION

Conversion Against Front Sight, Ignatius Piazza and Jennifer Piazza

- 97. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 96 of this Counterclaim as though set forth fully herein at length.
- 98. Through these Counter Defendants' conduct described above, Counter Defendants obtained Counter Claimants' property and have wrongfully asserted dominion over Counter Claimant's property; to wit: misappropriating and spending the loan proceeds under the CLA for purposes other than that for which it was intended.
- 99. Counter Defendants' wrongful conduct was in denial of, inconsistent with, and in defiance of Counter Claimant's rights and title to its money and/or property.
- 100. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.

///

FIFTH CAUSE OF ACTION Civil Conspiracy Against All Counter Defendants

- 101. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 100 of this Counterclaim as though set forth fully herein at length.
- 102. As set forth above, Counter Defendants Ignatius Piazza and Jennifer Piazza, both in their individual capacity and in their capacity as Trustees and/or beneficiaries of the VNV Trust Defendants, acted together in concert, in their individual capacities, to accomplish their unlawful objectives for the purpose of harming Counter Claimant.
- 103. While acting in their individual capacities and in their capacity as Trustees and/or beneficiaries of the VNV Trust Defendants, Ignatius Piazza and Jennifer Piazza conspired with Front Sight and the VNV Trust Defendants, using Front Sight and VNV Trust Defendants to achieve their unlawful objective of diverting monies from Front Sight that were needed to maintain Front Sight's solvency and its ability to meet its obligations under the CLA regarding timely completion of the Project and repayment of the loan, for their own individual advantage and benefit.
- 104. As a direct and proximate result of the Counter Defendants' acts, Counter Claimant has been damaged in an amount to be proven at trial.
- 105. Counter Defendants' conduct was malicious, oppressive and fraudulent under NRS42.005, entitling Counter Claimant to an award of punitive damages.
- 106. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.
- 107. Based on Counter Defendants' conduct and the inequitable result of allowing the transferred funds to remain in control of Counter Defendants, a constructive trust should be placed on all moneys transferred from Front Sight to the VNV Trust Defendants, as prayed for below.

28 ///

///

Page 37 of 41

SIXTH CAUSE OF ACTION Judicial Foreclosure Against Front Sight

108. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 107 of this Counterclaim as though set forth fully herein at length.

- 109. In July 2017, Counter Defendant Front Sight for good and valuable consideration executed and delivered the original Promissory Note to LVD Fund. On November 14, 2017, Counter Defendant Front Sight executed and delivered the Amended and Restated Promissory Note to LVD Fund. (Exhibit 7).
- 110. To secure the Note, on October 13, 2016, Counter Claimant LVD Fund recorded a Deed of Trust titled "Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents, and Fixture Filing," in the official records of Nye County, Nevada, as "DOC #860867." (Exhibit 1). On January 12, 2018, the "First Amendment to Construction Deed of Trust, Security Agreement and Fixture Filing," was recorded in the official records of Nye County, Nevada, as "DOC #886510." (Exhibit 2).
- 111. Counter Claimant LVD Fund is the owner and the holder of the note for value and has performed all obligation under the Promissory Note.
- 112. The encumbered Property is now owned by and in possession of the Counter Defendant Front Sight.
- 113. Counter Defendants have breached the Deed of Trust as discussed in detail above, which include but are not limited to: improper use of loan proceeds; failure to provide government approved plans; material delays in construction, material changes to cost, scope and timing of the construction; refusal to comply with regarding Senior Debt; failure to provide monthly project costs; failure to notify Lender of events of default; refusal to allow Lender to inspect books and records; diverting Front Sight assets out of Front Sight for the benefit the individual Counter Defendants; refusal to allow site inspections; failure to give Lender annual financial statements; and failure to provide EB5 documentation.
- 114. As of January 4, 2019 there remained due and owing under the Note approximately \$345,787.24 (excluding principal) as described in the Notice of Breach and Election to Sell Under Page **38** of **41**

the Deed of Trust. (Exhibit 6). Counter Defendants reserve the right to amend this Counterclaim up to the time of trial to include any additional amounts which become due and remain unpaid as a result of additional damages caused by Counter Defendants.

- 115. Counter Claimant is entitled to an order directing a foreclosure sale in the subject Property to abrogate any and all interest or claims that Counter Defendants might have in the subject Property.
- 116. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.

SEVENTH CAUSE OF ACTION

Waste Against Front Sight, Ignatius Piazza, and the VNV Trust Defendants

- 117. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 116 of this Counterclaim as though set forth fully herein at length.
 - 118. Counter Claimant LVD Fund (Lender) has a lien encumbering the subject Property.
 - 119. Counter Defendant Front Sight (Borrower) has possession of the Property.
- 120. Waste was committed to the property in bad faith, impairing its value, including but not limited to improperly using funds earmarked for development of the Property for the personal benefit of Counter Defendants Ignatius Piazza, Jennifer Piazza and the VNV Trust Defendants; selling unregistered securities which create substantial legal and financial liability to Front Sight, misappropriating Front Sight's assets for the personal benefit of Ignatius and Jennifer Piazza and other beneficiaries of the VNV Trust Defendants, and selling various instruments which include rights to Front Sight's resort property for highly reduced rates which further encumbers the Property, either directly or indirectly.
- 121. As a direct and proximate result of the waste committed by Counter Defendants, Counter Claimant has been injured in an amount to be proven at trial.
 - 122. Counter Claimant is entitled to treble damages under NRS 40.150.

Page 39 of 41

///

124. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.

PRAYER FOR RELIEF

WHEREFORE, all material allegations of Plaintiff's Second Amended Complaint having been denied, affirmative defenses having been stated, and counterclaims asserted, these responding Defendants now pray as follows:

- 1. That Plaintiff take nothing by way of its Second Amended Complaint on file herein and that the same be dismissed with prejudice;
- 2. For Judgment in favor of Counter Claimants against Counter Defendants, and each of them, in an amount in excess of Fifteen Thousand Dollars (\$15,000.00), subject to proof at trial;
 - 3 For appointment of a receiver over Counter Defendant Front Sight;
- 4. For an accounting from Counter Defendant Front Sight from October 6, 2016 forward, of any and all money paid and received, from all sources;
- 5. For an accounting from the Counter Defendant VNV Trusts from October 6, 2016 forward, of any and all money received from Counter Defendant Front Sight, and for all money distributed by the Counter Defendant Trusts since October 6, 2016.
- 6. For imposition of a constructive trust over the money transferred by Counter Defendant Front Sight to the VNV Trust Defendants in violation of Section 5.18 of the CLA, because the retention of said funds by the Counter Defendant Trusts against Counter Claimant LVD Fund's interests would be inequitable, and a constructive trust is essential to the effectuation of justice, and that restrictions be placed on such funds that limit their use to paying for the costs and expenses relating to completion of the Project.

Page **40** of **41**

1	7.	For injunctive relief pursuant to NRS 33.010 or as otherwise permitted by law or		
2	equity to enjoin Counter Defendant Front Sight from engaging in acts that further encumber			
3	the Property	and increase Counter Defendant Front Sight's actual or contingent liabilities in		
4	violation of t	violation of the CLA, including the sale of "credits," "points," "memberships," "certificates," or any		
5	other instrum	other instruments or products, including the sale of unregistered securities, that create contingent		
6	liabilities for Counter Defendant Front Sight and/or include the current or contingent right to conver			
7	said instruments directly or indirectly into ownership interests in Counter Defendant Front Sight or			
8	the Project.			
9	8.	For punitive damages pursuant to NRS 42.005.		
10	9.	For disgorgement of the funds misappropriated by Counter Defendant Front Sight and		
11	distributed to the other Counter Defendants;			
12	10.	For attorneys' fees and cost of suit incurred herein; and		
13	11.	For such other and further relief as the Court may deem just and proper.		
14	DATED this day of April, 2020.			
15	BAILEY * KENNEDY			
16				
17		By: <u>/s/ Andrea M. Champion</u> JOHN R. BAILEY		
18	JOHN K. BAILET JOSHUA M. DICKEY ANDREA M. CHAMPION			
19				
20				
21				
22				
23				
24				
25				
26				
27				
28		Page 41 of 41		

BAILEY * KENNEDY 8984 Sbanish Ridge Avenue Las Vegas, Neyada 89148-1302 702.562.8820

5/1/2020 4:46 PM Steven D. Grierson CLERK OF THE COURT 1 **MOT** JOHN R. BAILEY 2 Nevada Bar No. 0137 JOSHUA M. DICKEY 3 Nevada Bar No. 6621 ANDREA M. CHAMPION 4 Nevada Bar No. 13461 **BAILEY * KENNEDY** 5 8984 Spanish Ridge Avenue Las Vegas, Nevada 89148-1302 6 Telephone: 702.562.8820 Facsimile: 702.562.8821 7 JBailey@BaileyKennedy.com JDickey@BaileyKennedy.com 8 AChampion@BaileyKennedy.com C. KEITH GREER, ESQ. Cal. Bar. No. 135537 (Pro Hac Vice) 10 GREER AND ASSOCIATES, A PC 16855 West Bernardo Dr. Suite 255 11 San Diego, California 92127 Telephone: 858.613.6677 12 Facsimile: 858.613.6680 keith.greer@greerlaw.biz 13 Attorneys for Defendants 14 LAS VEGAS DEVELOPMENT FUND LLC; EB5 IMPACT CAPITAL REGIONAL CENTER 15 LLC; EB5 IMPACT ADVISORS LLC; ROBERT W. DZIUBLA; JON FLEMING; and 16 LINDA STANWOOD 17 DISTRICT COURT 18 CLARK COUNTY, NEVADA 19 20 HEARING DATE(S) ENTERED IN FRONT SIGHT MANAGEMENT LLC, a Case No. A-18-781084-B Nevada Limited Liability Company, ONYSSEY 21 Dept. No. XVI Plaintiff, 22 **DEFENDANT LAS VEGAS** VS. 23 DEVELOPMENT FUND, LLC'S LAS VEGAS DEVELOPMENT FUND LLC, a MOTION FOR CLARIFICATION ON 24 Nevada Limited Liability Company; et al, ORDER SHORTENING TIME 25 **HEARING REQUESTED** Defendant. 26 **DATE OF HEARING:** May 13, 2020 TIME OF HEARING: 10:30 a.m. 27 AND ALL RELATED COUNTERCLAIMS. 28 Page 1 of 10

Electronically Filed

Case Number: A-18-781084-B

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

DECLARATION OF ANDREA M. CHAMPION, ESQ.

- I, Andrea M. Champion, hereby declare as follows:
- 1. I am over eighteen (18) years of age and a resident of Clark County, Nevada.
- 2. I am counsel for Defendant/Counterclaimant Las Vegas Development Fund, LLC "(LVD Fund") in the above-captioned action.
- I have personal knowledge of and am competent to testify to the facts contained in 3. this Declaration. If called to do so, I would competently and truthfully testify to all matters set forth herein, except for those matters stated to be based upon information and belief.
- 4. I make this declaration in support of LVD Fund's Motion for Clarification (the "Motion").
- 5. On April 3, 2020, LVD Fund filed a Motion for Leave to Amend its Counterclaim (the "Motion for Leave to Amend.") Attached as Exhibit "A" to the Motion for Leave to Amend was a proposed Answer and First Amended Counterclaim (the "Exhibit").
- 6. Front Sight objected to certain information in the Exhibit being filed in the public domain, contending that it should be treated as confidential information (and not publicly available).
- 7. LVD Fund disagrees and contends that the Exhibit does not contain confidential or proprietary information.
- 8. The parties decided to remove the Exhibit from the public record and replace it with a redacted copy while they had the chance to brief this issue for the Court.
- 9. On April 17, 2020, the parties filed a Stipulation and Order to Replace Exhibit "A" to Defendants' Motion for Leave to Amend the Counterclaim.
- 10. Because the parties agreed that it was appropriate for the Court to decide whether the Exhibit needs to be redacted at, or near, the same time the Motion for Leave to Amend is heard, they stipulated that LVD Fund would file its Motion addressing the redaction issue on shortened time.
 - 11. The Motion for Leave to Amend is currently scheduled to be heard on May 6, 2020.
- 12. LVD Fund understands that this Court's May 6, 2020 hearing calendar is already full but the parties will be before this Court again the following week, on May 13, 2020.

/// 28

1	13.	Therefore, LVD Fund respectfully asks that this Motion be heard on May 13, 2020, or
2	at the Court's	s convenience.
3	14.	If this Motion is heard in the normal course of business, it is unlikely to be heard until
4	well after the	e Court decides LVD Fund's Motion for Leave to Amend.
5	15.	Assuming the Court grants the Motion for Leave to Amend, the Court's
6	determination	n of the issue presented in this Motion is necessary before LVD Fund files it Answer
7	and Amende	d Counterclaim.
8	16.	Therefore, an order shortening time is necessary.
9	17.	This declaration and request for an order shortening time is made in good faith, not
10	for purposes	of delay, and in compliance with EDCR 2.26.
11	D	Pated this 29 th day of April, 2020.
12		
13		_/s/ Andrea M. Champion
14		ANDREA M. CHAMPION
15		
16		ORDER SHORTENING TIME
17	It app	pearing to the satisfaction of the Court, and good cause appearing therefore, and
18	established b	by affidavit of counsel, it is hereby ORDERED that the time for the hearing on
19	DEFENDA	NT LAS VEGAS DEVELOPMENT FUND LLC'S MOTION FOR
20	CLARIFICA	ATION on file herein may be shortened and will be heard on the 13th day of May,
21	2020, at the l	hour of 10:30am, in Dept. 16 of the Eighth Judicial District Court.
22	Dated	this 1st day of April, 2020.
23		4
24		Junot C. William
25		DISTRICT COURT JUDGE
26		CG
27		
28		
		Page 3 of 10
		- -

I. INTRODUCTION

As this Court is well aware, one of the allegations in this case is that the Piazzas wrongfully diverted EB-5 investor funds intended for the Front Sight construction project for their own use. For the past two years, the parties have debated whether that allegation can be included in publicly filed briefs or whether the allegations and all supporting facts—including *how much* the Piazzas have transferred to their two family trusts—must be hidden from the public. Despite the existence of a Protective Order, multiple motions, oppositions, and Court orders, the parties still lack clear direction as to how to move forward. Therefore, LVD Fund brings this Motion to finally address this issue. The question presented is simple: may LVD Fund reference its allegations of fraudulent transfers in publicly filed documents or may Front Sight continue to hide its wrongdoing from the public?

The Court has twice decided this issue, and twice arrived at a different result. While LVD Fund respects that the Court is reluctant to bind the parties as to what may be filed in unredacted form during the remainder of this litigation, a decision is necessary to: (i) assist the parties in moving forward without engaging in unnecessary disputes; (ii) to prevent Front Sight from continuing to lodge unmeritorious attacks against LVD Fund for its alleged willful violations of the Protective Order; and (iii) to aid the Court in preserving valuable judicial resources by hearing the parties' dispute over the narrow issue presented. For these reasons and those set forth below, LVD Fund requests that the Court grant this Motion for Clarification.

II. THE PROCEDURAL HISTORY OF THIS ISSUE

At the inception of this litigation, Front Sight filed a Motion for Protective Order with this Court. LVD Fund quickly filed a Notice of Non-Opposition in response, agreeing that a Protective Order would benefit both parties.

On February 6, 2019, LVD Fund filed a Motion for Appointment of Receiver. Within its Motion, LVD Fund referenced Front Sight's tax returns, which were attached in redacted form as exhibits to the Motion. LVD Fund's Motion prompted Front Sight to file a Motion to Seal and/or Redact on February 15, 2019, asking that the court seal the tax records *and* redact all of LVD Fund's references to the tax records due to the "compelling privacy interest" and "proprietary business

Page 4 of 10

A year later, LVD Fund filed its Opposition to Jennifer Piazza and the VNV Trusts' Motion for Summary Judgment. Consistent with the Court's March 18, 2019 Order, LVD Fund filed excerpts from Front Sight's corporate tax returns *under seal* but referenced the amount of loans issued and distributions made to Front Sight shareholders (*i.e.*, Ignatius Piazza and the Dynasty Trusts) within the Opposition without redaction. On February 11, 2020, Front Sight filed its second Motion to Seal and/or Redact with the Court—again, asserting that the relevant portions of LVD Fund's Opposition should be redacted due to the "compelling privacy or safety interest," the "proprietary business information," and "Counterdefendants' personal information." (Mot. to Seal, 7:18–21, Feb. 11, 2020). LVD Fund filed its Opposition on February 14, 2020. The Court held a hearing on the Motion, and on February 18, 2020, issued a Minute Order directing that the Motion be "GRANTED for *time being*." (Min. Order, Feb. 18, 2020) (emphasis added).

On April 3, 2020, LVD Fund filed a Motion for Leave to Amend the Countercomplaint. Attached as Exhibit "A" to the Motion was a proposed Answer and First Amended Counterclaim (the "Proposed Amended Counterclaim"). Paragraphs 25, 47, and 76-81 of the Proposed Amended Counterclaim discussed LVD Fund's claims. Specifically, paragraph 25 of the Proposed Amended Counterclaim quoted Section 1.7 of the CLA and discussed the amount of loan proceeds actually spent on the construction of the Project (as discovered by LVD Fund in October 2018) but *did not* reference loan disbursements to Counterclaims. Paragraph 47 of the Proposed Amended Counterclaim similarly cited to Section 5.18 of the CLA and then discussed how the loan proceeds

Page 5 of 10

The Order issued by the Court on February 27, 2020 simply granted the Motion to Seal without reflecting the Court's intended temporary scope. (Order Granting Mot. to Seal, 2:1–3, Feb. 27, 2020).

disbursed to Counterdefendants violated Section 5.18 of the CLA. Finally, in paragraphs 76 through 81 of the Proposed Amended Counterclaim, LVD Fund set forth allegations of Front Sight's fraudulent transfers, including the amounts of those transfers from 2017 through 2019. LVD Fund believed, based on the Court's March 18, 2019 Order, that it had the right to publicly file this information within its Proposed Amended Counterclaim.

Front Sight not only took issue with the fact that LVD Fund had not filed the Proposed Amended Counterclaim under seal, but has now used the filing of the Proposed Amended Counterclaim as the basis to refuse to provide LVD Fund with the information necessary to support the first EB-5 investor's I-829 Petition to the USCIS. The parties agreed to temporarily resolve the matter by filing a stipulation and order to replace the Proposed Amended Counterclaim with a redacted version (redacting paragraphs 25, 47, and 76-81 in their entirety). (Stip. and Order to Replace Exhibit "A", ¶ 4, Apr. 17, 2020). Both parties reserved the right to assert their respective positions in future briefings and agreed that further direction from this Court was necessary to ensure the efficiency of future proceedings. (*Id.*).

LVD Fund now brings this Motion for Clarification. In doing so, LVD Fund respectfully requests that the Court define the contours of the requirement to seal and redact references to Counterdefendants' financial information in future filings.

III. ARGUMENT

A. Counterdefendants' Tax Returns Are Not Privileged.

While information contained in tax returns contains sensitive information, Nevada has explicitly refused to recognize a privilege for tax returns. *Agwara v. State Bar of Nev.*, 2017 133 Nev. Adv. Rep. 96, 406 P.3d 488, 493 (2017) ("[T]his state does not recognize a privilege for tax returns..."); *Hetter v. Eighth Jud. Dist. Ct.*, 110 Nev. 513, 520, 874 P.2d 762, 766 (1994); *McNair v. Eighth Jud. Dist. Ct.*, 110 Nev. 1285, 1289–90, 885 P.2d 576, 579 (1994) (rejecting argument that tax returns are privileged information); *see also Copper Sands Homeowners Ass'n v. Copper Sands Realty, LLC*, No. 2:10-cv-00510-GMN-GWF, 2012 U.S. Dist. LEXIS 44583, at *12 (D. Nev. Mar. 29, 2012) ("Under Nevada law, tax returns are not themselves privileged."). Instead, courts look to whether the information contained in the tax returns is relevant. *Hetter*, 110 Nev. at 520, 874 P.2d

Page 6 of 10

at766; *McNair*, 110 Nev. at 1290, 885 P.2d at 579. Thus, the extent to which LVD Fund may use Counterdefendants' tax returns depends on the relevance of the information contained therein.

B. <u>Counterdefendants' Financial Information Is Unquestionably Relevant.</u>

One of the many questions presented in this case is whether Front Sight and Counterdefendants breached the CLA by continuously diverting funds away from the EB-5 construction project and—contrary to the intent of the CLA—used those funds to pay for unrelated Front Sight costs, and to line the Piazza's pockets. The evidence, including Counterdefendants' tax returns, answers that question in the affirmative.

Information about Front Sight and the Piazza's taxes and bank records is relevant to show the Court, Front Sight investors, and the Nevada community, that the Piazzas did, in fact, improperly utilize Front Sight's assets for the lining of their own pockets. In this case, Front Sight insists that it lacked the funds necessary to complete the Project, while discovery has revealed that it was funneling millions of dollars to Counterdefendants for personal use. Front Sight is not the victim it has publicly portrayed itself to be. Front Sight should not be able to hide evidence of its malfeasance from the public.

Counterdefendants' tax returns show that the Piazzas diverted EB-5 loan proceeds to their family trusts and to the payment of pre-existing Front Sight debt, and the bank records show when these fraudulent transfers took place. This information is therefore relevant, and the Court should permit LVD Fund to reference the information in unreducted form.

C. Public Interest Demands That This Information Be Made Public.

Nevada Supreme Court Rules, Part VII, Rule 3(4) specifies that information may be sealed or redacted when "sealing or redaction is justified or required by [an] identified compelling circumstance." Nevada SCR 3.4(h). In sealing records, courts must use "the least restrictive means of sealing, and seal only as much of the record as is necessary to protect any confidential information." *Takeda Pharms. Am., Inc. v. Eighth Jud. Dist. Ct.*, No. 64614, 2013 Nev. Unpub. LEXIS 2000, at *1 (Nev. Dec. 31, 2013).

When deciding whether to seal or redact a record, courts begin with a "presumption in favor of public access to records and documents filed in [the] court." *Howard v. State*, 128 Nev. 736, 744

Page 7 of 10

291, P.3d 137, 142 (2012). A party may overcome this presumption only by "demonstrating sufficient grounds for denying access." *Id.*, 291 P.3d at 142. While all documents filed with the court are presumed to be publicly available, this presumption is especially strong in regards to the pleadings and motions filed with the court. Most jurisdictions have "recognized that certain documents play no role in the performance of [a court's] ... functions, ... [but], *a complaint—which forms the basis of a civil action and invokes the jurisdiction of the Court—is not such a document*. *In re Eastman Kodak Company's Application*, 2010 U.S. Dist. LEXIS 144955, at *1–2 (S.D.N.Y. June 15, 2020) (emphasis added). Indeed, "[t]here is a presumptive right of public access to pretrial motions of a nondiscovery nature, and the material filed in connection therewith." *Newton v. Hess Oil V.I. Corp.*, 68 V.I. 467, 476 (V.I. 2018).

The party seeking to seal or redact must show that "disclosure will work a clearly defined and serious injury' to the party seeking sealing. The presumption in favor of openness may be overcome only by a movant's showing that closure is 'essential to preserve higher values and is narrowly tailored to serve that interest." *Id.* at 476 (quoting *Beberman v. U.S. Dept. of State*, No. 2014-020, 2014 WL 12768904, *1 (D.V.I. May 21, 2014)). "Broad allegations of harm, unsubstantiated by specific examples or articulated reasoning, do not support a good cause showing." *Gillete Co. v. Dollar Shave Club*, No. 15-1158-LPS-CJB, 2017 U.S. Dist. LEXIS 148775, at *4–*5 (D. Del. Sept. 6, 2017) (quotation marks omitted). Likewise, "general assertions of 'privacy' and 'embarrassment' are insufficient to overcome the presumption of public access to judicial proceedings." *Hollinger Int'l Inc. v. Hollinger Inc.*, No. 04 C 698, 2005 U.S. Dist. LEXIS 30420, at *11 (N.D. Ill. Jan. 19, 2005); *see also Cipollone v. Liggett Group, Inc.*, 785 F.2d 1108, 1121 (3d Cir. 1986) (finding that "it may be especially difficult for a business enterprise, whose primary measure of well-being is presumably monetizable, to argue for a protective order on [the] ground[s of embarrassment].").

Here, none of the reasons Front Sight has proffered in support of redacting LVD Fund's factual allegations are sufficient to overcome the presumption of public access. Simply asserting that Counterdefendants' financial documents are "proprietary" and have a "compelling privacy interest" falls woefully short of the "specific examples" and "articulated reasoning" necessary to

Page 8 of 10

remove any and all reference to Counterdefendants' conduct from public access. LVD Fund's references to the specific amounts of money transferred from Front Sight to Counterdefendants is a far cry from "publish[ing] [confidential financial records] in the public domain" as originally feared by Front Sight. (Tr. Hr'g on Mot. for Protective Order, 4:19–20, Oct. 31, 2018)

LVD Fund is not seeking to splash Front Sight's trade secrets across the face of its pleadings and motions. Rather, LVD Fund is seeking to reference specific, relevant evidence regarding the solvency of Front Sight and the appropriateness of transfers made to the Piazza family within its public filings including, but not limited to, the Proposed Amended Counterclaim. LVD Fund will continue to file Counterdefendants' financial records under seal, thereby retaining the confidentiality of the documents themselves. However, the Court should permit LVD Fund to openly discuss the *evidence* of Counterdefendants' financial misconduct.

Counterdefendants' financial information is relevant to the issues in this case and relevant to the public's understanding of the Court's adjudication of this matter. *See, e.g., Gillette Co. v. Dollar Shave Club, Inc.*, No. 2017 U.S. Dist. LEXIS 148775, at *10–*11 (D. Dela. Sept. 6, 2017) (declining to redact documents where the "categories of information reflect the very crux of the parties' legal dispute."). Further, Front Sight willingly breached the CLA and defrauded LVD Fund into loaning Front Sight \$6,375,000 in EB-5 investment funds, and willingly sued LVD Fund, but is now seeking to avoid the repercussions of those choices. The judicial system does not exist at the pleasure of Front Sight and the Piazzas. Front Sight must "pay to play," and an open discussion of their financial information is essential to the resolution of this case. *See In re Eastman Kodak Company's Application*, 2010 U.S. Dist. LEXIS 144955, at *1–3 (S.D.N.Y. June 15, 2020) (chastising Kodak for its attempt to shield itself from counterclaims via sealing documents, and in doing so, turn the court "into its own arbitral forum—where Kodak sets the rules.").

IV. CONCLUSION

The law is clear: all documents filed with the court, especially pleadings and substantive motions, are to be publicly available. While public access is not without limit, it should only be restricted in very few situations. This is not one of those situations. While LVD Fund has no objection to filing Counterdefendants' confidential *documents* under seal, it objects to redacting all

Page 9 of 10

references to the damning evidence of Counterdefendants' financial malfeasance in pleadings and 1 2 motions. Such information must be publicly available, both to expose Counterdefendants' 3 misconduct and to provide the public insight into the Court's adjudication of this matter. Should the 4 Court find that some reduction is necessary to preserve Counterdefendants' privacy interests, LVD Fund requests that the redaction be limited to specific dollar amounts with the understanding that 5 general dollar amounts (i.e. hundreds, thousands, millions, etc. of dollars) be permitted, and not the 6 7 entirety of LVD Fund's allegations (as is currently redacted from the Proposed Amended Counterclaim). 8 9 For these reasons and those set forth above, LVD Fund respectfully requests that the Court grant the Motion for Clarification and determine that LVD Fund may file unredacted references to 10 11 Counterdefendants' financial information in the future. 12 DATED this 29th day of April, 2020. 13 **BAILEY KENNEDY** 14 15 By: /s/ Andrea M. Champion JOHN R. BAILEY 16 JOSHUA M. DICKEY ANDREA M. CHAMPION 17 Attorneys for Defendants 18 LAS VEGAS DEVELOPMENT FUND LLC; EB5 IMPACT CAPITAL 19 REGIONAL CENTER LLC; EB5 IMPACT ADVISORS LLC; ROBERT W. 20 DZIUBLA; JON FLEMING; and LINDA **STANWOOD** 21 22 23 24 25 26 27 28 Page 10 of 10

Electronically Filed 5/11/2020 3:25 PM Steven D. Grierson CLERK OF THE COURT

1 **OPPM** John P. Aldrich, Esq. 2 Nevada Bar No. 6877 Catherine Hernandez, Esq. 3 Nevada Bar No. 8410 Jamie S. Hendrickson, Esq. 4 Nevada Bar No. 12770 ALDRICH LAW FIRM, LTD. 5 7866 West Sahara Avenue Las Vegas, Nevada 89117 6 Telephone: (702) 853-5490 Facsimile: (702) 227-1975 7 Attorneys for Plaintiff/Counterdefendants 8 EIGHTH JUDICIAL DISTRICT COURT 9 **CLARK COUNTY, NEVADA** 10 FRONT SIGHT MANAGEMENT LLC, a Nevada Limited Liability Company, CASE NO.: A-18-781084-B 11 DEPT NO.: 16 Plaintiff, 12 **OPPOSITION TO DEFENDANT** VS. 13 LAS VEGAS DEVELOPMENT LAS VEGAS DEVELOPMENT FUND LLC, a **FUND LLC'S MOTION FOR** 14 Nevada Limited Liability Company; et al., **CLARIFICATION ON ORDER SHORTENING TIME** 15 Defendants. 16 AND ALL RELATED COUNTERCLAIMS. 17 18 Plaintiff FRONT SIGHT MANAGEMENT LLC ("Plaintiff") by and through its 19 attorneys, John P. Aldrich, Esq., Catherine Hernandez, Esq., and Jamie S. Hendrickson, Esq., of 20 the Aldrich Law Firm, Ltd., hereby opposes Defendant Las Vegas Development Fund LLC's 21 Motion for Clarification on Order Shortening Time. 22 23 24

This Opposition is made and based on the attached memorandum of points and authorities and supporting documentation, the papers and pleadings on file in this action, and any oral argument this Court may allow.

DATED this 11th day of May, 2020.

ALDRICH LAW FIRM, LTD.

/s/ Jamie S. Hendrickson
John P. Aldrich, Esq.
Nevada Bar No. 6877
Catherine Hernandez, Esq.
Nevada Bar No. 8410
Jamie S. Hendrickson, Esq.
Nevada Bar No. 12770
7866 West Sahara Avenue
Las Vegas, NV 89117
Tel (702) 853-5490
Fax (702) 226-1975
Attorneys for Plaintiff/Counterdefendant

MEMORANDUM OF POINTS AND AUTHORITIES

I.

INTRODUCTION

Defendants bring the instant Defendant Las Vegas Development Fund, LLC's Motion for Clarification on Order Shortening Time ("Motion for Clarification"), as a veiled Motion for Reconsideration on this Court's February 20, 2020, Minute Order granting Front Sight's Motion to Seal. The formal Order was entered on February 27, 2020 and Notice of Entry was filed on March 2, 2020. (See Notice of Entry of Order Granting Motion to Seal and/or Redact Portions of Defendants' Oppositions to Jennifer Piazza and the VNV Trusts' Motions for Summary Judgment to Protect Confidential Financial Information, attached hereto as **Exhibit 1.**)

Defendants seek to publish Front Sight's confidential tax and financial information in future motions and pleadings to allegedly prevent "Front Sight from hid[ing] its wrongdoing from the public." (Motion for Clarification, p. 4.) Front Sight's alleged financial malfeasance (so called by Defendants) concerns a series of transfers from Front Sight to the Dynasty Trusts, which Front Sight maintains are shareholder distributions to Dr. Piazza. Front Sight has nothing to hide, but does not wish to have its private financial information spread to the world at large as Defendants attempt to try this case in the forum of public opinion by repeatedly and baselessly accusing Dr. Piazza of misconduct. In reality, Front Sight is a privately-held, single member limited liability company. Dr. Piazza is the only member and manager of this limited liability company, which is a pass-through entity. As a pass-through entity, all of Front Sight's income passes through to Dr. Piazza for tax purposes. Therefore, Front Sight's funds are Dr. Piazza's funds. He cannot fraudulently transfer his own money to himself.

Defendants allege that as creditors of Front Sight via the Construction Loan Agreement ("CLA"), they were defrauded when Front Sight transferred funds to the Dynasty Trusts. Defendants allege that the transfers were fraudulent transfers under NRS 112.180 and NRS 112.190 because Front Sight was allegedly insolvent at the time of the transfers and because the transfers were allegedly made to related parties in violation of the CLA and without adequate consideration. Defendants further allege that the transfers consisted of EB-5 loan proceeds. None of these assertions is true. To the contrary, Front Sight had assets in excess of its liabilities at the time of the transfers. Front Sight posted net income for the years before and after the transfers. Additionally, LVDF is a secured creditor who suffered no possible harm from the transfers.

Finally, Supreme Court Rules require that Defendants redact or seal any references to Front Sight's private financial or tax information. Defendants attempt to argue that Front Sight must demonstrate that it will suffer a compelling harm should its financial and tax information be disclosed that outweighs the public's right to access legal documents. However, the very text of the Supreme Court Rules provide for redaction or sealing of tax information.

Because Defendants' Motion for Clarification is nothing more than a veiled request for an untimely motion for reconsideration without meeting the legal standard for such a motion, because Front Sight's private tax and financial information is not relevant, and because Front Sight's private tax and financial information must be sealed as a matter of law, Defendants' Motion for Clarification should be denied in its entirety.

II.

LEGAL ARGUMENT

A. FRONT SIGHT'S PRIVATE TAX AND FINANCIAL INFORMATION SHOULD BE REDACTED AS A MATTER OF LAW

Nevada Supreme Court Rules, Part VII, pertain to the sealing and redacting Court records. Pursuant to Rule 3(4):

The court may order the court files and records, or any part thereof, in a civil action to be sealed or redacted, provided the court makes and enters written findings that the specific sealing or redaction is justified by identified compelling privacy or safety interests that outweigh the public interest in access to the court record. The parties' agreement alone does not constitute a sufficient basis for the court to seal or redact court records. . .

Rule 3(4) also sets forth eight specific instances that warrant the sealing of records:

(b) The sealing or redaction furthers an order entered under NRCP 12(f) or JCRCP 12(f) or a protective order entered under NRCP 26(c) or JCRCP 12(c);

. . . .

(f) The sealing or redaction includes medical, mental health, or tax records;

. . . .

- (g) The sealing or redaction is necessary to protect intellectual proprietary or property interests such as trade secrets as defined in NRS 600A.030(5);
- (h) The sealing or redaction is justified or required by another identified compelling circumstance.

By the very letter of Supreme Court Rule 3, Front Sight's tax and financial information must be sealed or redacted as a matter of law. All of the information that Defendants seek to publish in future motions and pleadings comes from either Front Sight's tax returns or Front Sight's bank statements. Pursuant to S.C.R. 3.4(f), Defendants must seal any information derived from Front Sight's tax returns. Pursuant to S.C.R. 3.4(b), Defendants must seal any information obtained within discovery because it is covered by the Protective Order in this matter.

One has to wonder why Defendants insist on providing the specific financial information from either tax returns or bank statements if not to harass Front Sight when a more general presentation would suffice without disclosing private information.

Defendants present several cases that discuss a balancing test whereby the party seeking to seal or redact records must specify the compelling interest and specific harm that it could suffer if the information is redacted to overcome the public's right to full access to legal proceedings and court records. However, those cases are irrelevant because none specifically analyze S.C.R. 3 or pertain to redaction or sealing of tax records, and S.C.R. 3 specifically states that tax records overcome the presumption of the public's right to access information. Therefore, this Court should deny Defendants' Motion for Clarification by affirming its March 2, 2020,

Order Granting Motion to Seal, requiring Defendants' to redact Front Sight's confidential tax and financial information.

B. FRONT SIGHT'S TAX RETURNS ARE NOT RELEVANT

Defendants contend that Front Sight's private tax and financial information is relevant to their breach of contract and proposed fraud and fraudulent transfer claims. (Motion for Clarification, pp. 4-5.) Defendants make three claims: 1) Front Sight is insolvent; 2) Dr. Piazza transferred funds from Front Sight to the Dynasty Trusts; and, 3) at least some of the funds transferred from Front Sight to the Dynasty Trusts comprised EB-5 loan proceeds. *Id.* Only one of these claims is true: that Dr. Piazza made shareholder distributions to the Dynasty Trusts, which he is entitled to do. Defendants have offered no evidence to establish that Front Sight is insolvent. The only expert affidavit offered by Defendants to date only conclusively asserts that Front Sight is insolvent without offering any analysis for how that conclusion was determined. If Front Sight is not insolvent or did not make the alleged transfers with intent to defraud creditors, then Front Sight's and financial information is not relevant to any claim or defense in this matter.

1. Front Sight Is Not Insolvent

Front Sight cannot be insolvent for two important reasons. First, Front Sight's tax returns do not present its assets at fair market value. When Front Sight's real estate holdings and assets are calculated at fair market value, Front Sight's assets far exceed its liabilities. Even if Front Sight were insolvent in 2016-2018 based upon the information contained in Front Sight's tax returns, Front Sight's financial information from 2014-2016 remained largely consistent. On this basis, Front Sight's financial expert Douglas S. Winters, CPA, opined that Front Sight would have been insolvent in 2016 when LVDF loan Front Sight funds under the CLA. (*See* February 27, 2020 Expert Report of Douglas S. Winters, CPA, pp. 2-4, filed under seal as Exhibit 4 to

Piazza (filed March 3, 2020).)

Even if Front Sight's assets did not exceed its liabilities, Front Si

Even if Front Sight's assets did not exceed its liabilities, Front Sight has posted net income in 2016, 2017, and 2018, the years in which Defendants claim that Dr. Piazza fraudulently transferred funds to the Dynasty Trusts. A business that posts net income is not insolvent where its income is sufficient to meet its ongoing financial obligations. Because Front Sight's assets exceed its liabilities when measured a fair market value and because Front Sight consistently posts net income, it is not insolvent. One has to wonder how Front Sight remains in business in 2020 if it were insolvent in 2017 as Defendants contend. Finally, Front Sight has made all of its interest payments pursuant to the CLA and continues to do so. Because Front Sight is not insolvent, then the only other basis for Defendants' claim for fraudulent transfers rests upon whether Front Sight transferred money with intent to defraud creditors.

Reply to Opposition to Motion for Summary Judgment as to the Counterclaims Against Jennifer

2. The Transfers to the Dynasty Trusts Were Not Fraudulent

Defendants have presented no evidence to date that Front Sight fails to meet its financial obligations. Front Sight continues to make interest payments to LVDF. The CLA allows Front Sight to make transfers to shareholders where the transfers do not impair its ability to meet its obligations under the agreement. *See* Evid. Hrg. Exhibit 33, Section 5.18, at 0227. Even Defendants do not assert that Front Sight was in financial default of the CLA when it made transfers to the Dynasty Trusts in 2016-2018.

Moreover, Front Sight's alleged transfers to the Dynasty Trusts cannot defraud LVDF in any way because LVDF holds a security interest in the property. Front Sight cannot have made the transfers with intent to defraud LVDF because LVDF can foreclose on the Project to collect on the debt owed. If Front Sight transferred every penny of its profits to Dr. Piazza, LVDF

would still have a security interest against the land pursuant to the Construction Deed of Trust. No degree of insolvency would thwart LVDF's security interest, as it is undisputed that the value of the land upon which the project rests would not change based upon Front Sight's financial condition.

Furthermore, the alleged transfers could not have been made to thwart Front Sight's obligation to repay the EB-5 loan because the CLA prevents Front Sight from prepaying the loan until the EB-5 investors' I-829 petitions are approved by USCIS. *See* Evid. Hrg. Exhibit 33, Section 1.3, at 0206. Because Front Sight is not allowed to repay the principal balance on the EB-5 loan, pursuant to language in the CLA, is not delinquent on its interest payments, and cannot thwart LVDF's security interest in the project, Front Sight's transfers to the Dynasty Trusts cannot be fraudulent as a matter of law.

3. Front Sight's Expenditures on the Project Far Exceed EB-5 Proceeds

On January 22, 2020, this Court made a finding that Front Sight spent far more on the project than LVDF loaned in EB-5 proceeds. (*See* Findings of Fact, Conclusions of Law and Order Denying Defendant Las Vegas Development Funds, LLC's Motion to Dissolve Temporary Restraining Order and Appoint a Receiver, pp. 6, 7, attached hereto as **Exhibit 2**.) Therein the Court stated,

Based upon this *uncontroverted evidence*, the Court finds Front Sight's expenses on the Project *far exceed* the amount of the loan from Defendant LVDF.

Regarding alleged Breach #1, the Court concludes that Front Sight's expenses on the Project far exceed the amount of the loan from Defendant LVDF has **Defendant LVDF**'s assertion that Front Sight improperly used loan proceeds is without merit, and consequently, LVDF has failed to establish an alleged breach.

Id. at pp. 6, 7 (emphasis added). The Court has already made this finding. Accordingly, it is the law of the case. There is no argument or evidence that Front Sight transferred EB-5 loan

proceeds to the Dynasty Trusts. No expert has supported that contention, and the Court specifically found that Defendants' very assumption that Front Sight misused loan proceeds is without merit. Therefore, Defendants have no reason to reference specific transfers from Front Sight to the Dynasty Trusts. If anything, these transfers are shareholder distributions which are not relevant to this matter in any way.

Defendants cannot even assert that Front Sight's transfers to the Dynasty Trusts impaired Front Sight's ability to repay the loan under the CLA because the CLA prevents Front Sight from repaying the loan where investors' I-829 petitions have not received final adjudication. *See* Evid. Hrg. Exhibit 33, Section 1.3, at 0206. At this time, no investors have even submitted I-829 petitions. Once the First Investor submits his/her I-829 Petition later this month, it may not receive final adjudication for months. Therefore, Front Sight cannot begin to repay the loan.

Because Front Sight is neither insolvent, did not make transfers with intent to defraud LVDF, because Front Sight is not allowed to repay the principal balance on the CLA at this time, pursuant to language in the CLA, and because LVDF is a secured creditor, Front Sight's private tax and financial information is not relevant in this matter.

C. DEFENDANTS' MOTION FOR RECONSIDERATION IS UNTIMELY

Defendants' Motion amounts to a Motion for Reconsideration of this Court's February 20, 2020 Minute Order. The formal Order was filed on February 27, 2020 and Notice of Entry was filed on March 2, 2020. (**Exhibit 1.**) EDCR 2.24(b) requires that a motion to reconsider must be filed within ten (10) days of entry of the order about which a party seeks reconsideration. Defendants' Motion was filed long after that deadline expired. The Court should not consider this Motion because it is untimely. But even if the Court decides to consider

the Motion, it should be denied because the Court has already ruled on the sealing of the subject information, and that is now law of the case for this matter.

III.

CONCLUSION

Based on the foregoing, Plaintiff respectfully requests that the Court deny Defendant LVDF's Motion for Clarification in its entirety by preserving the status quo and requiring all references to Front Sight's tax information to be redacted in all future motions and/or pleadings.

DATED this 11th day of May, 2020.

ALDRICH LAW FIRM, LTD.

/s/ Jamie S. Hendrickson
John P. Aldrich, Esq.
Nevada Bar No. 6877
Catherine Hernandez, Esq.
Nevada Bar No. 8410
Jamie S. Hendrickson, Esq.
Nevada Bar No. 12770
7866 West Sahara Avenue
Las Vegas, NV 89117
Tel (702) 853-5490
Fax (702) 226-1975
Attorneys for Plaintiff/Counterdefendant

1 **CERTIFICATE OF SERVICE** 2 I HEREBY CERTIFY that on the 11th day of May, 2020, I caused the foregoing 3 OPPOSITION TO DEFENDANT LAS VEGAS DEVELOPMENT FUND LLC'S 4 MOTION FOR CLARIFICATION ON ORDER SHORTENING TIME to be electronically 5 filed and served with the Clerk of the Court using Wiznet which will send notification of such 6 filing to the email addresses denoted on the Electronic Mail Notice List, or by U.S. mail, postage 7 prepaid, if not included on the Electronic Mail Notice List, to the following parties: 8 Anthony T. Case, Esq. Kathryn Holbert, Esq. FARMER CASE & FEDOR 2190 E. Pebble Rd., Suite #205 10 Las Vegas, NV 89123 11 C. Keith Greer, Esq. 16855 West Bernardo Drive, Suite 255 12 San Diego, CA 92127 13 John R. Bailey, Esq. Joshua M. Dickey, Esq. 14 Andrea M. Champion, Esq. **BAILEY KENNEDY** 15 8984 Spanish Ridge Avenue Las Vegas, NV 89148 16 Attorneys for Defendants 17 18 19 /s/ T. Bixenmann An employee of ALDRICH LAW FIRM, LTD. 20 21 22 23 24

EXHIBIT 1

EXHIBIT 1

Electronically Filed 3/2/2020 9:39 AM Steven D. Grierson CLERK OF THE COURT

1	NEO	Otenas.
2	John P. Aldrich, Esq. Nevada Bar No. 6877	
2	Catherine Hernandez, Esq.	
3	Nevada Bar No. 8410	
	ALDRICH LAW FIRM, LTD.	
4	7866 West Sahara Avenue Las Vegas, NV 89117	
5	Telephone: (702) 853-5490	
	Facsimile: (702) 227-1975	
6	Attorneys for Plaintiff/Counterdefendants	
7	EIGHTH JUDICIAL D	ISTRICT COURT
	CLARK COUNTY	
8	EDON'E CICHE MANAGEMENT LLC	
9	FRONT SIGHT MANAGEMENT LLC, a Nevada Limited Liability Company,	CASE NO.: A-18-781084-B
	Nevada Emitted Elability Company,	DEPT NO.: 16
10	Plaintiff,	
11	VS.	NOTICE OF ENTRY OF ORDER
12	LAS VECAS DEVELODMENT ELIND LLC	
12	LAS VEGAS DEVELOPMENT FUND LLC, a Nevada Limited Liability Company; et al.,	
13		
1.4	Defendants.	
14		
15	AND ALL RELATED COUNTERCLAIMS.	
16		
16		
17	PLEASE TAKE NOTICE that an Order Gra	anting Motion to Seal and/or Redact Portions
18	of Defendants' Oppositions to Jennifer Piazza and	nd the VNV Trusts' Motions for Summary
19	Judgment to Protect Confidential Financial Informa	tion was entered by the Court in the above-
20	///	
21	///	
22	///	
23	///	
24		
4		
	1	
	Case Number: A-18	3-781084-B
l	J Case Mulliber. A-10	

captioned action on the 27th day of February, 2020, a true and correct copy of which is attached 1 2 hereto. DATED this 2nd day of March, 2020. 3 4 ALDRICH LAW FIRM, LTD. 5 /s/ John P. Aldrich John P. Aldrich, Esq. Nevada Bar No. 6877 6 Catherine Hernandez, Esq. 7 Nevada Bar No. 8410 7866 West Sahara Avenue 8 Las Vegas, Nevada 89117 Telephone: (702) 853-5490 9 Facsimile: (702) 227-1975 Attorneys for Plaintiff/Counterdefendants 10 11 **CERTIFICATE OF SERVICE** I HEREBY CERTIFY that on the 2nd day of March, 2020, I caused the foregoing 12 NOTICE OF ENTRY OF ORDER to be electronically filed and served with the Clerk of the 13 14 Court using Wiznet which will send notification of such filing to the email addresses denoted on 15 the Electronic Mail Notice List, or by U.S. mail, postage prepaid, if not included on the 16 Electronic Mail Notice List, to the following parties: 17 Anthony T. Case, Esq. Kathryn Holbert, Esq. 18 FARMER CASE & FEDOR 2190 E. Pebble Rd., Suite #205 19 Las Vegas, NV 89123 20 C. Keith Greer, Esq. 16855 West Bernardo Drive, Suite 255 21 San Diego, CA 92127 Attorneys for Defendants 22 /s/ T. Bixenmann 23 An employee of ALDRICH LAW FIRM, LTD. 24

Electronically Filed 2/27/2020 8:58 AM Steven D. Grierson CLERK OF THE COURT

ORDR

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

John P. Aldrich, Esq.

Nevada Bar No. 6877

Catherine Hernandez, Esq.

Nevada Bar No. 8410

ALDRICH LAW FIRM, LTD.

7866 West Sahara Avenue

Las Vegas, NV 89117

Telephone: (702) 853-5490

Facsimile: (702) 227-1975

Attorneys for Plaintiff/Counterdefendants

EIGHTH JUDICIAL DISTRICT COURT

CLARK COUNTY, NEVADA

FRONT SIGHT MANAGEMENT LLC, a Nevada Limited Liability Company,

Plaintiff,

VS.

LAS VEGAS DEVELOPMENT FUND LLC, a Nevada Limited Liability Company; et al.,

Defendants.

AND ALL RELATED COUNTERCLAIMS.

CASE NO.: A-18-781084-B

DEPT NO.: 16

ORDER GRANTING MOTION TO
SEAL AND/OR REDACT PORTIONS
OF DEFENDANTS' OPPOSITIONS
TO JENNIFER PIAZZA AND THE
VNV TRUSTS' MOTIONS FOR
SUMMARY JUDGMENT TO
PROTECT CONFIDENTIAL
FINANCIAL INFORMATION

This matter having come before the Court on February 18, 2020 at 9:00 a.m. on Plaintiff's Motion to Seal and/or Redact Portions of Defendants' Oppositions to Jennifer Piazza and the VNV Trusts' Motions for Summary Judgment to Protect Confidential Financial Information, John P. Aldrich, Esq. appearing on behalf of Plaintiff and Kathryn Holbert, Esq. appearing on behalf of Defendants, the Court having reviewed the pleadings on file herein, having heard oral argument by the parties, and for good cause appearing therefore,

1///

24

FEB 2 0 2029

IT IS HEREBY ORDERED that Plaintiff/Counterdefendants' Motion to Seal and/or Redact Portions of Defendants' Oppositions to Jennifer Piazza and the VNV Trusts' Motions for Summary Judgment to Protect Confidential Financial Information is GRANTED.

IT IS FURTHER ORDERED that the Clerk of Court shall seal the following portions of Defendant and Counterclaimant LVD Fund's Opposition to Counterdefendant Jennifer Piazza's Motion for Summary Judgment (filed on February 14, 2020):

- Page 3, lines 5-26;
- Page 5, lines 8-14;
- Page 6, line 22 through page 7, line 6;
- Page 7, lines 8-22;
- Page 7, line 25 through page 8, line 8;
- Page 10, line 25 through page 11, line 2;
- Page 11, lines 17-19;
- Page 11, lines 23-26;
- Page 13, line 24 through page 14, line 7;
- Page 15, lines 7-9; and
- Page 15, lines 23-24.

IT IS FURTHER ORDERED that the Clerk of Court shall seal the following portions of Defendant and Counterclaimant LVD Fund's Opposition to Counterdefendants VNV Dynasty Trust I and VNV Dynasty Trust II's Motion for Summary Judgment (filed on February 14, 2020):

- Page 3, lines 5-25;
- Page 5, lines 20-23;
- Page 6, lines 7-16;
- Page 6, line 18 through page 7, line 12;
- Page 7, line 15 through page 8, line 2;
- Page 10, lines 15-18;
- Page 11, lines 11-14;
- Page 12, lines 1-2;
- Page 13, lines 13-21;
- Page 14, lines 23-25;

///

ll l

- Page 15, lines 21-22; and
- Page 16, lines 8-15.

IT IS SO ORDERED.

DATED this 20 day of February, 2020.

DISTRICT COURT JUDGE

Respectfully submitted by:

ALDRICH LAW FIRM, LTD.

110 600-1

John P. Aldrich, Esq. Nevada Bar No. 6877 Catherine Hernandez, Esq. Nevada Bar No. 8410 7866 West Sahara Avenue Las Vegas, Nevada 89117

Tel: (702) 853-5490 Fax: (702) 227-1975 Attorneys for Plaintiff Approved as to form and content:

FARMER CASE & FEDOR

Anthony T. Case, Esq.
Nevada Bar No. 6589
Kathryn Holbert, Esq.
Nevada Bar No. 10084
2190 E. Pebble Rd., Suite #205

Las Vegas, NV 89123 Tel: (702) 579-3900 Fax: (702) 739-3001

Attorneys for Defendants

EXHIBIT 2

EXHIBIT 2

Electronically Filed 1/23/2020 11:45 AM Steven D. Grierson CLERK OF THE COURT

1	NEO	Stevent.
2	John P. Aldrich, Esq. Nevada Bar No. 6877	
3	Catherine Hernandez, Esq. Nevada Bar No. 8410	
4	ALDRICH LAW FIRM, LTD. 7866 West Sahara Avenue	
	Las Vegas, NV 89117	
5	Telephone: (702) 853-5490 Facsimile: (702) 227-1975	
6	Attorneys for Plaintiff/Counterdefendants	
7	EIGHTH JUDICIAL D CLARK COUNT	
8	FRONT SIGHT MANAGEMENT LLC, a	
9	Nevada Limited Liability Company,	CASE NO.: A-18-781084-B DEPT NO.: 16
10	Plaintiff,	
11	VS.	NOTICE OF ENTRY OF FINDINGS OF FACT, CONCLUSIONS OF
12	LAS VEGAS DEVELOPMENT FUND LLC, a	LAW, AND ORDER DENYING DEFENDANT LAS VEGAS
13	Nevada Limited Liability Company; et al.,	DEVELOPMENT FUND LLC'S MOTION TO DISSOLVE
	Defendants.	TEMPORARY RESTRAINING
14		ORDER AND TO APPOINT A RECEIVER
15 16	AND ALL RELATED COUNTERCLAIMS.	
17	PLEASE TAKE NOTICE that the Finding	gs of Fact, Conclusions of Law, and Order
18	Denying Defendant Las Vegas Development Fu	and LLC's Motion to Dissolve Temporary
19	Restraining Order and to Appoint a Receiver was en	ntered by the Court in the above-captioned
20	///	
21	///	
22	///	
23	1//	
24		
	1	

1	action on the 23 rd day of January, 2020, a true and correct copy of which is attached her	eto.
2	DATED this 23 rd day of January, 2020.	
3	ALDRICH LAW FIRM, LTD.	
4	/s/ John P. Aldrich	
5	John P. Aldrich, Esq. Nevada Bar No. 6877	
6	Catherine Hernandez, Esq. Nevada Bar No. 8410	
7	7866 West Sahara Avenue Las Vegas, Nevada 89117	
8	Telephone: (702) 853-5490 Facsimile: (702) 227-1975	
9	Attorneys for Plaintiff/Counterdefe	ndants
10		
11		
12		
13		
14		
15		
16		
17		
18		
19		
20		
21		
22		
23		
24		

CERTIFICATE OF SERVICE

1	
2	
3	
4	
5	
6	
7	
8	
9	
10	
11	
12	
13	
14	
15	
16	
17	
18	
19	
20	

21

22

23

24

I HEREBY CERTIFY that on the 23rd day of January, 2020, I caused the foregoing NOTICE OF ENTRY OF FINDINGS OF FACT, CONCLUSIONS OF LAW, AND ORDER DENYING DEFENDANT LAS VEGAS DEVELOPMENT FUND LLC'S MOTION TO DISSOLVE TEMPORARY RESTRAINING ORDER AND TO APPOINT A RECEIVER to be electronically filed and served with the Clerk of the Court using Wiznet which will send notification of such filing to the email addresses denoted on the Electronic Mail Notice List, or by U.S. mail, postage prepaid, if not included on the Electronic Mail Notice List,

Anthony T. Case, Esq. Kathryn Holbert, Esq. FARMER CASE & FEDOR 2190 E. Pebble Rd., Suite #205 Las Vegas, NV 89123

to the following parties:

C. Keith Greer, Esq. 16855 West Bernardo Drive, Suite 255 San Diego, CA 92127 Attorneys for Defendants

/s/ T. Bixenmann

An employee of ALDRICH LAW FIRM, LTD.

3

Electronically Filed 1/23/2020 10:50 AM Steven D. Grierson CLERK OF THE COURT

FFCL

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

John P. Aldrich, Esq. Nevada Bar No. 6877

Catherine Hernandez, Esq.

Nevada Bar No. 8410

ALDRICH LAW FIRM, LTD.

7866 West Sahara Avenue

Las Vegas, NV 89117

Telephone: (702) 853-5490 Facsimile: (702) 227-1975

Attorneys for Plaintiff/Counterdefendants

EIGHTH JUDICIAL DISTRICT COURT

CLARK COUNTY, NEVADA

FRONT SIGHT MANAGEMENT LLC, a Nevada Limited Liability Company,

Plaintiff,

VS.

LAS VEGAS DEVELOPMENT FUND LLC, a Nevada Limited Liability Company; et al.,

Defendants.

CASE NO.: A-18-781084-B

DEPT NO.: 16

FINDINGS OF FACT,
CONCLUSIONS OF LAW, AND
ORDER DENYING DEFENDANT
LAS VEGAS DEVELOPMENT FUND
LLC'S MOTION TO DISSOLVE
TEMPORARY RESTRAINING
ORDER AND TO APPOINT A
RECEIVER

AND ALL RELATED COUNTERCLAIMS.

17

16

18

19 20

21

22

23

24

This matter having come before the Court on September 20, 2019 and November 26, 2019 on Defendant Las Vegas Development Fund LLC's Motion to Dissolve Temporary Restraining Order and to Appoint Receiver, John P. Aldrich, Esq. appearing on behalf of Plaintiff and Kathryn Holbert, Esq. and C. Keith Greer, Esq., appearing on behalf of Defendants, the Court having reviewed the pleadings on file herein, having heard oral argument of the parties through their respective counsel, this Court makes the following Findings of Fact and Conclusions of Law.

JAN 1 0 2029

Insofar as any conclusion of law is deemed to have been or include a finding of fact, such a finding of fact is hereby included as a factual finding. Insofar as any finding of fact is deemed to have been or to include a conclusion of law such is included as a conclusion of law herein.

FINDINGS OF FACT

The Court makes the following Findings of Fact based on the evidence presented:

- In Section IIB of Defendant/Counterclaimant Las Vegas Development Fund, LLC's
 ("LVDF") Motion to Dissolve Temporary Restraining Order and Appoint a Receiver,
 Defendant LVDF asserts thirteen breaches of the Construction Loan Agreement
 ("CLA"):
 - a. Alleged Breach #1: Improper Use of Loan Proceeds CLA §1.7(e) (Motion,
 p. 10);
 - b. Alleged Breach #2: Failure to Provide Government Approved Plans CLA §3.2(b) (Motion, p. 10);
 - c. Alleged Breach #3: Failure to Timely Complete Construction CLA §5.1
 (Motion, p. 10);
 - d. Alleged Breach #4: Material Change of Costs, Scope or Timing of Work –
 CLA §5.2 (Motion, p. 11);
 - e. Alleged Breach #5: Refusal to Comply Regarding Senior Debt CLA §5.27 (Motion, p. 11);
 - f. Alleged Breach #6: Failure to Provide Monthly Project Costs CLA §3.2(a)
 (Motion, p. 11);

g.	Alleged	Breach	#7:	Failure	to	Notify	of	Event	of	Default	-	CLA	§5.10
	(Motion	, p. 11);											

- h. Alleged Breach #8: Refusal to Allow Inspection of Records CLA §5.4
 (Motion, p. 12);
- i. Alleged Breach #9: Refusal to Allow Inspection of the Project CLA §3.3 (Motion, p. 12);
- j. Alleged Breach #10: Failure to Provide EB-5 Information CLA §1.7(f)
 (Motion, p. 12);
- k. Alleged Breach #11: Non Payment of Default Interest CLA §1.2 (Motion, p. 12);
- Alleged Breach #12: Non Payment of Legal Fees CLA §8.2 (Motion, p. 12);
 and
- m. Alleged Breach #13: Failure to Comply with Applicable Laws (CLA §5.13) and Failure to Give Written Notice of Criminal Complaint (CLA §5.14) (Motion, p. 13).
- 2. The first allegation of breach focuses on the alleged misuse of loan proceeds by Plaintiff/Counter-Defendant, Front Sight Management, LLC (Front Sight). However, in its Opposition to Defendant/Counter-Claimant LVDF's Motion to Dissolve the TRO and Appoint a Receiver, Front Sight supplied evidence to establish project cost and expenditures which exceed the loan amounts advanced by LVDF.
- 3. There are four (4) paragraphs of the Construction Loan Agreement that relate to loan proceeds. They are as follows:

Section 1.7 EB-5 Program Requirements.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

(e) Borrower shall use the proceeds of the Loan solely for the purpose of funding directly, or advancing to Affiliates to pay, the costs of the Project, <u>in accordance with the terms and conditions of this Agreement</u>, as set forth in the Budge and the Project documents submitted to, and approved by, USCIS.

Section 3.7 Use of Loan Proceeds. Borrower shall use and apply the Loan proceeds solely to all or any number of the individual Project components in accordance with the Budge and also to pay some or all of any or all existing indebtedness encumbering the Project pursuant to a Permitted Encumbrance. Borrower shall use its best business judgment based upon then-current real estate market and availability of other financing resources to allocate the proceeds of the Loan in such a manner as to assure the full expenditure of the Loan proceeds advanced to Borrower. Borrower will comply with the requirements of the EB-5 Program and the other EB-5 Program covenants and requirements contained in this Agreement.

Section 4.29 Use of Loan Proceeds. The proceeds of the Loan shall be used to pay and obtain release of the existing liens on the Land, to pay for or reimburse Borrower for soft and hard costs related to the pre-construction, development, promotion, construction, development and operation of the Project in connection with the FSFTI Facility and the construction, development, operation, leasing and sale of the timeshare portion of the Project, all as more particularly described on Exhibit F, attached hereto. The Loan is made exclusively for business purposes in connection with holding, developing and financially managing real estate for profit, and none of the proceeds of the Loan will be used for the personal, family or agricultural purposes of the Borrower.

Using Loan Proceeds. Subject to Section 3.2, Borrower shall Section 5.3 use the Loan proceeds in its sole discretion to pay, or to reimburse Borrower for paying, costs and expenses incurred by Borrower in connection with the pre-construction, promotion, construction, development, operating and leasing of the Project on the Land and the equipping of the Improvements, together with the payoff and release of any existing liens and encumbrances on the Land. Borrower shall take all steps necessary to assure that Loan proceeds are used by its contractors and subcontractors to pay such costs and expenses which could otherwise constitute a mechanic's lien claim against the Project. Within thirty (30) days after the Completion Date, Borrower shall provide the documentation and supporting accounting records and contract documents necessary, in Lender's discretion, to demonstrate that between the Closing Date and the date of delivery of such documentation not less than the total amount of the Advances has been spent directly or indirectly on the Project substantially in a form acceptable to Lender for compliance with the EB-5 Program.

(Emphases added.)

4. Exhibit 47 to the Evidentiary Hearing Exhibits, Front Sight's "Response to Notice of Default dated July 30, 2018," shows project costs and expenditures well in excess of \$6.3 million LVDF advanced. In Exhibit C to that document, Front Sight provided

copies of QuickBooks monthly reports that showed the following Project costs and expenditures:

TIME PERIOD	TOTAL
October 2015 – December 2015	\$3,387,591.35
January 2016 - December 2016	\$7,466,570.24
January 2017 – December 2017	\$12,454,018.84
	\$23,308,180.43

5. Exhibit 48 to the Evidentiary Hearing Exhibits is Front Sight's "Additional Response to Notices of Default dated July 31, 2018, and August 24, 2018 and Initial Response to Notice of Default dated August 28, 2018." In that exhibit, Front Sight provided to Defendant Dziubla a multitude of documents showing the following expenses which were paid by Front Sight between the closing of the loan in October 2016 and June 30, 2017:

EXPENSE CATEGORY	TOTAL
Reimbursable construction costs prior to the closing date of the	\$994,336.56
Construction Loan Agreement	
Construction costs from the closing date of the Construction	\$1,031,728.10
Loan Agreement to June 30, 2017	,
Class Action lien payoff as of the time of closing of the	\$551,871.50
Construction Loan Agreement	·
Class action lien pay-down prior to the closing date of the	\$1,860,000.00
Construction Loan Agreement	
Holecek note paydown prior to the closing date of the	\$6,004,000.00
Construction Loan Agreement	
Holecek note paydown from the closing date of the Construction	\$1,422,000.00
Loan Agreement to June 30, 2017	
Project legal fees	\$81,551.25
Fees Paid to Chicago Title in connection with original closing	\$9,217.01
EB5 Impact Advisor fees	\$244,730.00
Fees paid to US Capital Partners evidencing efforts to secure	\$62,500.00
"Senior Debt" prior to securing construction line of credit from	,
Morales Construction	
Project consulting fees	\$82,550.00
	\$12,344,484.42

- 6. Adding construction costs prior to closing with construction costs from closing to June 30, 2017, plus the class action lien payoff as of the closing date of the CLA, Front Sight's expenses far exceed the US\$2.625M in EB5 funds delivered on or before June 30, 2017.
- 7. Exhibit 49 to the Evidentiary Hearing Exhibits is Front Sight's "EB-5 Documentation and Additional Information for the Period July 1, 2017, through October 31, 2018 Delivered Pursuant to Section 5.10(e) of the Construction Loan Agreement." In that exhibit, Front Sight provided to Defendant Dziubla several hundred additional pages of documents showing the following expenses which were paid by Front Sight:

EXPENSE CATEGORY	TOTAL
Construction costs from June 30, 2017, through and including July 1, 2018	\$2,088,490.00
Holecek note paydown from June 30, 2017, through and including July 1, 2018	\$1,896,000.00
Project legal fees from June 30, 2017, through and including July 1, 2018	\$14,116.00
Construction costs from July 1, 2018, through and including October 30, 2018	\$402,621.00
Holecek note paydown from July 1, 2018, through and including October 30, 2018	\$632,000.00
Project legal fees from July 1, 2018, through and including October 30, 2018	\$6,984.00
Construction costs from September 6, 2016, through and including August 24, 2018 (commercial revolving charge account of Front Sight established with Home Depot)	\$66,173.67
Construction costs from October 11, 2016, through and including July 13, 2018 (charged to the Visa credit card account of Front Sight established with City National Bank)	\$43,212.07
Construction costs from August 30, 2016, through and including February 20, 2018 (charged to the Premier Rewards Gold charge and credit card account of Front Sight established with American Express)	\$92,868.00
	\$5,242,464.74

Based on this uncontroverted evidence, the Court finds Front Sight's expenses on the
 Project far exceed the amount of the loan from Defendant LVDF.

- 9. As to the fourth breach alleged by Defendant/Counterclaimant LVDF, the alleged material change in size, scope, and timing of the project, it appears that the size of the classroom was reduced but not the overall size of the facility, and therefore, the Court finds that there is an issue of fact as to this alleged breach of the CLA.
- 10. Regarding the second, third, and fifth through thirteenth alleged breaches, as asserted by Defendant/Counterclaimant LVDF, the parties asserted multiple competing factual positions and made conflicting factual assertions regarding Defendant LVDF's allegations of breach of the CLA. Based on the state of the evidence as of the date of the hearing on the instant Motion, the Court finds that genuine issues of fact remain as to the second, third, and fifth through thirteenth alleged breaches, as asserted by Defendant/Counterclaimant LVDF.

CONCLUSIONS OF LAW AND ORDER

The Court makes the following Conclusions of Law:

- 1. Regarding alleged Breach #1, the Court concludes that Front Sight's expenses on the Project far exceed the amount of the loan from Defendant LVDF has Defendant LVDF's assertion that Front Sight improperly used loan proceeds is without merit, and consequently, LVDF has failed to establish this alleged breach.
- 2. As to the second, third, and fifth through thirteenth alleged breaches, as asserted by Defendant/Counterclaimant LVDF, the Court concludes that LVDF has not established that Plaintiff is in breach of the Construction Loan Agreement, and consequently, LVDF is not entitled to the relief it seeks by this Motion.
- Regarding the fourth alleged breach, pertaining to the reduction in the size of the
 Patriot Pavilion, because it appears that the size of the classroom was reduced but not the overall

- 1	
1	size of the facility, creating an issue of fact as to this alleged breach, the Court concludes that
2	LVDF has not established that Plaintiff is in breach of the construction Loan Agreement, and
3	consequently, LVDF is not entitled to the relief it seeks by this Motion.
4	ORDER
5	IT IS HEREBY ORDERED that Defendant Las Vegas Development Fund LLC's
6	Motion to Dissolve Temporary Restraining Order and to Appoint Receiver is DENIED.
7	IT IS SO ORDERED.
8	DATED this 22 day of January, 2020.
9	THE 1. D. N.
10	DISTRICT COURT JUDGE
11	9
12	Respectfully submitted by:
13	ALDRICH LAW FIRM, LTD.
14	ALDRICH EAW FIRM, ETD.
15	John P. Aldrich, Esq. Nevada Bar No. 6877
16	Catherine Hernandez, Esq. Nevada Bar No. 8410
17	7866 West Sahara Avenue Las Vegas, Nevada 89117
18	Tel: (702) 853-5490 Fax: (702) 227-1975
19	Attorneys for Plaintiff
20	
21	
22	
23	
24	
- 4	I .

Electronically Filed 5/13/2020 3:39 PM Steven D. Grierson CLERK OF THE COURT

		CEEKIK OF THE COOK
1	NEO	Atems.
2	John P. Aldrich, Esq. Nevada Bar No. 6877	
	Catherine Hernandez, Esq.	
3	Nevada Bar No. 8410 Jamie S. Hendrickson, Esq.	
4	Nevada Bar No. 12770	
_	ALDRICH LAW FIRM, LTD.	
5	7866 West Sahara Avenue Las Vegas, NV 89117	
6	Telephone: (702) 853-5490	
7	Facsimile: (702) 227-1975 Attorneys for Plaintiff/Counterdefendants	
,	Thorneys for Frankly/Counteracjenaums	
8	EIGHTH JUDICIAL D	
9	CLARK COUNTY	I, NEVADA
10	FRONT SIGHT MANAGEMENT LLC, a	CASE NO. A 10 701004 D
10	Nevada Limited Liability Company,	CASE NO.: A-18-781084-B DEPT NO.: 16
11	Plaintiff,	
12	vs.	NOTICE OF ENTRY OF
		STIPULATION AND ORDER TO
13	LAS VEGAS DEVELOPMENT FUND LLC, a Nevada Limited Liability Company; et al.,	EXTEND DISCOVERY DEADLINES AND CONTINUE
14		TRIAL (SECOND REQUEST)
15	Defendants.	
13		
16	AND ALL RELATED COUNTERCLAIMS.	
17		
18	DI EASE TAKE NOTICE that a Stimulation	n and Order to Extend Discovery Deadlines
10	•	·
19	and Continue Trial (Second Request) was entered b	y the Court in the above-captioned action on
20	///	
21	///	
22	///	
23	///	
24		
	1	

1	the 13 th day of May, 2020, a true and correct copy of	of which is attached hereto.
2	DATED this 13 th day of May, 2020.	
3		ALDRICH LAW FIRM, LTD.
4		/s/ John P. Aldrich John P. Aldrich, Esq.
5		Nevada Bar No. 6877 Catherine Hernandez, Esq.
6		Nevada Bar No. 8410 Jamie S. Hendrickson, Esq.
7		Nevada Bar No. 12770 7866 West Sahara Avenue
8		Las Vegas, Nevada 89117 Telephone: (702) 853-5490
9		Facsimile: (702) 227-1975 Attorneys for Plaintiff/Counterdefendants
10		, , , , , , , , , , , , , , , , , , ,
11		
12		
13		
14		
15		
16		
17		
18		
19		
20		
21		
22		
23		
24		

1	<u>CERTIFICATE OF SERVICE</u>
2	I HEREBY CERTIFY that on the 13 th day of May, 2020, I caused the foregoing
3	NOTICE OF ENTRY OF STIPULATION AND ORDER TO EXTEND DISCOVERY
4	DEADLINES AND CONTINUE TRIAL (SECOND REQUEST) to be electronically filed
5	and served with the Clerk of the Court using Wiznet which will send notification of such filing to
6	the email addresses denoted on the Electronic Mail Notice List, or by U.S. mail, postage prepaid,
7	if not included on the Electronic Mail Notice List, to the following parties:
8	Anthony T. Case, Esq. Kathryn Holbert, Esq. FARMER CASE & FEDOR
10	2190 E. Pebble Rd., Suite #205 Las Vegas, NV 89123
11 12	C. Keith Greer, Esq. 16855 West Bernardo Drive, Suite 255 San Diego, CA 92127
13 14 15 16	John R. Bailey, Esq. Joshua M. Dickey, Esq. Andrea M. Champion, Esq. BAILEY KENNEDY 8984 Spanish Ridge Avenue Las Vegas, NV 89148 Attorneys for Defendants
18 19	/s/ T. Bixenmann An employee of ALDRICH LAW FIRM, LTD.
20	
21	
22	
23	
24	

Electronically Filed 5/13/2020 3:14 PM Steven D. Grierson CLERK OF THE COURT

1 SAO John P. Aldrich, Esq. 2 Nevada Bar No. 6877 Catherine Hernandez, Esq. 3 Nevada Bar No. 8410 Jamie S. Hendrickson, Esq. Nevada Bar No. 12770 4 ALDRICH LAW FIRM, LTD. 5 7866 West Sahara Avenue Las Vegas, NV 89117 Telephone: (702) 853-5490 6 Facsimile: (702) 227-1975 7 Attorneys for Plaintiff/Counterdefendants 8 EIGHTH JUDICIAL DISTRICT COURT 9 **CLARK COUNTY, NEVADA** 10 FRONT SIGHT MANAGEMENT LLC, a CASE NO.: A-18-781084-B Nevada Limited Liability Company, DEPT NO.: 16 11 STIPULATION AND ORDER TO Plaintiff, EXTEND DISCOVERY DEADLINES 12 AND CONTINUE TRIAL VS. (SECOND REQUEST) 13 LAS VEGAS DEVELOPMENT FUND LLC, a Nevada Limited Liability Company; et al., 14 15 Defendants. 16 AND ALL RELATED COUNTERCLAIMS. 17 18 Plaintiff FRONT SIGHT MANAGEMENT LLC ("Plaintiff" or "Front Sight") and 19 Defendants LAS VEGAS DEVELOPMENT FUND LLC, EB5 IMPACT CAPITAL 20 REGIONAL CENTER LLC, EB5 IMPACT ADVISORS LLC, ROBERT W. DZIUBLA, JON 21 FLEMING, and LINDA STANWOOD (collectively, "Defendants" or the "EB5 Parties"), by 22 and through their respective counsel, and hereby stipulate and agree to extend the discovery 23

Production of Documents. These written discovery requests are and will be the subject of current and ongoing motion practice. Additionally, Plaintiff has served several sets of Interrogatories to all Defendants. Defendants' answers to those Interrogatories are also the subject of motion practice before the Court.

- 4. Defendants have served several sets of Requests for Production of Documents and Interrogatories to Front Sight, Ignatius Piazza, Jennifer Piazza, VNV Dynasty Trust I, and VNV Dynasty Trust II. Plaintiff and the Counterdefendants have all responded to Defendants' discovery requests. Defendants/Counterclaimant contend these discovery responses are deficient. If the parties are unable to resolve their discovery dispute, these discovery requests will be the subject of motion practice.
- 5. Defendants recently filed a Motion to Compel and/or for Temporary Restraining Order seeking additional documents from Plaintiff, asserting that the information sought is necessary so that the first EB-5 investor may file his I-829 Petition by Investor to Remove Conditions on Permanent Resident Status.
- 6. The parties have taken the following depositions:
 - a. Deposition of Jay Carter taken on February 12, 2020;
 - b. Deposition of David Keller taken on February 12, 2020;
 - Deposition of Person Most Knowledgeable of Empyrean West, LLC taken on February 12, 2020;
 - d. Deposition of Person Most Knowledgeable of Morales Construction, Inc. taken on March 16, 2020;
 - e. Deposition of Person Most Knowledgeable of All American Concrete and Masonry, Inc. taken on March 16, 2020; and

10

11

12

13

15

14

16

17

18

19

2021

22

23

24

- Deposition of Person Most Knowledgeable of Top Rank Builders, Inc. taken on March 16, 2020.
- Defendants issued subpoenas to Bank of America and Lucas Horsfall, LLP after the Court denied Plaintiff's motion to quash.
- 8. Defendants have served notices of intent to issue subpoenas to U.S. Capital and Summit Financial. The Court has since denied Plaintiff's motion to quash those subpoenas.
- 9. Defendants have an outstanding subpoena duces tecum to Morales Construction, Inc., All American Concrete and Masonry, Inc., and Top Rank Builders, Inc. that may become the subject of a motion for order showing cause if the subpoenaed parties refuse to respond to the subpoenas.
- On October 24, 2019, Plaintiff served its Designation of Expert Witnesses and on April
 2020, Plaintiff served its First Supplement to Designation of Expert Witnesses. On
 April 3, 2020, Defendants served their Designation of Expert Witnesses.

Remaining Discovery to be Completed:

The parties believe that the following discovery remains to be completed:

- 1. Depositions of the parties and witnesses identified by the parties;
- 2. Additional written discovery, including resolution of the parties' discovery disputes;
- 3. Additional subpoenas *duces tecum* to third-parties;
- 4. Supplemental initial expert disclosures;
- 5. Rebuttal expert disclosures;
- 6. Expert depositions; and
- 7. Other discovery as necessary.

Reasons Why Remaining Discovery Not Completed:

The President of the United States has declared a National Emergency in response to the COVID-19 pandemic. Likewise, the Governor of Nevada has declared a State of Emergency and Clark County has declared local emergencies.

In Nevada, Governor Sisolak has issued fourteen emergency directives, which have substantially impacted the ability of most businesses to operate normally and for people to travel for discovery purposes. On April 1, 2020, Governor Sisolak issued Directive 010 Stay at Home Order, which orders Nevadans to stay in their residences with limited exceptions. The Directive remains in effect until April 30, 2020, unless renewed by subsequent Directive.

Recognizing the severity of the COVID-19 pandemic, this Court has issued multiple administrative orders that change how the Court operates in an effort to do its part to contain the spread of the novel virus. Those administrative orders underscore the severity of the pandemic and reflect the Court's willingness to work with litigants to continue and extend matters, as necessary. Specifically, Administrative Order 20-09 stays "[a]ll pending, unexpired deadlines pursuant to NRCP 31 (depositions by written questions), 33 (interrogatories to parties), 34 (producing documents, electronically stored information and tangible things, or entering onto land, for inspection or other purposes), and/or 36 (requests for admissions) by 30 days" unless the time to respond to the written discovery expired prior to entry of the Court's Order. *See* Admin. Order 20-09 at pg. 4. Then, more recently, on April 17, 2020, the Court entered Administrative Order 20-13 that further tolls discovery deadlines from March 18, 2020, until thirty (30) days after the order expires, is modified, or rescinded. *See* Admin Order 20-13 at pg. 7. The Court has encouraged parties to proceed with discovery should proceed when at all possible. *See id*.

The parties have proceeded diligently with discovery thus far, including exchanging tens of thousands of pages of documents. Both parties have filed motions to compel on outstanding discovery issues and continue to work through discovery disputes. The parties will continue to

proceed with discovery when practicable. However, due to the COVID-19 pandemic, the parties agree that a continuance of discovery is necessary so that the parties can complete written discovery after the Court's Administrative Order expires (or is rescinded) and take depositions.

Moreover, on April 16, 2020, Plaintiff informed Defendants that in mid-March, in the midst of the growing COVID-19 pandemic and emergency orders, one of Plaintiff's experts advised that he would no longer be able to act as an expert in the case. Plaintiffs informed Defendants that it had located another expert but that Plaintiff's new expert would not be able to complete his report by April 20, 2020 (the parties' initial expert disclosure deadline once recalculated in light of Admin. Order 20-09). Although both parties have previously served a designation of expert witnesses, in light of recent events (including the COVID-19 pandemic and new counsel making an appearance to represent Defendants), on April 20, 2020, the parties mutually agreed that the initial expert disclosure deadline should be rescheduled so that the parties were able to amend their initial expert disclosures.

The parties now make this stipulation in good faith in order to complete the remaining discovery. At this time, based on the information available, they do not anticipate a further extension of discovery will be needed in the future and expect to be able to complete discovery by the below proposed dates.

Proposed Schedule for Completing Remaining Discovery:

The remaining discovery set forth above should be completed by September 3, 2020. The parties agree that the extension of the discovery deadlines is necessary to complete the parties' initial expert and rebuttal expert disclosures. Therefore, the parties request and stipulate that the Court continue the discovery deadlines permit the following proposed Discovery Scheduling Order:

///

///

1	EVENT DEADLINE	CURRENT DATE	PROPOSED DATE
2	Last day to complete discovery	July 3, 2020	October 1, 2020
3	Last day to file motions to amend pleadings or add parties	April 3, 2020	July 2, 2020
	Last day for initial expert disclosures	April 3, 2020	July 2, 2020
4	Last day for rebuttal expert disclosures	May 4, 2020	August 3, 2020
_	Last day to file dispositive motions	August 3, 2020	November 2, 2020
56	Current Trial Date:	- 1 - 11: 1 i	((0) 1:11 :
7	The parties' proposed extension of discovery	y deadlines by sixt	y (60) days will impact
8	the October 5, 2020 trial date. Therefore, the parties	further stipulate an	nd agree to continue trial
9	to the next available jury trial date after January 18	, 2021. This Stip	ulation is made in good

Dated this 30th day of April, 2020. Dated this 30th day of April, 2020.

ALDRICH LAW FIRM, LTD.

faith and not for purposes of delay.

BAILEY KENNEDY

/s/ John P. Aldrich John P. Aldrich, Esq. Nevada Bar No. 6877 Catherine Hernandez, Esq. Nevada Bar No. 8410 Jamie S. Hendrickson, Esq. Nevada Bar No. 12770 7866 West Sahara Avenue	/s/ Andrea M. Champion John R. Bailey, Esq. Nevada Bar No. 0137 Joshua M. Dickey, Esq. Nevada Bar No. 6621 Andrea M. Champion, Esq. Nevada Bar No. 13461
7866 West Sahara Avenue	Nevada Bar No. 13461 8984 Spanish Ridge Avenue
Las Vegas, Nevada 89117 Tel: (702) 853-5490	Las Vegas, Nevada 89148-1302
Fax: (702) 227-1975 Attorneys for Plaintiff/Counterdefendants	Tel: (702) 562-8820 Fax: (702) 562-8821 Attorneys for Defendants/Counterclaimant

ORDER

Having reviewed and considered the above Stipulation by the parties, and good cause appearing therefore,

IT IS HEREBY ORDERED that the discovery deadlines will be extended as agreed to by the parties as follows:

1	EVENT DEADLINE	CURRENT DATE	PROPOSED DATE
2	Last day to complete discovery	July 3, 2020	October 1, 2020
3	Last day to file motions to amend pleadings or add parties	April 3, 2020	July 2, 2020
	Last day for initial expert disclosures	April 3, 2020	July 2, 2020
4	Last day for rebuttal expert disclosures	May 4, 2020	August 3, 2020
5	Last day to file dispositive motions	August 3, 2020	November 2, 2020
6	IT IS FURTHER ORDERED that an Amendo	ed Scheduling Orde	er will issue. The trial
7	date will be moved to January 18, 2021 or as soon there	eafter as the Court	can accommodate it.
8	IT IS SO ORDERED.		
9	Dated this 13th day of April, 2020.		
10	7.	# : > 0	, ~

DISTRICT COURT JUDGE LB

Respectfully submitted by:

11

12

13

14

21

22

23

24

ALDRICH LAW FIRM, LTD.

/s/ John P. Aldrich 15 John P. Aldrich, Esq. Nevada Bar No. 6877 16 Catherine Hernandez, Esq. Nevada Bar No. 8410 Jamie S. Hendrickson, Esq. 17 Nevada Bar No. 12770 7866 West Sahara Avenue 18 Las Vegas, Nevada 89117 Tel: (702) 853-5490 19 Fax: (702) 227-1975 Attorneys for Plaintiff/Counterdefendants 20

Yes, you may e-sign on my behalf.

Thanks Traci.

Andrea Champion
BAILEY KENNEDY
8984 SPANISH RIDGE AVENUE
LAS VEGAS, NV 89148-1302
702.562.8820 (MAIN)
702.562.8821 (FAX)
702.789.4551 (DIRECT)
AChampion@BaileyKennedy.com

Electronically Filed 5/13/2020 3:40 PM Steven D. Grierson CLERK OF THE COURT

 $1 \parallel_{\mathbf{ARJT}}$

2

3

5

6

7

8

9

10

11

12

13

DISTRICT COURT

4 | CLARK COUNTY, NEVADA

FRONT SIGHT MANAGEMENT LLC, a Nevada Limited Liability Company,

lity Company,

Plaintiff,

| vs.

LAS VEGAS DEVELOPMENT FUND LLC, a Nevada Limited Liability Company; EB5 IMPACT CAPITAL REGIONAL CENTER LLC, a Nevada Limited Liability Company; B5 IMPACT ADVISORS LLC, a Nevada Limited Liability Company; ROBERT W. DZIUBLA, individually and as President and CEO of LAS VEGAS DEVELOPMENT FUND LLC and EB5

IMPACT ADVISORS LLC; JON
FLEMING, individually and as an
agent of LAS VEGAS DEVELOPMENT
FUND LLC and EB5 IMPACT ADVISORS

16 LLC; LINDA STANWOOD, individually and as Senior Vice President of LAS

VEGAS DEVELOPMENT FUND LLC and EB5 IMPACT ADVISORS LLC;

CHICAGO TITLE COMPANY, a
California corporation; DOES 1-

10, inclusive; and ROE CORPORATIONS
1-10, inclusive,

21

20

23

24

25

26

27

28

22 Defendants.
AND ALL RELATED CLAIMS

Case No. A-18-781084-B Dept No. XVI

> HEARING DATE(S) ENTERED IN ODYSSEY

AMENDED ORDER SETTING CIVIL JURY TRIAL, PRE-TRIAL/CALENDAR CALL, AND DEADLINES FOR MOTIONS; AMENDED DISCOVERY SCHEDULING ORDER

Pursuant to the Status Check re Trial Rescheduling held on April 30, 2020, the

1

Discovery Deadlines and Trial dates are hereby amended as follows:

Timothy C. Williams DISTRICT JUDGE DEPT XVI LAS VEGAS, NV 89155

Case Number: A-18-781084-B

IT IS HEREBY ORDERED that the parties will comply with the following deadlines:

B. DISCOVERY AND MOTION DEADLINES.

- 1. All parties shall complete discovery on or before **October 1, 2020. The Court will hear any discovery motions.** However, in the event it becomes necessary, the Court may request nominations for a stand-by special master for referrals of discovery issues on a motion-by-motion basis.
- All parties shall file motions to amend pleadings or add parties on or before
 July 2, 2020.
- 3. All parties shall make initial expert disclosures pursuant to NRCP 16.1(a)(2) on or before **July 2, 2020.**
- 4. All parties shall make rebuttal expert disclosures pursuant to NRCP 16.1(a)(2) on or before **August 3, 2020.**
 - 5. All parties shall file dispositive motions on or before **November 2, 2020.**
- 6. Counsel/parties in proper person are also directed to abide by EDCR 2.47 concerning the time for filing and noticing motions *in limine*. Except upon a showing of unforeseen extraordinary circumstances, the Court will not shorten time for the hearing of any such motions.

C. TRIAL AND INCIDENTAL DATES AND OBLIGATIONS.

- 1. A jury trial of the above-entitled case is set on a five week stack to begin, 22nd day of February, 2021, at 9:30 a.m.
 - 2. A pre-trial/calendar call will be held on February 11, 2021, at 10:30 a.m.
- 3. A status check re Trial Readiness is scheduled to be held on October 21, 2020 at9:00 am.

- 4. The Pre-Trial Memorandum must be filed no later than **February 18, 2021**, with a courtesy copy delivered to Department XVI. All parties, (Attorneys and parties in proper person) **MUST** comply with **All REQUIREMENTS** of EDCR 2.67, 2.68 and 2.69. Counsel should include in the Memorandum an identification of orders on all motions in limine or motions for partial summary judgment previously made, a summary of any anticipated legal issues remaining, a brief summary of the opinions to be offered by any witness to be called to offer opinion testimony as well as any objections to the opinion testimony.
- 5. All original depositions anticipated to be used in any manner during the trial must be delivered to the clerk prior to the firm trial date given at pre-trial/calendar call. If deposition testimony is anticipated to be used in lieu of live testimony, a designation (by page/line citation) of the portions of the testimony to be offered must be filed and served by facsimile or hand, two (2) judicial days prior to the firm trial date given at the pre-trial/calendar call. Any objections or counter-designations (by page/line citation) of testimony must be filed and served by facsimile or hand, one (1) judicial day prior to the firm trial date given at the pre-trial/calendar call. Counsel shall advise the clerk prior to publication.
- 6. In accordance with EDCR 2.67, counsel shall meet, review, and discuss exhibits. All exhibits must comply with EDCR 2.27. Two (2) sets must be three-hole punched and placed in three ring binders along with the exhibit list. The sets must be delivered to the clerk prior to the firm trial date given at the pre-trial/calendar call. Any demonstrative exhibits including exemplars anticipated to be used must be disclosed prior to the calendar call. Pursuant to EDCR 2.68, counsel shall be prepared to stipulate or make specific objections to individual proposed exhibits. Unless otherwise agreed to by the parties, demonstrative exhibits are marked for identification but not admitted into evidence.

7.	In accordance with EDCR 2.67, counsel shall meet, review, and discuss items to
be include	d in the Jury Notebook. Pursuant to EDCR 2.68, counsel shall be prepared to stipulate
or make sp	pecific objections to items to be included in the Jury Notebook.

8. In accordance with EDCR 2.67, counsel shall meet and discuss preinstructions to the jury, jury instructions, special interrogatories, if requested, and verdict forms. Each side shall provide the Court an agreed set of jury instructions and proposed form of verdict along with any additional proposed jury instructions with an electronic copy in Word format.

Failure of the designated trial attorney or any party appearing in proper person to appear for any court appearances or to comply with this Order shall result in any of the following: (1) dismissal of the action (2) default judgment; (3) monetary sanctions; (4) vacation of trial date; and/or any other appropriate remedy or sanction.

Counsel is asked to notify the Court Reporter at least two (2) weeks in advance if they are going to require daily copies of the transcripts of this trial or real time court reporting. Failure to do so may result in a delay in the production of the transcripts or the availability of real time court reporting.

Counsel is required to advise the Court immediately when the case settles or is otherwise resolved prior to trial. A stipulation which terminates a case by dismissal shall also indicate whether a Scheduling Order has been filed and, if a trial date has been set, the date of that trial. A copy should be given to Chambers.

DATED this 13^{th} day of May, 2020.

TIMOTAY C. WILLIAMS DISTRICT JUDGE

Timothy C. Williams DISTRICT JUDGE DEPT XVI LAS VEGAS, NV 89155

CERTIFICATE OF SERVICE

I hereby certify that on or about the date signed, I served a true and correct copy upon the parties by electronic transmission through the Eighth Judicial District Court E-Filing System in accordance with the mandatory electronic service requirements of Administrative Order 14-2 and the Nevada Electronic Filing and Conversation Rules.

/s/ Lynn Berkheimer

Lynn Berkheimer Judicial Executive Assistant

Timothy C. Williams DISTRICT JUDGE DEPT XVI LAS VEGAS, NV 89155

BAILEY * KENNEDY 8984 SPANISH RIDGE AVENUE LAS VEGAS, NEVADA 89148-1302 702.562.8820

5/18/2020 2:36 PM Steven D. Grierson CLERK OF THE COURT 1 **NEOJ** JOHN R. BAILEY 2 Nevada Bar No. 0137 JOSHUA M. DICKEY 3 Nevada Bar No. 6621 ANDREA M. CHAMPION 4 Nevada Bar No. 13461 **BAILEY * KENNEDY** 5 8984 Spanish Ridge Avenue Las Vegas, Nevada 89148-1302 6 Telephone: 702.562.8820 Facsimile: 702.562.8821 7 JBailey@BaileyKennedy.com JDickey@BaileyKennedy.com 8 AChampion@BaileyKennedy.com C. KEITH GREER, ESQ. Cal. Bar. No. 135537 (Pro Hac Vice) 10 GREER AND ASSOCIATES, APC 16855 West Bernardo Dr. Suite 255 11 San Diego, California 92127 Telephone: 858.613.6677 12 Facsimile: 858.613.6680 keith.greer@greerlaw.biz 13 Attorneys for Defendants 14 LAS VEGAS DEVELOPMENT FUND LLC; EB5 IMPACT CAPITAL REGIONAL CENTER 15 LLC; EB5 IMPACT ADVISORS LLC; ROBERT W. DZIUBLA; JON FLEMING; and 16 LINDA STANWOOD 17 DISTRICT COURT 18 CLARK COUNTY, NEVADA 19 FRONT SIGHT MANAGEMENT LLC, a Nevada Limited Liability Company, Case No. A-18-781084-B 20 Dept. No. XVI Plaintiff, 21 NOTICE OF ENTRY OF ORDER VS. 22 **GRANTING LAS VEGAS** LAS VEGAS DEVELOPMENT FUND LLC, a DEVELOPMENT FUND, LLC'S 23 Nevada Limited Liability Company; et al, MOTION TO COMPEL PRODUCTION OF DOCUMENTS OR, IN THE 24 ALTERNATIVE, MOTION FOR Defendants. PRELIMINARY INJUNCTION TO 25 ADDRESS FRONT SIGHT'S CONTINUING VIOLATION OF 26 **SECTION 5.10 OF THE** AND ALL RELATED COUNTERCLAIMS. CONSTRUCTION LOAN AGREEMENT 27 AND REQUEST FOR LIMITED RELIEF FROM THE PROTECTIVE ORDER 28 Page 1 of 3

Electronically Filed

Page 3 of 3

BAILEY * KENNEDY 8984 SPANISH RIDGE AVENUE LAS VEGAS, NEVADA 89148-1302 702.562.8820

5/15/2020 4:54 PM Steven D. Grierson CLERK OF THE COURT 1 **ORDR** JOHN R. BAILEY 2 Nevada Bar No. 0137 JOSHUA M. DICKEY 3 Nevada Bar No. 6621 ANDREA M. CHAMPION 4 Nevada Bar No. 13461 **BAILEY * KENNEDY** 5 8984 Spanish Ridge Avenue Las Vegas, Nevada 89148-1302 6 Telephone: 702.562.8820 Facsimile: 702.562.8821 7 JBailey@BaileyKennedy.com JDickey@BaileyKennedy.com 8 AChampion@BaileyKennedy.com C. KEITH GREER, ESQ. Cal. Bar. No. 135537 (Pro Hac Vice) 10 GREER AND ASSOCIATES, A PC 16855 West Bernardo Dr. Suite 255 11 San Diego, California 92127 Telephone: 858.613.6677 12 Facsimile: 858.613.6680 keith.greer@greerlaw.biz 13 Attorneys for Defendants 14 LAS VEGAS DEVELOPMENT FUND LLC; EB5 IMPACT CAPITAL REGIONAL CENTER 15 LLC; EB5 IMPACT ADVISORS LLC; ROBERT W. DZIUBLA; JON FLEMING; and 16 LINDA STANWOOD 17 DISTRICT COURT 18 CLARK COUNTY, NEVADA 19 FRONT SIGHT MANAGEMENT LLC, a Nevada Limited Liability Company, Case No. A-18-781084-B 20 Dept. No. XVI Plaintiff, 21 ORDER GRANTING LAS VEGAS **DEVELOPMENT FUND, LLC'S** VS. 22 MOTION TO COMPEL PRODUCTION LAS VEGAS DEVELOPMENT FUND LLC, a OF DOCUMENTS OR, IN THE 23 Nevada Limited Liability Company; et al, ALTERNATIVE, MOTION FOR PRELIMINARY INJUNCTION TO 24 ADDRESS FRONT SIGHT'S Defendants. CONTINUING VIOLATION OF 25 **SECTION 5.10 OF THE** CONSTRUCTION LOAN AGREEMENT 26 AND REQUEST FOR LIMITED RELIEF AND ALL RELATED COUNTERCLAIMS. FROM THE PROTECTIVE ORDER 27 28 Page 1 of 3

Electronically Filed

1 IT IS FURTHER ORDERED that, upon request, Front Sight shall forthwith supplement its 2 production of the Federally Required Information should any additional EB-5 investors need to file 3 an I-829 petition with USCIS in the future. IT IS FURTHER ORDERED that EB-5 investors (and/or his/her counsel) execute the 4 5 Acknowledgment to the Protective Order entered in this case before Las Vegas Development Fund, LLC may provide the information produced by Front Sight (i.e., the Federally Required Information) 6 7 to any EB-5 investor (or their counsel) and that the EB-5 investor (and/or their counsel) be allowed 8 to disseminate such information produced by Front Sight only to the Federal Government (i.e., 9 USCIS). 10 **IT IS SO ORDERED** this <u>15th</u> day of May, 2020. 11 12 TIMOTHY C. WILLIAMS DISTRICT CÖURT JUDGE 13 CG 14 Respectfully submitted by: 15 BAILEY KENNEDY, LLP 16 /s/ Andrea M. Champion JOHN R. BAILEY 17 Nevada Bar No. 0137 JOSHUA M. DICKEY 18 Nevada Bar No. 6621 Andrea M. Champion 19 Nevada Bar No. 13461 8984 Spanish Ridge Avenue 20 Las Vegas, Nevada 89148-1302 Telephone: 702.562.8820 21 Facsimile: 702.562.8821 JBailey@BaileyKennedy.com 22 JDickey@BaileyKennedy.com AChampion@BaileyKennedy.com 23 24 Attorneys for Defendants LAS VEGAS DEVELOPMENT FUND LLC; 25 EB5 IMPACT CAPITAL REGIONAL CENTER LLC; EB5 IMPACT ADVISORS LLC; ROBERT W. DZIUBLA; JON 26 FLEMING: and LINDA STANWOOD 27 28

BAILEY * KENNEDY 8984 Spanish Ridge Avenue Las Vegas, Neyada 89148-1302 702.562.8820

		Electronically Filed 6/4/2020 3:20 PM Steven D. Grierson
1	NEOJ	CLERK OF THE COURT
2	JOHN R. BAILEY Nevada Bar No. 0137	Blue
3	JOSHUA M. DICKEY	
3	Nevada Bar No. 6621 Andrea M. Champion	
4	Nevada Bar No. 13461 BAILEY KENNEDY	
5	8984 Spanish Ridge Avenue	
6	Las Vegas, Nevada 89148-1302 Telephone: 702.562.8820	
	Facsimile: 702.562.8821	
7	JBailey@BaileyKennedy.com JDickey@BaileyKennedy.com	
8	AChampion@BaileyKennedy.com	
9	C. KEITH GREER, ESQ.	
10	Cal. Bar. No. 135537 (<i>Pro Hac Vice</i>) GREER AND ASSOCIATES, APC	
	16855 West Bernardo Dr. Suite 255	
11	San Diego, California 92127 Telephone: 858.613.6677	
12	Facsimile: 858.613.6680	
13	keith.greer@greerlaw.biz	
14	Attorneys for Defendants LAS VEGAS DEVELOPMENT FUND LLC;	
	EB5 IMPACT CAPITAL REGIONAL CENTER	
15	LLC; EB5 IMPACT ADVISORS LLC; ROBERT W. DZIUBLA; JON FLEMING; and	
16	LINDA STANWOOD	
17	DISTRICT	COURT
18	CLARK COUN	TY, NEVADA
19	FRONT SIGHT MANAGEMENT LLC, a	1
20	Nevada Limited Liability Company,	Case No. A-18-781084-B Dept. No. XVI
	Plaintiff,	Bep. No. 11 1
21	vs.	NOTICE OF ENTRY OF ORDER
22	LAS VEGAS DEVELOPMENT FUND LLC, a	GRANTING DEFENDANT AND COUNTERCLAIMANT LAS VEGAS
23	Nevada Limited Liability Company; et al,	DEVELOPMENT FUND, LLC'S
24	Defendants.	NOTICE OF MOTION AND MOTION FOR LEAVE TO AMEND THE
25		COUNTERCOMPLAINT
26	-]
	AND ALL RELATED COUNTERCLAIMS.	
27		
28		
	Page 1	l of 3
	- 6	

Page 3 of 3

BAILEY * KENNEDY 8984 SPANISH RIDGE AVENUE LAS VEGAS, NEVADA 89148-1302 702.562.8820

1		Electronically Filed 6/4/2020 2:22 PM
1	ODDD	Steven D. Grierson CLERK OF THE COURT
1	ORDR JOHN R. BAILEY	Stevent Street
2	Nevada Bar No. 0137	
3	JOSHUA M. DICKEY Nevada Bar No. 6621	
3	ANDREA M. CHAMPION	
4	Nevada Bar No. 13461	
5	BAILEY	
	Las Vegas, Nevada 89148-1302	
6	Telephone: 702.562.8820 Facsimile: 702.562.8821	
7	JBailey@BaileyKennedy.com	
O	JDickey@BaileyKennedy.com	
8	AChampion@BaileyKennedy.com	
9	C. KEITH GREER, ESQ.	
10	Cal. Bar. No. 135537 (<i>Pro Hac Vice</i>) GREER AND ASSOCIATES, A PC	
10	16855 West Bernardo Dr. Suite 255	
11	San Diego, California 92127	
12	Telephone: 858.613.6677 Facsimile: 858.613.6680	
12	keith.greer@greerlaw.biz	
13	Attorneys for Defendants	
14	LAS VEGAS DEVELOPMENT FUND LLC;	
15	EB5 IMPACT CAPITAL REGIONAL CENTER LLC; EB5 IMPACT ADVISORS LLC; ROBERT	
13	W. DZIUBLA; JON FLEMING; and	
16	LINDA STANWOOD	
17	DISTRICT	COURT
18	CLARK COUNT	ΓY, NEVADA
19	FRONT SIGHT MANAGEMENT LLC, a	ı
	Nevada Limited Liability Company,	Case No. A-18-781084-B
20	D1-:4:00	Dept. No. XVI
21	Plaintiff,	ORDER GRANTING DEFENDANT AND
22	vs.	COUNTERCLAIMANT LAS VEGAS
22	LAS VEGAS DEVELOPMENT FUND LLC, a	DEVELOPMENT FUND, LLC'S NOTICE OF MOTION AND MOTION
23	Nevada Limited Liability Company; et al.,	FOR LEAVE TO AMEND THE
24	Defendants.	COUNTERCOMPLAINT
	Defendants.	
25		
26		
27	AND ALL RELATED COUNTERCLAIMS.	
28		
	Page 1	of 2
	1 100 -	

Case Number: A-18-781084-B

1	This matter came before the Court on M	Tay 13, 2020, at 10:30 a.m. on Defendant and
2	Counterclaimant Las Vegas Development Fund	, LLC's Notice of Motion and Motion for Leave to
3	Amend the Countercomplaint (the "Motion"). J	John P. Aldrich appeared on behalf of Plaintiff Front
4	Sight Management LLC; and John R. Bailey, A	andrea M. Champion, C. Keith Greer, and Kathryn
5	Holbert appeared on behalf of Defendants and C	Counterclaimant Las Vegas Development Fund, LLC.
6	The Court having reviewed the pleadings on file	e herein, having heard oral argument by the parties,
7	and for good cause appearing therefore,	
8	IT IS HEREBY ORDERED that the M	Iotion is GRANTED.
9	IT IS SO ORDERED this 4th day of .	June, 2020.
10		J: Hairm
11		HONORABLE TIMOTHY C. WILLIAMS
12		DISTRICT COURT JUDGE CG
13	Respectfully submitted by:	Approved as to form and content:
14	BAILEY KENNEDY, LLP	ALDRICH LAW FIRM, LTD.
15		
16	<u>/s/ Andrea M. Champion</u> JOHN R. BAILEY	<u>/s/ John P. Aldrich</u> John P. Aldrich
17	Nevada Bar No. 0137 Joshua M. Dickey	Nevada Bar No. 6877
18	Nevada Bar No. 6621	CATHERINE HERNANDEZ Nevada Bar No. 8410
19	Andrea M. Champion Nevada Bar No. 13461	7866 West Sahara Avenue
20	8984 Spanish Ridge Avenue Las Vegas, Nevada 89148-1302	Las Vegas, Nevada 89117 Tel: 702.853.5490
	Telephone: 702.562.8820	Fax: 702.227.1975 jaldrich@johnaldrichlawfirm.com
21	Facsimile: 702.562.8821 JBailey@BaileyKennedy.com	chernandez@johnaldrichlawfirm.com
22	JDickey@BaileyKennedy.com AChampion@BaileyKennedy.com	Attorneys for Plaintiff
23	Attorneys for Defendants	FRONT SIGHT MANAGEMENT LLC
24	LAS VEGAS DEVELOPMENT FUND LLC; EB5 IMPACT CAPITAL REGIONAL	
25	CENTER LLC; EB5 IMPACT ADVISORS LLC; ROBERT W. DZIUBLA; JON	
26	FLEMING; and LINDA STANWOOD	
27		
28		
	Pa	ge 2 of 2

BAILEY * KENNEDY 8984 SPANISH RIDGE AVENUE LAS VEGAS, NEVADA 89148-1302 702.562.8820

1	AACC	6/4/2020 4:40 PM Steven D. Grierson CLERK OF THE COURT
2	JOHN R. BAILEY Nevada Bar No. 0137	Deliver.
3	JOSHUA M. DICKEY Nevada Bar No. 6621	
4	Andrea M. Champion Nevada Bar No. 13461	
5	BAILEY * KENNEDY	
6	8984 Spanish Ridge Avenue Las Vegas, Nevada 89148-1302 Telephone: 702.562.8820	
7	Facsimile: 702.562.8821 JBailey@BaileyKennedy.com	
8	JDickey@BaileyKennedy.com AChampion@BaileyKennedy.com	
9	C. KEITH GREER, ESQ.	
10	Cal. Bar. No. 135537 (Pro Hac Vice) GREER AND ASSOCIATES, A PC	
11	16855 West Bernardo Dr. Suite 255 San Diego, California 92127	
12	Telephone: 858.613.6677 Facsimile: 858.613.6680	
13	keith.greer@greerlaw.biz	
14	Attorneys for Defendants LAS VEGAS DEVELOPMENT FUND LLC;	
15	EB5 IMPACT CAPITAL REGIONAL CENTER LLC; EB5 IMPACT ADVISORS LLC; ROBERT	
16	W. DZIUBLA; JON FLEMING; and LINDA STANWOOD	
17	DISTRICT	r court
18	CLARK COUN	
19	CLARK COON	II, NEVADA
	FRONT SIGHT MANAGEMENT LLC, a	G N A 10 501004 P
20	Nevada Limited Liability Company,	Case No. A-18-781084-B Dept. No. XVI
21	Plaintiff,	
22	VS.	DEFENDANTS' ANSWER TO
23	LAS VEGAS DEVELOPMENT FUND LLC, a Nevada Limited Liability Company; et al,	PLAINTIFF'S SECOND AMENDED COMPLAINT; AND FIRST AMENDED
24		COUNTERCLAIM
25	Defendants.	
26		-
27	AND ALL RELATED COUNTERCLAIMS.	
28		
	Page 1	of 42
	1	

Case Number: A-18-781084-B

Electronically Filed

28

///

Page 2 of 42

are or were officers of Defendants EB5IA, EB5IC, and LVDF. However, these responding

Defendants deny the remainder of the allegations in Paragraph 10 of Plaintiff's Second Amended Complaint.

GENERAL ALLEGATIONS

<u>Inducement of Front Sight to Fund Defendants' EB 5 Raise for the Development and</u>
<u>Construction of the Front Sight Resort Project in Detrimental Reliance on a Raise of \$75 Million</u>

- 11. These responding Defendants admit that Defendants and Plaintiff exchanged email correspondence. However, these responding Defendants deny Plaintiffs the remainder of the allegations in Paragraph 11 of Plaintiff's Second Amended Complaint.
- 12. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 12 of Plaintiff's Second Amended Complaint.
- 13. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 13 of Plaintiff's Second Amended Complaint.
- 14. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 14 of Plaintiff's Second Amended Complaint.
- 15. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 15 of Plaintiff's Second Amended Complaint.
- 16. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 16 of Plaintiff's Second Amended Complaint.
- 17. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 17 of Plaintiff's Second Amended Complaint.
- 18. These responding Defendants deny the allegations in Paragraph 18 of Plaintiff's Second Amended Complaint.

Page 3 of 42

26. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 26 of Plaintiff's Second Amended Complaint.

28 ///

26

2

3

4

5

6

7

8

9

10

12

13

14

15

17

18

19

20

21

22

23

24

25

26

27

28

///

- 27. These responding Defendants admit that the Regional Center Application was filed on or about April 14, 2014 and that the application was approved on or about July 27, 2015, and deny the remaining allegations in Paragraph 27 of Plaintiff's Second Amended Complaint.
- 28. These responding Defendants admit that the application for EB5 Impact Capital Regional Center, LLC was filed on April 15, 2014. However, these responding Defendants deny the remainder of the allegations in Paragraph 28 of Plaintiff's Second Amended Complaint.
- 29. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 29 of Plaintiff's Second Amended Complaint.
- 30. These responding Defendants admit that the application for EB5 Impact Capital Regional Center, LLC was approved on July 27, 2015. However, these responding Defendants deny the remainder of the allegations in Paragraph 30 of Plaintiff's Second Amended Complaint.
- 31. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 31 of Plaintiff's Second Amended Complaint.
- 32. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 32 of Plaintiff's Second Amended Complaint.
- 33. These responding Defendants admit to the existence of a website identified as "eb5impactcapital.com," and deny the allegations in Paragraph 33 of Plaintiff's Second Amended Complaint.
- 34. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 34 of Plaintiff's Second Amended Complaint.
- 35. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 35 of Plaintiff's Second Amended Complaint.

Page 5 of 42

- 36. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 36 of Plaintiff's Second Amended Complaint.
- 37. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 37 of Plaintiff's Second Amended Complaint.
- 38. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 38 of Plaintiff's Second Amended Complaint.
- 39. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the allegations in Paragraph 39 of Plaintiff's Second Amended Complaint.
- 40. These responding Defendants admit that LVD Fund has loaned Front Sight \$6,375,000 and deny the rest of the allegations in Paragraph 40 of Plaintiff's Second Amended Complaint.
- 41. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 41 of Plaintiff's Second Amended Complaint.
- 42. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the remainder of the allegations in Paragraph 42 of Plaintiff's Second Amended Complaint.
- 43. These responding Defendants deny the allegations in Paragraph 43 of Plaintiff's Second Amended Complaint.
- 44. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the allegations in Paragraph 44 of Plaintiff's Second Amended Complaint.
- 45. These responding Defendants deny the allegations in Paragraph 45 of Plaintiff's Second Amended Complaint.

Page 6 of 42

- 57. These responding Defendants admit that Defendants and Plaintiff attempted to resolve the issues regarding Plaintiff's Defaults regarding the Construction Loan Agreement.

 However, these responding Defendants deny the remainder of the allegations in Paragraph 57 of Plaintiff's Second Amended Complaint.
- 58. These responding Defendants admit that Defendant Las Vegas Development Fund recorded a Notice of Default on September 11, 2018. However, these responding Defendants deny the remainder of the allegations in Paragraph 58 of Plaintiff's Second Amended Complaint.
- 59. These responding Defendants admit that Defendants and Plaintiff exchanged correspondence. However, these responding Defendants deny the allegations in Paragraph 59 of Plaintiff's Second Amended Complaint.
- 60. These responding Defendants deny the allegations in Paragraph 60 of Plaintiff's Second Amended Complaint.
- 61. These responding Defendants admit that a Court order was entered regarding Plaintiff's Petition for Appointment of Receiver and for an Accounting. However, these responding Defendants deny the remainder of the allegations in Paragraph 61 of Plaintiff's Second Amended Complaint.
- 62. These responding Defendants admit they have complied with the Court order which was entered regarding Plaintiff's Petition for Appointment of Receiver and for an Accounting.

 However, these responding Defendants deny the remainder of the allegations in Paragraph 62 of Plaintiff's Second Amended Complaint.
- 63. These responding Defendants deny the allegations in Paragraph 63 of Plaintiff's Second Amended Complaint.
- 64. These responding Defendants admit Plaintiff is entitled to a \$36,000.00 offset. However, these responding Defendants deny the remainder of the allegations in Paragraph 64 of Plaintiff's Second Amended Complaint.
- 65. These responding Defendants admit Defendant EB5IA has been dissolved.

 However, these responding Defendants deny the remainder of the allegations in Paragraph 65 of Plaintiff's Second Amended Complaint.

Page 8 of 42

Page **9** of **42**

Page **10** of **42**

1	91.	These responding Defendants deny the allegations in Paragraph 91 of Plaintiff's	
2	Second Amended Complaint.		
3	92.	These responding Defendants deny the allegations in Paragraph 92 of Plaintiff's	
4	Second Amer	nded Complaint.	
5	93.	These responding Defendants deny the allegations in Paragraph 93 of Plaintiff's	
6	Second Amended Complaint.		
7	94.	These responding Defendants deny the allegations in Paragraph 94 of Plaintiff's	
8	Second Amended Complaint.		
9 10	FOURTH CAUSE OF ACTION (Civil Conspiracy Against All Defendants)		
11	95.	These responding Defendants repeat and re-allege their responses to each of the	
12	preceding and succeeding paragraphs as though fully set forth herein.		
13	96.	These responding Defendants deny the allegations in Paragraph 96 of Plaintiff's	
14	Second Amended Complaint.		
15	97.	These responding Defendants deny the allegations in Paragraph 97 of Plaintiff's	
16	Second Amended Complaint.		
17	98.	These responding Defendants deny the allegations in Paragraph 98 of Plaintiff's	
18	Second Amended Complaint.		
19	99.	These responding Defendants deny the allegations in Paragraph 99 of Plaintiff's	
20	Second Amer	nded Complaint.	
21		FIFTH CAUSE OF ACTION (Breach of Contract Against All Defendants EB5IA and LVDF)	
22		(Dreach of Contract Against An Defendants EDSIA and LVDI)	
23	100.	These responding Defendants repeat and re-allege their responses to each of the	
24	preceding and succeeding paragraphs as though fully set forth herein.		
25	101.	These responding Defendants admit the allegations in Paragraph 101 of Plaintiff's	
26	Second Amended Complaint.		
27	102.	These responding Defendants deny the allegations in Paragraph 102 of Plaintiff's	
28	Second Amer	nded Complaint.	
		Page 11 of 42	

103.	These responding Defendants deny the allegations in Paragraph 103 of Plaintiff's	
Second Amended Complaint.		
104.	These responding Defendants deny the allegations in Paragraph 104 of Plaintiff's	
Second Amer	nded Complaint.	
105.	These responding Defendants deny the allegations in Paragraph 105 of Plaintiff's	
Second Amended Complaint.		
106.	These responding Defendants deny the allegations in Paragraph 106 of Plaintiff's	
Second Amended Complaint.		
SIXTH CAUSE OF ACTION (Contractive Presch of Implied Covenant of Cood Feith and Feir Dealing		
(Contractual Breach of Implied Covenant of Good Faith and Fair Dealing Against the Entity Defendants)		
Plaintiff's Sixth Cause of Action has been dismissed as against Defendant EB5IC pursuant to this		
Court's Order filed April 9, 2019.		
107.	These responding Defendants repeat and re-allege their responses to each of the	
preceding and succeeding paragraphs as though fully set forth herein.		
108.	These responding Defendants admit the allegations in Paragraph 108 of Plaintiff's	
Second Amended Complaint.		
109.	These responding Defendants admit the allegations in Paragraph 109 of Plaintiff's	
Second Amer	nded Complaint.	
110.	These responding Defendants admit the allegations in Paragraph 110 of Plaintiff's	
Second Amended Complaint.		
111.	These responding Defendants deny the allegations in Paragraph 111 of Plaintiff's	
Second Amended Complaint.		
112.	These responding Defendants deny the allegations in Paragraph 112 of Plaintiff's	
Second Amer	nded Complaint.	
113.	These responding Defendants deny the allegations in Paragraph 113 of Plaintiff's	
Second Amended Complaint.		
///		
///	D 10 C10	
	Second Amer 104. Second Amer 105. Second Amer 106. Second Amer (C Plaintiff's Six Court's Order 107. preceding and 108. Second Amer 109. Second Amer 110. Second Amer 111. Second Amer 112. Second Amer 113. Second Amer 113.	

SEVENTH CAUSE OF ACTION

(Tortious Breach of Implied Covenant of Good Faith and Fair Dealing Against the Entity Defendants)

114-121. Plaintiff's Seventh Cause of Action has been dismissed as against the Entity Defendants pursuant to this Court's Order filed April 9, 2019.

EIGHTH CAUSE OF ACTION

(Intentional Interference With Prospective Economic Advantage Against the Entity Defendants and Defendant Dziubla)

Plaintiff's Eighth Cause of Action has been dismissed as against the Entity Defendants EB5IC and EB5IA pursuant to this Court's Order filed April 9, 2019. Therefore, Defendants Dziubla and LVD Fund respond as follows:

- 122. These responding Defendants repeat and re-allege their responses to each of the preceding and succeeding paragraphs as though fully set forth herein.
- 123. These responding Defendants lack sufficient information to admit or deny the allegations in Paragraph 123 of Plaintiff's Second Amended Complaint and, therefore, deny the same.
- 124. These responding Defendants deny the allegations in Paragraph 124 of Plaintiff's Second Amended Complaint.
- 125. These responding Defendants deny the allegations in Paragraph 125 of Plaintiff's Second Amended Complaint.
- 126. These responding Defendants deny the allegations in Paragraph 126 of Plaintiff's Second Amended Complaint.
- 127. These responding Defendants deny the allegations in Paragraph 127 of Plaintiff's Second Amended Complaint.
- 128. These responding Defendants deny the allegations in Paragraph 128 of Plaintiff's Second Amended Complaint.

NINTH CAUSE OF ACTION

(Unjust Enrichment Against All Defendants)

129-135. Plaintiff's Ninth Cause of Action has been dismissed as against all Defendants pursuant to this Court's Order filed April 9, 2019.

Page **13** of **42**

1 TENTH CAUSE OF ACTION (Negligent Misrepresentation Against all Defendants) 2 3 Plaintiff's Tenth Cause of Action has been dismissed as against Defendants Stanwood, Fleming, EB5IC, and LVDF pursuant to this Court's Order filed April 9, 2019. Therefore, Defendants EB5IA 4 and Dziubla respond as follows: 5 136. These responding Defendants repeat and re-allege their responses to each of the 6 preceding and succeeding paragraphs as though fully set forth herein. 7 137. These responding Defendants admit the allegations in Paragraph 137 of Plaintiff's 8 Second Amended Complaint. 9 These responding Defendants deny the allegations in Paragraph 138 of Plaintiff's 138. 10 Second Amended Complaint. 11 139. These responding Defendants deny the allegations in Paragraph 139 of Plaintiff's 12 Second Amended Complaint. 13 140. These responding Defendants deny the allegations in Paragraph 140 of Plaintiff's 14 Second Amended Complaint. 15 141. These responding Defendants deny the allegations in Paragraph 141 of Plaintiff's 16 Second Amended Complaint. 17 142. These responding Defendants deny the allegations in Paragraph 142 of Plaintiff's 18 Second Amended Complaint. 19 143. These responding Defendants deny the allegations in Paragraph 143 of Plaintiff's 20 Second Amended Complaint. 21 144. These responding Defendants deny the allegations in Paragraph 144 of Plaintiff's 22 Second Amended Complaint. 23 145. These responding Defendants deny the allegations in Paragraph 145 of Plaintiff's 24 Second Amended Complaint. 25 **ELEVENTH CAUSE OF ACTION** (Negligence Against All Defendants) 26 27 Plaintiff's Eleventh's Cause of Action has been dismissed as against all Defendants pursuant to this Court's Order filed April 9, 2019. 28

Page 14 of 42

1 TWELFTH CAUSE OF ACTION (Alter Ego Against All Defendants) 2 3 151-160. Plaintiff's Twelfth Cause of Action has been dismissed as against all Defendants 4 pursuant to this Court's Order filed April 9, 2019. 5 These responding Defendants, LAS VEGAS DEVELOPMENT FUND LLC; EB5 IMPACT 6 CAPITAL REGIONAL CENTER LLC; EB5 IMPACT ADVISORS LLC, a dissolved Nevada 7 Limited Liability Company; ROBERT W. DZIUBLA; JON FLEMING; and LINDA STANWOOD, 8 by and through their counsel of record, Bailey Kennedy, having fully and specifically responded to 9 each and every allegation set forth in Plaintiff's Second Amended Complaint, now assert the 10 following: 11 AFFIRMATIVE DEFENSES 12 FIRST AFFIRMATIVE DEFENSE 13 Plaintiff's Second Amended Complaint fails to state a claim for which relief can be granted 14 as against these responding Defendants. 15 SECOND AFFIRMATIVE DEFENSE 16 These responding Defendants generally deny all liability and all allegations of negligence or 17 wrongdoing. 18 THIRD AFFIRMATIVE DEFENSE 19 Any allegations or factual matters asserted by Plaintiff that are not specifically admitted are 20 hereby denied. 21 FOURTH AFFIRMATIVE DEFENSE 22 The claims referred to in Plaintiff's Second Amended Complaint, and the resulting damage-23 if any—to Plaintiff, was proximately caused or contributed to by Plaintiff's own negligence and, as 24 such, Plaintiff's negligence was greater than the negligence—if any—of these responding 25 Defendants and therefore, Plaintiff's recovery should be barred or diminished. 26 /// 27 /// 28 /// Page 15 of 42

1 FIFTH AFFIRMATIVE DEFENSE 2 If Plaintiff has been damaged as alleged, then said damages are the sole, direct, and 3 proximate result of actions and/or inactions of other named parties and/or third parties not presently 4 named herein over which these responding Defendants had no control. 5 SIXTH AFFIRMATIVE DEFENSE 6 These responding Defendants reserve the right to assert any and all defenses raised by any 7 other party to this action. 8 SEVENTH AFFIRMATIVE DEFENSE 9 These responding Defendants reserve the right to amend their Answer and/or assert 10 additional affirmative defenses based upon discovery as well as an investigation of the facts and 11 circumstances concerning the alleged incident that is the subject of Plaintiff's Amended Complaint. 12 EIGHTH AFFIRMATIVE DEFENSE 13 As a separate and distinct affirmative defense, these responding Defendants allege that, to the 14 extent that Plaintiff's Amended Complaint alleges violations of law, those alleged violations of law 15 are the result of the conduct or omissions of persons or entities other than these responding 16 Defendants. 17 NINTH AFFIRMATIVE DEFENSE 18 Plaintiff is barred from asserting any claims against these responding Defendants because 19 the alleged damages were the result of the intervening and/or superseding conduct of others. 20 TENTH AFFIRMATIVE DEFENSE 21 Plaintiff's claims are barred by the doctrine of laches and/or the statute of limitation. 22 ELEVENTH AFFIRMATIVE DEFENSE 23 These responding Defendants reserve the right to seek contribution and indemnity in the 24 event that these responding Defendants deem it appropriate to do so. 25 TWELFTH AFFIRMATIVE DEFENSE 26 As a separate and distinct affirmative defense, these responding Defendants allege that, 27 before the commencement of this action, these responding Defendants performed, satisfied, and 28 discharged all duties and obligations they may have owed to Plaintiff. Page 16 of 42

THIRTEENTH AFFIRMATIVE DEFENSE

Plaintiff's claims are barred by the doctrine of unclean hands.

FOURTEENTH AFFIRMATIVE DEFENSE

Plaintiff's claims are barred because Plaintiff was the first party to breach the contract and cannot maintain an action against the Defendants for a subsequent failure to perform.

FIFTEENTH AFFIRMATIVE DEFENSE

Plaintiff's claims are barred because the alleged tortious act by Defendants was justified and/or privileged.

SIXTEENTH AFFIRMATIVE DEFENSE

Plaintiff's claims are barred because all alleged injuries and damages, if any, were caused by the acts or omissions of Plaintiff.

SEVENTEENTH AFFIRMATIVE DEFENSE

Plaintiff's claims are barred because Defendants complied with applicable statutes and with the requirements and regulations of the State of Nevada.

FIRST AMENDED COUNTER CLAIM

1. This First Amended Counterclaim stems from Front Sight's misappropriation and diversion of construction loan proceeds for the personal benefit of its principal, Ignatius Piazza, his wife Jennifer Piazza, and beneficiaries of the VNV Trust Defendants, and Front Sight's breach of multiple material provisions of the Construction Loan Agreement (the "CLA")¹, including its failure to meet the construction schedule, material changes to the Project scope, failure to provide government approved construction plans, failure to obtain Senior Debt, failure to meet its reporting obligations to Lender under the CLA and EB-5 regulations, refusal to give Lender access to its books and records, refusal to allow a site inspection and answer questions by Lender's representatives, failure to pay default interest, further encumbering the Property by selling securities, and failure to pay Lender's legal fees relating to enforcing Borrower to comply with the terms of the CLA. Moreover, Borrower's recent actions of delaying construction, refusing to grant Lender's

Page 17 of 42

¹ "CLA" refers to the Construction Loan Agreement dated October 6, 2016, between Front Sight Management LLC ("Borrower") and Las Vegas Development Fund LLC ("Lender"). (See Dziubla Decl., Ex. 3).

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

representatives access to the property and concealing its books and records, raise serious questions regarding Front Sight's continued solvency (which is a required loan covenant) and thus, its ability to complete the Project.

2. This First Amended Counterclaim is further based upon Counter Defendants entering into a comprehensive scheme to defraud LVD Fund by falsely representing that Counter Defendant Front Sight had entered into a legitimate and bona fide \$36,000,000 "Loan Agreement – Construction Line of Credit" with Counter Defendant Morales Construction, Inc. ("Morales Construction"), that would have provided sufficient capital to make substantial progress toward completing the project. In reality, the "Loan Agreement" was a complete scam because all of the Counter Defendants knew Morales was not capable of fulfilling its obligation to extend tens of millions of dollars in credit, and none of the Counter Defendants ever intended to perform under the Loan Agreement.

I. PARTIES

- 3. Counter Claimant LAS VEGAS DEVELOPMENT FUND LLC (hereafter "LVD Fund" or "Lender") is a Nevada limited liability company with a principal place of business located in Nevada and has an interest and right in the Property through a certain Deed of Trust² that was by and between Front Sight and LVD FUND.
- FRONT SIGHT MANAGEMENT LLC (hereinafter as "Front Sight" or "Borrower") 4. is a Nevada limited liability company with a principal place of business located in Clark County, Nevada.
- 5. Counter Claimant is informed and believes, and on that basis alleges, Counter Defendant VNV DYNASTY TRUST I is a Nevada statutory trust, Nevada business, family trust, or other irrevocable trust that functions as an entity and that may claim title and ownership interest in the Property. Counter Claimant is informed and believes, and on that basis alleges, Counter

Page 18 of 42

² "Deed of Trust" refers to the "Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents, and Fixture Filing," recorded in the official records of Nye County, Nevada, as "DOC #860867" on October 13, 2016, a copy of which is attached as Exhibit 1, filed herewith, as amended by the "First Amendment to Construction Deed of Trust, Security Agreement and Fixture Filing," recorded in the official records of Nye County, Nevada, as "DOC #886510" on January 12, 2018, a copy of which is provided as Exhibit 2.

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

Defendant VNV DYNASTY TRUST I was organized and exists under the laws of Nevada and Counter Defendants IGNATIUS A. PIAZZA II and JENNIFER PIAZZA are trustees and/or beneficiaries of the VNV DYNASTY TRUST I.

- 6. Counter Claimant is informed and believes, and on that basis alleges, Counter Defendant VNV DYNASTY TRUST II is a Nevada statutory trust, Nevada business, family trust, or other irrevocable trust that functions as an entity and that may claim title and ownership interest in the Property. Counter Claimant is informed and believes, and on that basis alleges, Counter Defendant VNV DYNASTY TRUST II was organized and exists under the laws of Nevada and Counter Defendants IGNATIUS A. PIAZZA II and JENNIFER PIAZZA are trustees and/or beneficiaries of the VNV DYNASTY TRUST II. (Hereinafter, VNV DYNASTY TRUST I and VNV DYNASTY TRUST II are collectively referred to as the "VNV Trust Defendants" or "Trust Defendants").
- 7. Counter Claimant is informed and believes, and on that basis alleges, that Counter Defendant IGNATIUS A. PIAZZA II, ("Piazza"), is an individual who is, and at all times relevant hereto was, a resident of Sonoma County, California. Piazza is the managing member, or otherwise in control under another title, of Counter Defendant Front Sight Management, LLC and Trustee and/or beneficiary of VNV Trust Defendants.
- 8. Counter Claimant is informed and believes, and on that basis alleges, that DEFENDANT JENNIFER PIAZZA, is an individual who is, and at all times relevant hereto was, a resident of Sonoma County, California, and is Trustee and/or beneficiary of VNV Trust Defendants.
- 9. Counter Defendant MORALES CONSTRUCTION, INC. ("MORALES CONSTRUCTION") is a Nevada Corporation and licensed contractor with its principal place of business in Pahrump, Nevada.
- 10. Counter Defendant ALL AMERICAN CONCRETE & MASONRY INC. ("ALL AMERICAN CONCRETE") is a Nevada Corporation and licensed contractor with its principal place of business in Pahrump, Nevada.
- Counter Defendant TOP RANK BUILDERS INC. ("TOP RANK BUILDERS") is a 11. Nevada Corporation and licensed contractor with its principal place of business in Pahrump, Nevada. Page 19 of 42

12.	Counter Claimant is informed and believes, and on such basis alleges, that Counter
Defendant EI	FRAIN RENE MORALES-MORENO ("MORALES") is, and at all times relevant was,
a resident of l	Nye County, Nevada, and the principal and chief executive officer of MORALES
CONSTRUC	TION, ALL AMERICAN, and TOP RANK.

- 13. Counter Claimant is informed and believes, and on such basis alleges, that Counter Defendant MICHAEL GENE MEACHER ("MEACHER") is, and at all times relevant, was a resident of Nye County, Nevada, and the Vice President and Chief Operating Officer of Counter Defendant FRONT SIGHT.
- 14. Upon information and belief, each of the Counter Defendants sued herein as ROE Counter Defendants 1 through 10, inclusive, are beneficiaries or trustees of the Trust Defendants and claim an interest in the Property or are responsible in some manner for the events and happenings herein that Counter Claimant seeks to enjoin; that when the true names and capacities of such defendants become known, Counter Claimant will ask leave of this Court to amend this counterclaim to insert the true names, identities, and capacities together with proper charges and allegations.
- 15. Counter Claimant is informed and believes, and on that basis alleges, that Counter Defendants Front Sight and the VNV Trust Defendants are influenced and governed by Counter Defendant Ignatius Piazza, and they are so intertwined with one another as to be factually and legally indistinguishable. As such, the adherence to an LLC, corporate, or trust fiction of separate entities would, under the circumstances, sanction fraud and promote injustice.
- 16. As a result of Front Sight being the alter ego of Counter Defendant Ignatius Piazza, Ignatius Piazza is personally liable for the liabilities of Front Sight regarding the allegations set forth in this Counterclaim.

II. GENERAL ALLEGATIONS

17. The CLA was made to fund construction of the Front Sight Resort & Vacation Club ("FS Resort") and an expansion of the facilities and infrastructure of the Front Sight Firearms

Training Institute (the "Training Facilities") located on a 550-acre site in Pahrump, Nevada (the "Project"). The CLA dated October 6, 2016 (Exhibit 3) is the operative agreement for purposes of

Page 20 of 42

determining Front Sight's obligations as the "Borrower," and the remedies available to LVD Fund as the "Lender."

- 18. The "Project" is described as construction of the Front Sight Resort & Vacation Club ("FSRVC") and an expansion of the facilities and infrastructure of the Front Sight Firearms Training Institute ("FSFTI") (the "Facilities") located in a 550 acre site in Pahrump, Nevada. The Facilities will include 102 timeshare residential units, up to 150 luxury timeshare RV pads, an 85,000 square foot restaurant, retail, classroom, and office building (to be known as the Patriot Pavilion) and related infrastructure and amenities, all of which will be located at One Front Sight Road, Pahrump, Nevada 89041.
- Immigrant Investor Program, known as "EB-5." The EB 5 Immigrant Investor Program, which is administered by the United States Citizenship and Immigration Services ("USCIS"), provides certain immigrant investors, who can demonstrate that their investments are creating jobs in this country, with a potential avenue to lawful permanent residency in the United States. The program sets aside EB-5 visas for participants who invest in commercial enterprises approved by USCIS, frequently administered by entities called "regional centers." Each investor is required to invest a minimum of \$500,000 and, through the EB-5 Immigrant Investor Program, is anticipated to receive permanent foreign resident status within the United States assuming compliance with the EB-5 program requirements and creation of 10 US jobs per investor. Material departures from the USCIS approved plans for the Project, including delays in construction, and diversion of funds from the Project to general corporate or personal uses, are all significant breaches of the CLA and potentially jeopardize the immigration status of the EB-5 Investors.
- 20. According to the USCIS, the Immigrant Investor Program, also known as "EB-5," was created to stimulate the U.S. economy through job creation and capital investment from immigrant investors by creating a new commercial enterprise or investing in a troubled business. In this case, the immigrant investors are attempting to gain lawful permanent residence for themselves and their families by participating in a Regional Center Pilot Program, which requires them to make a capital investment of \$500,000, since this region is deemed to be a Targeted Employment Area Page 21 of 42

which failure continues more than 5 days following notice of such failure from Lender."

24. In the event of default, Lender can, inter alia: suspend the obligation to make further advances of funds (CLA §6.2(b)); foreclose on the Deed of Trust (CLA §6.2(e)); and "take over and complete such construction in accordance with the Plans, with such changes therein as Lender may, in its discretion, deem appropriate, all at the risk, cost and expense of Borrower." (CLA §6.3). [emphasis added]

BORROWER'S BREACHES AND DEFAULT UNDER THE CLA

A. Breach Number 1: Improper Use of Loan Proceeds - CLA § 1.7(e)

- 25. Section 1.7(e) of the CLA provides that "Borrower shall use the proceeds of the Loan solely for the purpose of funding directly, or advancing to Affiliates to pay, the costs of the Project, in accordance with the terms and conditions of this Agreement, as set forth in the Budget and the Project documents submitted to, and approved by, USCIS." However, in its October 30, 2018 prove-up to LVD Fund regarding EB-5 compliance, Front Sight revealed that although it has spent all of the \$6,375,000 in loan proceeds since the initial disbursement in October 2016, only approximately \$2,690,000 of the proceeds were actually spent on construction of the EB-5 project.
- 26. Counter Claimants are informed and believe, and thereon allege, that more than \$3,675,000 of EB-5 loan proceeds have been diverted to fund matters that are not related to completion of the approved EB-5 plan, such as payment of Front Sight's general overhead expenses, thereby severely prejudicing the EB-5 investors.
- 27. Counter Claimants are informed and believe, and thereon allege, that during the past two years, while Front Sight has been using EB-5 (CLA) loan proceeds to pay its general overhead operating costs, pay off a pre-existing loan for which Ignatius Piazza and Jennifer Piazza are personal guarantors, and disburse multi-million shareholder distributions to Counter Defendants Ignatius Piazza, Jennifer Piazza, and the VNV Trust Defendants.

B. Breach Number 2: Failure to Provide Government Approved Plans-CLA §3.2(b)

28. Section 3.2 (b)(I) of the CLA requires that, prior to the Commencement Date, Front Sight provide LVD Fund with "Plans, in the form previously submitted to Lender, as finally Page 23 of 42

approved for construction by the Project Architect and the applicable Governmental Authority." (Exhibit 3, pg. 20). The "Commencement Date" for the Project is defined in the First Amendment to Loan Agreement effective July 1, 2017 as "October 6, 2016." (Exhibit 4). This is to include "a schedule listing all Contractors, and primary contracts relating to the Project having a contract sum in excess of \$250,000 for any such Contractor, and construction contracts, subcontracts and schedules relating to the Project." (Id. CLA §3.2(b)(ii)). In a letter dated August 28, 2018, Robert Dziubla, on behalf of LVD Fund, gave notice to Front Sight that it was in default for failure to provide construction plans and the related lists of contractors, licenses, agreements, and permits relating to the construction as required under §§3.2(b)(I) and (ii) of the CLA. Front Sight remains in default under these provisions of the CLA.

C. Breach Number 3: Failure to Timely Complete Construction - CLA § 5.1

- 29. Pursuant to Section 5.1 of the CLA, Front Sight was required to complete construction by the "Completion Date" which is defined as "the date that is no later than thirty-six (36) months from the Commencement Date." (Exhibit 3 pg. 3). Pursuant to the First Amendment to the Loan Agreement, the "Commencement Date" is defined as "October 4, 2016." (Exhibit 4, §1). Therefore, construction of the project must be completed on or before October 4, 2019.
- 30. Front Sight has explicitly acknowledged in writing that it is in default of this requirement, warning LVD Fund in a letter dated August 25, 2018 that ". . . the foreclosure killed the project when it was 18 months away from being completed." Even by Counter Defendant Front Sight's written projection as of August 25, 2018, the Project would not be completed by the contractual Completion Date of October 4, 2019, i.e., 36 months after the commencement date as stated in the First Amendment to Loan Agreement.
- 31. This is a material event of Default, and it is particularly prejudicial to the EB-5 investors who risk losing their EB-5 benefits if the project is not completed in accordance with the schedule approved by the USCIS.

D. Breach Number 4: Material Change of Costs, Scope, or Timing of Work - CLA § 5.2

32. Section 5.2 of the CLA states in pertinent part:

///

Page 24 of 42

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

Project, showing changes in or variations from the original Estimated Construction Cost Statement, as soon as such changes are known to Borrower. Borrower shall deliver to Lender a revised construction schedule, if and when any target date set forth therein has been delayed by twenty (20) consecutive days or more, or when the aggregate of all such delays equals thirty (30) days or more. Borrower shall not make or consent to any change or modification in such Plans, contracts or subcontracts, and no work shall be performed with respect to any such change or modification, without the prior written consent of Lender, if (I) such change or modification would in any material way alter the design or structure of the Project or change the rentable area thereof in any way, or increase or decrease the Project cost by \$250,000 or more (after taking into account cost savings and any insurance proceeds of Borrower received by Lender) for any single change or modification, or (ii) the aggregate amount of all changes and modifications exceeds \$500,000 (after taking into account cost savings and any insurance proceeds of Borrower received by Lender). Borrower shall promptly furnish Lender with a copy of all changes or modifications in the Plans, contracts or subcontracts for the Project prior to any Advance used to fund such change or modification whether or not Lender's consent to such change or

modification is required hereby."

"Borrower shall deliver to Lender revised, estimated costs of the

33. Front Sight has made multiple material changes to the plans and schedule without obtaining written consent from LVD Fund, including, inter alia, reducing the size of the "Patriot Pavilion" from 85,000 square feet, as represented to USCIS, to approximately 25,000 - 30,000 square feet, while also modifying plans to eliminate foundations. Counter Claimants are informed and believe, and thereon allege, that this change by Front Sight is a material change in the construction plans, in breach of the CLA.

E. Breach Number 5: Refusal to Comply Regarding Senior Debt - CLA § 5.27

34. Under the CLA, Front Sight was required to obtain Senior Debt from a traditional construction lender, originally by March 31, 2016 (Exhibit 3 at pg. 11 "Senior Debt" defined), then was given an extension to December 31, 2017 (Exhibit 4 at ¶4), and then was given an extension to June 30, 2018 (Exhibit 5 at ¶1). To date, Front Sight has not secured Senior Debt that meets the requirements of the CLA.

Page 25 of 42

F. Breach Number 6: Failure to Provide Monthly Project Costs - CLA § 3.2(a)

35. Front Sight has not delivered the required Monthly Evidence of Project Costs. "From and after the date of the first Advance of the Loan, Borrower shall deliver to Lender on a monthly basis evidence of the Project costs funded during the preceding month." (CLA § 3.2(a)). Counter Defendant Front Sight has not delivered a single monthly Project cost report.

G. Breach Number 7: Failure to Notify of Event of Default - CLA § 5.10

36. Section 5.10(d) of the CLA requires the Borrower to notify Lender of the occurrence of an Event of Default. "Within five (5) Business Days after the occurrence of any event actually known to Borrower which constitutes a Default or an Event of Default, notice of such occurrence, together with a detailed statement of the steps being taken to cure such event, and the estimated date, if known, on which such action will be taken." Front Sight has failed to notify LVD Fund of either (1) the existence of certain events of default; or (2) a detailed statement of the steps being taken to cure the event of default.

H. Breach Number 8: Refusal to Allow Inspection of Records - CLA § 5.4

37. Section 5.4 of the CLA provides:

Keeping of Records. Borrower shall set up and maintain accurate and complete books, accounts and records pertaining to the Project. Borrower will permit representatives of Lender to have reasonable access to and to inspect and copy such books, records and contracts of Borrower and to inspect the Project and to discuss Borrower's affairs, finances and accounts with any of its principal officers, all at such times and as often as may reasonably be requested by Lender.

- 38. LVD Fund made a demand to Inspect the Books and Records by Notice of Default and Letter dated July 30, 2018.
- 39. Front Sight explicitly refused to comply with this obligation under the CLA, as stated in the letter from Ignatius Piazza dated August 20, 2018. It states "Borrower is not in breach; thus, there will be no inspections. In the Notice; you have included a "Notice of Inspections" which alleges that "[P]ursuant to articles 3.3 and 5.4 of the CLA, we hereby serve you notice that we and our representatives will inspect the Project and your books and records on Monday, August 27." As set forth above and below herein, we contend that Borrower is not in breach or default of any of its Page 26 of 42

obligations under the Loan Agreement; thus, Borrower will not authorize any inspections whatsoever by Lender or its representatives of the Project or its books and records on the proposed date of August 27 [2018], or at any other time."

40. The right of inspection with advance notice pursuant to §3.3 and §5.4 of the CLA is not contingent on whether there is an Event of Default. Front Sight's refusal to permit the inspection constitutes a separate Event of Default acknowledged in writing by Front Sight.

I. Breach Number 9: Refusal to Allow Inspection of the Project - CLA § 3.3

41. Section 3.3 of the CLA provides:

Inspections: Lender and its representatives shall have access to the Project at all reasonable times and shall have the right to enter the Project to conduct such inspections thereof as they shall deem necessary or desirable for the protection of Lender's interests; provided, however, that for so long as no Event of Default shall have occurred and be continuing, Lender shall provide to borrower prior to the notice of not less than seventy-two (72) hours of any such inspections and such inspection shall be subject to the rights of club members (i.e., owners of timeshare interests) and any tenants under any applicable leases."

- 42. As discussed in the section above, on July 30, 2018, LVD Fund made a demand to Front Sight for permission to inspect the Project, with more than 72 hours notice, even though Events of Default negated the need for advanced notice. In response, Front Sight explicitly refused to comply with this obligation under the CLA, stating: "Borrower will not authorize any inspections whatsoever by Lender or its representatives of the Project or its books and records on the proposed date of August 27 [2018], or at any other time."
- 43. This is a material breach of the CLA justifying court intervention because the right of inspection is necessary for Lender to determine, inter alia, appropriate use of loan proceeds, construction progress, and possible impairment of security, which is necessary for Lender to protect its interests.

J. Breach Number 10: Failure to Provide EB-5 Information - CLA § 1.7(f)

44. In order to verify continuing eligibility for participation in the EB-5 Investor Program with the USCIS, Front Sight was required to submit certain EB-5 information on a continuing basis Page 27 of 42

45. Front Sight has failed to provide the required EB-5 Information. It is necessary to give Lender access to the information needed in order to meet its obligations to its EB-5 investors so the investors don't lose their investment and their path to citizenship.

K. Breach Number 12: Transferring Assets to Related Parties - CLA § 5.18

- 46. Section 5.18 of the CLA provides that: "Borrower shall not directly or indirectly, prior to completion of all of the improvements or the Completion Date, (a) make any distribution of money or property to any Related Party, or make or advance to any Related Party, or (b) make any loan or advance to any Related Party, or . . . (d) pay any fees or other compensation . . . to itself or to any Related Party, if any such payment in (a) through (d), inclusive, might adversely affect Borrower's ability to repay the loan in accordance with its terms . . ."
- 47. In violation of § 5.18, Counter Defendant Ignatius Piazza removed and converted away from Front Sight in 2016-2017 (\$ as income to Ignatius Piazza and the VNV Trust Defendants and \$ in "loans" from Front Sight). Then, in 2017-2018, Ignatius Piazza removed and converted another \$ out for himself and the VNV Trust Defendants in 2017.
- 48. Counter Claimant LVD Fund is informed and believes that Ignatius Piazza has transferred additional funds from Front Sight to himself, his wife Jennifer Piazza (either directly or indirectly) and the VNV Trust Defendants in violation of §5.18, which have yet to be disclosed.
- 49. Counter Claimants are informed and believe, and thereon allege, that Counter

 Defendants Ignatius Piazza and Jennifer Piazza—both individually, as Trustees of the VNV Trust

 Page 28 of 42

Defendants, and/or as beneficiaries of the VNV Trust Defendants—knew about the source of the transferred funds, and that transferring such funds violated the CLA, and with such knowledge endorsed and aided in the removal of funds from Front Sight, and directly benefitted from the funds through the VNV Trust Defendants and by reduction in debts that Ignatius Piazza and Jennifer Piazza had personally guaranteed.

50. Counter Defendants have now diverted out of Front Sight, for their personal benefit, enough capital to have completed the Front Sight Resort Project well within the time constraints approved by the USCIS for the EB-5 Project. By diverting profits generated by Front Sight's operations to themselves, their trusts, and using EB-5 investor funds to pay Front Sight's operating expenses and pre-existing loans, Counter Defendants Ignatius Piazza and Jennifer Piazza misappropriated loan proceeds and endangered Front Sight's solvency.

L. Breach Number 11: Non Payment of Default Interest - CLA § 1.2

- 51. Section 1.2 of the CLA provides that if there is an Event of Default, interest shall be charged at the "Default Rate." The "Default Rate" is defined as "the lesser of five percent (5%) per annum in excess of the Loan Rate or the maximum lawful rate of interest which may be charged." (Exhibit 3, CLA, pg. 4, "Default Rate Defined.") Because Front Sight is in default under multiple provisions of the CLA as detailed above, the Default Rate provisions of Section 1.2 were properly triggered.
- 52. Front Sight has failed and refused to pay the Default Rate despite the demand therefore. As a result of failing to pay default interest rates, **Front Sight is in monetary default under the terms of the CLA.**

M. Breach Number 12: Non Payment of Legal Fees - CLA § 8.2

53. Section 8.2(a) of the CLA provides that "Borrower agrees to pay and reimburse Lender upon demand for all reasonable expenses paid or incurred by Lender (including reasonable fees and expenses of legal counsel) in connection with the collection and enforcement of the Loan Documents, or any of them." This obligation was specifically reaffirmed in ¶7 of the First Amendment to the Loan Agreement (Exhibit 4), with respect to failure to provide the EB-5 Information. LVD Fund has incurred legal fees in connection with the Notices of Default and has Page 29 of 42

made demand of payment therefore from Front Sight. To date, Front Sight has refused to pay such fees and this constitutes a monetary default under §6.1(b) of the CLA. LVD Fund has also incurred attorneys' fees and costs in excess of \$625,000 in defense of this action and pursuing its rights and remedies under the CLA and Deed of Trust, for which Front Sight is contractually liable.

N. Breach Number 13: Wrongfully Encumbering the Property.

- 54. Section 5.7 of the CLA provides that "[w]ithout the prior written consent of Lender, Borrower shall not voluntarily or involuntarily agree to, cause, suffer or permit any sale, conveyance, lease, mortgage, grant, lien, encumbrance, security interest, pledge, assignment or transfer of: (a) the Project or any part or portion thereof, or (b) any ownership interest in Borrower, direct or indirect, legal or equitable (including the issuance, sale, redemption, or repurchase of any such interest, the distribution of treasury stock, or the payment of any indebtedness owed to Borrower by any managers, subsidiaries, Affiliates or owners of equity interests or debentures)."
- 55. In breach of this provision of the CLA, Counter Defendants Front Sight and Ignatius Piazza have been selling, and continue to sell, "credits," "points," "memberships," "certificates," and other instruments and products, including the sale of unregistered securities, that create contingent liabilities for Counter Defendant Front Sight and/or include the current or contingent rights to convert said instruments directly or indirectly into ownership interests in Counter Defendant Front Sight or the Project.
- 56. As a result of the multiple breaches outlined above, on January 4, 2019, LVD Fund filed the "Notice of Breach, Default and Election to Sell Under the Deed of Trust" with the Nye County Recorder (DOC #905512, attached hereto as Exhibit 6).
- 57. Counter Defendant Front Sight thereafter has failed to correct any of the previously cited breaches and Events of Default under the CLA, and has further breached the CLA by failing to provide Counter Claimant LVD Fund with financial statements within 75 days of the end of calendar year 2018, as identified in § 5.10 of the CLA, despite Counter Claimant making the demand for said financial statements by letter dated March 25, 2019.

27 ///

28 ///

Page 30 of 42

Material Misrepresentations Regarding the Morales Construction Line of Credit

- 58. By October 2017, Front Sight was in breach of the CLA. Front Sight had failed to timely obtain Senior Debt and provide LVD Fund with the EB5 documentation required under the CLA. Thereafter, Front Sight concocted a scheme to further defraud LVD Fund and to convince LVD Fund to continue working with Front Sight to fund the project.
- 59. Specifically, in or about October 2017, Counter Defendants Front Sight, Piazza, Meacher, Morales, and the Morales Entities (i.e., Morales Construction, All American Concrete and Top Rank Builders) entered into a comprehensive scheme to further defraud LVD Fund. The scheme involved Front Sight and the Morales Entities entering into a fictitious \$36 million loan agreement to give the false appearance that Front Sight had access to enough credit to complete the Project.
- 60. Counter Defendants carried out the fraudulent scheme with the intent that LVD Fund would rely on this false appearance of access to credit and believe that the credit would in fact be utilized for construction of the Project. Counter Defendants further intended that the fictitious loan agreement would give LVD Fund a false sense of security so that it would release funds it was withholding from Front Sight (pursuant to §3.1 of the CLA), and facilitate continued solicitation of additional EB-5 investors by using the loan agreement to give an appearance that Front Sight was putting more money into construction than it really was.
- 61. In furtherance of the fraudulent scheme, on October 31, 2017, Front Sight entered into the purported "Loan Agreement Construction Line of Credit" ("Loan Agreement") with the Morales Entities. (*See* Exhibit 8). The Loan Agreement was executed by Counter Defendant Morales. Per the terms of the Loan Agreement, the Morales Entities were to provide Front Sight with up to \$36,000,000 of credit to be applied towards completing the Project.
- 62. Counter Defendants Front Sight, Piazza, Meacher, Morales, and the Morales Entities caused this "Loan Agreement" to be executed with no intent to ever utilize the credit line, and with knowledge that the Morales Entities were not capable of extending or carrying the amount of credit purportedly available under the agreement's terms.

28 ///

63. On October 31, 2017, Meacher represented to LVD Fund that: 1 2 "Attached please find fully executed documents between Front Sight Management and our three primary contractors. This Construction 3 Line of Credit and associated Promissory Note extends to Front Sight up to \$36,000,000 in construction credit pursuant to the terms of the 4 agreements . . . 5 These documents and the attached construction line of credit along with the upcoming Letter of Commitment from USCP should jump 6 start the marketing in both China and India. Please release the funds for the investor you now hold and give me the vehicle by 7 which we send the funds for Dr. Shah's marketing road show that we promised with his next closing. Also light a fire under David and 8 Kyle. Get them to put some serious effort to close the 26 investors in China who are currently looking for another project. There are now 9 no excuse [sic] for not closing more of these EB-5 investors." (Emphasis added) 10 64. Counter Claimant is informed and believes, and thereon alleges, that in return for the 11 12 Morales Entities entering into the fraudulent Loan Agreement, Front Sight agreed to contract with 13 the Morales Entities to perform construction work on the Project. Morales, as the owner of the 14 Morales Entities, personally benefitted from the profit generated by the millions of dollars received 15 from Front Sight. 16 65. Rather than the construction funding coming from the Morales Entities pursuant to 17 the Loan Agreement, the Counter Defendants agreed that the funds were to come solely from LVD 18 Fund. The Loan Agreement was simply a ruse to lull LVD Fund into soliciting more EB-5 funds, 19 with the intent that the false appearance of Front Sight having a \$36 million line of credit would 20 result in a greater number of EB-5 investors coming forward. 21 FIRST CAUSE OF ACTION Fraud by Front Sight, Morales, Piazza, Meacher, Morales, and the Morales Entities 22 23 67. Counter Claimant repeats and realleges each and every allegation contained in 24 paragraphs 1 through 66 of this Counterclaim as though set forth fully herein at length. 25 68. When Counter Defendants made the misrepresentations set forth above, they knew 26 them to be false. 27 69. Counter Defendants made the misrepresentations knowing that Counter Claimant and members of the Class would rely on said misrepresentations. 28

Page 32 of 42

25

26

27

28

1

2

3

4

5

6

7

8

9

- 70. LVD Fund did in fact rely on said misrepresentations to its detriment. Had LVD Fund known the true facts, it would not have released the funds it was holding pursuant to §3.1 of the CLA and would not have solicited additional EB-5 investors for the Front Sight Project.
- 71. As a direct and proximate result of the fraud and intentional misrepresentations made by the Counter Defendants, Counter Claimant LVD Fund has sustained damages well in excess of the fifteen thousand dollar (\$15,000) jurisdictional limit of this court.
- 72. The conduct of Counter Defendants, and each of them, as described herein, was malicious, oppressive, and fraudulent under NRS 42.005, entitling Counter Claimant to an award of punitive damages.
- 73. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.
- 74. LVD Fund also is entitled to attorney's fees under Section 8.2 of the Construction Loan Agreement for enforcement of the contract.

SECOND CAUSE OF ACTION Fraudulent Transfers - NRS §§ 112.180 and 112.190 Against Front Sight, VNV Dynasty Trust I and VNV Dynasty Trust II

- 75. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 74 of this Counterclaim as though set forth fully herein at length.
- 76. Pursuant to the CLA § 5.18, Front Sight was prohibited from making certain related party transactions or transfers if such transfers would impair the ability of Front Sight to repay the construction loan under the CLA.
- 77. Despite being insolvent at year end 2016, Front Sight made an undocumented "loan to shareholder" of in excess of \$ in FY 2016.
- 78. The "loan to shareholder" was in fact a disguised distribution of over \$ for the benefit of the shareholder.

Page 33 of 42

Page 34 of 42

88. Pursuant to NRS 112.210, LVD Fund seeks: (a) avoidance of the transfers and loan to shareholder; (b) an attachment or garnishment against the asset transferred or other property of the transferee pursuant to NRS 31.010 to 31.460, inclusive, and (c) subject to applicable principles of equity and in accordance with applicable rules of civil procedure: (1) an injunction against further disposition by the debtor or a transferee, or both, of the assets transferred or of other property; (2) appointment of a receiver to take charge of the assets transferred or of other property of the transferee; or (3) any other relief the circumstances may require.

THIRD CAUSE OF ACTION

Intentional Interference with Contractual Relationships Against Ignatius Piazza, Jennifer Piazza, and VNV Trust Defendants.

- 89. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 88 of this Counterclaim as though set forth fully herein at length.
- 90. Front Sight and LVD Fund entered into a written Construction Loan Agreement (Ex. 3), along with a First Amendment in July 2017 (Ex. 4), and a Second Amendment in February 2018. (Ex. 5).
- 91. Counter Defendants had knowledge of the valid contract or had reason to know of its existence;
- 92. These Counter Defendants committed intentional acts intended or designed to disrupt the contractual relationship or to cause the contracting party to breach the contract, including but not limited to, inducing Front Sight to improperly use funds for the personal benefit of Counter Defendants Ignatius Piazza, Jennifer Piazza, and VNV Trust Defendants.
 - 93. Front Sight did in fact breach the contract as stated specifically above.
 - 94. The breach was caused by the wrongful and unjustified conduct.

Page 35 of 42

///

Ģ	95.	As a direct and proximate result of Counter Defendants' intentional acts to induce
Front Si	ght to	breach the CLA, Counter Claimant sustained damages in the amount to be proven at
trial.		

96. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.

FOURTH CAUSE OF ACTION Conversion Against Front Sight, Ignatius Piazza and Jennifer Piazza

- 97. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 96 of this Counterclaim as though set forth fully herein at length.
- 98. Through these Counter Defendants' conduct described above, Counter Defendants obtained Counter Claimants' property and have wrongfully asserted dominion over Counter Claimant's property; to wit: misappropriating and spending the loan proceeds under the CLA for purposes other than that for which it was intended.
- 99. Counter Defendants' wrongful conduct was in denial of, inconsistent with, and in defiance of Counter Claimant's rights and title to its money and/or property.
- 100. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.

FIFTH CAUSE OF ACTION Civil Conspiracy Against All Counter Defendants

101. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 100 of this Counterclaim as though set forth fully herein at length.

Page **36** of **42**

- 103. While acting in their individual capacities and in their capacity as Trustees and/or beneficiaries of the VNV Trust Defendants, Ignatius Piazza and Jennifer Piazza conspired with Front Sight and the VNV Trust Defendants, using Front Sight and VNV Trust Defendants to achieve their unlawful objective of diverting monies from Front Sight that were needed to maintain Front Sight's solvency and its ability to meet its obligations under the CLA regarding timely completion of the Project and repayment of the loan, for their own individual advantage and benefit.
- 104. As a direct and proximate result of the Counter Defendants' acts, Counter Claimant has been damaged in an amount to be proven at trial.
- 105. Counter Defendants' conduct was malicious, oppressive, and fraudulent under NRS42.005, entitling Counter Claimant to an award of punitive damages.
- 106. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.
- 107. Based on Counter Defendants' conduct and the inequitable result of allowing the transferred funds to remain in control of Counter Defendants, a constructive trust should be placed on all monies transferred from Front Sight to the VNV Trust Defendants, as prayed for below.

SIXTH CAUSE OF ACTION Judicial Foreclosure Against Front Sight

- 108. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 107 of this Counterclaim as though set forth fully herein at length.
- 109. In July 2017, Counter Defendant Front Sight for good and valuable consideration executed and delivered the original Promissory Note to LVD Fund. On November 14, 2017, Counter Page 37 of 42

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

Defendant Front Sight executed and delivered the Amended and Restated Promissory Note to LVD Fund. (Exhibit 7).

- 110. To secure the Note, on October 13, 2016, Counter Claimant LVD Fund recorded a Deed of Trust titled "Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents, and Fixture Filing," in the official records of Nye County, Nevada, as "DOC #860867." (Exhibit 1). On January 12, 2018, the "First Amendment to Construction Deed of Trust, Security Agreement and Fixture Filing," was recorded in the official records of Nye County, Nevada, as "DOC #886510." (Exhibit 2).
- 111. Counter Claimant LVD Fund is the owner and the holder of the note for value and has performed all obligation under the Promissory Note.
- 112. The encumbered Property is now owned by and in possession of Counter Defendant Front Sight.
- 113. Counter Defendants have breached the Deed of Trust as discussed in detail above, which include but are not limited to: improper use of loan proceeds; failure to provide government approved plans; material delays in construction; material changes to cost, scope, and timing of the construction; refusal to comply with regarding Senior Debt; failure to provide monthly project costs; failure to notify Lender of events of default; refusal to allow Lender to inspect books and records; diverting Front Sight assets out of Front Sight for the benefit the individual Counter Defendants; refusal to allow site inspections; failure to give Lender annual financial statements; and failure to provide EB5 documentation.
- 114. As of January 4, 2019 there remained due and owing under the Note approximately \$345,787.24 (excluding principal) as described in the Notice of Breach and Election to Sell Under the Deed of Trust. (Exhibit 6). Counter Defendants reserve the right to amend this Counterclaim up to the time of trial to include any additional amounts which become due and remain unpaid as a result of additional damages caused by Counter Defendants.
- Counter Claimant is entitled to an order directing a foreclosure sale in the subject Property to abrogate any and all interest or claims that Counter Defendants might have in the subject Property.

Page 38 of 42

116. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.

SEVENTH CAUSE OF ACTION

Waste Against Front Sight, Ignatius Piazza, and the VNV Trust Defendants

- 117. Counter Claimant repeats and realleges each and every allegation contained in paragraphs 1 through 116 of this Counterclaim as though set forth fully herein at length.
 - 118. Counter Claimant LVD Fund (Lender) has a lien encumbering the subject Property.
 - 119. Counter Defendant Front Sight (Borrower) has possession of the Property.
- 120. Waste was committed to the property in bad faith, impairing its value, including but not limited to improperly using funds earmarked for development of the Property for the personal benefit of Counter Defendants Ignatius Piazza, Jennifer Piazza, and the VNV Trust Defendants; selling unregistered securities which create substantial legal and financial liability to Front Sight, misappropriating Front Sight's assets for the personal benefit of Ignatius and Jennifer Piazza and other beneficiaries of the VNV Trust Defendants, and selling various instruments which include rights to Front Sight's resort property for highly reduced rates which further encumbers the Property, either directly or indirectly.
- 121. As a direct and proximate result of the waste committed by Counter Defendants, Counter Claimant has been injured in an amount to be proven at trial.
 - 122. Counter Claimant is entitled to treble damages under NRS 40.150.
- 123. Counter Defendants' conduct was malicious, oppressive, and fraudulent under NRS42.005, entitling Counter Claimant to an award of punitive damages.
- 124. As a result of Counter Defendants' actions, Counter Claimant has been required to retain the services of an attorney in order to pursue this claim against said Counter Defendants, and each of them, and is therefore entitled to be compensated for any and all costs incurred in the

Page 39 of 42

prosecution of this action, including without limitation, any and all reasonable costs and attorney's fees.

PRAYER FOR RELIEF

WHEREFORE, all material allegations of Plaintiff's Second Amended Complaint having been denied, affirmative defenses having been stated, and counterclaims asserted, these responding Defendants now pray as follows:

- 1. That Plaintiff take nothing by way of its Second Amended Complaint on file herein and that the same be dismissed with prejudice;
- 2. For Judgment in favor of Counter Claimants against Counter Defendants, and each of them, in an amount in excess of Fifteen Thousand Dollars (\$15,000.00), subject to proof at trial;
 - For appointment of a receiver over Counter Defendant Front Sight;
- 4. For an accounting from Counter Defendant Front Sight from October 6, 2016 forward, of any and all money paid and received, from all sources;
- 5. For an accounting from the Counter Defendant VNV Trusts from October 6, 2016 forward, of any and all money received from Counter Defendant Front Sight, and for all money distributed by the Counter Defendant Trusts since October 6, 2016.
- 6. For imposition of a constructive trust over the money transferred by Counter Defendant Front Sight to the VNV Trust Defendants in violation of Section 5.18 of the CLA, because the retention of said funds by the Counter Defendant Trusts against Counter Claimant LVD Fund's interests would be inequitable, and a constructive trust is essential to the effectuation of justice, and that restrictions be placed on such funds that limit their use to paying for the costs and expenses relating to completion of the Project.
- 7. For injunctive relief pursuant to NRS 33.010 or as otherwise permitted by law or equity to enjoin Counter Defendant Front Sight from engaging in acts that further encumber the Property and increase Counter Defendant Front Sight's actual or contingent liabilities in violation of the CLA, including the sale of "credits," "points," "memberships," "certificates," or any other instruments or products, including the sale of unregistered securities, that create contingent liabilities for Counter Defendant Front Sight and/or include the current or contingent right to convert Page 40 of 42

1	said instruments directly or indirectly into ownership interests in Counter Defendant Front Sight or			
2	the Project.			
3	8.	For punitive damages pursuant to NRS 42.005.		
4	9.	For disgorgement of the funds misappropriated by Counter Defendant Front Sight and		
5	distributed to	o the other Counter Defendants;		
6	10.	For attorneys' fees and cost of suit incurred herein; and		
7	11.	For such other and further relief as the Court may deem just and proper.		
8	DATED this 4th day of June, 2020.			
9		BAILEY * KENNEDY		
10				
11		By: <u>/s/ Andrea M. Champion</u> JOHN R. BAILEY		
12		JOHN R. BAILEY JOSHUA M. DICKEY ANDREA M. CHAMPION		
13		ANDREA M. CHAMPION		
14				
15				
16				
17				
18				
19				
20				
21				
22				
23				
24				
25				
26				
27				
28		D 41 C42		
		Page 41 of 42		
	I			

EXHIBIT 1

RECORDING REQUESTED BY: AFTER RECORDING, RETURN TO:

LAS VEGAS DEVELOPMENT FUND LLC C70 EB5 Impact Capital PO BOX 3003 Incline Village, NV 89450 APN 045-481-05 045-481-06

DOC #860867

Official Records Nye County NV
Deborah Beatty - Recorder
10/13/2016 08:32:24 AM
Requested By: CHICAGO TIMESHARE ESRecorded By: to RPTT:\$0
Recording Fee: \$51.00
Non Conformity Fee: \$25.00
Page 1 of 38

57285-NBU /93090176-U26 Sp

Space above this line for Recorder's use

CONSTRUCTION DEED OF TRUST, SECURITY AGREEMENT, ASSIGNMENT OF LEASES AND RENTS, AND FIXTURE FILING

This Document serves as a Fixture Filing under the Uniform Commercial Code, as amended from time to time, covers goods that are or become fixtures on the land, and is to be filed in the real property records of Nye County, Nevada.

THIS CONSTRUCTION DEED OF TRUST, SECURITY AGREEMENT, ASSIGNMENT OF LEASES AND RENTS, AND FIXTURE FILING (the "Deed of Trust") is made and entered into effective as of October Ob., 2016 by FRONT SIGHT MANAGEMENT, LLC, a Nevada limited liability company ("Grantor"), whose address is 1 Front Sight Road, Pahrump, Nevada 89061, to Chicago Title Company ("Trustee") whose address is 725 S. Figueroa Street, Suite 200, Los Angeles, California 90017, for the benefit of Las Vegas Development Fund LLC, a Nevada limited liability company ("Lender"), as beneficiary, whose address is P.O. Box 3003, 916 Southwood Blvd., Suite 1G, Incline Village, Nevada 89450.

To secure the full and timely payment of the secured indebtedness (as hereinafter defined), and in further consideration of the premises and for the purposes herein recited, and to secure the payment, performance and observance by Grantor of the covenants and conditions contained herein, in the Note (as hereinafter defined) and in all other agreements, documents and instruments (the "Other Documents") now or hereafter governing, securing, or guaranteeing the Loan (as hereinafter defined) evidenced by the Note (the Note, this Deed of Trust and the Other Documents being sometimes hereinafter collectively referred to as the "Loan Documents"), Grantor GRANTS, BARGAINS, SELLS, ASSIGNS and CONVEYS unto Trustee, in trust, for the benefit of Lender, WITH POWER OF SALE, AND RIGHT OF ALL ENTRY and possession of the following described land, real property interests, buildings, improvements, fixtures and other personal property:

(a) All that tract or parcel of land and other real property interests in Nye County, Nevada, more particularly described in Exhibit A attached hereto and made a part hereof (the "Land"), and all buildings and improvements of every kind and description now or hereafter erected or placed on the Land (the "Improvements"), and all right, title and interest of Grantor, now owned or hereafter acquired, in and to (i) all streets, roads, alleys, easements, rights-of-way, licenses, rights of ingress and egress, vehicle parking rights and public places, existing or

proposed, abutting, adjacent, used in connection with or pertaining to the Land or the Improvements; (ii) any strips or gores between the Land and abutting or adjacent property; and (iii) all options to purchase the Land or the Improvements or any portion thereof or interest therein, and any greater estate in the Land or the Improvements, including any and all water and water rights up to two hundred (200) acre-feet only per year, timber, crops and mineral interests on or pertaining to the Land;

- (b) All materials intended for construction, reconstruction, alteration and repair of the Improvements, all of which materials shall be deemed to be included within the premises hereby conveyed immediately upon the delivery thereof to the Land, and all fixtures and articles of personal property now or hereafter owned by Grantor and attached to or contained in and used in connection with the aforesaid Land and Improvements, including, but not limited to, all furniture, furnishings, apparatus, machinery, equipment, motors, elevators, fittings, radiators, ranges, refrigerators, awnings, shades, screens, blinds, carpeting, office equipment and other furnishings and all plumbing, heating, lighting, cooking, laundry, ventilating, refrigerating, incinerating, air conditioning and sprinkler equipment, telephone systems, televisions and television systems, computer systems and fixtures and appurtenances thereto and all renewals or replacements thereof or articles in substitution thereof, whether or not the same are or shall be attached to the Land and Improvements in any manner, but specifically excluding any and all firearms and related ammunition inventory owned or held by Grantor on the Land (the "Accessories");
- All (i) plans and specifications for the Improvements; (ii) Grantor's rights, but not liability for any breach by Grantor, under all commitments, insurance policies, contracts and agreements for the design, development, construction, operation or inspection of the Improvements and other contracts related to the Land, Improvements and Accessories or the operation thereof and related to the sale of any Land comprising the Improvements; (iii) deposits (including, but not limited to, Grantor's rights in tenants' security deposits, deposits with respect to utility services to the Land and Improvements, and any deposits or reserves hereunder or under any other Loan Document for taxes, insurance or otherwise), rebates or refunds of impact fees or other taxes, assessments or charges, money, accounts, instruments, documents, notes and chattel paper arising from or by virtue of any transactions related to the Land, Improvements and Accessories, and any account or deposit account from which Grantor may from time to time authorize Lender to debit and/or credit payments due with respect to the Loan; (iv) permits, licenses, franchises, certificates, development rights, commitments and rights for utilities, and other rights and privileges obtained in connection with the Land, Improvements and Accessories; (v) leases, rents, royalties, bonuses, issues, profits, revenues and other benefits of the Land, Improvements and Accessories; (vi) engineering, accounting, title, legal and other technical or business data concerning the Land, Improvements and Accessories which are in the possession of Grantor or in which Grantor can otherwise grant a security interest; (vii) all lists and contact information concerning then current members of the Front Sight Vacation Club and Resort, and all booklets, brochures and advertising materials for current members of the Front Sight Vacation Club and Resort.
- (d) All (i) proceeds (cash or non-cash) of or arising from all or any portion of the properties, rights, titles and interests referred to in paragraphs (a), (b) and (c) above, including, but not limited to, proceeds of any sale, lease or other disposition thereof, proceeds of each policy of insurance relating thereto (including premium refunds), proceeds of the taking thereof or of any

rights appurtenant thereto, including change of grade of streets, curb cuts or other rights of access, by condemnation, eminent domain or transfer in lieu thereof for public or quasi-public use under any law, and proceeds arising out of any damage thereto; and (ii) other interests of every kind and character which Grantor now has or hereafter acquires in, to or for the benefit of the properties, rights, titles and interests referred to in paragraphs (a), (b) and (c) above and all property used or useful in connection therewith, including, but not limited to, rights of ingress and egress and remainders, reversions and reversionary rights or interests;

TO HAVE AND TO HOLD the foregoing rights, interests and properties, and all rights, estates, powers and privileges appurtenant thereto (collectively, the "Property") unto Trustee, its successors in trust, forever, with power of sale, and Grantor does hereby bind itself, its successors, and assigns, to WARRANT AND FOREVER DEFEND the title to the Property unto Trustee against every person whomsoever lawfully claiming or to claim the same or any part thereof.

Grantor, as debtor, hereby grants to Lender, as secured party, a security interest in all of the property described in paragraphs (a), (b), (c) and (d) above which constitutes personal property or fixtures (collectively, the "Collateral") to secure the obligations of Grantor under the Note and the other Loan Documents. This Deed of Trust constitutes a security agreement under the Uniform Commercial Code, as amended from time to time, in effect in the state in which the Land is situated, or under the Uniform Commercial Code in force in any other state to the extent the same is applicable law, and Lender shall have all of the rights of a secured party thereunder in addition to its right hereunder or otherwise. This Deed of Trust may secure an obligation incurred for the construction of an improvement on the Land and as such constitutes a "construction mortgage" under the Uniform Commercial Code, as amended from time to time, in effect in the state in which the Land is situated.

Grantor covenants, represents and agrees to and with Trustee and Lender as follows:

ARTICLE I The Loan

- 1.1 Loan. The indebtedness secured by this Deed of Trust is the result of a loan in the original principal amount of up to Seventy-Five Million Dollars \$75,000,000 (the "Loan") provided by Lender to Grantor. The Loan is evidenced by (a) that certain Construction Loan Agreement (together with any extensions, revisions, modifications or amendments hereafter made, the "Loan Agreement"), of even date herewith, by and between Grantor and Lender, and (b) that certain Promissory Note executed by Grantor of even date herewith, payable to the order of Lender in the maximum original principal amount of the Loan (together with any extensions, revisions, modifications or amendments hereafter made, the "Note").
- 1.2 <u>Use of Loan Proceeds.</u> The Loan evidenced by the Note is solely for business and commercial purposes, and is not for personal, family, household or agricultural purposes. The Property forms no part of any property owned, used or claimed by Grantor as a residence or business homestead and is not exempt from forced sale under the laws of the state in which the Property is situated. Grantor hereby disclaims and renounces each and every claim to all or any part of the Property as a homestead.

- 1.3 Payment of Note. Grantor will pay principal and interest on the Loan in accordance with the Loan Documents, including the Loan Agreement, the Note and this Deed of Trust.
- Amount Secured. This Deed of Trust secures and enforces the payment and performance of the Note and the other Loan Documents, and all indebtedness, liabilities, duties, covenants, promises and other obligations, whether joint or several, direct or indirect, fixed or contingent, liquidated or unliquidated, and the cost of collection of all such amounts, owed by Grantor to Lender now or hereafter incurred or arising pursuant to or permitted by the provisions of the Note, this Deed of Trust or any other Loan Document. This Deed of Trust also secures all present and future loan disbursements (future advances) made by Lender under the Note (it being contemplated by Grantor and Lender that such future indebtedness may be incurred), plus interest thereon, all charges and expenses of collection incurred by Lender, including court costs and reasonable attorneys' fees, and all other sums from time to time owing to Lender by Grantor under the Loan Documents. The indebtedness referred to in this Section 1.4 is hereinafter sometimes referred to as the "secured indebtedness" or the "indebtedness secured hereby".
- 1.5 <u>Defined Terms.</u> All capitalized terms not otherwise defined herein shall have the meanings assigned to them in the Note, the terms and provisions of which are incorporated herein by reference, or if such capitalized term is not defined in the Note, the capitalized term shall have the meaning assigned to it in the Loan Agreement.
- 1.6 <u>Subordination to Senior Debt</u>. Lender agrees that this Deed of Trust shall be subordinated to the Senior Debt and to other Permitted Encumbrances, as such terms are defined in the Loan Agreement, and that such subordination of this Deed of Trust to the Senior Debt and other Permitted Encumbrances shall be in accordance with the applicable provisions of the Loan Agreement.

ARTICLE II Release

If and when Grantor has paid and performed all of the secured indebtedness, and no further advances are to be made under the Note, Trustee, upon request by Lender, will provide a reconveyance of the Property from the lien of this Deed of Trust and termination statements for filed financing statements, if any, to Grantor. Grantor shall be responsible for the recordation of such reconveyance and the payment of any recording and filing costs. Upon the recording of such reconveyance and the filing of such termination statements, the absolute assignment of rents set forth below shall automatically terminate and become null and void.

ARTICLE III Grantor's Representations and Warranties

Grantor represents and warrants to Lender that:

3.1 Organization. Grantor (a) is a limited liability company duly organized with a legal status separate from its affiliates, validly existing, and in good standing under the laws of the state of its formation or existence, and (b) has complied with all conditions prerequisite to its doing business in the state in which the Land is situated.

- 3.2 <u>Authority; Power to Carry on Business; Licenses.</u> Grantor has all requisite power and authority to execute and deliver the Loan Documents to which it is a party, to receive the Loan, to grant and convey the security interests contemplated under this Deed of Trust and to perform its obligations under the Note, this Deed of Trust, the other Loan Documents, and all such action has been duly and validly authorized by all necessary limited liability company proceedings on its part. Grantor has all requisite power and authority to own and operate its properties and to carry on its business as now conducted and as presently planned to be conducted. Grantor has all licenses, permits, consents and governmental approvals or authorizations necessary to carry on its business as now conducted or as presently planned to be conducted.
- 3.3 Execution and Binding Effect. The Loan Documents to which Grantor is a party have been duly and validly executed and delivered by Grantor and constitute legal, valid and binding obligations of Grantor, enforceable in accordance with their terms.
- 3.4 <u>Authorizations and Filings</u>. No authorization, consent, approval, license, exemption or other action by, and no registration, qualification, designation, declaration or filing with, any Governmental Authority is or will be necessary or advisable in connection with the execution and delivery of the Note, this Deed of Trust and the other Loan Documents, the consummation of the transactions contemplated herein or therein, or the performance of or compliance by Grantor with the terms and conditions herein or therein.
- 3.5 Execution and Delivery. Neither the execution and delivery of the Note, this Deed of Trust or the other Loan Documents and the consummation of the transactions herein or therein contemplated, nor performance of or compliance with the terms and conditions hereof or thereof will (a) violate any applicable law, (b) conflict with or result in a breach of or a default under the organizational documents of Grantor, (c) conflict with or result in a breach of or a default under any agreement or instrument to which Grantor is a party or by which it or any of its properties (now owned or acquired in the future) may be subject or bound, or (d) result in the creation or imposition of any lien or encumbrance upon any property (owned or leased) of Grantor (other than the liens created by this Deed of Trust or the other Loan Documents).
- Title to Property. Grantor represents and warrants that it has good and indefeasible title to the Land and Improvements (and any fixtures) in fee simple and has title to any appurtenant easements and interests described above and has the right to convey and encumber the same, that title to such property is free and clear of all liens, encumbrances and claims whatsoever except for (a) the Permitted Encumbrances, (b) the liens and security interests evidenced by this Deed of Trust, (c) statutory liens for real estate taxes and assessments on the Property which are not yet delinquent or due and payable, (d) other liens and security interests (if any) in favor of Lender, and (e) the matters set forth in Schedule B of the final mortgagee title policy insuring this Deed of Trust, as approved by Lender, and that it will warrant and defend the title to such property against the claims of all persons or parties. As to the Collateral, Grantor represents and warrants that it has title to such property, free and clear of all liens, encumbrances, and claims whatsoever except for the liens and security interests (if any) in favor of the lender of the Senior Debt and the Permitted Encumbrances, that it has the right to convey and encumber such property and that it will warrant and defend title to such property against the claims of all persons or parties.

- 3.7 <u>Financial Information</u>. Any financial information provided by Grantor to Lender as of the date hereof is accurate and complete and has been prepared in accordance with generally accepted accounting principles consistently applied.
- 3.8 Adequate Access. The Land has adequate rights of access to public road and rights of way, as shown in the survey(s) furnished to Lender.
- 3.9 <u>Utilities</u>. All utility services necessary for the development of the Land and the Property are available at the boundaries of the Land, including electric and natural gas facilities, telephone service, water supply, storm and sanitary sewer facilities.
- 3.10 Zoning. The current and anticipated use of the Land complies with all applicable zoning ordinances, regulations and restrictive covenants affecting the Land without the existence of any variance, non-complying use, nonconforming use or other special exception, all use restrictions of any Governmental Authority having jurisdiction have been satisfied, and no violation of any law or regulation exists with respect thereto.
- 3.11 <u>Endangered Species and Historical Sites Disclosure</u>. There are no threatened or endangered species or their habitat affecting the Property, and there are no cemeteries, burial grounds, or archeological or historical sites on the Property.
- 3.12 <u>Jurisdictional Wetlands or Waters of the U.S.</u> There are no jurisdictional wetlands or "waters of the U.S." located on any part of the Property.
- 3.13 Special Assessment Districts and Other Reimbursement Obligations. The Property is not located in a utility district, flood control district or other special assessment district, except for the Grantor-disclosed drainage channels that go across the Land and are considered "flood zone areas" on which areas no construction is contemplated or planned. There are no special assessments, special taxes, pro-rata or other reimbursement obligations applicable to the Property.
- 3.14 <u>Property Disclosure</u>. Grantor has fully disclosed the existence, presence or applicability to the Property of the following: existing gas or oil wells and applicable municipal set-back requirements; special use permits; development permits, plans and plats; existing water wells and confirmation of water rights; drainage channels considered "flood zone areas" on or near which no construction is contemplated or planned; any water features and/or dams located on or adjacent to the Property; wetlands or other environmental permits; and any other licenses, permits or approvals necessary for the ownership or operation of the Property.
- 3.15 <u>Foreign Person Disclosure</u>. Grantor is not a "foreign person" within the meaning of the Internal Revenue Code, as amended, Sections 1445 and 7701 or the regulations promulgated thereunder.
- 3.16 OFAC Disclosure. Neither Grantor nor any of its affiliates, nor any of their respective partners, members, shareholders or other equity owners, and none of their respective employees, officers, directors, representatives or agents is, nor will they become, a person or entity with whom U.S. persons or entities are restricted from doing business under regulations of the Office of Foreign Asset Control ("OFAC"), of the Department of the Treasury (including those named on OFAC's Specially Designated and Blocked Persons List) or under any statute, executive

order (including the September 24, 2001, Executive Order Blocking Property and Prohibiting Transactions with Persons Who Commit, Threaten to Commit, or Support Terrorism), or other governmental action and is not and will not engage in any dealings or transactions or be otherwise associated with such persons or entities.

- 3.17 <u>No Material Adverse Change</u>. Since the date of the most recent financial statements provided by Grantor to Lender, there has been no material adverse change in the financial condition, business or properties of Grantor.
- 3.18 No Event of Default: Compliance with Instruments. No event has occurred and is continuing, and no condition exists, which constitutes an Event of Default (as hereinafter defined) or with the passage of time would constitute an Event of Default. Grantor is not in violation of any term of its organizational documents. Grantor is not in violation of any agreement or instrument to which it is a party or by which it or any of its properties (now owned or hereafter acquired) may be subject or bound.
- 3.19 <u>Litigation</u>. There is no pending, contemplated or, to Grantor's knowledge, threatened action, suit or proceeding by or before any Governmental Authority against or affecting Grantor or the Property or any portion thereof.
- 3.20 <u>Laws</u>. Grantor is not in violation of any law, which violation is reasonably likely to have a material adverse effect on the financial condition, business or properties of Grantor.
- 3.21 Accurate and Complete Disclosure. No representation or warranty made by Grantor under this Deed of Trust or under the other Loan Documents and no statement made by Grantor in any financial statement, certificate, report, exhibit or document furnished by Grantor to Lender pursuant to or in connection with the Note, this Deed of Trust or the other Loan Documents is false or misleading in any material respect (including by omission of material information necessary to make such representation, warranty or statement not misleading). Grantor is not aware of any facts which have not been disclosed to Lender in writing by or on behalf of Grantor which would be reasonably likely to have a material adverse effect on the financial condition, business or properties of Grantor. The representations and warranties set forth herein are to survive the delivery of the Loan Documents and the making of the Loan.

ARTICLE IV Affirmative Covenants

Grantor covenants to Lender as follows:

- 4.1 <u>Preservation of Existence and Franchises</u>. Grantor, and each signatory to this Deed of Trust that signs on Grantor's behalf, will preserve and keep in full force and effect its existence (separate and apart from its affiliates), good standing, rights, franchises, trade names, trademarks and other associated goodwill whether existing at common law or as a federal or state registration.
- 4.2 <u>Compliance with Licensing Bodies</u>. Grantor shall maintain all certificates of compliance and authority and licenses that are necessary or required by any Governmental Authority or licensing authority having jurisdiction over Grantor or the Property for the current and anticipated use or operation of the Property.

4.3 INTENTIONALLY OMITTED.

- 4.4 Other Taxes, Utilities and Liens. (a) Grantor shall pay or cause to be paid, when and as due, all real and personal property taxes, assessments, water rates, dues, charges, fines and impositions of every nature whatsoever imposed, levied or assessed or to be imposed, levied or assessed upon or against the Property or any part thereof, or upon the interest of Lender in the Property, as well as all income taxes, assessments and other governmental charges lawfully levied and imposed by the United States of America or any state, county, municipality, assessment district, or other taxing authority upon Grantor or in respect of the Property or any part thereof, or any charge which, if unpaid, might become a lien or charge upon the Property prior to or equal to the lien of this Deed of Trust for any amounts secured hereby or would have priority or equality with this Deed of Trust in distribution of the proceeds of any foreclosure sale of the Property or any part thereof; provided, however, Grantor shall have the right to contest any such taxes, assessments, rates, dues, charges, fine or impositions if the execution or other enforcement of any lien or charge upon the Property is and continues to be effectively stayed or bonded in a manner satisfactory to Lender, the validity and amount of such taxes, assessments, rates, dues, charges, fines or impositions are being actively contested in good faith and by appropriate lawful proceedings and such liens or charges do not, in the aggregate, materially detract from the value of the Property or materially impair the use thereof and the operation of Grantor's business.
 - (b) Grantor shall promptly pay or cause to be paid all charges by utility companies, whether public or private, for electricity, gas, water, sewer and other utilities.
 - (c) Grantor shall promptly pay or cause to be paid and will not suffer any mechanics, laborer's, statutory or other lien which might or could be prior to or equal to the lien of this Deed of Trust to be created or to remain outstanding upon any of the Property; provided, however, such a lien may be filed against the Property if the execution or other enforcement of any such lien is and continues to be effectively stayed or bonded in a manner satisfactory to Lender for the full amount thereof, the validity and amount of the lien secured thereby are being actively contested in good faith and by appropriate lawful proceedings and such liens do not, in the aggregate, materially detract from the value of the Property or materially impair the use thereof and the operation of Grantor's business.
- 4.5 Reimbursement. Grantor agrees that if it shall fail to pay or cause to be paid when due any tax, assessment or charge levied or assessed against the Property, any utility charge, whether public or private, or any insurance premium, or if it shall fail to procure the insurance coverage and the delivery of the insurance certificates required hereunder, or if it shall fail to pay any other charge or fee required hereunder, then Lender, at its option and in addition to any other rights or remedies set forth herein, may (but shall have no obligation to) pay or procure the same. Grantor shall reimburse Lender upon demand for any sums of money paid by Lender pursuant to this Section 4.5, together with interest on each such payment at the rate set forth in the Note. All such sums so expended by Lender, and the interest thereon, shall become part of the secured indebtedness.
- 4.6 <u>Further Assurances</u>. Grantor agrees to execute and deliver to Lender, concurrently with the execution of this Deed of Trust and upon the request of Lender from time to time hereafter,

all financing statements, control agreements and other documents required to perfect and maintain the security interests created hereby.

- 4.7 Fees and Expenses. Grantor shall pay or reimburse Lender and Trustee for all reasonable attorneys' fees, costs and expenses incurred by Lender or the Trustee in any action, legal proceeding or dispute of any kind which affects the Loan, the interest created herein, the Property or the Collateral, including but not limited to, any foreclosure of this Deed of Trust, enforcement of payment of the Note and other secured indebtedness, any condemnation action involving the Property, any bankruptcy proceeding or any action to protect the security hereof or to enforce Lender's rights and remedies hereunder. Any such amounts paid by Lender or Trustee shall be due and payable upon demand and shall become part of the secured indebtedness.
- 4.8 <u>Maintenance of Property</u>. Grantor shall maintain the Property in good condition and repair, reasonable wear and tear excepted.
- 4.9 Compliance with Applicable Laws. Grantor shall comply with all applicable laws including, without limitation, all laws applicable to the use of the Property; provided, however, that Grantor shall have the ability to contest any alleged failure to conform to or comply with such laws so long as such obligations shall be contested by appropriate proceedings pursued in good faith and any penalties or other adverse effect of its nonperformance shall be stayed or otherwise not in effect. Grantor will do, or cause to be done, all such things as may be required by law in order fully to protect the security and all rights of Lender under this Deed of Trust. Grantor shall not cause or permit the lien of this Deed of Trust to be impaired in any way.
- 4.10 <u>Inspection</u>. Grantor shall permit Lender, or its agents, at any and all reasonable times, to enter and pass through or over the Property for the purpose of appraising, inspecting or evaluating the same at Lender's cost and expense, provided that any such appraisal, inspection or evaluation does not unreasonably interfere with or adversely affect Grantor's operations and shall otherwise be in accordance with the provisions of Section 3.3 of the Loan Agreement.
- 4.11 Releases and Waivers. Grantor agrees that no release by Lender of any of Grantor's successors in title from liability on the secured indebtedness, no release by Lender of any portion of the Property or the Collateral, no subordination of lien, no forbearance on the part of Lender to collect on the secured indebtedness or any part thereof, no waiver of any right granted or remedy available to Lender, and no action taken or not taken by Lender shall in any way diminish Grantor's obligation to Lender or have the effect of releasing Grantor, or any successor to Grantor, from full responsibility to Lender for the complete discharge of each and every of Grantor's obligations hereunder or under the Note, any other Loan Document or any other secured indebtedness.
- 4.12 <u>Insurance</u>. Grantor shall, at all times until the Note and all other sums due from Grantor to Lender have been fully repaid, maintain, or cause to be maintained, in full force and effect (and shall furnish to Lender copies of), property insurance, liability insurance and workers compensation insurance that are consistent with policies issued from a reputable carrier in Southern Nevada for businesses such as that operated by Borrower. Borrower shall not take any action that would void or otherwise impair any coverages required hereby or that would result in any denial or limitation of such coverages.

4.13 <u>Condemnation.</u> In the event that any or all of the Property shall be condemned and taken under the power of eminent domain, Grantor shall give immediate written notice to Lender and Lender shall have the right to receive and collect all damages awarded by reason of such taking, and the right to such damages hereby is assigned to Lender who shall have the discretion to apply the amount so received, or any part thereof, to the indebtedness secured hereby and if payable in installments, applied in the inverse order of maturity of such installments, or to any alteration, repair or restoration of the Property by Grantor.

4.14 Condemnation and Insurance Proceeds.

- Assignment to Lender. The proceeds of any award or claim for damages. direct or consequential, in connection with any condemnation or other taking of or damage or injury to the Property, or any part of it, or for conveyance in lieu of condemnation, are assigned to and shall be paid to Lender, who shall hold them in a non-interest-bearing general account regardless of whether Lender's security is impaired. All causes of action, whether accrued before or after the date of this Deed of Trust, of all types for damages or injury to the Property or any part of it, or in connection with any transaction financed by funds lent to Grantor by Lender and secured by this Deed of Trust, or in connection with or affecting the Property or any part of it, including, without limitation, causes of action arising in tort or contract or in equity, are assigned to Lender as additional security, and the proceeds shall be paid to Lender. Lender, at its option may appear in and prosecute in its own name any action or proceeding to enforce any such cause of action and may make any compromise or settlement of such action. Grantor shall notify Lender in writing immediately on obtaining knowledge of any casualty damage to the Property or damage in any other manner in excess of Ten Thousand Dollars (\$10,000) or knowledge of the institution of any proceeding relating to condemnation or other taking of or damage or injury to all or any portion of the Property. Lender, in its sole and absolute discretion, may participate in any such proceedings and may join Grantor in adjusting any loss covered by insurance. Grantor covenants and agrees with Lender, at Lender's request, to make, execute and deliver at Grantor's expense, any and all assignments and other instruments sufficient for the purpose of assigning the aforesaid award or awards, causes of action, or claims of damages or proceeds to Lender free, clear, and discharged of any and all encumbrances of any kind or nature;
- (b) <u>Insurance Payments</u>. All compensation, awards, proceeds, damages, claims, insurance recoveries, rights of action, and payments that Grantor may receive or to which Lender may become entitled with respect to the Property if any damage or injury occurs to the Property, other than by a partial condemnation or other partial taking of the Property, shall be paid over to Lender and shall be applied first toward reimbursement of all costs and expenses of Lender in connection with their recovery and disbursement, and shall then be applied as follows:
 - (i) Lender shall consent to the application of such payments to the restoration of the Property so damaged only if Grantor has met all the following conditions (a breach of one of which shall constitute a default under this Deed of Trust, the Note, and any other Loan Documents): (1) no Event of Default exists under any of the terms, covenants, and conditions of the Loan Documents; (2) all

then-existing Leases affected in any way by such damage will continue in full force and effect; (3) the insurance or award proceeds, plus any sums that Grantor may contribute for such purpose, shall be sufficient to fully restore and rebuild the Property under then current Government Requirements (defined below); and (4) all restoration of the Improvements so damaged or destroyed shall be made with reasonable promptness and shall be of a value at least equal to the value of the Improvements so damaged or destroyed before any such damage or destruction; or

- (ii) If fewer than all conditions (1) through (4) in Section 4.14(b)(i) are satisfied, then such payments shall be applied in the sole and absolute discretion of Lender (1) to the payment or prepayment with any applicable prepayment premium, of any secured indebtedness in such order as Lender may determine, or (2) to the reimbursement of Grantor's expenses incurred in the rebuilding and restoration of the Property. If Lender elects under this Section 4.14(b)(ii) to make any funds available to restore the Property, then all of conditions Section 4.14(b)(i) shall apply, except for such conditions that Lender, in its sole and absolute discretion, may waive.
- (iii) "Governmental Requirements" shall mean any and all laws, statutes, codes, ordinances, regulations, enactments, decrees, judgments and orders of any Governmental Authority.
- (iv) <u>Material Loss Not Covered</u>. If any material part of the Property is damaged or destroyed and the loss, measured by the replacement cost of the Improvements according to then-current Government Requirements, is not adequately covered by insurance proceeds collected or in the process of collection, Grantor shall deposit with Lender, within ten (10) days after Lender's request, the amount of the loss not so covered.
- (c) <u>Total Condemnation Payments</u>. All compensation, awards, proceeds, damages, claims, insurance recoveries, rights of action, and payments the Grantor may receive or to which Grantor may become entitled with respect to the Property in the event of a total condemnation or other total taking of the Property shall be paid over to Lender and shall be applied first to reimbursement of all Lender's costs and expenses in connection with their recovery, and shall then be applied to the payment of any indebtedness secured hereby by such order as Lender may determine, until the secured indebtedness has been paid and satisfied in full. Any surplus remaining after payment and satisfaction of the indebtedness secured by this Deed of Trust shall be paid to Grantor as its interest may then appear.
- (d) Partial Condemnation Payments. All compensation, awards, proceeds, damages, claims, insurance recoveries, rights of action and payments ("funds") that Grantor may receive or to which Grantor may become entitled with respect to the Property in the event of a partial condemnation or other partial taking of the Property, unless Grantor and Lender otherwise agree in writing, shall be divided into two portions, one equal to the principal balance of the Note at the time of receipt of such funds and the other equal to the amount by which such funds exceed the principal balance of the Note at the time of receipt

of such funds. The first such portion shall be applied to the indebtedness secured hereby, whether or not then due, including but not limited to principal, accrued interest, and advances and in such order or combination as Lender may determine, with the balance of the funds paid to Grantor. Any dispute as to the fair market value of the Property shall be settled by arbitration in accordance with the Real Estate Valuation Arbitration Rules of the American Arbitration Association.

- (c) No Cure of Waiver of Default. Any application of such amounts or any portion of it to any secured indebtedness shall not be construed to cure or waive any Event of Default or notice of default under this Deed of Trust or invalidate any act done under any such default or notice.
- 4.15 <u>Use of Property</u>. (a) Grantor shall use or permit the Property to be used solely for the purpose of operating the Front Sight Firearms Training Institute and the Front Sight Resort and Vacation Club complex, and Grantor shall not use or permit the use of the Property for any other principal use without Lender's prior written consent. Grantor shall not use or permit the use of the Property or any part thereof for any other purpose which in the reasonable opinion of Lender would adversely affect the then value or character of the Property or any part thereof.
 - (b) Grantor shall not suffer or permit the Property or any portion thereof to be used by the public, as such, without restriction or in such manner as might reasonably tend to impair Grantor's title to the Property or any portion thereof, or in such manner as might reasonably make possible a claim or claims of adverse usage or adverse possession by the public, as such, or of implied dedication of the Property or any portion thereof.
- 4.16 Taxes on Note and Deed of Trust. Grantor shall promptly pay all income, franchise and other taxes owing by Grantor and any stamp, documentary, recordation and transfer taxes or other taxes (unless such payment by Grantor is prohibited by law) which may be required to be paid with respect to the Note, this Deed of Trust or any other instrument evidencing or securing any of the secured indebtedness. In the event of the enactment after this date of any law of any Governmental Authority applicable to Lender, the Note, the Property or this Deed of Trust deducting from the value of property for the purpose of taxation any lien or security interest thereon, or imposing upon Lender the payment of the whole or any part of the taxes or assessments or charges or liens herein required to be paid by Grantor, or changing in any way the laws relating to the taxation of deeds of trust or mortgages or security agreements or debts secured by deeds of trust or mortgages or security agreements or the interest of the mortgagee or secured party in the property covered thereby, or the manner of collection of such taxes, so as to affect this Deed of Trust or the indebtedness secured hereby or Lender, then, and in any such event, Grantor, upon demand by Lender, shall pay such taxes, assessments, charges or liens, or reimburse Lender therefore.
- 4.17 <u>Authorization to File Financing Statements; Power of Attorney.</u> Grantor hereby authorizes Lender at any time and from time to time to file and authenticate any initial financing statements, amendments thereto and continuation statements with or without signature of Grantor as authorized by applicable law, as applicable to all or any part of the Collateral. For purposes of such filings, Grantor agrees to furnish any information requested by Lender promptly upon Lender's request. Grantor also ratifies its authorization for Lender to have filed any initial

financing statements, amendments thereto or continuation statements, if filed prior to the date of this Deed of Trust. Grantor hereby irrevocably constitutes and appoints Lender and any officer or agent of Lender, with full power of substitution, as its true and lawful attorneys-in-fact with full irrevocable power and authority in the place and the stead of Grantor or in the name of Grantor to execute in the name of Grantor or authenticate any such documents and otherwise to carry out the purposes of this Section 4.17, to the extent that the authorization above by Grantor is not sufficient. To the extent permitted by law, Grantor hereby ratifies all acts said attorneys-in-fact have lawfully done in the past or shall lawfully do or cause to be done in the future by virtue hereof. This power of attorney is a power coupled with an interest and shall be irrevocable.

4.18 INTENTIONALLY OMITTED.

- GRANTOR SHALL INDEMNIFY AND HOLD Indemnification. (a) HARMLESS LENDER AND TRUSTEE FROM AND AGAINST, AND REIMBURSE THEM ON DEMAND FOR, ANY AND ALL INDEMNIFIED MATTERS (AS HEREINAFTER DEFINED), IN ALL CASES WHETHER OR NOT CAUSED BY OR ARISING, IN WHOLE OR IN PART, OUT OF THE COMPARATIVE, CONTRIBUTORY OR SOLE NEGLIGENCE OF LENDER OR TRUSTEE. FOR PURPOSES OF THIS SECTION 4.19, THE TERMS "LENDER" AND "TRUSTEE" SHALL INCLUDE THE DIRECTORS, OFFICERS, PARTNERS, EMPLOYEES AND AGENTS OF LENDER AND TRUSTEE, RESPECTIVELY, AND ANY PERSONS OWNED OR CONTROLLED BY, OWNING OR CONTROLLING, OR UNDER COMMON CONTROL OR AFFILIATED WITH LENDER OR TRUSTEE, RESPECTIVELY. WITHOUT LIMITATION, THE FOREGOING INDEMNITIES SHALL APPLY TO EACH INDEMNIFIED PERSON WITH RESPECT TO MATTERS WHICH IN WHOLE OR IN PART ARE CAUSED BY OR ARISE OUT OF THE NEGLIGENCE OF SUCH (AND/OR ANY OTHER) INDEMNIFIED PERSON. HOWEVER, SUCH INDEMNITIES SHALL NOT APPLY TO A PARTICULAR INDEMNIFIED PERSON TO THE EXTENT THAT THE SUBJECT OF THE INDEMNIFICATION IS CAUSED BY OR ARISES OUT OF THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF THAT INDEMNIFIED PERSON. ANY AMOUNT TO BE PAID UNDER THIS <u>SECTION 4.19</u> BY GRANTOR TO LENDER AND/OR TRUSTEE SHALL BE A DEMAND OBLIGATION OWING BY GRANTOR (WHICH GRANTOR HEREBY PROMISES TO PAY) TO LENDER AND/OR TRUSTEE PURSUANT TO THIS DEED OF TRUST. NOTHING IN THIS SECTION 4.19. ELSEWHERE IN THIS DEED OF TRUST OR IN ANY OTHER LOAN DOCUMENT SHALL LIMIT OR IMPAIR ANY RIGHTS OR REMEDIES OF LENDER AND/OR TRUSTEE (INCLUDING WITHOUT LIMITATION ANY RIGHTS OF CONTRIBUTION OR INDEMNIFICATION) AGAINST GRANTOR OR ANY OTHER PERSON UNDER ANY OTHER PROVISION OF THIS DEED OF TRUST, ANY OTHER LOAN DOCUMENT, ANY OTHER AGREEMENT OR ANY APPLICABLE FEDERAL, STATE OR LOCAL LAW, STATUTE, ORDINANCE, CODE, RULE, REGULATION, LICENSE, PERMIT, ORDER OR DECREE.
 - (b) As used herein, the term "Indemnified Matters" means any and all claims, demands, liabilities (including strict liability), losses, damages (including consequential damages), causes of action, judgments, penalties, fines, costs and expenses (including without limitation, reasonable fees and expenses of attorneys and other professional consultants and experts, and of the investigation and defense of any claim, whether or not

such claim is ultimately defeated, and the settlement of any claim or judgment including all value paid or given in settlement) of every kind, known or unknown, foreseeable or unforeseeable, which may be imposed upon, asserted against or incurred or paid by Lender and/or Trustee at any time and from time to time, whenever imposed, asserted or incurred, because of, resulting from, in connection with, or arising out of any transaction, act, omission, event or circumstance in any way connected with the Property or with this Deed of Trust or any other Loan Document, including but not limited to any bodily injury or death or property damage occurring in or upon or in the vicinity of the Property through any cause whatsoever at any time on or before the Release Date (as hereinafter defined), any act performed or omitted to be performed hereunder or under any other Loan Document, any breach by Grantor of any representation, warranty, covenant, agreement or condition contained in this Deed of Trust or in any other Loan Document or any Event of Default, except to the extent caused by the gross negligence or intentional misconduct of Lender, its agents, employees and/or representatives. The term "Release Date" as used herein means the earlier of the following two dates: (i) the date on which the indebtedness and obligations secured hereby have been paid and performed in full and this Deed of Trust has been fully reconveyed and released, or (ii) the date on which the lien of this Deed of Trust is fully and finally foreclosed or a conveyance by deed in lieu of such foreclosure is fully and finally effective, and possession of the Property has been given to the purchaser or grantee free of occupancy and claims to occupancy by Grantor and Grantor's heirs, devisees, representatives, successors and assigns; provided, that if such payment, performance, release, foreclosure or conveyance is challenged, in bankruptcy proceedings or otherwise, then the Release Date shall be deemed not to have occurred until such challenge is rejected, dismissed or withdrawn with prejudice. The indemnities in this Section 4.19 shall not terminate upon the Release Date or upon the release, foreclosure or other termination of this Deed of Trust but will survive the Release Date, foreclosure of this Deed of Trust or conveyance in lieu of foreclosure, the repayment of the secured indebtedness, the discharge and release of this Deed of Trust and the other Loan Documents, any bankruptcy or other debtor relief proceeding, and any other event whatsoever.

4.20 Payment of Costs. Grantor shall (a) pay all reasonable legal fees incurred by Lender in connection with the preparation of the Loan Documents (including any amendments thereto or consents, releases, or waivers granted thereunder); (b) reimburse Lender, promptly upon demand, for all amounts expended, advanced, or incurred by Lender to satisfy any obligation of Grantor under the Loan Documents, which amounts shall include all court costs, reasonable attorneys' fees (including, without limitation, for trial, appeal, or other proceedings), fees of auditors and accountants and other investigation expenses reasonably incurred by Lender in connection with any such matters; and (c) pay any and all other costs and expenses of performing or complying with any and all of the obligations under the Note, this Deed of Trust and under the other Loan Documents. All of the foregoing listed fees, costs and expenses are collectively called herein, the "Expenses." Except to the extent that the Expenses are included within the definition of "indebtedness secured hereby," the payment of such Expenses shall not be credited, in any way and to any extent, against any installment on or portion of the indebtedness secured hereby.

ARTICLE V Negative Covenants

Grantor covenants to Lender as follows:

- 5.1 <u>Liens.</u> Grantor shall not at any time create, incur, assume or permit to exist any lien or encumbrance on or against the Property or agree to become liable to do so, except for (a) the Permitted Encumbrances, (b) the liens and security interests evidenced by this Deed of Trust, (c) statutory liens for real estate taxes and assessments on the Property which are not yet delinquent, (d) other liens and security interests (if any) in favor of Lender, and (e) the matters set forth in Schedule B of the final mortgagee title policy insuring this Deed of Trust as approved by Lender.
- 5.2 <u>Indebtedness.</u> With respect to the Property, Grantor shall not at any time, create, incur, assume or suffer to exist any indebtedness, except (a) the indebtedness under the Permitted Encumbrances, (b) indebtedness under the Note or any other Loan Document or any other document, instrument or agreement between Grantor and Lender, and (c) current accounts payable, accrued expenses and other expenses arising out of transactions (other than borrowing) in the ordinary course of business.
- 5.3 <u>Guaranties and Contingent Liabilities.</u> Grantor shall not at any time directly or indirectly become or be liable in respect of any guaranty or contingent obligation, or assume, guarantee, become surety for, endorse or otherwise agree, become or remain directly or contingently liable upon or with respect to any obligation or liability of any person or entity (other than Grantor), except (i) by endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of business, or (ii) by indemnity agreements given by Grantor to a title insurance company or a bonding company in connection with any project being constructed or sold by Grantor, including the Project.
- 5.4 <u>Loans and Investments</u>. Grantor shall not at any time make or suffer to remain outstanding any loan or advance to, or purchase, acquire, or own any stock, bonds, notes or securities of, or any partnership interest (whether general or limited) in, or any other interest in, or make any capital contribution or loan to, any person or entity (other than Grantor), or agree, become or remain liable to do any of the foregoing.

5.5 INTENTIONALLY OMITTED.

5.6 Self-Dealing. Grantor shall not enter into or carry out any transaction (including, without limitation, purchasing property or services from or selling property or services to) with any Affiliate (as hereinafter defined) except (a) officers, managers, members, employees and affiliates of Grantor may render services to Grantor for compensation at the same rates generally paid by companies engaged in the same or similar businesses for the same or similar services; and (b) Grantor may enter into and carry out other transactions with Affiliates if in the ordinary course of business, pursuant to the reasonable requirements of Grantor's business upon terms that are fair and reasonable and no less favorable to Grantor than Grantor would obtain in a comparable arm's-length transaction. "Affiliate" means, with respect to any individual or entity (each, a "Person"), another Person that directly, or indirectly through one or more intermediaries, Controls or is

Controlled by or is under common Control with the Person specified. "Control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of a Person, whether through the ability to exercise voting power, by contract or otherwise. "Controlling" or "Controlled" have meanings correlative thereto.

- 5.7 <u>Disposition of Property</u>. Except in connection with the obligations with respect to the Senior Debt and related agreements, Grantor shall not (a) sell, convey, pledge, assign, lease, abandon or otherwise transfer or dispose of, voluntarily or involuntarily (any of the foregoing being referred to in this Section 5.7 as a transaction and any set of related transactions constituting but a single transaction), all or any portion of the Property or any interest therein or enter into any agreement to do so, or (b) subdivide the Property, submit the Property, or any portion thereof, to condominium or other multiple form of ownership, or dedicate any portion of the Property to public ownership. Lender hereby consents to Grantor taking actions to secure the Senior Debt as such transactions are reasonably necessary for the development of the Project, including the time share units and the RV resort, as provided in the Loan Agreement and the Budget.
- 5.8 Ownership and Control. Grantor shall not cause or permit any change in the ownership (whether direct or indirect) of Grantor from that in existence on the date hereof.
- 5.9 Merger; Consolidation; Business Acquisitions. Grantor shall not merge or agree to merge with or into or consolidate with any other person or entity. Grantor shall not form any subsidiaries or acquire any material portion of the stock, other equity interests or assets or business of any other person or entity.
- 5.10 <u>Change in Zoning; Easements; Restrictions.</u> Grantor shall not seek or acquiesce in any annexation of the Property or any zoning reclassification of all or any portion of the Land or Property or grant or consent to any easement, dedication, plat, or restriction (or allow any easement to become enforceable by prescription), or any amendment or modification thereof, covering all or any portion of the Land or Property, without Lender's prior written consent. Lender hereby agrees that it will not unreasonably withhold or delay consent to Grantor taking actions that would otherwise violate the foregoing provisions so long as such transactions are reasonably necessary for the development of the Project, including the time share units and the RV resort as provided in the Loan Agreement and the Budget.
- 5.11 <u>Drilling</u>, Grantor shall not, without Lender's prior written consent, permit any drilling or exploration for, or extraction, removal, or production of, any minerals from, the surface or subsurface of the Land regardless of the depth thereof or the method of mining or extraction therefrom.
- 5.12 <u>Waste</u>; <u>Alterations</u>. Grantor shall not commit or permit any waste or impairment of the Property and shall not (subject to the provisions of <u>Sections 4.8 and 4.9</u> hereof), without Lender's prior written consent, which consent shall not be unreasonably delayed or withheld, make or permit to be made any alterations or additions to the Property of a material nature other than those alterations and additions consisting of the Improvements that shall constitute the accommodations and other facilities of the project known as the Front Sight Resort and Vacation Club. Subject to the foregoing and in no way constituting a waiver thereof, in the event Lender

were to give such consent, then any alterations or additions to the Property would be at Grantor's sole cost and expense.

ARTICLE VI Events of Default

- 6.1 Events of Default. An "Event of Default" means the occurrence or existence of one or more of the following events or conditions (whatever the reason for such Event of Default and whether voluntary, involuntary or effected by operation of law):
 - (a) Grantor defaults in any payment of principal or interest on the Loan by the date due according to the terms of the Loan Agreement or of the Note, and such default remains uncured for a period of ten (10) days after the payment became due; provided, however, that there is no cure period for payments due on the Maturity Date; or
 - (b) Grantor defaults in the payment of undisputed fees or other amounts payable to or on behalf of Lender pursuant to the Note, this Deed of Trust or under any other Loan Documents, other than as described in Section 6.1(a) above, and such default continues unremedied for a period of ten (10) days after notice thereof from Lender to Grantor; or
 - (c) Grantor defaults in the performance or observance of any agreement, covenant or condition required to be performed or observed by Grantor under the terms of this Deed of Trust, or any other Loan Document, other than a default described elsewhere in this Section, and such default continues unremedied for a period of thirty (30) days after notice from Lender to Grantor thereof provided that, if cure cannot reasonably be effected within such 30-day period, such failure shall not be an event of default hereunder so long as Grantor promptly (in any event, within ten (10) days after such notice of default from Lender) commences cure, and thereafter diligently (in any event, within ninety (90) days after receipt of such notice of default from Lender) prosecutes such cure to completion; and provided further, that notwithstanding the 30-day cure period or extended cure period described above in this subparagraph (c), if a different notice or cure period is specified under any Loan Document or under any provision of the Loan Documents as to any such failure or breach, the specific Loan Document or provision shall control, and Grantor shall have no more time to cure the failure or breach than is allowed under the specific Loan Document or provision as to such failure or breach; or
 - (d) Any representation or warranty made by Grantor in this Agreement or by Grantor or an Affiliate, if made in connection with the Loan, in any of the other Loan Documents, or in any certificate or document furnished under the terms of this Agreement or in connection with the Loan, shall be untrue or incomplete in any material respect when made or deemed made or restated hereunder unless such representation or warranty was not known by Grantor to be untrue or incomplete at the time made and such representation or warranty is corrected by Grantor and disclosed by Grantor to Lender; or
 - (e) Lender's security interest or lien under this Deed of Trust is or shall become unperfected or invalid; or

- (f) Grantor defaults under any term, covenant or condition of any of the Note or of any of the other Loan Documents to which Grantor is a party, other than a default described elsewhere in this Section, after the expiration of any notice or grace period, if any, provided therein;
- (g) Work on the Project, once commenced, shall be substantially abandoned, or shall, by reason of Grantor's fault, be unreasonably delayed or discontinued for a period of fifteen (15) consecutive days, or construction shall be delayed for any reason whatsoever to the extent that Completion cannot, in the reasonable judgment of Lender, be accomplished prior to the Completion Date;
- Any of Grantor, or any Related Party who is a party to any of the Loan Documents, shall file a petition for bankruptcy; or shall apply for, consent to or permit the appointment of a receiver, custodian, trustee or liquidator for it or any of its property or assets; or shall generally fail to, or admit in writing its inability to, pay its debts as they mature; or shall make a general assignment for the benefit of creditors or shall be adjudicated bankrupt or insolvent; or shall take other similar action for the benefit or protection of its creditors; or shall give notice to any governmental body of insolvency or of pending insolvency or suspension of operations; or shall file a voluntary petition in bankruptcy or a petition or an answer seeking reorganization or an arrangement with creditors, or to take advantage of any bankruptcy, reorganization, insolvency, readjustment of debt, rearrangement, dissolution, liquidation or other similar debtor relief law or statute; or shall file an answer admitting the material allegations of a petition filed against it in any proceeding under any such law or statute; or shall be dissolved, liquidated, terminated or merged; or shall effect a plan or other arrangement with creditors; or a trustee, receiver, liquidator or custodian shall be appointed for it or for any of its property or assets and shall not be discharged within ninety (90) days after the date of his appointment; or a petition in involuntary bankruptcy or similar proceedings is filed against it and is not dismissed within ninety (90) days after the date of its filing; or
- (i) Lender determines that the remaining undisbursed Loan proceeds, together with the proceeds of any Senior Debt, are insufficient to fully pay all of the then-unpaid costs of the Project and the estimated expenses of completion (including the Interest Reserve), and Grantor fails to either (i) deposit with Lender, within three (3) Business Days following demand, sufficient funds to permit Lender to pay said excess costs as the same become payable or (ii) pay said excess costs directly and deliver to Lender unconditional mechanics' lien waivers therefor (or paid receipts for non-lienable items), at Lender's option; or
- (j) except to the extent otherwise permitted pursuant to the terms and conditions of the Loan Agreement or this Deed of Trust, the sale, lease, transfer or further encumbrance (whether by operation of law or otherwise) (and whether at one time or in or pursuant to a series of events) of (A) the Property or any part thereof or any interest therein, or (B) more than forty-nine percent (49%) in the aggregate of any direct or indirect ownership interest in Grantor; or

- (k) A default occurs with respect to the Senior Debt and remains uncured after the expiration of any applicable notice or grace period; or
- (l) A default occurs in the performance of Grantor's obligations in any of Section 5.6, 5.7, 5.8, 5.10, 5.13, 5.16, 5.18, 5.19, 5.22, 5.23 or 5.24, of the Loan Agreement;
- (m) The General Contract is terminated by either party thereto or either party thereto shall fail to perform its obligations (after any applicable notice and cure period) under the General Contract; or
- (n) Any uncured default by Grantor occurs and remains uncured under the Management Agreement; or
- (o) Any failure by Grantor to timely deliver the EB-5 information, which failure continues more than five (5) business days following notice of such failure by Lender.
- 6.2 Remedies of Lender. Upon the occurrence of an Event of Default, unless such Event of Default is subsequently waived in writing by Lender, Lender may, without notice and without prejudice to any other right or remedy Lender may have, exercise from time to time any of the rights and remedies available under the Note, this Deed of Trust or any other Loan Document or under applicable law.

ARTICLE VII Rights and Remedies

- Acceleration of Loan. Upon the occurrence of an uncured Event of Default specified in Section 6.1 hereof, the entire unpaid balance of the indebtedness secured hereby (including all accrued interest and all other sums secured hereby) shall, at the option of Lender, become immediately due and payable without notice, demand, presentment, notice of nonpayment or nonperformance, protest, notice of protest, notice of intent to accelerate, notice of acceleration. or any other notice or any other action, all of which are hereby waived by Grantor and all other parties obligated in any manner whatsoever on the indebtedness secured hereby. If an uncured Event of Default specified in Subsection (h) of Section 6.1 hereof occurs and continues or exists, the entire unpaid balance of the indebtedness secured hereby (including all accrued interest and all other sums secured hereby) shall automatically become immediately due and payable without notice, demand, presentment, notice of nonpayment or nonperformance, protest, notice of protest, notice of intent to accelerate, notice of acceleration, or any other notice or any other action, all of which are hereby waived by Grantor and all other parties obligated in any manner whatsoever on the indebtedness secured hereby. The failure to exercise any remedy available to the Lender shall not be deemed to be a waiver of any rights or remedies of the Lender under the Loan Documents, at law or in equity.
- 7.2 Foreclosure Power of Sale. Upon the occurrence of any uncured Event of Default, Lender may request Trustee to proceed with foreclosure under the power of sale which is hereby conferred, such foreclosure to be accomplished in accordance with the following provisions:
 - (a) Foreclosure; Power of Sale. Trustee, if and as directed by Lender, shall have all of the rights and may exercise all of the powers set forth in applicable law of the

State of Nevada. Trustee may sell the Property in its entirety or in parcels, and by one or by several sales, as deemed appropriate by Trustee in its sole and absolute discretion. If Trustee chooses to have more than one foreclosure sale, Trustee may cause the foreclosure sales to be held simultaneously or successively, on the same day, or on such different days and at such different times as Trustee may elect. Trustee shall receive and apply the proceeds from the sale of the Property, or any portion thereof, in accordance with Nevada law. Before any foreclosure sale, Lender or Trustee shall give such notice of default and election to sell as may be required by law. After the lapse of such time as may then be required by law following the recordation of such notice of default, and notice of sale having been given as then required by law, Trustee shall sell the property being sold at a public auction to be held at the time and place specified in the notice of sale. Neither Trustee nor Lender shall have any obligation to make demand on Grantor before any foreclosure sale. From time to time in accordance with then-applicable law, Trustee may, and in any event at Lender's request shall, postpone any foreclosure sale by public announcement at the time and place noticed for that sale. At any foreclosure sale, Trustee shall sell to the highest bidder at public auction for cash in lawful money of the United States (or cash equivalents acceptable to Trustee to the extent permitted by applicable law), payable at the time of sale. Trustee shall execute and deliver to the purchaser(s) a deed or deeds conveying the property being sold without any covenant or warranty whatsoever, expressed or implied. The recitals in any such deed of any matters of fact, including any facts bearing upon the regularity or validity of any foreclosure sale, shall be conclusive proof of their truthfulness. Any such deed shall be conclusive against all persons as to the facts recited therein. Any Person, including Trustee or Lender, may purchase at such sale, and any bid by Lender may be, in whole or in part, in the form of cancellation of all or any part of the Obligations.

- (b) <u>Judicial Action</u>. Lender and Trustee, if and as directed by Lender, shall have the right to bring an action in any court of competent jurisdiction for foreclosure of this Deed of Trust a deficiency judgment as provided by law, or for specific enforcement of any of the covenants or agreements of this Deed of Trust.
- Collection of Rents. Upon the occurrence of an Event of Default, the license granted to Grantor to collect the Rents (defined below) shall be automatically and immediately revoked, without further notice to or demand upon Grantor. Lender may, but shall not be obligated to, exercise any or all of the rights and remedies provided in Nevada Law and perform any or all obligations of the landlord under any or all of the Leases (defined below), and Lender may, but shall not be obligated to, exercise and enforce any or all of Grantor's rights under the Leases. Without limiting the generality of the foregoing, Lender may notify the tenants under the Leases that all Rents are to be paid to Lender, and following such notice all Rents shall be paid directly to Lender and not to Grantor or any other Person other than as directed by Lender, it being understood that a demand by Lender on any tenant under a Lease for the payment of Rent shall be sufficient to warrant payment by such tenant of Rent to Lender without the necessity of further consent by Grantor. Grantor hereby irrevocably authorizes and directs the tenants under the Leases to pay all Rents to Lender instead of to Grantor, upon receipt of written notice from Lender, without the necessity of any inquiry of Grantor and without the necessity of determining the existence or non-existence of an Event of Default. Grantor hereby appoints Lender as

Grantor's attorney-in-fact with full power of substitution, which appointment shall take effect upon the occurrence of an Event of Default and is coupled with an interest and is irrevocable prior to the full and final payment and performance of the indebtedness secured hereby, in Grantor's name or in Lender's name: (i) to endorse all checks and other instruments received in payment of Rents and to deposit the same in any account selected by Lender; (ii) to give receipts and releases in relation thereto; (iii) to institute, prosecute and/or settle actions for the recovery of Rents; (iv) to modify the terms of any Leases including terms relating to the Rents payable thereunder; (v) to cancel any Leases; (vi) to enter into new Leases; and (vii) to do all other acts and things with respect to the Leases and Rents which Lender may deem necessary or desirable to protect the security for the secured indebtedness. Any Rents received shall be applied first to pay all of Lender's costs and expenses and next in reduction of the other secured indebtedness. Grantor shall pay, on demand, to Lender, the amount of any deficiency between (1) the Rents received by Lender, and (2) all Expenses incurred together with interest thereon as provided in this Deed of Trust and the other Loan Documents.

- (d) Taking Possession or Control of the Property. As a matter of right without regard to the adequacy of the security, and to the extent permitted by law without notice to Grantor, Lender shall be entitled, upon application to a court of competent jurisdiction, to the immediate appointment of a receiver for all or any part of the Property and the Rents, whether such receivership may be incidental to a proposed sale of the Property or otherwise, and Grantor hereby consents to the appointment of such a receiver and agrees that such receiver shall have all of the rights and powers granted to Lender pursuant to Section 7.2(c). In addition, to the extent permitted by law, and with or without the appointment of a receiver, or an application therefor, Lender may (i) enter upon, and take possession of (and Grantor shall surrender actual possession of), the Property or any part thereof, without notice to Grantor and without bringing any legal action or proceeding, or, if necessary by force, legal proceedings, ejectment or otherwise, and (ii) remove and exclude Grantor and its agents and employees therefrom.
- Management of the Property. Upon obtaining possession of the Property or upon the appointment of a receiver as described in Section 7.2(d), Lender, Trustee or the receiver, as the case may be, may, at its sole option, (i) make all necessary or proper repairs and additions to or upon the Property, (ii) operate, maintain, control, make secure and preserve the Property, and (iii) complete the construction of any unfinished Improvements on the Property and, in connection therewith, continue any and all outstanding contracts for the erection and completion of such Improvements and make and enter into any further contracts which may be necessary, either in their or its own name or in the name of Grantor (the costs of completing such Improvements shall be Expenses secured by this Deed of Trust and shall accrue interest as provided in the Note). Lender, Trustee or such receiver shall be under no liability for, or by reason of, any such taking of possession, entry, holding, removal, maintaining, operation or management, except for Lender's, Trustee's or Receiver's negligence, gross negligence or willful misconduct. The exercise of the remedies provided in this Section shall not cure or waive any Event of Default, and the enforcement of such remedies, once commenced, shall continue for so long as Lender shall elect, notwithstanding the fact that the exercise of such remedies may have, for a time, cured the original Event of Default.

- (f) <u>Cooperation of Grantor</u>. Grantor agrees to cooperate fully with Lender's management of the Property, including, without limitation, providing full access to the Property and all collateral.
- Uniform Commercial Code. Lender may proceed under the Uniform Commercial Code as to all or any part of the Collateral, and in conjunction therewith may exercise all of the rights, remedies and powers of a secured creditor under the Uniform Commercial Code. Upon the occurrence of any uncured Event of Default, Grantor shall assemble all of the Collateral and make the same available within the Improvements or at such other location required by Lender. Any notification required by the Uniform Commercial Code shall be deemed reasonably and properly given if sent in accordance with the notice provisions of this Deed of Trust at least ten (10) days before any sale or other disposition of the Collateral. Disposition of the Collateral shall be deemed commercially reasonable if made pursuant to a public sale advertised at least twice in a newspaper of general circulation in the community where the Property is located. It shall be deemed commercially reasonable for the Trustee to dispose of the Collateral without giving any warranties as to the Collateral and specifically disclaiming all disposition warranties. Alternatively, Lender may choose to dispose of some or all of the Property, in any combination consisting of both Collateral and Real Property, in one sale to be held in accordance with the law and procedures applicable to real property, as permitted by Article 9 of the Uniform Commercial Code. Grantor agrees that such a sale of Collateral together with Real Property constitutes a commercially reasonable sale of the Collateral.
- (g) Application of Proceeds. Unless otherwise provided by applicable law, all proceeds from the sale of the Property or any part thereof pursuant to the rights and remedies set forth in this Article VII and any other proceeds received by Lender from the exercise of any of its other rights and remedies hereunder or under the other Loan Documents shall be applied first to pay all Expenses and next in reduction of the other secured indebtedness, in such manner and order as Lender may elect.
- (h) Other Remedies. Lender shall have the right from time to time to protect, exercise and enforce any legal or equitable remedy against Grantor provided under the Loan Documents or by applicable laws.
- 7.3 Remedies Cumulative. All remedies provided in this Deed of Trust, in the Note and in the other Loan Documents are cumulative and may, at the election of Lender, be exercised alternatively, successively, or in any manner and are in addition to any other rights provided by law.
- 7.4 Suits to Protect the Property. Lender and Trustee shall have power (a) to institute and maintain such suits and proceedings as they may deem expedient to prevent any impairment of the Property or the Collateral by any acts which may be unlawful or any violation of this Deed of Trust, (b) to preserve or protect their interest in the Property and the Collateral, and (c) to restrain the enforcement of or compliance with any legislation or other Governmental Requirement, rule or order that may be unconstitutional or otherwise invalid, if the enforcement of or compliance with, such Governmental Requirement, rule or order would impair the security hereunder or be prejudicial to the interest of Lender.

ARTICLE VIII Waivers

- 8.1 Waiver of Certain Rights. To the full extent permitted by applicable law, Grantor agrees that Grantor will not at any time insist upon, plead, claim or take the benefit or advantage of any law now or hereafter in force providing for any appraisement, valuation, stay, extension or redemption, homestead, moratorium, reinstatement, marshaling or forbearance, and Grantor, for Grantor, Grantor's heirs, devisees, representatives, successors and assigns, and for any and all persons ever claiming any interest in the Property, to the extent permitted by applicable law, hereby waives and releases all rights of redemption, valuation, appraisement, stay of execution, notice of intention to mature or declare due the whole of the secured indebtedness, and all rights to a marshaling of assets of Grantor, including the Property, or to a sale in inverse order of alienation in the event of foreclosure of the liens and/or security interests hereby created. Grantor shall not have or assert any right under any statute or rule of law pertaining to the marshaling of assets, sale in inverse order of alienation, the exemption of homestead, the administration of estates of decedents, or other matters whatsoever to defeat, reduce or affect the right of Lender under the terms of this Deed of Trust to a sale of the Property for the collection of the secured indebtedness without any prior or different resort for collection, or the right of Lender under the terms of this Deed of Trust to the payment of the secured indebtedness out of the proceeds of sale of the Property in preference to every other claimant whatsoever except for the Senior Debt. Grantor waives any right or remedy which Grantor may have or be able to assert pursuant to any provision of Nevada law, including, but not limited to, the rights or remedies pertaining to the rights and remedies of sureties. If any law referred to in this Section 8.1 and now in force, of which Grantor or Grantor's heirs, devisees, representatives, successors or assigns or any other persons claiming any interest in the Property might take advantage despite this Section, shall hereafter be repealed or cease to be in force, such law shall not thereafter be deemed to preclude the application of this Section 8.1.
- 8.2 Waivers and Agreements Regarding Remedies. To the fullest extent permitted by applicable law, Grantor hereby waives any right to bring or utilize any defense, counterclaim or setoff, other than one which denies the existence or sufficiency of the facts upon which any foreclosure action is grounded. If any defense, counterclaim or setoff, other than one permitted by the preceding clause, is timely raised in a foreclosure action, such defense, counterclaim or setoff shall be dismissed. If such defense, counterclaim or setoff is based on a claim which could be tried in an action for money damages, such claim may be brought in a separate action which shall not thereafter be consolidated with the foreclosure action. The bringing of such separate action for money damages shall not be deemed to afford any grounds for staying the foreclosure action.

ARTICLE IX Environmental Warranties, Representations, Covenants and Indemnification Provisions

- 9.1 <u>Definitions</u>, As used in this Article IX, the following definitions shall apply:
- (a) Environmental Activity. The existence, use, storage, Release, threatened Release, generation, processing, abatement, removal, or disposal of any Hazardous Substance on, to, or from the Property or the handling, transportation, treatment, or disposal of any Hazardous Substance arranged by or on behalf of any Indemnitor.

- (b) Environmental Claims. Any and all governmental and third-party actions, suits, demands, demand letters, claims, liens, notices of non-compliance or violation, investigations, proceedings, consent orders, or consent agreements relating in any way to the presence or Use of any Hazardous Substance on the Property or the Release or threatened Release of any Hazardous Substance to or from the Property or the violation of any Environmental Requirement or any Environmental Permit applicable to the Property or which otherwise relate to any Environmental Activity, including, without limitation, (i) those of or brought by any Governmental Authority for enforcement, cleanup, removal, response, remedial, or other actions or damages pursuant to any applicable Environmental Requirement, and (ii) those of or brought by any third party seeking damages, contribution, indemnification, cost recovery, compensation, or injunctive relief arising in connection with any Environmental Requirement, any Hazardous Substance or from any alleged injury or threat of injury to property, human health, or the environment resulting or allegedly resulting from any Environmental Activity.
- Environmental Damages. All claims, judgments, damages, losses, (c) penalties, fines, liabilities (including strict liability), encumbrances, liens, costs, and expenses imposed upon, incurred by, or imposed any party in connection with or arising from (i) any Environmental Activity, (ii) any Environmental Claim, (iii) all costs and expenses of investigation and defense of any Environmental Claim, whether or not such Environmental Claim is ultimately defeated, or (iv) any good faith settlement or agreed judgment, including, without limitation, reasonable attorneys' fees, disbursements, and consultants' fees incurred as a result of an Environmental Claim or a violation of any Environmental Requirement pertaining to any Indemnitor or the Property (regardless of whether the existence or alleged existence of such Hazardous Substance or the violation or alleged violation of such Environmental Requirement arose prior to any Indemnitor's Use of such Property). "Environmental Damages" shall also include, without limitation, (A) damages for personal injury or injury to property or natural resources occurring upon or off of the Property, (B) fees incurred for the services of attorneys, consultants, contractors, experts, and laboratories, and all other costs incurred in connection with the investigation of the presence or alleged presence of Hazardous Substances on, about, or under the Property, the removal or remediation of any Hazardous Substances, or the violation or alleged violation of any Environmental Requirements, including, without limitation, costs and expenses for the preparation of any feasibility studies or reports or the performance of any cleanup, remediation, removal, response, abatement, containment, closure, restoration, or monitoring work required by any Governmental Authority or necessary in defense of any Environmental Claim, (C) reasonable attorneys' fees, costs, and expenses incurred in enforcing this Article IX or collecting any sums due hereunder, (D) liability to any third person or Governmental Authority to indemnify such person or entity for costs expended in connection with the items referenced above, and (E) diminution in the value of the Property.
- (d) <u>Environmental Laws</u>. All federal, state or local laws, statutes, rules, regulations, ordinances, permits, licenses and determinations of any Governmental Authority having jurisdiction over any Indemnitor, the Property, or any user or occupant of the Property, and relating to health, industrial hygiene and/or the environment, now existing or hereafter in effect, including, without limitation, the Comprehensive

Environmental Response, Compensation and Liability Act of 1980, as amended (42 U.S.C. § 9601, et seq.) ("CERCLA"), the Solid Waste Disposal Act, as amended (42 U.S.C. § 6901, et seq.), the Hazardous Materials Transportation Act, as amended (49 U.S.C. § 1801, et seq.), the Clean Air Act, as amended (42 U.S.C. § 7401, et seq.), the Federal Water Pollution Control Act, as amended (33 U.S.C. § 1251, et seq.), the Toxic Substances Control Act, as amended (15 U.S.C. § 2601, et seq.), the Safe Drinking Water Act, as amended (42 U.S.C. § 300f, et seq.), the Atomic Energy Act, as amended (42 U.S.C. § 2014, et seq.), the Federal Insecticide, Fungicide and Rodenticide Act, as amended (7 U.S.C. § 136, et seq.), the Oil Pollution Act of 1990, as amended (33 U.S.C. § 2701, et seq.), the Emergency Planning and Community Right-to-Know Act of 1986, as amended (42 U.S.C. § 11001, et seq.), the Occupational Safety and Health Act, as amended (29 U.S.C. § 651, et seq.), and the Endangered Species Act, and any corresponding state laws, statutes, regulations or ordinances.

- (e) <u>Environmental Permits</u>. All permits, approvals, identification numbers, licenses, and other authorizations required under any applicable Environmental Requirement.
- (f) Environmental Requirements. All Environmental Laws and all rules, regulations, guidelines, standards, orders, decrees, permits, licenses, concessions, and franchises promulgated pursuant thereto, and/or other restrictions or requirements of any Governmental Authority relating to health, industrial hygiene and/or the environment, and all applicable judicial, regulatory, or administrative decisions, decrees, judgments, or orders thereunder, as may be amended from time to time.
- (g) Governmental Authority. Any governmental authority (federal, state, county, district, municipal, city or otherwise), including, without limitation, the United States of America, any state of the United States of America, and any subdivision of any of the foregoing, and any agency, department, commission, board, office, authority, instrumentality, bureau, or court now or hereafter in effect, having jurisdiction over the Property, or over any Indemnitor or any occupant or user of the Property, or any of their respective businesses, operations, assets, or properties.
- (h) <u>Hazardous Substance</u>. Any substance, product, material, element, compound, chemical or waste, whether solid, liquid or gaseous (i) the presence or Release of which requires reporting, investigation, or remediation under any Environmental Requirement, (ii) which is defined, listed, classified or regulated as a "hazardous waste," "hazardous substance," "extremely hazardous waste," "restricted hazardous waste," "hazardous material," "toxic substance," "regulated substance," or other similar or related term under or in any Environmental Requirement, (iii) which is toxic, radioactive, or otherwise classified as hazardous or toxic and is or becomes regulated by any Governmental Authority as a threat to human health or the environment, (iv) the presence of which on or about the Property causes or threatens to cause a nuisance upon the Property or to adjacent property, (v) the presence of which on adjacent properties could constitute a trespass by any Indemnitor, (vi) which is asbestos, (vii) which is polychlorinated biphenyls, (viii) which contains petroleum or any petroleum-derived product,

- (ix) underground storage tanks, whether empty, filled or partially filled with any substance, or (x) any radioactive materials, urea formaldehyde foam insulation, or radon.
- (i) <u>Indemnitees</u>. Lender, any assignee of Lender with respect to all or any portion of the Loan, and all of their respective subsidiaries, affiliates, shareholders, partners, members, directors, officers, agents, attorneys, and employees, and their respective successors and assigns, and "Indemnitee" means any one of the Indemnitees.
 - (j) <u>Indemnitors</u>. Grantor and its successors and assigns.
- (k) Release. Any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, seeping, migrating, dumping, or disposing into the environment (including the abandonment or discarding of barrels, containers, drums, tanks or other closed receptacles containing any Hazardous Substance).
- (1) <u>Use</u>. Use, ownership, development, construction, maintenance, management, operation, or occupancy (of the Property).
- INDEMNIFICATION. GRANTOR HEREBY ASSUMES LIABILITY FOR, AND HEREBY AGREES TO AND SHALL INDEMNIFY, DEFEND (AT TRIAL AND APPELLATE LEVELS, ADMINISTRATIVE PROCEEDINGS AND ARBITRATIONS, WITH ATTORNEYS, CONSULTANTS AND EXPERTS ACCEPTABLE TO LENDER). SAVE, AND HOLD HARMLESS EACH INDEMNITEE FROM AND AGAINST ANY AND ALL ENVIRONMENTAL DAMAGES AND ENVIRONMENTAL CLAIMS IMPOSED UPON, ASSERTED OR AWARDED AGAINST OR INCURRED BY THE PROPERTY OR ANY INDEMNITEE, UNLESS, AND TO THE EXTENT, SUCH ENVIRONMENTAL DAMAGES OR ENVIRONMENTAL CLAIMS ARE FINALLY DETERMINED TO HAVE ARISEN SOLELY AND DIRECTLY FROM THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF INDEMNITEES. THIS OBLIGATION SHALL INCLUDE ANY CLAIMS RESULTING FROM THE NEGLIGENCE OR ALLEGED NEGLIGENCE OF ANY INDEMNITEE. THIS OBLIGATION SHALL INCLUDE, WITHOUT LIMITATION, (I) THE BURDEN OF DEFENDING ALL CLAIMS, SUITS, AND ADMINISTRATIVE PROCEEDINGS (WITH COUNSEL REASONABLY APPROVED BY INDEMNITEES), EVEN IF SUCH CLAIMS, SUITS, OR PROCEEDINGS ARE GROUNDLESS, FALSE, FRAUDULENT, OR FRIVOLOUS, AND CONDUCTING ALL NEGOTIATIONS OF ANY DESCRIPTION, (II) PAYING AND DISCHARGING, WHEN AND AS THE SAME SHALL BECOME DUE, ANY AND ALL JUDGMENTS, PENALTIES, OR OTHER SUMS DUE AGAINST ANY INDEMNITEE, (III) PAYING AND DISCHARGING, WHEN AND AS THE SAME SHALL BECOME DUE, ALL COSTS OF REMOVAL AND/OR REMEDIATION OF ANY KIND, AND PROMPTLY DISPOSING OF SUCH HAZARDOUS SUBSTANCES (WHETHER OR NOT SUCH HAZARDOUS SUBSTANCE MAY BE LEGALLY ALLOWED TO REMAIN UPON, ABOUT, OR BENEATH THE PROPERTY IF REMOVAL OR REMEDIATION IS, IN LENDER'S DISCRETION, PRUDENT), (IV) PAYING AND DISCHARGING, WHEN AND AS THE SAME SHALL BECOME DUE, ALL COSTS OF DETERMINING WHETHER THE PROPERTY IS IN COMPLIANCE, AND PROMPTLY CAUSING THE PROPERTY TO BE IN COMPLIANCE, WITH ALL APPLICABLE ENVIRONMENTAL

REQUIREMENTS, (V) PAYING AND DISCHARGING, WHEN AND AS THE SAME SHALL BECOME DUE, ALL COSTS ASSOCIATED WITH CLAIMS FOR DAMAGES TO PERSONS, PROPERTY, OR NATURAL RESOURCES, AND (VI) PAYING AND DISCHARGING, WHEN AND AS THE SAME SHALL BECOME DUE, INDEMNITEES' REASONABLE ATTORNEYS' FEES, CONSULTANTS' FEES, AND COURT COSTS. ANY INDEMNITEE, AT ITS EXPENSE (OR AT GRANTOR'S EXPENSE IF GRANTOR'S COUNSEL OR INDEMNITEE REASONABLY BELIEVES A CONFLICT EXISTS IN DUAL REPRESENTATION), MAY EMPLOY ADDITIONAL COUNSEL OF ITS CHOICE TO ASSOCIATE WITH COUNSEL EMPLOYED BY GRANTOR; AND, IF AN EVENT OF DEFAULT EXISTS, ANY INDEMNITEE MAY IN GOOD FAITH SETTLE ANY CLAIM (INCLUDING ANY ENVIRONMENTAL CLAIM) AGAINST IT, WHETHER OR NOT SUBJECT TO INDEMNIFICATION HEREUNDER, WITHOUT THE CONSENT OR JOINDER OF GRANTOR OR ANY OTHER PARTY.

- 9.3 <u>SURVIVAL</u>. THIS ARTICLE IX, INCLUDING THE INDEMNITY CONTAINED HEREIN, SHALL SURVIVE THE RELEASE OF THE LIEN OF THIS DEED OF TRUST OR THE EXTINGUISHMENT OF THE LIEN BY FORECLOSURE OR ACTION IN LIEU THEREOF.
- Rights Under Environmental Requirements and Other Rights. Nothing in this Deed of Trust or in any other Loan Document shall limit or impair any claims, rights or remedies of Lender or any other Indemnitee against Grantor or any other person under any Environmental Requirement or otherwise at law or in equity, including any claims for fraud, misrepresentation, waste or breach of contract other than this Deed of Trust, and any rights of contribution or indemnification. In addition to any other rights or remedies Lender may have under this Deed of Trust or the other Loan Documents, at law or in equity, upon any breach or default by Grantor under this Deed of Trust, Lender may pursue any remedies available to it under Nevada Law. Without limiting any of the remedies provided herein or in the other Loan Documents, Grantor acknowledges and agrees that the provisions of this Article IX are environmental provisions, made by Grantor relating to the real property security, and that Grantor's failure to comply with the terms of this Deed of Trust is a breach of contract such that Lender shall have the remedies provided under Nevada Law for the recovery of damages and for the enforcement thereof. Lender's action for the recovery of damages or enforcement of this Deed of Trust shall not constitute an action within the meaning of any provision of law limiting the right to a deficiency or a deficiency judgment.

ARTICLE X Assignment of Leases and Rents

10.1 Absolute Assignment. In order to provide a source of future payment of the secured indebtedness, Grantor hereby absolutely and unconditionally grants, transfers, conveys, sells, sets over and assigns to Lender all of Grantor's right, title and interest now existing and hereafter arising in and to the leases, subleases, concessions, licenses, franchises, occupancy agreements, tenancies, subtenancies and other agreements, either oral or written, now existing and hereafter arising which affect the units constituting the Front Sight Resort and Vacation Club, together with any and all security deposits, guaranties of the lessees' or tenants' obligations (including any and all security therefor), and other security under any such leases, subleases, concessions, licenses,

franchises, occupancy agreements, tenancies, subtenancies and other agreements, and all supporting obligations, letters of credit (whether tangible or electronic) and letter of credit rights guaranteeing or supporting any of the foregoing (all of the foregoing, and any and all extensions, modifications and renewals thereof, shall be referred to, collectively, as (the "Leases"), and hereby gives to and confers upon Lender the right to collect all the income, rents, issues, profits, royalties and proceeds from the Leases and any business conducted at the Front Sight Resort and Vacation Club Units (but specifically excluding any income, rents, issues, profits, royalties and proceeds from any Leases and any other business conducted by or on behalf of FSFTI) and any and all prepaid rent and security deposits thereunder (collectively, the "Rents"). The term "Rents" includes, but is not limited to, all minimum rents, additional rents, percentage rents, deficiency rents, common area maintenance charges, lease termination payments, refunds of any type, prepayment of rents, settlements of litigation, settlements of past due rents, and liquidated damages following default, and all proceeds payable under any policy of insurance covering loss of rents, together with any and all rights and claims of any kind that Grantor may have against any tenant under the Leases or any other occupant of the units constituting the Front Sight Resort and Vacation Club. This Deed of Trust is intended by Lender and Grantor to create and shall be construed to create an absolute unconditional and presently effective assignment to Lender of all of Grantor's right, title and interest in and to the Leases and the Rents and shall not be deemed merely to create a security interest therein for the payment of any indebtedness or the performance of any obligations under the Loan Documents. Grantor irrevocably appoints Lender its true and lawful attorney at the option of Lender at any time to demand, receive and enforce payment, to give receipts, releases and satisfactions and to sue, either in the name of Grantor or in the name of Lender, for all such Rents and apply the same to the secured indebtedness.

- 10.2 Revocable License to Collect. Notwithstanding the foregoing assignment of Rents, so long as no Event of Default remains uncured, Grantor shall have a revocable license, to collect all Rents, and to retain any portion thereof not required to pay the expenses of the Property or the obligations secured thereby. Upon any Event of Default, Grantor's license to collect and retain Rents shall terminate automatically and without the necessity for any notice.
- Collection and Application of Rents by Lender, While any Event of Default remains uncured, (a) Lender may at any time, without notice, in person, by agent or by court-appointed receiver, and without regard to the adequacy of any security for the secured indebtedness, enter upon any portion of the Property and/or, with or without taking possession thereof, in its own name sue for or otherwise collect Rents (including past due amounts); and (b) without demand by Lender, Grantor shall promptly deliver to Lender all prepaid rents, deposits relating to Leases or Rents, and all other Rents then held by or thereafter collected by Grantor whether prior to or during the continuance of any Event of Default. Any Rents collected by or delivered to Lender may be applied by Lender against the secured indebtedness, less all Expenses, including reasonable attorneys' fees and disbursements, in such order as Lender shall determine in its sole and absolute discretion. No application of Rents against any secured indebtedness or other action taken by Lender under this Article X shall be deemed or construed to cure or waive any Event of Default, or to invalidate any other action taken in response to such Event of Default, or to make Lender a mortgagee-in-possession of the Property. In no event shall the assignment of Rents or Leases cause the secured indebtedness to be reduced by an amount greater than the Rents actually received by Lender and applied by Lender to the secured indebtedness, whether before, during or after (1) an Event of Default or (2) a suspension or revocation of the license granted to Grantor in this

Article X with regard to the Rents. Grantor and Lender specifically intend that the assignment of Rents and Leases contained in this Deed of Trust is not intended to result in a pro tanto reduction of the secured indebtedness, nor is it intended to constitute a payment of, or with respect to, the secured indebtedness, and, therefore, Grantor and Lender specifically intend that the secured indebtedness shall not be reduced by the value of the Rents and Leases assigned hereby. Such reduction shall occur only if, and to the extent that, Lender actually receives Rents and applies such Rents to the secured indebtedness. Grantor agrees that the value of the license granted with regard to the Rents equals the value of the absolute assignment of Rents to Lender.

- 10.4 <u>Direction to Tenants</u>. Grantor hereby irrevocably authorizes and directs the tenants under all Leases to pay all amounts owing to Grantor thereunder to Lender following receipt of any written notice from Lender that states that an Event of Default remains uncured and that all such amounts are to be paid to Lender. Grantor further authorizes and directs all such tenants to pay all such amounts to Lender without any right or obligation to inquire as to the validity of Lender's notice and regardless of the fact that Grantor has notified any such tenants that Lender's notice is invalid or has directed any such tenants not to pay such amounts to Lender.
- 10.5 <u>Termination.</u> The assignment contained in this Article X will terminate upon the full reconveyance of this Deed of Trust.

ARTICLE XI General Conditions

11.1 Concerning the Trustee.

- (a) <u>Trustee</u>. Trustee shall be deemed to have accepted the terms of this trust when this Deed of Trust, duly executed and acknowledged, is made a public record as provided by law. Trustee shall not be obligated to notify any party to this Deed of Trust of any pending sale under any other deed of trust or of any action or proceeding in which Grantor, Lender, or Trustee is a party, unless such sale relates to or reasonably might affect the Property, this Deed of Trust, Lender's security for the payment and performance of the secured indebtedness, or the rights or powers of Lender or Trustee under the Loan Documents, or unless such action or proceeding has been instituted by Trustee against the Property, Grantor, or Lender.
- (b) Power of Trustee to Reconvey or Consent. At any time, without liability and without notice to Grantor, on Lender's written request and presentation of the Note and this Deed of Trust to Trustee for endorsement, and without altering or affecting (i) the personal liability of Grantor or any other person for the payment of the secured indebtedness, or (ii) the lien of this Deed of Trust on the remainder of the Property as security for the repayment of the full amount of the secured indebtedness then or later secured by this Deed of Trust, (iii) or any right or power of Lender or Trustee with respect to the remainder of the Property, Trustee may (1) reconvey or release any part of the Property from the lien of this Deed of Trust; (2) approve the preparation or filing of any map or plat of the Property; (3) join in the granting of any easement burdening the Property; or (4) enter into any extension or subordination agreement affecting the Property or the lien of this Deed of Trust.

- Substitution of Trustee. Lender, at Lender's option, may from time to time, by written instrument, substitute a successor or successors to any Trustee named in or acting under this Deed of Trust, which instrument, when executed and acknowledged by Lender and recorded in the office of the Recorder of the county or counties in which the Property is located, shall constitute conclusive proof of the proper substitution of such successor Trustee or Trustees. The successor Trustee or Trustees shall, without conveyance from the predecessor Trustee, succeed to all right, title, estate, powers, and duties of such predecessor Trustee, including, without limitation, the power to reconvey the Property. To be effective, the instrument must contain the names of the original Grantor, Trustee, and Lender under this Deed of Trust, the book and page or instrument or document number at which, and the county in which, this Deed of Trust is recorded, and the name and address of the substitute Trustee. If any notice of default has been recorded under this Deed of Trust, this power of substitution cannot be exercised until all costs, fees, and expenses of the then acting Trustee have been paid. On such payment, the then acting Trustee shall endorse receipt of the payment on the instrument of substitution. The procedure provided in this paragraph for substitution of Trustees is not exclusive of other provisions for substitution provided by applicable law.
- (d) No Representation by Trustee or Lender. By accepting or approving anything required to be observed, performed, or fulfilled or to be given to Trustee or Lender pursuant to the Loan Documents, including without limitation, any officer's certificate, balance sheet, statement of profit and loss or other financial statement, survey, appraisal, or insurance policy, neither Trustee nor Lender shall be deemed to have warranted, consented to, or affirmed the sufficiency, legality, effectiveness, or legal effect of the same, or of any term, provision, or condition thereof, and such acceptance or approval thereof shall not be or constitute any warranty or affirmation with respect thereto by Trustee or Lender.
- No Liability of Trustee. Trustee shall not be liable for any error of judgment or act done by Trustee in good faith, or be otherwise responsible or accountable under any circumstances whatsoever (including Trustee's negligence), except for Trustee's gross negligence or willful misconduct. Trustee shall have the right to rely on any instrument, document or signature authorizing or supporting any action taken or proposed to be taken by Trustee hereunder, believed by Trustee in good faith to be genuine. All moneys received by Trustee shall, until used or applied as herein provided, be held in trust for the purposes for which they were received, but need not be segregated in any manner from any other moneys (except to the extent required by law), and Trustee shall be under no liability for interest on any moneys received by Trustee hereunder. Grantor hereby ratifies and confirms any and all acts which the herein named Trustee or its successor or successors, substitute or substitutes, in this trust, shall do lawfully by virtue hereof. Grantor will reimburse Trustee for, and indemnify and save Trustee harmless against, any and ail liability and expenses which may be incurred by Trustee (including as a result of Trustee's negligence) in the performance of its duties. The foregoing indemnity shall not terminate upon discharge of the secured indebtedness or foreclosure, or release or other termination, of this Deed of Trust.

- 11.2 <u>Number and Gender.</u> Words in the singular used herein shall be deemed to include the plural and words in the plural shall be deemed to include the singular, unless in each instance the context requires otherwise; and words of any gender shall be deemed to include the masculine, feminine and neuter.
- Notices, All notices or other communications required or permitted to be given pursuant to this Deed of Trust shall be in writing and shall be considered as properly given (a) if mailed by first class United States mail, postage prepaid, registered or certified with return receipt requested; (b) by delivering same in person to the intended addressee; (c) by delivery to a reputable independent third party commercial delivery service for same day or next day delivery and providing for evidence of receipt at the office of the intended addressee; or (d) by facsimile to the addressee with evidence of receipt at the addressee's facsimile number, if any. Notice so mailed shall be effective three (3) days after its deposit with the United States Postal Service or any successor thereto; notice given by personal delivery shall be effective only if and when received by the addressee; notice sent by such a commercial delivery service shall be effective upon delivery to the recipient (if sent for same day delivery) or the first business day following delivery to such commercial delivery service (if for next day delivery); and notice given by other means shall be effective only if and when received at the office or designated place or machine of the intended addressee. For purposes of notice, the addresses of the parties shall be as set forth herein; provided, however, that either party shall have the right to change its address for notice hereunder to any other location within the continental United States by the giving of ten (10) days' prior written notice to the other party in the manner set forth herein.

Grantor hereby requests that a copy of any notice of default and any notice of sale hereunder be mailed to Grantor at the address set forth on the first page of this Deed of Trust. That address is also the mailing address of Grantor as debtor under the UCC. Lender's address given on the first page of this Deed of Trust is the address for Lender as secured party under the UCC.

- 11.4 <u>Invalidation of Provisions</u>. Invalidation of any one or more of the provisions of this Deed of Trust shall in no way affect any of the other provisions hereof, which shall remain in full force and effect.
- 11.5 <u>Headings</u>. The captions and headings herein are inserted only as a matter of convenience and for reference and in no way define, limit, or describe the scope of this Deed of Trust or the intent of any provision hereof.
- 11.6 GOVERNING LAW AND VENUE. THIS DEED OF TRUST SHALL BE GOVERNED BY, AND CONSTRUED, APPLIED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEVADA AND APPLICABLE LAWS OF THE UNITED STATES OF AMERICA, WITHOUT REGARD TO CONFLICTS OF LAW. GRANTOR AGREES THAT THIS DEED OF TRUST IS PERFORMABLE IN NYE COUNTY, NEVADA. GRANTOR STIPULATES THAT CLARK COUNTY, NEVADA, IS PROPER VENUE FOR ANY ACTION OR PROCEEDING INVOLVING THIS AGREEMENT, TO THE EXCLUSION OF ALL OTHER VENUES. GRANTOR WAIVES ANY OBJECTION BASED ON FORUM NON CONVENIENS AND ANY OBJECTION TO VENUE OF ANY ACTION INSTITUTED UNDER THIS DEED OF TRUST, AND

CONSENTS TO THE GRANTING OF SUCH LEGAL OR EQUITABLE RELIEF AS IS DEEMED APPROPRIATE BY THE COURT.

- 11.7 No Third-Party Beneficiary. Grantor and Lender acknowledge that this Deed of Trust is made solely for the benefit of the parties hereto and their respective successors and assigns, and no third party should or may assume that any third-party beneficiary rights are extended or created hereby.
- shall be binding upon Grantor, and the heirs, devisees, representatives, successors and assigns of Grantor (and all references in this Deed of Trust to Grantor shall be deemed to include all such heirs, devisees, representatives, successors and assigns of Grantor), and shall inure to the benefit of Trustee and Lender and shall constitute covenants running with the Land. Lender may, from time to time and without notice to Grantor, assign, participate or otherwise transfer all or any portion of the Loan secured hereby, the Note, this Deed of Trust (and the lien created hereby) and the other Loan Documents (and Lender's rights and interests thereunder), in whole or in part, and the term "Lender" shall include Lender's successors and assigns and any subsequent holder(s) of the Note secured hereby or any assignee or transferee thereof whether by operation of law or otherwise.
- No Usury Intended. Grantor and Lender intend to comply strictly with applicable usury laws. All agreements between Grantor and Lender, whether now existing or hereafter arising and whether written or oral, are hereby limited so that in no contingency, whether by reason of the disbursement of the principal amount of the Loan, demand, prepayment or acceleration of the maturity of the Note or otherwise, shall the interest contracted for, charged, received, paid or agreed to be paid to Lender (including any other compensation, however denominated, held or deemed to be interest) exceed the maximum amount of interest permitted under applicable federal and Nevada law that may be contracted for, charged, received, paid or agreed to be paid to Lender (including any compensation, however denominated, held or deemed to be interest) (the "Maximum Lawful Rate"). If, from any circumstance whatsoever, interest (and any compensation, however denominated, held or deemed to be interest) would otherwise be payable to Lender in excess of the Maximum Lawful Rate, the interest and any such other compensation payable or paid to Lender shall be reduced to the Maximum Lawful Rate; and if from any circumstance Lender shall ever receive interest or anything of value deemed interest by applicable law in excess of the Maximum Lawful Rate, an amount equal to any such excessive interest shall be applied to the reduction of the principal of the Note and not to the payment of interest, or if such excessive interest exceeds the unpaid principal balance of the Note, such excess shall be refunded to Grantor. All interest (including any other compensation, however denominated, held or deemed to be interest) paid or agreed to be paid to Lender shall, to the extent permitted by applicable law, be amortized, prorated, allocated and spread in equal parts through the full stated term of the Note, including renewals or forbearance periods, so that the rate or amount of interest on the Note shall not exceed the Maximum Lawful Rate; and in the event the Note is paid in full by Grantor prior to the end of the full stated term of the Note and the interest (including any other compensation, however denominated, held or deemed to be interest) received for the actual period of the existence of the Note exceeds the Maximum Lawful Rate, Lender shall refund to Grantor the amount of the excess or shall credit the amount of the excess against amounts owing under the Note. Grantor hereby agrees that as a condition precedent to any claim seeking usury penalties against Lender,

Grantor will provide written notice to Lender, advising Lender in reasonable detail of the nature and amount of the violation, and Lender shall have sixty (60) days after receipt of such notice in which to correct such usury violation, if any, by either refunding such excess interest to Grantor or crediting such excess interest against the Note and/or any other indebtedness then owing by Grantor to Lender.

- 11.10 WAIVER OF JURY TRIAL. AS A SPECIFICALLY BARGAINED INDUCEMENT FOR LENDER TO MAKE THE LOAN TO GRANTOR, TO THE FULLEST EXTENT NOW OR HEREAFTER PERMITTED BY LAW, GRANTOR AND LENDER HEREBY WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY GRANTOR OR LENDER AGAINST THE OTHER TO ENFORCE THIS DEED OF TRUST, TO COLLECT DAMAGES FOR THE BREACH OF THIS DEED OF TRUST, OR WHICH IN ANY OTHER WAY ARISE OUT OF, ARE CONNECTED TO OR ARE RELATED TO THIS DEED OF TRUST. ANY SUCH ACTION SHALL BE TRIED BY THE JUDGE WITHOUT A JURY.
- 11.11 ENTIRE AGREEMENT. THE NOTE, THIS DEED OF TRUST AND THE OTHER LOAN DOCUMENTS CONTAIN THE FINAL, ENTIRE AGREEMENT BETWEEN THE PARTIES HERETO RELATING TO THE SUBJECT MATTER HEREOF AND THEREOF AND ALL PRIOR AGREEMENTS, WHETHER WRITTEN OR ORAL, RELATIVE HERETO AND THERETO WHICH ARE NOT CONTAINED HEREIN OR THEREIN ARE SUPERSEDED AND TERMINATED HEREBY. THE NOTE, THIS DEED OF TRUST AND THE LOAN DOCUMENTS MAY NOT BE CONTRADICTED BY EVIDENCE OF PRIOR, CONTEMPORANEOUS, OR SUBSEQUENT ORAL AGREEMENTS OR DISCUSSIONS OF THE PARTIES HERETO, EXCEPT AS INCORPORATED IN WRITING INTO THE LOAN ARE NO REPRESENTATIONS, UNDERSTANDINGS. DOCUMENTS. THERE STIPULATIONS, AGREEMENTS OR PROMISES, ORAL OR WRITTEN, WITH RESPECT TO THE MATTERS ADDRESSED IN THE LOAN DOCUMENTS.
- 11.12 No Waiver by Lender or Trustee. No course of dealing or conduct by or among Lender, Trustee and Grantor shall be effective to amend, modify or change any provisions of this Deed of Trust or the other Loan Documents. No failure or delay by Lender or Trustee to insist upon the strict performance of any term, covenant or agreement of this Deed of Trust or of any of the other Loan Documents, or to exercise any right, power or remedy consequent upon a breach thereof, shall constitute a waiver of any such term, covenant or agreement or of any such breach, or preclude Lender or Trustee from exercising any such right, power or remedy at any later time or times. By accepting payment after the due date of any of the secured indebtedness, neither Lender nor Trustee shall be deemed to waive the right either to require prompt payment when due of all other secured indebtedness, or to declare an Event of Default for failure to make prompt payment of any such other secured indebtedness. Neither Grantor nor any other person now or hereafter obligated for the payment of the whole or any part of the secured indebtedness shall be relieved of such liability by reason of (a) the failure of Lender to comply with any request of Grantor or of any other Person to take action to foreclose this Deed of Trust or otherwise enforce any of the provisions of this Deed of Trust, or (b) any agreement or stipulation between any subsequent owner or owners of the Property and Lender, or (c) Lender's extending the time of payment or modifying the terms of this Deed of Trust or any of the other Loan Documents without first having obtained the consent of Grantor or such other Person. Regardless of consideration,

and without the necessity for any notice to or consent by the holder of any subordinate Lien on the Property, Lender may release any Person at any time liable for any of the secured indebtedness or any part of the security for the Obligations and may extend the time of payment or otherwise modify the terms of this Deed of Trust or any of the other Loan Documents without in any way impairing or affecting the lien of this Deed of Trust or the priority of this Deed of Trust over any subordinate lien. The holder of any subordinate Lien shall have no right to terminate any Lease regardless of whether or not such Lease is subordinate to this Deed of Trust. Lender may resort to the security or collateral described in this Deed of Trust or any of the other Loan Documents in such order and manner as Lender may elect in its sole discretion.

- 11.13 Attorneys' Fees; Expenses. Grantor shall reimburse Lender for all attorneys' fees and expenses, and all other costs and expenses, arising from and after the date hereof, incurred by Lender in connection with the enforcement of Lender's rights under this Agreement and each of the other Loan Documents, including, without limitation, attorneys' fees and expenses and other costs and expenses for trial, appellate proceedings, out-of-court negotiations, workouts and settlements, and for enforcement of rights under any state or federal statute, including, without limitation, attorneys' fees, costs and expenses incurred in bankruptcy and insolvency proceedings such as (but not limited to) in connection with seeking relief from stay in a bankruptcy proceeding. The term "expenses," as used in the preceding sentence, includes any expenses incurred by Lender in connection with any of the out-of-court, state, federal or bankruptcy proceedings referenced above, including but not limited to the fees and expenses of any appraisers, consultants and expert witnesses retained or consulted by Lender in connection with any of those proceedings. Lender shall also be entitled to its attorneys' fees, costs and expenses incurred in any post-judgment proceedings to collect and enforce the judgment. Grantor will upon demand pay to Lender the amount of any and all expenses, including the fees and expenses of its counsel and of any experts and agents, which Lender may incur in connection with (a) the administration of this Agreement, (b) the custody or preservation of, or the sale of, collection from, or other realization upon, any of the Property. (c) the exercise or enforcement of any of the rights of Lender hereunder, and/or (d) the failure by Grantor to perform or observe any of the provisions hereof. This provision is separate and severable and shall survive the merger of this Agreement into any judgment on this Agreement.
- 11.14 INDEMNIFICATION. GRANTOR HEREBY ACKNOWLEDGES AND AGREES THAT THIS DEED OF TRUST CONTAINS CERTAIN INDEMNIFICATION PROVISIONS, INCLUDING, BUT NOT LIMITED TO, SECTIONS 4.19, 9.2 and 11.1 HEREOF WHICH MAY, IN CERTAIN INSTANCES, INCLUDE INDEMNIFICATION BY GRANTOR OR OTHERS AGAINST LENDER'S OR TRUSTEE'S OWN NEGLIGENCE.
- 11.15 <u>Subrogation</u>. Lender shall be subrogated, for further security, to the lien, although released of record, of any and all encumbrances paid out of the proceeds of the loan evidenced by the Loan Documents.

[SIGNATURE PAGE FOLLOWS]

EXECUTED effective as of the date first above written.

GRANTOR:

FRONT SIGHT MANAGEMENT, LLC a Nevada limited liability company

	By:	tius Piazza, Manager
STATE OF CALIFORNIA)	0
COUNTY OF) SS)	
On , befo	ore me,	, Notary Public
personally appeared Ignatius Piazza,	, Manager, Front Sig	ht Management, LLC, a Nevada limited actory evidence to be the person(s) whose
executed the same in his/her/their ar	uthorized capacity(ie	d acknowledged to me that he/she/theyes), and that by his/her/their signature(s)
on the instrument the person(s), or the instrument.	he entity upon behal	f of which the person(s) acted, executed
		aws of the State of California that the
foregoing paragraph is true and corre	ect.	
Witness my hand and official seal.		**Please See Attach**
1	[Seal]	X California Aciatów ledgment
(Signature)		California Jurat

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document.

STATE OF CALLY	-ornia)			
COUNTY OF <u>San</u>	Francisso	•		
On OCT 06,	20,16 be	fore me, $>$.	Fakeri	
		a Notary Public in	and for said Cour	ity and State
personally appeared, Ig	INOUTIUS PI	1azza -		
\wedge	X	X	, who proved	to me on the
basis of satisfactory evide instrument and acknowled authorized capacity(ies), at the entity upon behalf of w	edged to me that had had that by his/her/the	he/she/they execution signature(s) on	ited the same in the instrument the	his/ber/their
I certify under PENALTY	Y OF PERJURY un	der the laws of t	he State of Calife	ornia that the
foregoing paragraph is true	e and correct.	16	P. FA	
WITNESS my hand and of	fficial seal.	NRO	COMM. #. Notary Public San Mater My Comm. Expire	· California
Signature:		7	5 79 20 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	

(Notary Seal)

EXHIBIT "A" LEGAL DESCRIPTION

PARCEL 1:

A TRACT OF LAND BEING A PORTION OF TRACT 38, OF THE FRACTION TOWNSHIP 22 SOUTH, RANGE 54 EAST, M.D.M. AS SHOWN BY THE INDEPENDENT RE-SURVEY AND SURVEY WITH TRACT SEGREGATION FILED WITH THE BUREAU OF LAND MANAGEMENT ON MAY 10, 1935, ALL SITUATED IN NYE COUNTY, NEVADA, MORE PARTICULARLY DESCRIBED AS FOLLOWS:

BEGINNING AT THE NORTH BOUNDARY CORNER OF THE MOST EASTERLY BOUNDARY LINE OF SAID TRACT 38, BEING THE CORNER KNOWN AS "AP11" OF TRACT 38 AS SHOWN BY SAID BUREAU OF LAND MANAGEMENT SURVEY:

THENCE ALONG THE BOUNDARY LINES OF SAID TRACT 38 ON THE FOLLOWING THREE (3) COURSES: 1) SOUTH 89° 55′ 56" WEST, 1318.50 FEET;

THENCE 2) NORTH 00° 48' 15" WEST, 1309.00 FEET;

THENCE 3) NORTH 89° 19' 08" WEST, 1310.94 FEET;

THENCE SOUTH 07° 25' 58" WEST, 864.51 FEET; SOUTH 51° 50' 25" EAST, 540.22 FEET; THENCE SOUTH 85° 06' 44" EAST, 391.56 FEET; SOUTH 44° 07' 13" EAST, 886.99 FEET; THENCE SOUTH 32° 07' 51" EAST, 909.73 FEET TO A POINT ON THE BOUNDARY LINE OF SAID TRACT 38;

THENCE SOUTH 89° 59' 28" EAST ALONG SAID BOUNDARY LINE OF TRACT 38, 861.95 FEET; THENCE NORTH 00° 48' 57" WEST ALONG SAID BOUNDARY LINE OF TRACT 38, 1308.90 FEET TO THE POINT OF BEGINNING.

MORE COMMONLY KNOW AS: LOT 1 PER RECORD OF SURVEY FOR BOUNDARY LINE ADJUSTMENT MAP, FILE NUMBER 645836, RECORDED DECEMBER 28, 2005.

NOTE: THE ABOVE METES AND BOUNDS DESCRIPTION APPEARED PREVIOUSLY IN THAT CERTAIN DOCUMENT RECORDED DECEMBER 28, 2005 AS INSTRUMENT NO. 645837, OF OFFICIAL RECORDS, NYE COUNTY, NEVADA.

PARCEL 2:

A TRACT OF LAND BEING A PORTION OF TRACT 38, OF THE FRACTION TOWNSHIP 22 SOUTH, RANGE 54 EAST, M.D.M. AS SHOWN BY THE INDEPENDENT RE-SURVEY AND SURVEY WITH TRACT SEGREGATION FILED WITH THE BUREAU OF LAND MANAGEMENT ON MAY 10, 1935, ALL SITUATED IN NYE COUNTY, NEVADA, MORE PARTICULARLY DESCRIBED AS FOLLOWS:

COMMENCING AT THE NORTH BOUNDARY CORNER OF THE MOST EASTERLY BOUNDARY LINE OF SAID TRACT 38, BEING THE CORNER KNOWN AS "AP11" OF TRACT 38 AS SHOWN BY SAID BUREAU OF LAND MANAGEMENT SURVEY;

THENCE ALONG THE BOUNDARY LINES OF SAID TRACT 38 ON THE FOLLOWING THREE (3) COURSES: 1) SOUTH 89° 55' 56" WEST, 1318.50 FEET TO "AP12" OF SAID TRACT 38;

THENCE 2) NORTH 00° 48' 15" WEST, 1309.00 FEET TO "AP13" OF SAID TRACT 38;

THENCE 3) NORTH 89° 19' 08" WEST, 1310.94 FEET TO THE POINT OF BEGINNING OF THE

TRACT OF LAND DESCRIBED HEREIN;

THENCE SOUTH 07° 25' 58" WEST, 864.51 FEET; SOUTH 51° 50' 25" EAST, 540.22 FEET;

THENCE SOUTH 85° 06' 44" EAST, 391.56 FEET; SOUTH 44° 07' 13" EAST, 886.99 FEET;

THENCE SOUTH 32° 07' 51" EAST, 909.73 FEET TO A POINT ON THE BOUNDARY LINE OF SAID TRACT 38;

THENCE ALONG SAID BOUNDARY LINE OF TRACT 38 ON THE FOLLOWING ELEVEN (11)

COURSES: 1) NORTH 89° 59' 28" WEST, 456.95 FEET;

THENCE 2) SOUTH 00° 19' 21" EAST, 2632.07 FEET;

THENCE 3) NORTH 89° 43' 00" WEST, 2650.49 FEET;

THENCE 4) NORTH 00° 00' 22" WEST, 2637.91 FEET;

THENCE 5) NORTH 89° 33' 52" WEST, 2645.16 FEET;

THENCE 6) NORTH 00° 21' 41" EAST, 2638.39 FEET;

THENCE 7) SOUTH 89° 18' 43" EAST, 1308.09 FEET;

THENCE 8) NORTH 01° 14' 10" EAST, 1318.86 FEET;

THENCE 9) SOUTH 88° 49' 59" EAST, 1266.00 FEET;

THENCE 10) SOUTH 00° 32' 57" EAST, 1307.62 FEET;

THENCE 11) SOUTH 89° 19' 08" EAST, 1302.28 FEET TO THE POINT OF BEGINNING.

MORE COMMONLY KNOW AS: LOT 2 PER RECORD OF SURVEY FOR BOUNDARY LINE ADJUSTMENT MAP, FILE NUMBER 645836, RECORDED DECEMBER 28, 2005.

NOTE: THE ABOVE METES AND BOUNDS DESCRIPTION PREVIOUSLY APPEARED IN THAT CERTAIN DOCUMENT RECORDED DECEMBER 28, 2005 AS INSTRUMENT NO. 645838 OF NYE COUNTY, NEVADA.

End of Legal Description

EXHIBIT 2

APA 045-48)-05 And Oc RECORDING REQUESTED BY: AFTER RECORDING, RETURN TO: ROBERT DZIUBLA LAS VEGAS DEVELOPMENT FUND, LLC

ROBERT DZIUBLA
LAS VEGAS DEVELOPMENT FUND, LLC
916 SOUTHWOOD BLVD., SUITE 1G
INCLINE VILLAGE, NV 89450

DOC #886510

Official Records Nye County NV Deborah Beatty - Recorder 01/12/2018 01:26:10 PM Requested By: FNTG NCS (LAS VEGAS)

Recorded By: MJ_RPTT:\$0
Recording Fee: \$35,00
Non Conformity Fee: \$

Page 1 of 6

Space above this line for Recorder's use

FIRST AMENDMENT TO CONSTRUCTION DEED OF TRUST, SECURITY AGREEMENT AND FIXTURE FILING

This Document serves as a Fixture Filing under the Uniform Commercial Code, as amended from time to time, covers goods that are or become fixtures on the land, and is to be filed in the real property records of Nye County, Nevada.

THIS FIRST AMENDMENT TO CONSTRUCTION DEED OF TRUST, SECURITY AGREEMENT AND FIXTURE FILING (this "First Amendment") is made and entered into effective as of July 1, 2017 (the "Effective Date") by FRONT SIGHT MANAGEMENT, LLC, a Nevada limited liability company ("Grantor"), whose address is 1 Front Sight Road, Pahrump, Nevada 89061, to Chicago Title Company ("Trustee") whose address is 725 S. Figueroa Street, Suite 200, Los Angeles, California 90017, for the benefit of Las Vegas Development Fund LLC, a Nevada limited liability company ("Lender"), as beneficiary, whose address is P.O. Box 3003, 916 Southwood Blvd., Suite 1G, Incline Village, Nevada 89450. Lender and Grantor and their respective permitted successors and assigns are sometimes referred to in this First Amendment individually as a "Party" and collectively as the "Parties".

RECITALS

A. Lender and Borrower entered into that certain Construction Loan Agreement dated as of October 4, 2016 (the "Original Loan Agreement"). Pursuant to the Original Loan Agreement, Borrower executed a Promissory Note dated October 6, 2016 (the "Original Note") and a Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated October 6, 2016 and recorded under Document #860867 on October 13, 2016 in the Official Records of Nye County, Nevada (the "Original Deed of Trust"). The Original Loan Agreement was amended by a First Amendment to Loan Agreement effective as of July 1, 2017 (the Original Loan Agreement") and the Original Note was replaced and superseded by an Amended and Restated Promissory Note effective as of July 1, 2017 (the "Promissory Note"). The Original Deed of Trust as amended by this First Amendment to Deed of Trust is referred to herein as the "Deed of Trust".

TAKS IS BEING RECORDED AT THE REQUEST CACHICAGO TITLE AS AN <u>ACCOMMODATION ONLY</u>
WITH NO LIABILITY

Signed in Counterpart AM 045-481-05 Am OK
RECORDING REQUESTED BY:
AFTER RECORDING, RETURN TO:
ROBERT DZIUBLA
LAS VEGAS DEVELOPMENT FUND, LLC
916 SOUTHWOOD BLVD., SUITE 1G
INCLINE VILLAGE, NV 89450

Space above this line for Recorder's use

FIRST AMENDMENT TO CONSTRUCTION DEED OF TRUST, SECURITY AGREEMENT AND FIXTURE FILING

This Document serves as a Fixture Filing under the Uniform Commercial Code, as amended from time to time, covers goods that are or become fixtures on the land, and is to be filed in the real property records of Nye County, Nevada.

THIS FIRST AMENDMENT TO CONSTRUCTION DEED OF TRUST, SECURITY AGREEMENT AND FIXTURE FILING (this "First Amendment") is made and entered into effective as of July 1, 2017 (the "Effective Date") by FRONT SIGHT MANAGEMENT, LLC, a Nevada limited liability company ("Grantor"), whose address is 1 Front Sight Road, Pahrump, Nevada 89061, to Chicago Title Company ("Trustee") whose address is 725 S. Figueroa Street, Suite 200, Los Angeles, California 90017, for the benefit of Las Vegas Development Fund LLC, a Nevada limited liability company ("Lender"), as beneficiary, whose address is P.O. Box 3003, 916 Southwood Blvd., Suite 1G, Incline Village, Nevada 89450. Lender and Grantor and their respective permitted successors and assigns are sometimes referred to in this First Amendment individually as a "Party" and collectively as the "Parties".

RECITALS

A. Lender and Borrower entered into that certain Construction Loan Agreement dated as of October 4, 2016 (the "Original Loan Agreement"). Pursuant to the Original Loan Agreement, Borrower executed a Promissory Note dated October 6, 2016 (the "Original Note") and a Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated October 6, 2016 and recorded under Document #860867 on October 13, 2016 in the Official Records of Nye County, Nevada (the "Original Deed of Trust"). The Original Loan Agreement was amended by a First Amendment to Loan Agreement effective as of July 1, 2017 (the Original Loan Agreement") and the Original Note was replaced and superseded by an Amended and Restated Promissory Note effective as of July 1, 2017 (the "Promissory Note"). The Original Deed of Trust as amended by this First Amendment to Deed of Trust is referred to herein as the "Deed of Trust".

THIS IS BEING RECORDED AT THE REQUEST.

CHICAGO TITLE AS AN ACCOMMODATION ONLY

WITH HO LIABILITY

Signed in Counterpart B. The Parties desire to amend the Original Deed of Trust to modify the rights and obligations of the Parties as further set forth below.

AGREEMENT

NOW THEREFORE, in consideration of the foregoing Recitals, which are hereby incorporated into the operative provisions of this First Amendment by this reference, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties amend the Original Deed of Trust as follows:

- 1. <u>Defined Terms</u>. Initially capitalized terms not defined herein shall have the respective meanings assigned to such terms in the Original Deed of Trust.
- Amendment and Restatement of Article I of the Original Deed of Trust.
 Article I of the Original Deed of Trust is hereby amended and restated from and after the Effective Date as follows:

"ARTICLE I The Loan

- 1. Loan. The indebtedness secured by this Deed of Trust is the result of a loan in the original principal amount of up to Fifty Million Dollars \$50,000,000 (the "Loan") provided by Lender to Grantor. The Loan is evidenced by (a) that certain Construction Loan Agreement dated October 6, 2016, by and between Grantor and Lender, as amended by that certain First Amendment to Loan Agreement (as amended, together with any further extensions, revisions, modifications or amendments thereto, the "Loan Agreement"), dated as of the Effective Date, by and between Grantor and Lender, and (b) that certain Amended and Restated Promissory Note executed dated as of the Effective Date, by Grantor, payable to the order of Lender in the maximum original principal amount of the Loan (together with any extensions, revisions, modifications or amendments hereafter made, the "Note")."
- Agreement Ratified. Except as specifically amended or modified herein, each and every term, covenant, and condition of the Deed of Trust as amended is hereby ratified and shall remain in full force and effect.
- 4. Governing Law. This instrument shall be interpreted and construed in accordance with the laws of the State of Nevada.
- 5. <u>Binding Agreement</u>. This First Amendment shall be binding upon and inure to the benefit of the Parties hereto and their respective successors and permitted assigns.
- 6. <u>Counterparts</u>. This First Amendment may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same document.

[SIGNATURES APPEAR ON FOLLOWING PAGE]

IN WITNESS WHEREOF, Lender and Borrower have signed this First Amendment as of the First Amendment Effective Date.

BORROWER:

FRONT SIGHT MANAGEMENT, LLC, A Nevada Limited Liability Company

By:

Name: Ignatius Piazza

Title: Manager

LENDER:

LAS VEGAS DEVELOPMENT FUND, LLC, A Nevada Limited Liability Company

By:_

Name:

Title:

PLEASE SEE ATTACHED CALIFORNIA ALL-PURPOSE ACKNOWLEDGEMENT FORM

CALIFORNIA ALL-PURPOSE ACKNOWLEDGMEN		
A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document.		
State of California County of Socoma. On Nov. 14, 2017 before me, Jo Date personally appeared Tanatius And	olm A Russell NotAppy Public. Here Insert Name and Title of the Officer	
who proved to me on the basis of satisfactory evidence to the within instrument and acknowledged to me that authorized capacity(ies), and that by his/her/their signs	te to be the person(s) whose name(s) is/are subscribed the/she/they executed the same in his/ber/their ature(s) on the instrument the person(s), or the entity	
JOHN A. RUSSELL. Notary Public - California Sonoma County Commission # 2171851 My Comm. Expires Nov 13, 2020	I certify under PENALTY OF PERJURY under the laws of the State of California that the foregoing paragraph is true and correct. WITNESS my hand and official seal.	
Place Notary Seal and/or Stamp Above OPTI	Signature of Notary Public ONAL	
Completing this information can deter alteration of the document or fraudulent reattachment of this form to an unintended document.		
Description of Attached Document Title or Type of Document: First Amendment To Contraction Deed of Trust Security Document Date: 11/14/2017 Number of Pages: 3		
Signer(s) Other Than Named Above:		
Capacity(ies) Claimed by Signer(s) Signer's Name: Corporate Officer — Title(s): Partner — Limited General Individual Attorney in Fact Guardian of Conservator Other: Signer is Representing:	Signer's Name: Corporate Officer – Title(s): Partner – Limited General Individual Attorney in Fact Trustee Guardian of Conservator Other: Signer is Representing:	

©2017 National Notary Association

IN WITNESS WHEREOF, Lender and Borrower have signed this First Amendment as of the First Amendment Effective Date.

- - 3⁸

BORROWER:	
FRONT SIGHT MANAGEMENT, LLC, A Nevada Limited Liability Company	
Ву:	
Name: Ignatius Piazza	
Title: Manager	
LENDER:	
LENDEK:	
LAS VEGAS DEVELOPMENT FUND, LLC, A Nevada Limited Liability Company	
By: Rebert Dzsubbe	SEE ATTACHED
Title: Pasid 7 & CFO	

ACKNOWLEDGMENT

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document.

attached, and not the truthfulness, accuracy, or validity of that document.	
State of California County of <u>Sun Diego</u>	
On January 67 7018 before me, Tyler (insert r	name and title of the officer)
who proved to me on the basis of satisfactory evidence to be subscribed to the within instrument and acknowledged to make their authorized capacity(ies), and that by his/her/their person(s), or the entity upon behalf of which the person(s) at I certify under PENALTY OF PERJURY under the laws of the paragraph is true and correct.	e that (he)she/they executed the same in eir signature(s) on the instrument the acted, executed the instrument.
WITNESS my hand and official seal.	TYLER GAWRONSKI COMM. # 2166736 UNDTARY PUBLIC-CALIFORNIA UNDTARY PUBLIC PUBLIC PUBLIC PUBLIC PUBLIC PUBLIC PUBLIC PUBLICA UNDTARY PUBLIC PUBLIC PUBLIC PUBLIC PUBLIC PUBLIC P
Signature (Seal)	

EXHIBIT 3

CONSTRUCTION LOAN AGREEMENT

by and between

FRONT SIGHT MANAGEMENT LLC

a Nevada limited liability company as Borrower

and

LAS VEGAS DEVELOPMENT FUND LLC,

a Nevada limited liability company, as Lender

Dated: October 6, 2016

TABLE OF CONTENTS

RECITALS Recital A Recital B _____1 Recital C DEFINITIONS **Definitions** ARTICLE I LOAN Section 1.1 Section 1.2 Section 1.3 Prepayment14 Section 1.4 Payments ______14 Section 1.5 Section 1.6 Extension of Maturity Date......15 Section 1.7 ARTICLE II CONDITIONS OF BORROWING Section 2.1 Section 2.2 Section 2.3 Section 2.4 ARTICLE III ADVANCES OF LOAN PROCEEDS Section 3.1 Section 3.2 Section 3.3 Inspections20 Lender's Responsibilities......20 Section 3.4 Procedures for Advances21 Section 3.5 Section 3.6 Section 3.7 ARTICLE IV REPRESENTATIONS AND WARRANTIES Section 4.1 Authority......22 Section 4.2 No Approvals22 Section 4.3 i

Page

TABLE OF CONTENTS

(continued)

		Page
¥		
Section 4.4	Legal and Valid Obligations	23
Section 4.5	Litigation	23
Section 4.6	Title	23
Section 4.7	Defects and Hazards	23
Section 4.8	Payment of Taxes	23
Section 4.9	Agreements	
Section 4.10	No Defaults under Loan Documents or Other Agreements	24
Section 4.11	Boundary Lines; Conformance with Governmental Requirements and	
	Restrictions	
Section 4.12	Project Costs	
Section 4.13	Utilities	25
Section 4.14	Personal Property	25
Section 4.15	Condemnation	25
Section 4.16	Separate Lots	25
Section 4.17	Federal Reserve Regulations	25
Section 4.18	Investment Company Act	26
Section 4.19	Unregistered Securities	26
Section 4.20	Accuracy of Information	26
Section 4.21	ERISA Compliance	26
Section 4.22	Compliance	26
Section 4.23	Consents	27
Section 4.24	Environmental Laws	27
Section 4.25	Anti-Terrorism Regulations	27
Section 4.26	Subsidiaries	28
Section 4.27	Leases	28
Section 4.28	Ownership and Control of Borrower	
Section 4.29	Use of Loan Proceeds	28
Section 4.30	Insurance	29
	ARTICLE V	
(A) COVEN	ANTS OF BORROWER	30
Section 5.1	Completing Construction	30
Section 5.2	Changing Costs, Scope or Timing of Work	30
Section 5.3	Using Loan Proceeds	31
Section 5.4	Keeping of Records	31
Section 5.5	Providing Evidence of Completion	.,31
Section 5.6	Maintaining Insurance Coverage	
Section 5.7	Transferring, Conveying or Encumbering the Project or Borrower's	
	Ownership Interests	32
Section 5.8	Complying with the Loan Documents and Other Documents	
Section 5.9	Updated Appraisals	
Section 5.10	Reporting Requirements	

TABLE OF CONTENTS

(continued)

		Page
Section 5.11	Taxes and Claims	3.4
Section 5.11	Maintain Existence	
Section 5.12	Compliance with Applicable Laws	
Section 5.13	Notice	
Section 5.14	Contingent Liability	
Section 5.15	Merger, Consolidation, and Management	
Section 5.17	Loss of Note or other Loan Documents	
Section 5.17	Distributions	
	Permits and Licenses.	
Section 5.19	Patriot Act	
Section 5.20		
Section 5.21	Related Party Transactions	
Section 5.22	Leases	
Section 5.23	Debt; Operations and Fundamental Changes of Borrower	
Section 5.24	Accessibility Regulation	
Section 5.25	Reports and Returns	
Section 5.26	Management Agreement	
Section 5.27	Senior Debt	39
(B) COVENA	NT OF LENDER	39
. ,		
	ARTICLE VI	
	DEFAULTS	
Section 6.1	Events of Default	39
Section 6.2	Rights and Remedies	41
Section 6.3	Completion of Project by Lender	42
	ARTICLE VII	
	INTEREST, FEES AND EXPENSES	
Section 7.1	Interest; Fees; and Expenses	43
Section 7.2	Authorization to Make Loan Advances to Cure Borrower's Defau	
	ADTICLE STIL	
	ARTICLE VIII MISCELLANEOUS	
Section 8.1	Waiver and Amendment	4.4
Section 8.1		
	Expenses and Indemnities	
Section 8.3	Binding Effect; Waivers; Cumulative Rights and Remedies	
Section 8.4	Incorporation By Reference	
Section 8.5	Survival	
Section 8.6 Section 8.7	Governing Law; Waiver of Jury Trial; Jurisdiction	
	Counterparts	
Section 8.8	Notices	
Section 8.9	No Third Party Reliance	48

TABLE OF CONTENTS (continued)

(continuea)		
		<u>Page</u>
40		_
0 1 0 10	Y 4 4	40
Section 8.10	Lender Assignment	
Section 8.11	Time of the Essence	
Section 8.12	No Oral Modifications	48
Section 8.13	Captions	48
Section 8.14	Borrower-Lender Relationship	48

EXHIBITS

EXHIBIT A	Budget
EXHIBIT B	Draw Request
EXHIBIT C	Draw Request Certificate
EXHIBIT D	Legal Description
EXHIBIT E	Estimated Construction Cost Statement
EXHIBIT F	Improvements
EXHIBIT G	Ownership and Control

CONSTRUCTION LOAN AGREEMENT

.

THIS CONSTRUCTION LOAN AGREEMENT (this "Agreement") is made and entered into this Fourth day of October, 2016, by and between FRONT SIGHT MANAGEMENT LLC, a Nevada limited liability company ("Borrower"), and LAS VEGAS DEVELOPMENT FUND, LLC, a Nevada limited liability company ("Lender"), with respect to the following facts:

RECITALS

- A. Borrower has requested that the Lender provide the Loan (as hereinafter defined) to Borrower in the principal sum of up to SEVENTY-FIVE MILLION DOLLARS (\$75,000,000) for the purpose of paying off Existing Liens (as hereinafter defined) and financing the construction of the Improvements (as hereinafter defined) in accordance with the Budget (as hereinafter defined).
- Borrower has secured, or will endeavor to secure, financing for the remaining B. costs of the Project (as hereinafter defined) as follows: (1) other equity contributions from Borrower in the amount of \$75,000,000, in the aggregate, on the date the Loan is made, which includes the enterprise value of Borrower's primary business operation, Front Sight Firearms Training Institute ("FSFTI"), which enterprise value was determined by Bring & Company, Inc. to be \$50,000,000 on September 21, 2015, (2) Borrower's equity in the land on which the Project will be located, as well as the land and improvements on the adjacent parcel which is the location of the existing FSFTI, which land is currently encumbered, inter alia, by a deed of trust with a current outstanding balance of approximately \$4,661,446.07 as of September 10, 2016, and (3) such additional construction financing as may be secured by Borrower at a date subsequent to the date of this Agreement, with the understanding that any and all liens securing such additional construction financing will be superior to the liens securing the Loan evidenced by this Agreement. The land on which the Project will be located has an appraised value of \$25,000,000. The enterprise value of FSFTI and the value of the land and improvements thereon are referred to collectively as the "Borrower's Equity".
- C. Based upon (a) the efforts made by Lender to comply with the rules and regulations promulgated under the EB-5 Program (as hereinafter defined) and (b) the materials and submissions developed by Lender, the Loan has been structured for the payment of costs relating to the development of the Project as set forth in the Budget approved by Lender.
- NOW, THEREFORE, in consideration of the mutual covenants and agreements hereinafter set forth, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

DEFINITIONS

For purposes of this Agreement, the following terms shall have the following respective meanings, unless expressly established otherwise:

"Accessibility Regulation" means any federal, state or local law, statute, code, ordinance, rule, regulation or requirement, including, without limitation, under the United States Americans With Disabilities Act of 1990, as amended (the "ADA"), relating to accessibility to facilities or properties for disabled, handicapped and/or physically-challenged persons, or other persons covered by the ADA.

"Advances" means (i) any portion of the Loan advanced by Lender to or for the benefit of Borrower in accordance with Article III of this Agreement; (ii) any advance by Lender to protect the Project or the lien of the Loan Documents, including Protective Advances; and (iii) any other advance by Lender required or permitted under this Agreement.

"Affiliate" means any Person, together with any other Person directly or indirectly controlling, controlled by or under common control with such Person. A Person shall be deemed to control another Person if the controlling Person (i) owns 25% or more of the voting securities (or other ownership interests) of the controlled Person, or (ii) possesses, directly or indirectly, the power to direct or cause the direction of the management or policies of the controlled Person, whether through ownership of stock, by contract or otherwise.

"Agreement" means this Construction Loan Agreement, including any amendments and/or supplements hereto executed by and between Borrower and Lender.

"Anti-Terrorism Laws" means any laws relating to terrorism or money laundering, including Executive Order No. 13224, the USA Patriot Act, the Laws comprising or implementing the Bank Secrecy Act, and the Law administered by the United States Treasury Department's Office of Foreign Asset Control (as any of the foregoing Laws may from time to time be amended, renewed, extended, or replaced).

"Architect's Agreement" means that certain agreement to be entered into by and between Borrower and the Project Architect for design of the construction work with respect to the Improvements and other services related to the same.

"Blocked Person" has the meaning assigned to it in Section 4.26(b).

"Borrower" has the meaning assigned to it in the introductory paragraph hereof.

"Borrower Equity" means other equity contributions from the Borrower as described in Recital B above.

"Borrower Operating Agreement" means that certain Amended and Restated Operating Agreement dated February 16, 2012.

"Borrower's Organizational Documents" means all formation documents of Borrower, including the LLC-1, including any amendments thereof and supplements thereto.

"Budget" means an itemized statement of actual and estimated costs to be incurred by Borrower with respect to the construction of the Improvements, and other non-construction costs relating to the Project and Borrower's business operations as set forth in Exhibit A, attached

hereto and made a part hereof, signed and sworn to by Borrower, as the same may be amended or supplemented from time to time, specifically allocated to be paid with Loan proceeds.

"Business Day" means any day other than a Saturday, a Sunday, or a legal holiday on which banks in Las Vegas, Nevada, are not open for business.

"Closing Date" means the date that this Agreement is executed, which shall be the date on or about which Lender reasonably anticipates making the initial Advance to the Borrower pursuant to the terms and conditions of this Agreement.

"Code" means the Internal Revenue Code of 1986, as amended.

.

"Collateral" means (a) all of the collateral covered by the Deed of Trust, this Agreement or any other Loan Document, and (b) all accessions to, substitutions for and replacements, products and proceeds of any of the foregoing, including, but not limited to, proceeds of any insurance policies, claims against third parties, and condemnation or requisition payments with respect to all or any of the foregoing.

"Commencement Date" means the date following installation of the required infrastructure on the Land and on which construction of the buildings that will constitute the Front Sight Resort and Vacation Club units commences.

"Commitment" means an amount not to exceed seventy five million dollars (\$75,000,000). Such Commitment shall be reduced by any principal payments made by or on behalf of Borrower or any principal reductions otherwise required under and pursuant to the Loan Documents.

"Completion" means that (i) such portion of the Improvements are substantially completed in accordance with the Plans, as reasonably approved by Lender, paid for in full, free of all mechanics', labor, materialmen's and other similar lien claims, and substantial completion has been certified by the Project Architect; (ii) a certificate of substantial completion for such Improvements has been signed by Borrower delivered to Lender; (iii) Lender has received acceptable evidence that all Governmental Requirements and all private restrictions and covenants relating to such Improvements have been complied with or satisfied (if applicable) and that final certificates of occupancy for such Improvements, if applicable, have been issued by all appropriate governmental authorities; and (iv) the requirements in Section 5.7 have been satisfied.

"Completion Date" means the date that is no later than thirty-six (36) months from the Commencement Date. The Completion Date shall be subject to extensions for delays resulting from Force Majeure (as defined herein), provided, that Borrower gives written notice to Lender immediately upon becoming aware of the occurrence of any Force Majeure condition and, provided further, that the aggregate period of any and all such Force Majeure delays shall not exceed ninety (90) days.

"Contractor" means any person, party or entity which has a contract or subcontract under which payment may be required for any work done, material supplied or services furnished in connection with acquiring, constructing, financing, equipping and/or developing the Project.

"Control" means the power to direct or cause the direction of the management and policies of a Restricted Party or any other Person, directly or indirectly, whether through the ownership of voting securities or other beneficial interests, by contract or otherwise.

"<u>Deed of Trust</u>" means the Deed of Trust, Assignment of Leases and Rents, Security Agreement and Fixture Filing of even date herewith, encumbering the Project, executed by Borrower in favor of Lender to secure the Loan, including any amendments, modifications and/or supplements thereto.

"<u>Default</u>" means any event which, with the giving of notice to Borrower or the lapse of time, or both, would constitute an Event of Default.

"<u>Default Rate</u>" means the lesser of five percent (5%) per annum in excess of the Loan Rate or the maximum lawful rate of interest which may be charged, if any.

"<u>Draw Request</u>" means a spreadsheet summary of the Budget, provided by Borrower for an Advance of Loan proceeds under this Agreement in the form of <u>Exhibit B</u> attached hereto, together with a Draw Request Certification. General Contractor shall be signatory to each Draw Request to acknowledge its approval of the terms therein.

"<u>Draw Request Certification</u>" means a certification from Borrower to accompany all Advances for Loan proceeds under this Agreement, in the form of <u>Exhibit C</u> attached hereto.

"EB-5 Information" means such documents, certificates, and accounting information required to be submitted by Borrower to Lender pursuant to Article V(A), Section 5.10 hereof for purposes of documenting compliance by Borrower with certain aspects of this Agreement and the EB-5 Program.

"EB-5 Investors" has the meaning assigned to it in Section 1.7(b).

"EB-5 Program" has the meaning assigned to it in Section 1.7(b).

"Environmental Impact Study" means the Phase I Environmental Impact Study previously delivered to Lender and prepared by GeoTek, Inc., and dated as of February 2, 2012.

"Environmental Law" means all federal, state, regional, county and local statutes, regulations, ordinances, rules, regulations and policies, all court and administrative orders and decrees and arbitration awards, and the common law, which pertain to environmental matters or contamination of any type whatsoever, including, but not limited to, those relating to the presence, manufacture, processing, use, distribution, treatment, storage, disposal, generation or transportation of Hazardous Substances; air, water (including surface water, groundwater, and stormwater) or soil (including subsoil) contamination or pollution; the presence or Release of

Hazardous Substances, protection of wildlife, endangered species, wetlands or natural resources; health and safety of employees and other persons; and notification requirements relating to the foregoing, including, without limitation, the following statutes, and regulations adopted thereunder: the Comprehensive Environmental Response, Compensation and Liability Act, as amended by the Superfund Amendments and Reauthorization Act of 1986, 42 U.S.C. § 9601 et seq. ("CERCLA"); the Solid Waste Disposal Act, as amended by the Resource Conservation Recovery Act and the Hazardous and Solid Waste Amendments of 1984, 42 U.S.C. § 6901 et seq. ("RCRA"); the Federal Water Pollution Control Act, as amended by the Clean Water Act of 1977, 33 U.S.C. § 1251 et seq.; the Clean Air Act, as amended, 42 U.S.C. § 7401 et seq.; the Toxic Substances Control Act, 15 U.S.C. § 2601 et seq. ("TSCA"); the Safe Drinking Water Act, 42 U.S.C. §§ 300f through 300j; the Hazardous Materials Transportation Act, 49 U.S.C. § 1801 et seq.; the Oil Pollution Act of 1990, 33 U.S.C. § 2701 et seq.; the Emergency Planning and Community Right-to-Know Act, 42 U.S.C. § 11001 et seq.; and the Occupational Safety and Health Act, 19 U.S.C. § 6251 et seq., as each of the foregoing may be amended from time to time.

"Environmental Liability" means any claim, demand, obligation, cause of action, allegation, order, violation, damage, injury, judgment, penalty or fine, cost of enforcement, cost of remedial action, diminution in value or any other cost or expense whatsoever, including reasonable attorneys' fees and disbursements, resulting from the presence or use of Hazardous Substances, the violation or alleged violation of any Environmental Law, or the imposition of any Environmental Lien.

"Environmental Lien" means a Security Interest in favor of any third party for: (a) any liability under an Environmental Law; or (b) damages arising from or costs incurred by such third party in response to a Release or threatened Release of any Hazardous Substance or constituent into the environment.

"Equipment" means all furniture, fixtures and equipment directly acquired by Borrower with the proceeds of the Loan and located or to be located in or on, and used in connection with, the management, maintenance or operation of, the Land and the Improvements.

"ERISA" means the Employee Retirement Income Security Act of 1974, as the same may from time to time be amended, and the rules and regulations promulgated thereunder by any governmental agency or authority, as from time to time in effect.

"ERISA Affiliate" means any trade or business (whether or not incorporated) which is a member of a group of which Borrower is a member and which is under common control within the meaning of Section 414 of the Code, as amended from time to time, and the regulations promulgated and rulings issued thereunder.

"<u>Escrow Account</u>" means the account that has been established for the benefit of Lender with the Escrow Agent.

"Escrow Administrator" means NES Financial Corp., a California corporation.

"Escrow Agent" means Time Escrow, Inc, 3055 Wilshire Boulevard, Suite 1150, Los Angeles, Ca. 90010

"Escrow Agreement" means that certain agreement between Escrow Agent and the Lender effective as of July 1, 2016.

"Estimated Construction Cost Statement" means an itemized statement of actual and estimated costs of the Project, in the form of Exhibit E attached hereto and hereby made a part hereof, signed by Borrower, General Contractor and the Project Architect, as the same may be amended or supplemented, and consistent with the items enumerated in the Budget.

"Event of Default" has the meaning assigned to it in Section 6.1 hereof.

"Excluded Taxes" means, in the case of Lender, taxes imposed on its overall net income, and franchise taxes imposed on it, by (a) the United States of America or any subdivision thereof, (b) the jurisdiction under the laws of which Lender is incorporated or organized, (c) the jurisdiction in which the Lender's principal executive office is located, or (d) any foreign government or subdivision thereof.

"Extension Fee" means a non-refundable fee in the amount of one-percent (1%) of the then-existing outstanding principal balance of the Loan as the date of the first day of the Extension Term, payable by Borrower to Lender on or before the first day of the Extension Term. "Extension Term" has the meaning assigned to it in Section 1.6.

"Fees" means the Extension Fee.

"<u>First Option Maturity Date</u>" as set forth in <u>Section 1.6</u>, shall be the date twenty-four (24) months after the Initial Maturity Date.

"Fiscal Year" means the period of January 1 of any year through December 31 of such year.

"Force Majeure" means any act of God; strikes, shortage or unavailability of labor or materials; lockouts or labor difficulty, explosion; sabotage; accident; riot or civil commotion; act of war; fire or other casualty; adverse weather conditions; governmental delays; legal requirements; and other causes beyond the reasonable control of Borrower.

"GAAP" means generally accepted accounting principles consistently applied and maintained throughout the period indicated and consistent with the financial statements delivered to Lender pursuant to Article V. Whenever any accounting term is used herein and is not otherwise defined, it shall be interpreted in accordance with GAAP.

"General Contract" means one or more agreements by and between Borrower and General Contractor for the construction of the Project.

"General Contractor" means one or more general contractors duly licensed in the State of Nevada and selected by Borrower, as identified to Lender on or before the Commencement Date.

"Governmental Authority" means any court, board, agency, commission, office or authority of any nature whatsoever or any governmental unit (federal, state, commonwealth, county, district, municipal, city or otherwise) whether now or hereafter in existence.

"Governmental Requirements" means all Laws, statutes, codes, ordinances, and governmental rules, regulations and requirements of a Governmental Authority applicable to Borrower, Lender or the Project, including, without limitation, Environmental Laws, and the requirements of the ADA, and all covenants, agreements, restrictions and encumbrances contained in any instruments, either of record or known to Borrower, at any time in force affecting the Project or any part thereof, including any which may (i) require repairs, modifications or alterations in or to the Project or any part thereof, or (ii) in any way limit the use and enjoyment thereof.

"Hazardous Substance(s)" means any substance, chemical, compound, product, solid, gas, liquid, waste, byproduct, pollutant, contaminant, or material which is defined or regulated under any Environmental Law, and includes, without limitation, (a) mold, asbestos, polychlorinated biphenyls, and petroleum (including petroleum products or derivatives, crude oil or any fraction thereof), and (b) any material classified or regulated as "hazardous waste" pursuant to RCRA.

"Holdback" means 25% (\$125,000) of each EB-5 Investor's subscription, held in the Escrow Account for Lender's benefit, which will ultimately either be made available for refund to an investor if the Release Condition is not satisfied or, if the Release Condition is satisfied, be made available for an Advance to Borrower.

"Management Agreement" means that certain Club Management Agreement to be entered into by and between Borrower, Front Sight Resort and Vacation Club Members Association, Inc., and Manager

"Manager" means LaTour Hotels and Resorts, Inc., a California corporation, and any successor manager approved by Lender.

"Improvements" means the buildings and improvements, including the existing facilities used by FSFTI in the operation of its business (the "FSFTI Facility") as well as all structures or improvements to be built on the Land, including, without limitation, the items described in Exhibit F, attached hereto, which is based on the description of the Project in the Executive Summary of the Business Plan, as well as site work, landscaping, parking areas, access drives, offices, and common areas which are to be placed or constructed upon, above or below the Land.

"Indebtedness" means in all cases without duplication, all items of indebtedness or liability of Borrower other than the Obligations, at any time which in accordance with GAAP would be included in determining total liabilities as shown on the liability side of a consolidated balance sheet of Borrower as of the date of determination, including: (a) indebtedness for

borrowed money; (b) obligations under direct or indirect guaranties of indebtedness or obligations of others referred to in clause (a) above; (c) any indebtedness secured by any Security Interest on the property of such entity; and (d) liabilities in respect of unfunded vested benefits under any Plan for which the minimum funding standards of Section 302 of ERISA have not been met.

"Indemnified Parties" has the meaning assigned to it in Section 8.2(b).

"Initial Maturity Date" means the date sixty (60) months after the first disbursement of funds by Lender to Borrower under this Agreement.

"Initial Term" means that period of time commencing on the Closing Date and ending sixty (60) months after the first disbursement of funds by Lender to Borrower under this Agreement.

"Interest Reserve" means a portion of the proceeds of the Loan allocated to pay interest on the Loan through Completion of the Improvements, the initial and continuing amount of which shall be the equivalent of three months' worth of interest, calculated at the Loan Rate, on the then-outstanding principal balance of the Loan.

"<u>Land</u>" means approximately 550 acres of land located in Nye County, Nevada, more specifically described on <u>Exhibit D</u>, attached hereto and made a part hereof by this reference.

"Late Charge" has the meaning assigned to it in Section 1.2.

"<u>Laws</u>" means all federal, state and local laws, statutes, codes, ordinances, rules and regulations, including judicial opinions and presidential authority in the applicable jurisdiction.

"Lease" means any lease, sublease or sub-sublease, letting, license, concession or other agreement (whether written or oral and whether now or hereafter in effect), pursuant to which any Person is granted a possessory interest in, or right to use or occupy, all or any portion of any space in the Project, and every modification, amendment or other agreement relating to such lease, sub-sublease or other agreement entered into in connection with such lease, sublease, sub-sublease or other agreement and every guarantee of the performance and observance of the covenants, conditions and agreements to be performed and observed by the other party thereto.

"<u>Lender-Approved Appraisal</u>" means that certain appraisal dated as of October 8, 2014, prepared by Hospitality Real Estate Counselors and previously provided to, and accepted and approved by, Lender.

"Liens" means any mortgage, deed of trust, deed to secure debt, lien (statutory or otherwise, but excluding liens for ad valorem taxes that are not delinquent), pledge, hypothecation, easement, restrictive covenant, preference, assignment, security interest, or any other encumbrance, charge or transfer of, or any agreement to enter into or create any of the foregoing, on or affecting all or any portion of the Project or any interest therein, or any direct or indirect interest in Borrower, including any conditional sale or other title retention agreement,

any financing lease having substantially the same economic effect as any of the foregoing, the filing of any financing statement, and mechanic's, materialman's, construction and other similar liens and encumbrances.

"Loan" means collectively, the loan of the proceeds of the Note by Lender to Borrower in Advances to be made pursuant to the terms of this Agreement in the maximum aggregate principal amount not to exceed the Commitment.

"Loan Documents" means all documents now or hereafter entered into which evidence, secure and/or govern the Loan and/or any of the Obligations, including, but not limited to, this Agreement, the Note and the Deed of Trust.

"Loan Rate" means the interest rate applicable to the Loan as calculated at an annual rate of 6% during the Initial Term and, if extended, 7% during the Extension Term.

"Material Adverse Occurrence" means any occurrence of whatsoever nature (including, without limitation, any firm, final and unappealable adverse determination in any litigation, arbitration or governmental investigation or proceeding) which Lender shall reasonably determine could materially adversely affect the then-present or prospective financial condition or operations of Borrower, the value of the Project or any other material Collateral securing repayment of the Loan, or impair the ability of Borrower to perform its obligations as and when required under any of the Loan Documents.

"Material Subcontractor" means any subcontractor providing materials, supplies or labor to the Project under a contract or multiple contracts in the aggregate amount of \$250,000 or more.

"Maturity Date" means the Initial Maturity Date, subject to being extended as set forth in Section 1.6 below.

"Note" means the Promissory Note of even date herewith, executed and delivered by Borrower to the order of Lender, in the original maximum principal amount of the Commitment, including any amendments and/or restatements thereof and supplements thereto executed by Borrower and Lender.

"Obligations" means, collectively: (i) Borrower's obligations for the payment of the Loan, interest and other charges, and all Fees; (ii) the performance of all other obligations of Borrower contained herein; (iii) the payment and performance of each and every obligation of Borrower contained in any other Loan Document; and (iv) the performance of each and every obligation of Borrower contained in any renewal, extension, amendment, modification, consolidation, change of, or substitution or replacement for, all or any part hereof, the Note or any other Loan Document.

"Operating Budget" means a detailed listing of all anticipated annual income and expenses from and for managing, maintaining and operating the Project (or any portion thereof) for its first full or partial Fiscal Year and for each succeeding Fiscal Year of operation, prepared by Borrower and in form and substance reasonably acceptable to Lender.

"Operating Expenses" means actual operating expenses of the Project paid in cash by Borrower during such period, other than those operating expenses that are paid from Advances made by Lender to Borrower pursuant to this Agreement.

"Operating Statement" means, for any period, a current, detailed statement of income and expenses from and for managing, maintaining and operating the Project for such period, in form and substance reasonably acceptable to Lender.

"Other Taxes" means any present or future stamp or documentary taxes and any other excise or property taxes, charges or similar levies which arise from any payment made hereunder or under the Note or from the execution or delivery of, or otherwise with respect to, this Agreement or the Note.

"Permitted Encumbrances" mean the Liens, charges and encumbrances on title to the Land listed on Schedule B, I to the Title Commitment on the Closing Date and such other matters of title thereafter approved by Lender in writing. For the avoidance of doubt, any liens, charges and/or encumbrances securing the Senior Debt shall be considered as "Permitted Encumbrances" as and when such liens, charges and/or encumbrances are granted by Borrower in favor of the provider of the Senior Debt.

"Person" means an individual, corporation, partnership, limited liability company, joint venture, trust or unincorporated organization.

"Plan" means each employee benefit plan covered by Title IV of ERISA whether now in existence or hereafter instituted, of Borrower or any ERISA Affiliate.

"Plans" mean the final construction plans for the Improvements, including drawings, specifications, details and manuals, as approved by the applicable Governmental Authority responsible for reviewing and approving construction plans for compliance with applicable Governmental Requirements.

"Project" means the Land, the Improvements and the Equipment.

"Project Architect" means K.C. Camis Architect or such other architect(s) duly licensed in accordance with the laws of Nevada as may be selected by Borrower.

"Protective Advance" means all necessary costs and expenses (including reasonable attorneys' fees and disbursements) reasonably incurred by Lender in order to remedy an Event of Default under the Loan Documents, which Event of Default, by its nature, may impair any portion of the collateral for the Loan or the value of such collateral, interfere with the enforceability or enforcement of the Loan Documents, or otherwise materially impair the payment of the Loan, and other Obligations (including, without limitation, the costs of unpaid insurance premiums, foreclosure costs, costs of collection, costs incurred in bankruptcy proceedings and other costs incurred in enforcing any of the Loan Documents.

"Regional Center" means EB5 Impact Capital Regional Center, LLC, a Nevada limited liability company.

"Related Party" means any one or more of the following: (a) an Affiliate of Borrower, or (b) any of the shareholders, partners, members or other equity holders of Borrower, and any Affiliate thereof.

"Release" means, without limitation, (a) any intentional, unintentional, knowing or unknowing presence, spilling, leaking, pumping, pouring, emitting, emptying, discharging, migrating, injecting, escaping, leaching, dumping or disposing any Hazardous Substance at, on or into the indoor or outdoor environment or otherwise in, onto, from or about the air, water (including surface waters and groundwater), soils, subsoils or any other surface or media on-site or off-site, and (b) the abandonment or discarding of barrels, drums, containers, underground tanks, or any other receptacles ever containing any Hazardous Substances.

"Release Condition" means approval of an EB-5 Investor's I-526 Immigrant Petition by the USCIS.

"Restricted Party" means Borrower, and any shareholder, partner, member or nonmember manager, or any direct or indirect legal or beneficial owner, of Borrower, from time to time.

"Security Interest" means any lien, pledge, mortgage, encumbrance, charge or security interest of any kind whatsoever (including, without limitation, the lien or retained security title of a conditional vendor) whether arising under a security instrument or as a matter of law, judicial process or otherwise or the agreement by Borrower, or any Subsidiary of Borrower, to grant any lien, security interest or pledge, mortgage or encumber any asset.

"Senior Debt" means the additional loan that will be sought by Borrower, and which Borrower will use it best efforts to obtain, from a traditional financial institution specializing in financing projects such as the Project. Although the Senior Debt would be funded subsequent to this Loan, Lender agrees to subordinate its Deed of Trust to the new Senior Debt, so long as the Borrower is not in default and all of the following conditions are met;

- (a) The loan shall be evidenced by a promissory note not in excess of Fifty Million and no/100 United States Dollars (US\$50,000,000.00).
- (b) The loan proceeds shall be disbursed in payment, or in reimbursement for payment, of the construction and development of the Project.
- (c) The loan shall contain provisions concerning disbursement procedures, mechanisms to protect against mechanics liens and related matters as are customarily found in construction loans made by institutional lenders and Lender shall be provided with copies of such documents showing the progress of construction and the disbursement of funds as are provided to senior lender.

Borrower shall obtain such Senior debt no later than December 31, 2016

"Subscription Conditions" means, for each EB-5 Investor, the following conditions: (a) Lender's receipt of completed subscription documents; (b) deposit of the entire subscription price into the Escrow Account; (c) proof of I-526 filing with the USCIS; and (d) EB-5 Investor receipt of notice of acceptance from Lender.

"Subsidiary" means any corporation or other entity of which more than 50% of the outstanding capital stock or interests having ordinary voting power to elect a majority of the board of directors or the board of governors or otherwise to Control the activities of such entity (irrespective of whether or not at the time other class or classes of the equity of such entity shall or might have voting power upon the occurrence of any contingency) is at the time directly or indirectly owned by Borrower or by one or more other Subsidiaries.

"<u>Taxes</u>" means any and all present or future taxes, duties, levies, imposts, deductions, charges or withholdings, and any and all liabilities with respect to the foregoing, but excluding Excluded Taxes and Other Taxes.

"<u>Title Commitment</u>" means that certain ALTA Plain Language Commitment Number NCS-753020-HHLV, dated as of September 16, 2015, and prepared by Title Company.

"Title Company" means Chicago Title Insurance Company.

"<u>USA Patriot Act</u>" means the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Public Law 107-56, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

"<u>USCIS</u>" means the United States Citizenship and Immigration Services of the United States Department of Homeland Security.

ARTICLE I

LOAN

Section 1.1 Principal. Subject to the terms and conditions of this Agreement, Lender agrees to lend to Borrower and Borrower agrees to borrow from Lender, the proceeds of the Loan, from time to time in accordance with the terms hereof until the Maturity Date, for the purpose of refinancing, developing and constructing the Project; provided, however, Lender shall not be obligated to make any Advance if, after giving effect to such Advance, the sum of Lender's aggregate Advances then outstanding would exceed the Commitment. Each Advance shall bear interest at the Loan Rate, computed on each Advance from the date it is made by Lender as more fully described in Section 1.2 below. In no event shall Lender be obligated hereunder to lend to Borrower more than Borrower has qualified to receive under the terms of Article III hereof.

All Advances made by Lender shall be evidenced by the Note. The entire principal balance of the Note shall mature and be payable at the Maturity Date.

Lender shall enter in its records the amount of each Advance, the rate of interest borne on such Advances and the payments of the principal balance and interest received by Lender, and such records shall be conclusive evidence of the subject matter thereof, absent manifest error.

Section 1.2 <u>Interest</u>. Borrower shall pay to Lender interest on the Note computed at the Loan Rate.

In the event that the interest and/or charges in the nature of interest, if any, provided for by this Agreement or by any other Loan Document, shall contravene a legal or statutory limitation applicable to the Loan, if any, Borrower shall pay only such amounts as would legally be permitted; provided, however, that if the defense of usury and all similar defenses are unavailable to Borrower, Borrower shall pay all amounts provided for herein. If, for any reason, amounts in excess of the amounts permitted in the foregoing sentence shall have been paid, received, collected or applied hereunder, whether by reason of acceleration or otherwise, then, and in that event, any such excess amounts shall be applied to principal, unless principal has been fully paid, in which event such excess amount shall be refunded to Borrower.

Interest at the Loan Rate shall accrue on each and every Advance from and after the date it is made by Lender to Borrower. Notwithstanding the foregoing, and for the avoidance of doubt, it is expressly acknowledged and agreed between Lender and Borrower that Borrower shall pay interest only on such amounts as actually have been disbursed directly to Borrower and that Borrower shall not pay interest on any amounts retained by Lender, including, but not limited to, the Holdback and the Interest Reserve. Principal and interest shall be paid by Borrower in accordance with the Note. Interest shall be paid monthly. Interest shall be computed at the Loan Rate and shall be computed on the basis of a 365 day year, but shall be charged for the actual number of days principal is unpaid. If all unpaid Advances made by Lender have not been repaid on or before the Initial Maturity Date, or the First Option Maturity Date, as applicable, or if an Event of Default occurs pursuant to this Agreement (and is not cured in compliance with the terms of this Agreement) or any other Loan Document or if all amounts due under the Loan Documents otherwise become due and payable in accordance with the terms and conditions of the applicable Loan Documents, then the entire unpaid balance of all Advances made by Lender and all other Obligations shall (without notice to or demand upon Borrower) at the sole option of Lender become due and payable on said date, together with all unpaid, accrued interest thereon, and with interest computed at the Default Rate during the continuance of the Event of Default or if such Event of Default is not cured or waived, then from and after that date until all Advances are paid in full. In such event, interest at the Default Rate shall be payable on the fifth (5th) day of each calendar month.

In the event that Borrower fails to make any required payment of principal or interest on the Note (other than the balloon payment at the Maturity Date) on or before the fifth (5th) Business Day following the due date thereof, Borrower shall pay to Lender, in addition to interest at the Loan Rate, a late payment charge equal to three percent (3%) of the amount of the overdue payment (each, a "Late Charge"), for the purpose of reimbursing Lender for a portion of the expense incident to handling the overdue payment. The Late Charge shall apply individually to all payments past due and there will be no daily prorated adjustment. This provision shall not be deemed to excuse a late payment or be deemed a waiver of any other rights

Lender may have including the right to declare the entire unpaid principal and/or interest immediately due and payable. Borrower agrees that the Late Charge is a provision for liquidated damages and represents a fair and reasonable estimate of the damages Lender will incur by reason of the late payment considering all circumstances known to Borrower and Lender on the date hereof. Borrower further agrees that proof of actual damages will be difficult or impossible.

Section 1.3 Prepayment. Subject to the following sentence, Borrower may prepay the Loan, in whole or in part, without any prepayment penalty or premium, at any time during either the Initial Term or the Extension Term. Notwithstanding the foregoing, Borrower shall not repay any portion of the Loan corresponding to that portion of an Advance made by Lender to Borrower with the funds received from a Class B member of Lender until such time as said Class B member of Lender shall have received final adjudication of his or her I-829 petition removing conditions for permanent residency in the United States.

Section 1.4 Payments. All payments and prepayments of principal of, and interest on, the Note and all Fees, expenses and other Obligations under the Loan Documents payable to Lender shall be made, without deduction, set off, or counterclaim, in immediately available funds not later than 2:00 p.m., Pacific time on the dates due, to Lender at the office specified by it from time to time, for the benefit of Lender, except as otherwise specifically provided in this Agreement. Funds received on any day after 2:00 p.m., Pacific Time shall be deemed to have been received on the next Business Day. Whenever any payment to be made hereunder or on the Note shall be stated to be due on a day which is not a Business Day, such payment shall be made on the next succeeding Business Day and such extension of time shall be included in the computation of any interest or fees. Borrower hereby authorizes Lender, at the discretion of Lender, to make an Advance in order to pay, on behalf of Borrower, any amount due on the Note or pursuant to any of the other Loan Documents without further action on the part of Borrower and regardless of whether Borrower is able to comply with the terms, conditions and covenants of this Agreement at the time of such Advance.

(i) So long as no Event of Default has occurred and is continuing, all payments received by Lender (including the proceeds of Advances for such payments) for application to the principal, interest, fees, costs and expenses due to Lender shall be applied in the following order:

First, to any costs and expenses due hereunder, and any Fees due to Lender;

Second, to any unpaid interest then due under the Loan Documents;

Third, to all Obligations; and

Fourth, to the unpaid principal balance of the Note.

(ii) After an Event of Default has occurred and is continuing, all amounts received by Lender shall be applied in the following order:

First, to any costs and expenses due hereunder, and any Fees due to Lender;

Second, to costs and expenses of preserving the Collateral, preserving Lender's security interests therein and enforcement of the rights of Lender under the Loan Documents;

Third, to any unpaid interest then due under the Loan Documents

Fourth to all Obligations; and

Fifth, to the unpaid principal balance of the Note.

All amounts received by Lender (whether the result of payment transmitted by Borrower or otherwise) on account of payment of interest on or principal of the Note, or other payments due under this Agreement or any other Loan Documents, as the case may be, shall be so applied by it pursuant to this Section.

- Section 1.5 Fees. In addition to the interest and other consideration to Lender herein, Borrower agrees to pay to Lender the Fees, as and when due in accordance with the terms of the Loan Documents. No termination or reduction of the Commitment and no failure of Borrower to satisfy the conditions set forth in Article II shall entitle Borrower to a refund of any portion of such Fees.
- Section 1.6 <u>Extension of Maturity Date</u>. At the option of Borrower, the Initial Maturity Date may be extended until the First Option Maturity Date (the "<u>Extension Term</u>"), if all of the following conditions are satisfied, in the sole discretion of the Lender:
 - (a) Borrower gives written notice of its request for an extension to Lender by no earlier than one hundred twenty (120) days and by no later than ninety (90) days prior to the Initial Maturity Date;
 - (b) The Project has been issued a certificate of occupancy by the competent Governmental Authority.
 - (c) Payment by Borrower of the Extension Fee;
 - (d) As of the date of request and on the Initial Maturity Date, there exists no Default or Event of Default;
 - (e) As of the date of request and on the Initial Maturity Date, no material adverse change in the financial condition of the Borrower has occurred;
 - (f) The delivery from Borrower to Lender of all financial information relating to Borrower in accordance with Section 5.10(a) hereof;
 - (g) There shall have been no material adverse change to the physical condition of the Project, and the Project shall be free and clear of all Liens (other than the lien of the Deed of Trust, the Permitted Encumbrances and the lien of ad valorem taxes that are not delinquent) unless approved in writing by Lender;

- (h) Borrower delivers to Lender an Architect's certificate of completion of the construction works of the Project and a copy of the plans and drawings for said construction works, if not previously provided to Lender.
- (i) Borrower delivers to Lender an endorsement to or reissuance of the existing Title Policy, bringing current the effective date of such coverage, stating that the coverage afforded by the Title Policy shall not be affected because of such extension and showing that there have been no Liens against the Project from and after the date hereof, unless consented to in writing by Lender;

In the event that, for any reason, Borrower fails to satisfy all of the foregoing conditions, the Loan shall mature and be due and payable in full on the Initial Maturity Date, as determined by Lender in its sole discretion.

Section 1.7 EB-5 Program Requirements.

- (a) Lender hereby represents and warrants that, as of the date of this Agreement, pursuant to that certain letter from the United States Citizenship and Immigration Services ("<u>USCIS</u>") dated July 27, 2015, EB5 Impact Capital Regional Center, LLC, a Nevada limited liability company ("<u>Regional Center</u>") is authorized to act as an approved and federally-designated "regional center" under the Immigrant Investor Pilot Program created by Section 610 of Public Law 102-395 (Oct. 6, 1992), which has been extended through September 30, 2016. The Regional Center may provide construction financing in Nye County, Nevada. The Regional Center will sponsor the Project for EB-5 Program purposes.
- The Loan will be comprised of investments made into Lender anticipated to be in the approximate amount of Five Hundred Thousand Dollars (\$500,000) (which amount may be increased pursuant to newly enacted EB-5 visa legislation) per immigrant investor (collectively, the "EB-5 Investors") who seeks to obtain permanent residence in the United States under the EB-5 Immigrant Investor Program created by Section 203(b)(5) of the Immigration and Nationality Act (INA) ("EB-5 Program"). Such funds shall initially be held by Escrow Agent in the Escrow Account which will be administered and maintained by the Escrow Administrator). Upon satisfaction of each EB-5 Investor's Subscription Conditions, 75% (\$375,000) of his or her subscription will be released to the Lender by the Escrow Administrator, pursuant to the Escrow Agreement, and made available for an Advance to Borrower upon Borrower's request. The remaining 25% (\$125,000) of each EB-5 Investor's subscription ("Holdback") will be held in the Escrow Account for the Lender's benefit until the corresponding EB-5 Investor's I-526 Immigrant Petition is either approved or finally adjudicated and denied The Holdback may be released to the Lender by the Escrow Administrator, upon the Lender's written direction, after the corresponding EB-5 Investor's I-526 Immigrant Petition is approved by the USCIS ("Release Condition"). At such time, the Holdback may be released and made available for an Advance to Borrower. If the Release Condition is not satisfied, the Holdback will be released to the Lender and may be made available for refund to the corresponding EB-5 Investor. As

more fully set forth in Section 1.2 above, Lender shall not charge Borrower interest on the amount of the Holdback.

- (c) Lender and Regional Center shall be responsible for monitoring at Lender's sole cost and expense the required submissions made by the EB-5 Investors and/or their legal counsel for purposes of seeking to obtain the necessary approvals from the USCIS for the EB-5 Investors' conditional residency in the United States in accordance with the requirements of the EB-5 Program.
- (d) Borrower acknowledges that, for compliance with the rules and regulations promulgated under the EB-5 Program, Lender is required to deploy its funds with third parties located within the approved geographic area of the Regional Center. Based upon the materials provided by Borrower, Lender has determined to allocate the Loan to Borrower as a qualified third party in compliance with the rules and regulations promulgated under the EB-5 Program. Lender acknowledges that the Project has been approved by USCIS as an "Exemplar Form I-526 Petition Project."
- (e) Borrower shall use the proceeds of the Loan solely for the purpose of funding directly, or advancing to Affiliates to pay, the costs of the Project, in accordance with the terms and conditions of this Agreement, as set forth in the Budget and the Project documents submitted to, and approved by, USCIS.
- (f) Borrower shall submit to Lender the EB-5 Information. Failure of Borrower to use the proceeds of the Loan in accordance with the terms and conditions of this Agreement or to provide the EB-5 Information shall be a default pursuant to Section 6.1.

ARTICLE II

CONDITIONS OF BORROWING

Lender shall not be required to make any Advance hereunder until the pre-closing requirements, conditions and other requirements set forth below have been completed and fulfilled to the satisfaction of Lender, at Borrower's sole cost and expense.

- Section 2.1 <u>Pre-Closing Requirements</u>. On or prior to the Closing Date (except as otherwise provided in this Section), Borrower shall provide to Lender, except as otherwise instructed, each of the following, in form and substance acceptable to Lender:
 - (a) A copy of the Title Commitment.
 - (b) A schedule listing all primary contracts relating to the Project having a contract sum in excess of \$250,000.
 - (c) One (1) copy of the recent, certified ALTA/ACSM Survey of the Land, which has already been delivered to, and approved by, Lender. Soil reports on the Land, as already delivered to, and approved by, Lender.

- (d) The Environmental Impact Study already delivered to, and approved by, Lender. Certificates of insurance, together with paid receipts, indicating that all insurance currently required under the terms of <u>Section 5.6 hereof</u>.
- (e) That certain Development Agreement adopted pursuant to Nye County Ordinance No. 378, recorded on August 3, 2009, as Document Number 731349 in the Official Records of the Nye County, Nevada, evidencing the applicable zoning with respect to the Property.
- (f) A copy of Borrower's Organizational Documents, certified as true, correct and complete by a manager of Borrower authorized to do so, together with (i) a current certificate of existence/good standing from the jurisdiction in which Borrower was organized (and from the jurisdiction in which the Land is located, if different from the jurisdiction in which Borrower was organized), and (ii) resolutions and/or consents of those parties necessary to authorize the transaction contemplated hereby.
- (g) A flood-zone certification indicating that the Project is not located in a flood plain or any other flood-prone area as designated by any governmental agency; provided, however, that if the Project is so located, Borrower shall provide proof of flood insurance to Lender as required by Lender.
- (h) A proposed Operating Budget for the Project for its first Fiscal Year of operation in the form previously submitted to the Nevada Real Estate Division in connection with the registration of the Front Sight Resort and Vacation Club.
- (i) Letters from the suppliers confirming the availability of water, storm and sanitary sewer, gas, electric and telephone utilities for the Project.
- (j) A copy of each non-cancellable agreement relating to the management, operation or maintenance of the Property and of each such agreement which cannot be cancelled upon notice of thirty (30) days or less.
- Section 2.2 <u>Loan Documents</u>. On or before the Closing Date, Borrower shall execute and deliver (or cause to be executed and delivered) to Lender, the Loan Documents, in form and substance reasonably acceptable to Lender and to its counsel, in their sole but reasonable discretion, to evidence and secure the Loan. Lender may designate which of the Loan Documents are to be placed of record, the order of recording thereof, and the offices in which the same are to be filed and/or recorded. Borrower shall pay all filing, documentary, intangible, recording and/or registration taxes and/or fees due upon the Note, if any, the Deed of Trust, any financing statements and/or the other Loan Documents
- Section 2.3 <u>Title Insurance</u>. On or prior to the Closing Date, Lender shall have received the Title Policy, proforma or a marked-up commitment to issue the Title Policy, signed by an officer of the Title Company, in form and substance satisfactory to Lender and including all endorsements as required by Lender. Title Company will provide priority insurance over all possible mechanics' lien claims, despite the fact that construction of the Improvements may have commenced prior to the recording of the Deed of Trust.

Section 2.4 Opinion of Borrower's Attorneys. Lender shall have received from outside counsel for Borrower one or more current written opinions, in form and substance acceptable to Lender, addressed to Lender, covering matters such as due formation, authorization, execution and delivery of the Loan Documents.

ARTICLE III

ADVANCES OF LOAN PROCEEDS

Section 3.1 General. Provided no Default or Event of Default has occurred and is continuing, the Loan proceeds shall be advanced by the Lender for the benefit of Borrower in accordance with the terms and conditions set forth in this Article III. All monies advanced by the Lender with respect to the Project shall constitute a loan made to Borrower under this Agreement, evidenced by the Note and secured by the other Loan Documents, and interest shall be computed thereon, as prescribed by this Agreement and the Note, from the date the Loan account is charged with the amount of the Advance.

Lender will advise Borrower within five (5) business days every time Lender has received a new EB-5 Investor's funds into the Escrow Account. If so requested by Borrower, Lender will make an Advance of as little as \$375,000, which represents the available funds from each new EB-5 Investor pursuant to the terms of this Agreement.

Lender reserves the right to make Advances of amounts on the Note which are allocated to any of the designated items in the Budget for soft or hard costs related to the development of the Project, and construction of the Improvements or otherwise with respect to the Project or for such other purposes or in such different proportions as Lender may, in its reasonable discretion, deem necessary or advisable; <u>provided</u>, <u>however</u>, after the occurrence of an Event of Default, Lender may act in its sole discretion.

No Advance shall constitute a waiver of any condition precedent to the obligation of Lender to make any further Advance, or preclude Lender from thereafter declaring the failure of Borrower to satisfy any such condition precedent to be an Event of Default. All conditions precedent to the obligation of Lender to make any Advance are imposed hereby solely for the benefit of Lender, and no other party may require satisfaction of any such condition precedent or shall be entitled to assume that Lender will make or refuse to make any Advance in the absence of strict compliance with such condition precedent. Provided no Default or Event of Default has occurred, Lender may waive any requirement of this Agreement for any Advance which Lender, in its reasonable discretion, determines is not material.

Lender will Advance to itself, when due, from the Interest Reserve, without further order or request from Borrower, all interest payable to Lender under the terms hereof or of the Note (so long as the conditions to such advances have been satisfied, or waived by Lender, and sufficient funds remain in the Interest Reserve).

Section 3.2 Draw Requests.

- (a) From and after the date of the first Advance of the Loan, Borrower shall deliver to Lender on a monthly basis evidence of the Project costs funded during the preceding month (whether from Loan proceeds or otherwise), and, to the extent that Borrower is seeking an additional Advance, the Draw Request approved by General Contractor, together with an itemized summary and copies of all invoices included in such disbursement.
- (b) Prior to the date of the first Advance of the Loan, Borrower shall, in addition to satisfying all other conditions for an Advance in this Section, provide to Lender the insurance required per Section 5.6 of this Agreement, Prior to the Commencement Date, Borrower shall, in addition to satisfying all other conditions for an Advance in this Section, provide to Lender (i) Plans, in the form previously submitted to Lender, as finally approved for construction by the Project Architect and the applicable Governmental Authority, (ii) a schedule listing all Contractors, and primary contracts relating to the Project having a contract sum in excess of \$250,000 for any such Contractor, and all other major subcontracts, and such engineering, architectural, and construction contracts, subcontracts and schedules relating to the Project. If requested by Lender, Borrower shall also furnish to Lender copies of such contracts with each Contractor; and (iii) a list of all agreements, licenses and permits relating to the construction, development and operation of the Project.

Section 3.3 <u>Inspections</u>. Lender and its representatives shall have access to the Project at all reasonable times and shall have the right to enter the Project and to conduct such inspections thereof as they shall deem necessary or desirable for the protection of Lender's interests; <u>provided</u>, <u>however</u>, that for so long as no Event of Default shall have occurred and be continuing, Lender shall provide to Borrower prior notice of not less than seventy-two (72) hours of any such inspections and such inspections shall be subject to the rights of Club Members (i.e., owners of Timeshare Interests) and any tenants under any applicable Leases.

Borrower shall be responsible for making its own inspections of the Project during the course of such construction work and shall determine to its own satisfaction that the work done and materials supplied are in accordance with applicable contracts with its contractors. By advancing funds after any inspection of the Project by Lender, Lender shall not be deemed to waive any Event of Default, waive any right to require construction defects to be corrected, or acknowledge that all construction conforms to the Plans.

Section 3.4 Lender's Responsibilities. It is expressly understood and agreed that Lender does not assume any liability or responsibility for the sufficiency of the Loan proceeds to complete the Project, for protection of the Project, for the adequacy of the Plans, the compliance of the Project and/or Plans with Governmental Requirements, for the satisfactory completion of the Project, for inspection during construction or to notify Borrower, General Contractor or any other party of any construction defects, for the adequacy of the Interest Reserve, for the adequacy or accuracy of the Budget, for any representations made by Borrower, or for any acts on the part of Borrower or its contractors to be performed in connection with the construction of the Project.

Section 3.5 Procedures for Advances.

- (a) Request for Advances. Any request by Borrower for an Advance shall be made at least five (5) Business Days prior to the date of the requested Advance.
- (b) <u>Direct Advances</u>. At Lender's option after the occurrence of an Event of Default, Lender may (i) make any Advances directly to any person, including the General Contractor, to whom Lender determines that payment is due and (ii) make advances to any person to whom Lender determines that payment should be made in order to cure or to prevent the occurrence of any Default. Any of the aforesaid Advances shall be deemed advanced under the Note as of the date on which funds are transferred by Lender.
- (c) <u>Title endorsement</u>. Prior to each additional advance, Lender shall have received a commitment from the Title Company to issue an endorsement showing the increased loan amount in the same priority as the previously advanced amounts, provided, however, that upon the closing of the Senior Debt, the priority of the lien of the Deed of Trust reflected in any subsequent endorsements shall be subordinated to the lien of the deed of trust or similar instrument securing the Senior Debt.
- (d) <u>Joint Checks</u>. All Advances of Loan proceeds made to a party other than Borrower hereunder shall be made pursuant to a joint check executed by Lender and Borrower, with proper endorsements included.

Section 3.6 Stop Notices.

- (a) General. Borrower shall not cause any "stop notice" or similar notice to be filed or served on Lender with respect to the Project. Borrower shall defend, indemnify and hold Lender and its officers, directors, agents and employees harmless from and against all claims, damages, loss, liability, costs and expenses (including reasonable attorneys' fees) arising from or relating to any stop notice, the compliance therewith and the defense thereof. Lender may require Borrower to provide a release bond for any stop notice, which bond shall be subject to Lender's review and approval and/or may take such action with respect to any stop notice as Lender may deem appropriate in Lender's reasonable discretion and Lender may withhold such amounts from disbursement in connection with the Loan as Lender may elect in Lender's reasonable discretion, to the extent Lender is obligated to withhold funds pursuant to applicable law or any demands made in connection with any stop notice.
- (b) <u>Notices</u>. Upon an Event of Default and so long as it is continuing, Borrower irrevocably appoints Lender as its attorney-in-fact, coupled with an interest and with full power of substitution, to file for record, at the Borrower's cost and expense and in Borrower's name, any notices of completion, notices of cessation of labor, or any other notices that Lender considers necessary or desirable to protect its security.
- Section 3.7 <u>Use of Loan Proceeds</u>. Borrower shall use and apply the Loan proceeds solely to all or any number of the individual Project components in accordance with the Budget and also to pay some or all of any or all existing indebtedness encumbering the Project pursuant

to a Permitted Encumbrance. Borrower shall use its best business judgment based upon thencurrent real estate market and availability of other financing resources to allocate the proceeds of the Loan in such a manner as to assure the full expenditure of the Loan proceeds advanced to Borrower. Borrower will comply with the requirements of the EB-5 Program and the other EB-5 Program covenants and requirements contained in this Agreement.

ARTICLE IV

REPRESENTATIONS AND WARRANTIES

(A) Borrower represents, warrants and covenants to Lender that:

Section 4.1 <u>Borrower's Formation and Powers</u>. Borrower is a limited liability company duly organized, validly existing and in good standing under the laws of the State of Nevada, and qualified and authorized to do business in all jurisdictions in which the conduct of its business and affairs requires it to be so qualified. Borrower has all power, authority, permits, consents, authorizations and licenses necessary to carry on its business, to construct, equip, own and operate the Project and to execute, deliver and perform its obligations under this Agreement and the other Loan Documents; all consents necessary to authorize the execution, delivery and performance of this Agreement and the other Loan Documents have been duly adopted and are in full force and effect; and this Agreement and the other Loan Documents have been duly executed and delivered by Borrower.

The execution, delivery and performance by Borrower of this Agreement and other Loan Documents to which Borrower is a party have been duly authorized by all necessary action and do not and will not (i) violate any provision of any laws, rule, regulation (including, without limitation, Regulation U of the Board of Governors of the Federal Reserve System), order, writ, judgment, injunction, decree, determination or award presently in effect having applicability to Borrower or of Borrower's Organizational Documents, (ii) result in a breach of or constitute a default under any indenture or loan or credit agreement or any other agreement, lease or instrument to which Borrower is a party or by which it or its properties may be bound or affected, or (iii) result in or require the creation or imposition of any Security Interest in any of its properties pursuant to the provisions of any agreement or other document binding upon or applicable to Borrower or any of its properties, except pursuant to the Loan Documents.

- Section 4.2 <u>Authority</u>. The execution, delivery and performance by Borrower of this Agreement and other Loan Documents to which Borrower is a party have been duly authorized by all necessary limited liability company action.
- Section 4.3 No Approvals. As of the date of the first Advance of the Loan, no authorization, consent, approval, license, exemption of or filing or registration with any court or governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, is or will be necessary to the valid execution, delivery or performance by Borrower of this Agreement, the Note, or any other Loan Documents to which Borrower is a party.
- Section 4.4 <u>Legal and Valid Obligations</u>. This Agreement, the Note, and the other Loan Documents to which Borrower are a party constitute the legal, valid and binding

obligations of Borrower, enforceable against Borrower in accordance with their respective terms, subject to bankruptcy and insolvency laws and other laws generally affecting the enforceability of creditor's rights generally and subject to limitations on the availability of equitable remedies.

Litigation. There are no actions, suits or proceedings (whether or not purportedly on behalf of Borrower) pending or, to the knowledge of Borrower, threatened against Borrower or affecting any of the Project or Borrower's other assets (if any), at law or in equity or before any federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign, which contests the validity or enforceability of this Agreement or any of the other Loan Documents or the transactions contemplated hereby or as a result of which Borrower may become subject to any judgment or liability which if determined adversely to Borrower, would constitute a Material Adverse Occurrence as to Borrower. There are no actions, suits or proceedings pending or, to the knowledge of Borrower, threatened against Borrower or its other assets (if any), at law or in equity or before any federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign. As of the date of the first Advance of the Loan, Borrower shall not be in default with respect to any final judgment, writ, injunction, decree, rule or regulations of any court, arbitrator or federal, state, municipal or other governmental department, commission, board, bureau, agency or instrumentality, domestic or foreign.

Section 4.6 <u>Title</u>. Borrower has good, marketable and insurable fee simple title to the Land, and good title to the rest of the Project, subject to no lien, charge, mortgage, deed of trust, restriction or encumbrance, except Permitted Encumbrances. There are no mechanics', materialman's or other similar Liens which have been filed for work, labor or materials affecting the Project which are or may be Liens prior to, or equal, or subordinate to the Liens created by the Loan Documents. None of the Permitted Encumbrances, individually or in the aggregate, (a) impair the use or intended operations of the Project, or (b) impair Borrower's ability to pay its Obligations in a timely manner.

Section 4.7 <u>Defects and Hazards</u>. Borrower does not know of any defects, facts or conditions affecting the Land that would make it unsuitable for the use contemplated hereunder or of any abnormal hazards (including soils and groundwater contamination, earth movement or slippage) affecting the Land that have not been previously disclosed in writing to Lender by Borrower.

Section 4.8 Payment of Taxes. There have been filed all federal, state and local tax returns with respect to Borrower and its direct and indirect business operations which are required to be filed. Borrower has paid or caused to be paid to the respective taxing authorities all taxes as shown on such returns or on any assessments received by it to the extent that such taxes have become due. Borrower knows of no proposed material tax assessment against Borrower, and Borrower is not obligated by any other agreement, tax treaty, instrument or otherwise to contribute to the payment of taxes owed by any other person or entity. All material tax liabilities are adequately provided for or reserved against on the books of Borrower, as appropriate.

Section 4.9 Agreements.

The Borrower's Organizational Documents shall be in full force and effect as of the date of the first Advance of the Loan and, as of said date of the first Advance of the Loan, shall be free from any default on the part of Borrower. Each of (a) the Architect's Agreement, and (b) the General Contract, shall be in full force and effect before the Commencement Date and, as of said Commencement Date, shall be free from any default on the part of Borrower. Borrower, is not in default in the performance, observance or fulfillment of any of the obligations, covenants or conditions contained in any agreement or instrument to which Borrower, is a party the effect of which default would constitute a Material Adverse Occurrence as to Borrower. The General Contract shall require the General Contractor to construct the Project in accordance with the Plans in the form which have been delivered to Lender, without any change thereto not disclosed to Lender. The General Contract shall constitute the entire agreement between Borrower and the General Contractor with respect to the construction of the Project. As of the Commencement Date, Borrower shall have performed all of its obligations required by the General Contract and/or Architect's Agreement, and shall have paid all sums under the General Contract and/or Architect's Agreement which are required to be performed or paid as of the date hereof.

Section 4.10 No Defaults under Loan Documents or Other Agreements. No Default or Event of Default has occurred under any of the Loan Documents. As of the date of the first Advance of the Loan, Borrower shall not be in default in the payment of the principal or interest on any of its Indebtedness for borrowed money; and no event has occurred, or will occur, which, with the lapse of time or the giving of notice or both, would constitute an Event of Default under the Loan Documents. Borrower is not obligated for the payment of any commission or other fee with respect to the purchase of the Land and Improvements or, if Borrower is so obligated, such commission or other fee has been paid in full.

Section 4.11 Boundary Lines; Conformance with Governmental Requirements and Restrictions. The exterior lines of the Improvements are, and at all times will be, within the boundary lines of the Land. Borrower has examined and is familiar with all applicable covenants, conditions, restrictions and reservations, and with all applicable Governmental Requirements, including, but not limited to, building codes and zoning, environmental, hazardous substance, energy and pollution control laws, ordinances and regulations affecting the Project. Borrower has obtained all permits which are necessary for the construction of the Project in accordance with the Plans and in accordance with all applicable building, environmental, subdivision, land use and zoning laws, including all permits for the Improvements, annexation agreements, plot plan approvals, subdivision approvals (including the approval and recordation of any required subdivision map), environmental approvals (including a negative declaration or an environmental impact report if required under applicable law), sewer and water permits and zoning and land use entitlements. Borrower has obtained all approvals of the parties required in connection with the construction of the Project pursuant to any license, easement or restriction affecting the Land. The Project will in all respects conform to and comply with said covenants, conditions, restrictions, reservations and Governmental Requirements.

Section 4.12 Project Costs. On a line by line and total basis, the Project costs shown

on the Estimated Construction Cost Statement are, to the best of Borrower's knowledge, true, correct and complete in all material respects, and represent the total of all costs which Borrower expects to pay to complete the Project.

- Section 4.13 <u>Utilities</u>. Telephone services, gas, electric power, storm sewers, sanitary sewer and water facilities are available to the boundaries of the Land, adequate to serve the Project and not subject to any conditions that would prevent the use of the Project for its intended purposes. All streets and easements necessary for construction and operation of the Project are available to the boundaries of the Land.
- Section 4.14 Personal Property. Borrower is now and shall continue to be the sole owner of the Equipment free from any lien, security interest or adverse claim of any kind whatsoever, except for the Permitted Encumbrances and any liens or security interests in favor of Lender.
- Section 4.15 <u>Condemnation</u>. No condemnation proceeding or moratorium is pending or, to the best of Borrower's knowledge, threatened against the Land which would impair the construction, use, sale or occupancy of the Land or its Improvements.
- Section 4.16 <u>Separate Lots</u>. The Land is comprised of two (2) parcels which constitute separate tax lots and do not constitute a portion of any other tax lot not a part of such Land.
- Section 4.17 Federal Reserve Regulations. No portion of the Loan hereunder will be used to purchase or carry any "margin stock" as defined in Regulation U of the Board of Governors of the Federal Reserve System of the United States or for the purpose of reducing or retiring any indebtedness which was originally incurred to purchase or carry any margin security or for any other purpose which might constitute this transaction as a "purpose credit" within the meaning of said Regulation U. No portion of the Loan hereunder will be used for any purpose that violates, or which is inconsistent with, the provisions of Regulation X of the Board of Governors of the Federal Reserve System or any other regulation of said Board of Governors.
- Section 4.18 Investment Company Act. Borrower is not an "investment company," or an "affiliated person" of, or a "promoter" or "principal underwriter" for, an "investment company," as such terms are defined in the Investment Company Act of 1940, as amended. The making of the Loan, the application of the proceeds and repayment thereof by Borrower and the performance of the transactions contemplated by this Agreement will not violate any provision of said Act, or any rule, regulation or order issued by the Securities and Exchange Commission thereunder. Furthermore, Borrower is not subject to regulation under the Interstate Commerce Act, or any federal or state statue or regulation limiting its ability to incur indebtedness for money borrowed.
- Section 4.19 <u>Unregistered Securities</u>. Borrower has not: (a) issued any unregistered securities in violation of the registration requirements of Section 5 of the Securities Act of 1933, as amended, or any other law; or (b) violated any rule, regulation or requirement under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, in

either case where the effect of such violation would constitute a Material Adverse Occurrence as to Borrower.

Section 4.20 Accuracy of Information. All factual information heretofore or herewith prepared by Borrower or a Related Party of Borrower and furnished by or on behalf of Borrower to Lender for purposes of or in connection with this Agreement or any transaction contemplated hereby is true and accurate in every material respect on the date as of which such information is dated or certified and no such information contains any material misstatement of fact or omits to state a material fact or any fact necessary to make the statements contained therein not misleading as of such date. To the best of Borrower's knowledge, all factual information heretofore or herewith prepared by a Person other than Borrower or a Related Party of Borrower and furnished by or on behalf of Borrower to Lender for purposes of or in connection with this Agreement or any transaction contemplated hereby is true and accurate in every material respect on the date as of which such information is dated or certified and no such information contains any material misstatement of fact or omits to state a material fact or any fact necessary to make the statements contained therein not misleading as of such date.

Section 4.21 ERISA Compliance. Borrower has not adopted a Plan. As of the date hereof and throughout the term of the Loan (i) Borrower is not and will not be an "employee benefit plan," as defined in Section 3(3) of ERISA, subject to Title I of ERISA, (ii) none of the assets of Borrower constitutes or will constitute, by virtue of the application of 29 C.F.R. Section 2510.3-101(f) as modified by Section 3(42) of ERISA, "plan assets" of one or more such plans within the meaning of 29 C.F.R. Section 2510.3-101, (iii) Borrower is not and will not be a "governmental plan" within the meaning of Section 3(32) ERISA, and (iv) transactions by or with Borrower are not and will not be subject to state statutes regulating investment of, and fiduciary obligations with respect to, governmental plans.

Section 4.22 Compliance. Borrower:

- (a) is in compliance and conformity with all Governmental Requirements the violation of which, individually or in the aggregate, would constitute a Material Adverse Occurrence as to Borrower; and
- (b) has not received and does not anticipate the receipt of any order or notice of any violation or claim of violation of any Governmental Requirement which would constitute a Material Adverse Occurrence as to Borrower.
- Section 4.23 Consents. To the extent that any franchises, licenses, permits, certificates, authorizations, approvals or consents from any federal, state or local (domestic or foreign) government, commission, bureau or agency are material to the present conduct of the business and operations of Borrower or are required for the acquisition, ownership, operation or maintenance by Borrower of the Project, such franchises, licenses, permits, certificates, authorizations, approvals and consents have been, or will be upon Completion, validly granted, in full force and effect, and be valid and sufficient authorization therefor.
- Section 4.24 Environmental Laws. Except as specifically disclosed in the Environmental Impact Study previously delivered to, and approved by, Lender, Borrower: (a)

has not received any notice or otherwise learned of any Environmental Liability relating to the Project which would individually or in the aggregate constitute a Material Adverse Occurrence as to Borrower arising in connection with (i) any non-compliance or alleged non-compliance with or violation of the requirements of any Environmental Law, or (ii) the Release or threatened Release of any Hazardous Substance, or other substance into the environment; (b) has no knowledge of any threatened or actual liability in connection with the Release or threatened Release of any Hazardous Substance, or other substance into the environment relating to the Project which would individually or in the aggregate constitute a Material Adverse Occurrence; or (c) has not received any notice or otherwise learned of any federal or state investigation evaluating whether any remedial action is needed to respond to a Release or threatened Release of any Hazardous Substances into the environment where such liability individually or in the aggregate for all such liabilities would constitute a Material Adverse Occurrence as to Borrower. Borrower has not received any notice of any violation or alleged non-compliance of any Environmental Laws relating to the Project where such violation would constitute a Material Adverse Occurrence as to Borrower.

Section 4.25 Anti-Terrorism Regulations.

- (a) General. None of Borrower, or, to the best of Borrower's knowledge, any Affiliate thereof is in violation of any Anti-Terrorism Law or engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law
- (b) Executive Order No. 13224. None of the Borrower, or, to the best of Borrower's knowledge, any Affiliate thereof or their respective agents acting or benefiting in any capacity in connection with the Loan or other transactions hereunder, is any of the following (each a "Blocked Person"):
 - (i) a Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224;
 - (ii) a Person owned or controlled by, or acting for or on behalf of, any Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224;
 - (iii) a Person or entity with which Lender is prohibited from dealing or otherwise engaging in any transaction by any Anti-Terrorism Law;
 - (iv) a Person or entity that commits, threatens or conspires to commit or supports "terrorism" as defined in Executive Order No. 13224;
 - (v) a Person or entity that is named as a "specially designated national" on the most current list published by the U.S. Treasury Department Office of Foreign Asset Control at its official website or any replacement website or other replacement official publication of such list; or

- (vi) a person or entity who is affiliated or associated with a person or entity listed above.
- (c) None of Borrower, or, to the best of Borrower's knowledge, any Affiliate thereof nor any of their agents acting in any capacity in connection with the Loan, any letters of credit or other transactions hereunder (i) conducts any business or engages in making or receiving any contribution of funds, goods or services to or for the benefit of any Blocked Person, or (ii) deals in, or otherwise engages in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224.
- (d) Neither Borrower nor, to the best of Borrower's knowledge, any Affiliate thereof nor any person owning an interest therein, are a "Special Designated National" or "Blocked Person" as those terms are defined in the office of Foreign Asset Control Regulations (31 C.F.R. § 500 et. seq.).
- Section 4.26 Subsidiaries. Borrower has no Subsidiaries.
- Section 4.27 <u>Leases</u>. Except as provided below in <u>Section 5.24</u>, there is no Lease in effect relating to the Project, or any portion of the Project, except for Leases that have been approved in writing by Lender.
- Section 4.28 Ownership and Control of Borrower. As of the date of this Agreement, the direct owners of Borrower are set forth on Exhibit Gto this Agreement and the indirect owners set forth thereon are accurate and complete.
- Section 4.29 <u>Use of Loan Proceeds</u>. The proceeds of the Loan shall be used to pay and obtain release of the existing liens on the Land, to pay for or reimburse Borrower for soft and hard costs related to the pre-construction, development, promotion, construction, development and operation of the Project in connection with the FSFTI Facility and the construction, development, operation, leasing and sale of the timeshare portion of the Project, all as more particularly described on Exhibit <u>F</u>, attached hereto. The Loan is made exclusively for business purposes in connection with holding, developing and financially managing real estate for profit, and none of the proceeds of the Loan will be used for the personal, family or agricultural purposes of the Borrower.
- Section 4.30 <u>Insurance</u>. Borrower has obtained or caused to be obtained the insurance required pursuant to <u>Section 5.6</u> of this Agreement.
 - (B) Lender represents, warrants and covenants to Borrower that:
 - (i) <u>General</u>. None of Lender, or, to the best of Lender's knowledge, any Affiliate or member thereof is in violation of any Anti-Terrorism Law or engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law.

- (ii) Executive Order No. 13224. None of the Lender, or, to the best of Lender's knowledge, any Affiliate or member thereof or their respective agents acting or benefiting in any capacity in connection with the Loan or other transactions hereunder, is a Blocked Person.
- (iii) None of Lender, or, to the best of Lender's knowledge, any Affiliate or member thereof nor any of their agents acting in any capacity in connection with the Loan, any letters of credit or other transactions hereunder (a) conducts any business or engages in making or receiving any contribution of funds, goods or services to or for the benefit of any Blocked Person, or (b) deals in, or otherwise engages in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224.
- (iv) Neither Lender nor, to the best of Lender's knowledge, any Affiliate or member thereof nor any person owning an interest therein, are a "Special Designated National" or "Blocked Person" as those terms are defined in the office of Foreign Asset Control Regulations (31 C.F.R. § 500 et. seq.).

THE WARRANTIES AND REPRESENTATIONS IN THIS <u>ARTICLE IV</u>, AND ANY ADDITIONAL WARRANTIES AND REPRESENTATIONS CONTAINED HEREIN AND IN THE OTHER LOAN DOCUMENTS, SHALL BE DEEMED TO HAVE BEEN RENEWED AND RESTATED BY BORROWER AT THE TIME OF EACH REQUEST BY BORROWER FOR AN ADVANCE OF LOAN PROCEEDS.

ARTICLE V

(A) COVENANTS OF BORROWER

While this Agreement is in effect, and until Lender has been paid in full the principal of and interest on all Advances made by Lender hereunder and under the other Loan Documents, Borrower agrees to comply with, observe and keep the following covenants and agreements:

Section 5.1 Completing Construction. Borrower shall commence construction of the Improvements no later than the Commencement Date. Borrower shall not become a party to any contract, other than the General Contract and the Architect's Agreement, for the performance of any work on the Project or for the supplying of any labor, materials or services for construction of the Improvements, except upon such terms and with such parties as shall be approved in writing by Lender and subject to the condition that each Contractor party to any approved contract execute and deliver to Lender agreements in form acceptable to Lender that either subordinate or waive the lien rights of such Contractors to the liens of Lender under the Loan Documents including, without limitation, the liens of Lender evidenced by the Deed of Trust; provided, however, Borrower may, without the approval of Lender, enter into one or more contracts with any Contractor whose contracts' aggregate amount with such Contractor is less than \$250,000. Borrower shall provide to Lender a list, within thirty (30) days after the end of each calendar quarter, of all Contractors, including all subcontractors, who have entered into written contracts directly with either Borrower or the General Contractor during the preceding calendar quarter. No approval by Lender of any contract or change order shall make Lender

responsible for the adequacy, form or content of such contract or change order. Borrower shall expeditiously complete and fully pay for the construction of the Project in a good and workmanlike manner and in accordance with the Plans submitted or to be submitted to and approved by Lender, and in compliance with all applicable Governmental Requirements, and any covenants, conditions, restrictions and reservations applicable thereto so that Completion of the construction of the Project occurs on or before the Completion Date. Borrower assumes full responsibility for the compliance of the Plans and the Project with all Governmental Requirements, covenants, conditions, restrictions and reservations, and with sound building and engineering practices, and, notwithstanding any approvals by Lender, Lender shall have no obligation or responsibility whatsoever for the Plans or any other matter incident to the Project or the construction of the Improvements. Borrower shall correct or cause to be corrected (a) any defect in the Improvements, (b) any departure in the construction of the Improvements from the Plans or Governmental Requirements, and (c) any encroachment by any part of the Improvements or any other structure located on the Land on any building line, easement, property line or restricted area that is in violation of any applicable Governmental Requirements, or any private agreements for which Borrower has not received a waiver approved by the Lender in writing. Borrower shall cause all roads necessary for the utilization of the Project for its intended purposes to be completed and dedicated (if dedication thereof is required by any governmental authority), the bearing capacity of the soil on the Land to be made sufficient to support the Improvements, and sufficient local utilities to be made available to the Project and installed at costs (if any) set out in the Budget, on or before the Completion Date.

Section 5.2 Changing Costs, Scope or Timing of Work. Borrower shall deliver to Lender revised, estimated costs of the Project, showing changes in or variations from the original Estimated Construction Cost Statement, as soon as such changes are known to Borrower. Borrower shall deliver to Lender a revised construction schedule, if and when any target date set forth therein has been delayed by twenty (20) consecutive days or more, or when the aggregate of all such delays equals thirty (30) days or more.

Borrower shall not make or consent to any change or modification in such Plans, contracts or subcontracts, and no work shall be performed with respect to any such change or modification, without the prior written consent of Lender, if (i) such change or modification would in any material way alter the design or structure of the Project or change the rentable area thereof in any way, or increase or decrease the Project cost by \$250,000 or more (after taking into account cost savings and any insurance proceeds of Borrower received by Lender) for any single change or modification, or (ii) the aggregate amount of all changes and modifications exceeds \$500,000 (after taking into account cost savings and any insurance proceeds of Borrower received by Lender). Borrower shall promptly furnish Lender with a copy of all changes or modifications in the Plans, contracts or subcontracts for the Project prior to any Advance used to fund such change or modification whether or not Lender's consent to such change or modification is required hereby.

Section 5.3 <u>Using Loan Proceeds</u>. Subject to <u>Section 3.2</u>, Borrower shall use the Loan proceeds in its sole discretion to pay, or to reimburse Borrower for paying, costs and expenses incurred by Borrower in connection with the pre-construction, promotion, construction, development, operating and leasing of the Project on the Land and the equipping of the

Improvements, together with the payoff and release of any existing liens and encumbrances on the Land. Borrower shall take all steps necessary to assure that Loan proceeds are used by its contractors and subcontractors to pay such costs and expenses which could otherwise constitute a mechanic's lien claim against the Project. Within thirty (30) days after the Completion Date, Borrower shall provide the documentation and supporting accounting records and contract documents necessary, in Lender's discretion, to demonstrate that between the Closing Date and the date of delivery of such documentation not less than the total amount of the Advances has been spent directly or indirectly on the Project substantially in a form acceptable to Lender for compliance with the EB-5 Program.

Section 5.4 Keeping of Records. Borrower shall set up and maintain accurate and complete books, accounts and records pertaining to the Project. Borrower will permit representatives of Lender to have reasonable access to and to inspect and copy such books, records and contracts of Borrower and to inspect the Project and to discuss Borrower's affairs, finances and accounts with any of its principal officers, all at such times and as often as may reasonably be requested by Lender. Any such inspection by Lender shall be for the sole benefit and protection of Lender, and Lender shall have no obligation to disclose the results thereof to Borrower or to any third party. When a Default or Event of Default exists, Lender may do any of the foregoing during normal business hours without advance notice or other limitation.

Section 5.5 Providing Evidence of Completion. Upon completion of the Improvements, Borrower shall furnish Lender with all items required to evidence Completion, certification from the Project Architect that the Project has been completed in accordance with the approved Plans; the evidence of insurance required by Section 5.6 hereof; copies of all warranties covering materials, equipment and appliances included within the Project; copies of all licenses and permits required for operation of the Project, if not previously provided to Lender; and photographs of the completed Improvements as well as evidence that such Improvements have been fully completed, including all punch-list items, as well as such other documents and materials required pursuant to Section 3.7(b) hereof.

Section 5.6 Maintaining Insurance Coverage. Borrower shall, at all times until the Note and all other sums due from Borrower to Lender have been fully repaid, maintain, or cause to be maintained, in full force and effect (and shall furnish to Lender copies of), property insurance, liability insurance and workers compensation insurance that are consistent with policies issued from a reputable carrier in Southern Nevada for businesses such as that operated by Borrower. Borrower shall not take any action that would void or otherwise impair any coverages required hereby or that would result in any denial or limitation of such coverages.

Section 5.7 <u>Transferring, Conveying or Encumbering the Project or Borrower's Ownership Interests</u>. Without the prior written consent of Lender, Borrower shall not voluntarily or involuntarily agree to, cause, suffer or permit any sale, conveyance, lease, mortgage, grant, lien, encumbrance, security interest, pledge, assignment or transfer of: (a) the Project or any part or portion thereof, or (b) any ownership interest in Borrower, direct or indirect, legal or equitable (including the issuance, sale, redemption, or repurchase of any such interest, the distribution of treasury stock, or the payment of any indebtedness owed to Borrower by any managers, subsidiaries, Affiliates or owners of equity interests or debentures), except for

the following:

- (i) Permitted Encumbrances;
- (ii) Leases permitted by Section 5.24 of this Agreement;
- (iii) Routine, nonexclusive utility easements entered into by Borrower in the ordinary course of business which do not subject the Property to any material financial obligations;
- (iv) In connection with transfers of member interests in Borrower made for estate planning purposes that do not affect the management or control of Borrower and are in compliance with the requirements set forth in the Loan Documents;
- (v) In the event that that Borrower obtains any Senior Debt, Borrower shall be permitted to secure such Senior Debt with a deed of trust and other applicable security documents that are senior to the lien of the Deed of Trust and Lender shall cooperate with Borrower and the provider of the Senior Debt to cause the lien of the Deed of Trust to be subordinated to the lien of any deed of trust and other applicable security documents that are to secure such Senior Debt. For the avoidance of doubt, Lender does hereby covenant and agree with Borrower that Lender shall execute, or cause to be executed, any and all documentation reasonably required by the provider of the Senior Debt, in order to cause the aforementioned subordination of the lien of the Deed of Trust, including, but not limited to, any form of subordination and intercreditor agreement that may be required by the provider of the Senior Debt.

Furthermore, the Loan may not be assumed by any Person without Lender's prior written consent.

Section 5.8 Complying with the Loan Documents and Other Documents. Borrower shall comply with and perform all of its obligations under the Loan Documents, and under all other contracts and agreements to which Borrower is a party relating to the ownership, occupancy, use, construction or management of the Project, and shall comply with all requests by Lender which are consistent with the terms thereof.

Section 5.9 <u>Updated Appraisals</u>. Borrower agrees that Lender shall have the right to obtain, at Lender's expense, an updated appraisal of the Project at any time that (a) an Event of Default shall have occurred and be continuing hereunder, or (b) after an occurrence of an Event of Default Lender determines, in its sole discretion, that the security for the Loan has been physically or financially impaired in any material manner. Without limitation of the foregoing, Lender may, at Lender's expense, elect to obtain such an appraisal from time to time regardless of whether any of the events described in (a) and (b) above have occurred. In the event that Lender shall elect to obtain an appraisal, Lender may immediately commission an appraiser acceptable to Lender, at Lender's cost and expense, to prepare the appraisal. Borrower shall cooperate with Lender and the appraiser in obtaining the necessary information to prepare such appraisal. In the event that Borrower fails to cooperate with Lender in obtaining such an

appraisal, such event shall constitute an Event of Default hereunder and Lender shall be entitled to exercise all remedies available to it hereunder. In the event such appraisal is required by reason of the damage or destruction of a portion of the Project, the fair market value shall be calculated on the Project after restoration of the Improvements.

Section 5.10 Reporting Requirements. Borrower shall furnish to Lender the following:

(a) Financial Statements.

As soon as available and in any event within seventy-five (75) days after the end of each calendar year, unaudited consolidated annual financial statements of Borrower prepared in accordance with GAAP (or another accounting basis reasonably acceptable to Lender) consistently applied, certified by Borrower, which financial statements shall include a balance sheet and related statements of income, retained earnings and cash flow (including, without limitation, the information necessary to determine Borrower's cash flow with respect to the previous calendar year), and a detailed list of real estate owned, directly or indirectly, by Borrower. The financial statements of Borrower shall be accompanied by a no default certificate in form acceptable to Lender certifying that Borrower has no knowledge of the occurrence of any event which constitutes a Default or an Event of Default under this Agreement.

- (b) Operating Statements and Operating Budgets. Starting with the month after the calendar month in which Completion occurs, Borrower shall deliver to Lender such Operating Statements and annual Operating Budget as are provided to Borrower by the Management Company prior to the start of each Fiscal Year thereof starting with the Fiscal Year in which Completion occurs. All such Operating Statements shall be certified as true, correct and complete by Borrower and shall be prepared in accordance with GAAP.
- (c) <u>Litigation and Other Proceedings</u>. Promptly in writing, notice of (i) all litigation against Borrower in which the amount sought to be recovered exceeds \$50,000 except in cases when the claim is covered by insurance and the insurance company has agreed to assume the defense of the claim and (ii) all proceedings before any Governmental Authority affecting Borrower which, if adversely determined, would constitute a Material Adverse Occurrence as to Borrower.
- (d) <u>Defaults</u>. Within five (5) Business Days after the occurrence of any event actually known to Borrower which constitutes a Default or an Event of Default, notice of such occurrence, together with a detailed statement of the steps being taken to cure such event, and the estimated date, if known, on which such action will be taken.
- (e) Additional EB-5 reporting. Without limiting the foregoing, information to be provided to Lender by Borrower prior to October 31 of each year, shall specifically include:

- (i) Annual report of expenditures on the project, showing amounts at least equal to the amount of money Lender has disbursed to Borrower have been spent on the Project.; this will include appropriate backup documentation, such as copies of major invoices & payment receipts, major contracts, bank statements, etc.
- (ii) Annual report of payroll records and 1-9 records Borrower shall require its contractors to provide quarterly employment records (form 941) so that the information available for its submissions to Lender.
- (iii) Annual report of actual number of full-time jobs (35 hours per week minimum) at the Project.
 - (iv) Federal / state quarterly employment tax returns.
- (v) Annual limited liability company income tax returns for the prior calendar year.
- Section 5.11 <u>Taxes and Claims</u>. Borrower shall pay and discharge all taxes, when due, assessments and other governmental charges upon the Project, as well as all claims for labor and materials which, if unpaid, might become a lien or charge upon the Project; <u>provided</u>, <u>however</u>, that Borrower shall have the right to contest the amount, validity and/or applicability of any of the foregoing which is being contested in good faith and by proper proceedings, and strictly in accordance with the terms of the Deed of Trust.
- Section 5.12 <u>Maintain Existence</u>. Borrower shall preserve and maintain its name, existence, rights and privileges in the jurisdiction of its organization and qualify and remain qualified in each jurisdiction in which such qualification is necessary in view of its business and operations.
- Section 5.13 Compliance with Applicable Laws. Borrower shall promptly and faithfully comply with, conform to and obey all present and future Governmental Requirements, including, but not limited to, all Environmental Laws, where failure to so do might have a Material Adverse Occurrence; provided, however, that Borrower shall have the ability to contest any alleged failure to conform to or comply with such Governmental Requirements so long as such obligations shall be contested by appropriate proceedings pursued in good faith and any penalties or other adverse effect of its nonperformance shall be stayed or otherwise not in effect.
- Section 5.14 Notice. Borrower shall give prompt written notice to Lender (a) of any action or proceeding instituted by or against Borrower, in any federal or state court or before or by any commission or other regulatory body, federal, state or local, or any such proceedings threatened against Borrower which, if adversely determined, would be a Material Adverse Occurrence as to Borrower or where the amount is \$250,000 or more, (b) any material loss or unusual depreciation of any material asset and the amount of the same, (c) any material dispute that may arise between Borrower or any subsidiary and any Governmental Authority, (d) any

labor controversy resulting or likely to result in a strike or work stoppage against Borrower or any subsidiary, (e) any proposal by any governmental unit to acquire all or any part of Borrower or any subsidiary's assets or business, (f) any change in Borrower's place of business or location of any fixtures or other assets, and (g) any other matters which has resulted or is reasonably likely to result in a Material Adverse Occurrence.

Section 5.15 Contingent Liability. Borrower shall not assume, guarantee, endorse or otherwise become directly or contingently liable in connection with any obligation of any Person (other than Borrower), except (i) by endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of business, or (ii) by indemnity agreements given by Borrower to a title insurance company or a bonding company in connection with any project being constructed or sold by Borrower, including the Project.

Section 5.16 Merger, Consolidation, and Management. Borrower and its subsidiaries shall not dissolve, merge or consolidate into any Person, convert into any other type of Person, form or dispose of any subsidiary of Affiliate, or permit any other Person to merge into it, or acquire (in a transaction analogous in purpose or effect to a consolidation or merger) all or substantially all of the assets of any other Person. Furthermore, Borrower shall not make a material change to its management without a minimum of thirty (30) days' prior written notice to Lender.

Section 5.17 Loss of Note or other Loan Documents. Upon notice from Lender of the loss, theft, or destruction of the Note and upon receipt of an affidavit of lost note and an indemnity reasonably satisfactory to Borrower from Lender, or in the case of mutilation of a Note, upon surrender of such mutilated Note, Borrower shall make and deliver a new note of like tenor in lieu of the then-to-be-superseded Note. If any of the other Loan Documents were lost or mutilated, Borrower agrees to execute and deliver replacement Loan Documents in the same form of such Loan Document(s) that were lost or mutilated.

Section 5.18 <u>Distributions</u>. Borrower shall not, directly or indirectly, prior to the later to occur of Completion of all of the Improvements or the Completion Date, (a) make any distribution of money or property to any Related Party, or (b) make any loan or advance to any Related Party, or (c) pay any principal or interest on any indebtedness due any Related Party, or (d) pay any fees or other compensation (other than payments made to the General Contractor or Manager pursuant to the Budget in conformance with the terms of this Agreement or payments made to any principal of Borrower in accordance with prior historical distributions) to itself or to any Related Party, if any such payment in (a) through (d), inclusive, might adversely affect Borrower's ability to repay the loan in accordance with its terms, provided, however, after Completion of all of the Improvements and the Completion Date, Borrower may make the distributions, loans, advances and payments described in (a), (b), (c) and (d) as long as no Event of Default has occurred and is continuing, and Borrower is in compliance with the reporting requirements set forth in Section 5.12 of this Agreement and further provided that no such payments described in (a) through (d) above shall be made at any time if such payment might adversely affect the ability of the Borrower repay the Loan in accordance with its terms.

Section 5.19 <u>Permits and Licenses</u>. Borrower shall promptly obtain and comply with all necessary licenses, permits and approvals from, and has satisfied the requirements of, all governmental entities necessary to commence and complete construction of the Improvements.

Section 5.20 Patriot Act. Borrower shall not (i) conduct any business or engage in any transaction or dealing with any Blocked Person, including the making or receiving of any contribution of funds, goods or services to or for the benefit of any Blocked Person; (ii) deal in, or otherwise engage in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224; or (iii) engage in or conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in Executive Order No. 13224, the USA Patriot Act or any other Anti-Terrorism Law. Borrower shall deliver to Lender any certification or other evidence requested from time to time by Lender in its sole discretion, confirming Borrower's compliance with this Section. Borrower shall not authorize or enable any of their respective Affiliates or agents to do any of the foregoing in this Section.

Section 5.21 Related Party Transactions. Without Lender's prior written consent, Borrower shall not enter into, or be a party to, any contract or other transaction with a Related Party (including transactions involving the purchase, sale, or exchange of property, the rendering of services or sale of stock) except in the ordinary course of business and upon fair and reasonable terms no less favorable to itself than it would obtain in a comparable arms-length All Lender-approved Related Party agreements shall satisfy the following conditions: (i) the contract or other agreement therefore is terminable immediately at the election of Lender after the occurrence of an Event of Default, (ii) such agreements must provide that any amount payable by Borrower under any such contract or agreement with a Related Party is expressly made subordinate by Borrower and Related Party to Borrower's payment Obligations under the Loan Documents, and (iii) the contract or transaction is not otherwise prohibited by this Agreement or any of the other Loan Documents. In addition to the foregoing restrictions, Borrower shall not make any distributions of any type to any Related Party, except for salary and reimbursement of reasonable expenses actually incurred and consistent with the Budget, whether such proposed additional distribution is in the form of dividends, interest, returns of capital, distributions of profit, bonuses or any other form of distribution.

Section 5.22 <u>Leases</u>. Borrower shall not enter into any Lease, and shall not substantially modify or terminate any Lease, unless (a) Lender has been notified on or prior to the date of execution of such Lease, and (b) such Lease is made subordinate to the lien, operation and effect of the Deed of Trust pursuant to a subordination, non-disturbance and attornment agreement satisfactory each Lease is entered into in the ordinary course of Borrower's business. Borrower shall comply in all respects with the terms, the covenants, agreements, conditions and requirements of each of the Leases, as, when and in the manner required thereby. Borrower shall promptly notify Lender when Borrower receives notice of any default by Borrower as landlord under any Lease.

Section 5.23 Debt; Operations and Fundamental Changes of Borrower. Borrower:

- (a) will not own any asset other than (i) the Project and (ii) incidental personal property necessary for the ownership, operation, management and financing of the Project;
- (b) will not engage in any business other than the ownership, development, management, leasing, operation and sale of the Project and will conduct and operate its business in substantially the manner as presently anticipated to be conducted and operated;
- (c) except for the Senior Debt, will not incur any debt, secured or unsecured, direct or contingent (including guaranteeing any obligation), other than (i) the Loan, and (ii) trade payables or accrued expenses incurred in the ordinary course of business of operating the Project provided the same are paid when due in accordance with customary trade terms or contested in good faith by appropriate proceedings with adequate reserves as determined by Lender;
- (d) will not and will not permit any subsidiary of Borrower to create, incur, or suffer any security interest upon any of its present or future assets that are collateralized by Lender, other than (i) in favor or Lender, (ii) in favor of the provider of the Senior Debt or (iii) liens or claims of materialmen, mechanics, carriers, warehousemen, or processors arising by operation of law in the ordinary course of business and securing obligations that are either paid when due or contested in good faith by appropriate proceedings with adequate reserves as determined by Lender.
- (e) will not permit the Project or any portion thereof to secure any debt whatsoever (senior, subordinate or pari passu) other than the Loan and/or any Senior Debt;
- (f) will not make any loans or advances to any third party other than financing to timeshare purchasers, and will not acquire obligations or securities of its Affiliates:
- (g) will not make any payment on any indebtedness in violation of any subordination agreement made by the holder of such indebtedness;
 - (h) will not purchase or acquire obligations owed by third parties;
- (i) will remain solvent and pay its debts and liabilities (including, as applicable, shared personnel and overhead expenses) from Borrower's assets as the same shall become due:
- (j) will do all things necessary to observe organizational formalities and preserve Borrower's existence;

- (k) will at all times hold itself out to the public as a legal entity separate and distinct from any other entity (including any Affiliate of Borrower or any constituent party of Borrower), shall correct any known misunderstanding regarding its status as a separate entity, shall conduct business in its own name, and shall not identify itself or any of its Affiliates as a division or part of the other;
- (l) will maintain or have access to adequate capital for the normal obligations reasonably foreseeable in a business of its size and character and in light of its contemplated business operations;
- (m) will not make or consent to any material change in its method of accounting (including the basis of application of GAAP) or in its tax elections under any Law:
- (n) will not sell, lease, transfer or otherwise dispose of all or any material part of its present or future assets other than the sale of timeshare interests in the Project;
- (o) will, to the fullest extent permitted by law, not seek or effect the liquidation, dissolution, winding up, consolidation or merger, in whole or in part, of Borrower; and
- (p) except as otherwise permitted in this Agreement or any of the other Loan Documents, will not commingle its funds and other assets with those of any Affiliate or constituent party of Borrower or any other Person, and will hold all of its assets in its own name.
- Section 5.24 <u>Accessibility Regulation</u>. Borrower shall comply with all Accessibility Regulations which are applicable to the Project.
- Section 5.25 Reports and Returns. Borrower shall file with the appropriate Governmental Authority every report and notice required by all Laws (including tax returns, levies, and assessments) on or before the initial due date or the extended due date, if there exists a valid extension for filing without interest or penalties.
- Section 5.26 <u>Management Agreement</u>. Borrower shall deliver to Lender a copy of the fully-executed Management Agreement at least thirty (30) days before the issuance of any certificates of occupancy for the Project.
- Section 5.27 Senior Debt. Borrower will use its best efforts to obtain Senior Debt as defined herein. Borrower and Lender expect that the Senior Debt documents will impose provisions concerning disbursement procedures, mechanisms to protect against mechanics liens and related matters as are customarily found in construction loans made by institutional lenders, which procedures also tend to help protect Lender. If Borrower has not obtained such Senior Debt by March 31, 2017, Borrower agrees that Lender may impose provisions concerning such matters similar to those customarily found in construction loans made by institutional lenders. In addition, Borrower will execute and deliver, upon request by Lender, such assignments of contracts relating to the Project as Lender shall request, including, but not limited to, the

Management Agreement, documents concerning the construction of the Project and any leases.

(B) COVENANT OF LENDER

Lender shall not (i) conduct any business or engage in any transaction or dealing with any Blocked Person, including the making or receiving of any contribution of funds, goods or services to or for the benefit of any Blocked Person; (ii) deal in, or otherwise engage in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224; or (iii) engage in or conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in Executive Order No. 13224, the USA Patriot Act or any other Anti-Terrorism Law. Lender shall deliver to Borrower any certification or other evidence requested from time to time by Borrower in its sole discretion, confirming Lender's compliance with this Section. Lender shall not authorize or enable any of their respective Affiliates, members or agents to do any of the foregoing in this Section.

ARTICLE VI

DEFAULTS

- Section 6.1 Events of Default. Any of the following events shall constitute an Event of Default under this Agreement (each an "Event of Default"):
 - (a) Borrower shall default in any payment of principal or interest due according to the terms hereof or of the Note, and such default shall remain uncured for a period of five (5) days after the payment became due, provided, however, there is no cure period for payments due on the Maturity Date;
 - (b) Borrower shall default in the payment of undisputed fees or other amounts payable to Lender hereunder or under any other Loan Document other than as set forth in subsection (a) above, and such default continues unremedied for a period of ten (10) days after notice from Lender to Borrower thereof.;
 - (c) Borrower shall default in the performance or observance of any agreement, covenant or condition required to be performed or observed by Borrower under the terms of this Agreement, or any other Loan Document, other than a default described elsewhere in this Section, and such default continues unremedied for a period of thirty (30) days after notice from Lender to Borrower thereof provided that, if cure cannot reasonably be effected within such 30-day period, such failure shall not be an event of default hereunder so long as Borrower promptly (in any event, within ten (10) days after such notice of default from Lender) commences cure, and thereafter diligently (in any event, within ninety (90) days after receipt of such notice of default from Lender) prosecutes such cure to completion; and provided further, however, that notwithstanding the 30-day cure period or extended cure period described above in this subparagraph (c), if a different notice or cure period is specified under any Loan Document or under any provision of the Loan Documents as to any such failure or breach, the specific Loan Document or provision shall control, and Borrower shall have no more time to cure the

failure or breach than is allowed under the specific Loan Document or provision as to such failure or breach;

- (d) Any representation or warranty made by Borrower in this Agreement or by Borrower or an Affiliate, if made in connection with the Loan, in any of the other Loan Documents, or in any certificate or document furnished under the terms of this Agreement or in connection with the Loan, shall be untrue or incomplete in any material respect when made or deemed made or restated hereunder unless such representation or warranty was not known by Borrower to be untrue or incomplete at the time made and such representation or warranty is corrected by Borrower and disclosed by Borrower to Lender;
- (e) Borrower shall be in default under any term, covenant or condition of any of the Note or of any of the other Loan Documents to which Borrower is a party, other than a default described elsewhere in this Section, after the expiration of any notice or grace period, if any, provided therein;
- (f) Work on the Project, once commenced, shall be substantially abandoned, or shall, by reason of Borrower's fault, be unreasonably delayed or discontinued for a period of fifteen (15) consecutive days, or construction shall be delayed for any reason whatsoever to the extent that Completion cannot, in the reasonable judgment of Lender, be accomplished prior to the Completion Date;
- Any of Borrower, or any Related Party who is a party to any of the Loan Documents, shall file a petition for bankruptcy; or shall apply for, consent to or permit the appointment of a receiver, custodian, trustee or liquidator for it or any of its property or assets; or shall generally fail to, or admit in writing its inability to, pay its debts as they mature; or shall make a general assignment for the benefit of creditors or shall be adjudicated bankrupt or insolvent; or shall take other similar action for the benefit or protection of its creditors; or shall give notice to any governmental body of insolvency or of pending insolvency or suspension of operations; or shall file a voluntary petition in bankruptcy or a petition or an answer seeking reorganization or an arrangement with creditors, or to take advantage of any bankruptcy, reorganization, insolvency, readjustment of debt, rearrangement, dissolution, liquidation or other similar debtor relief law or statute; or shall file an answer admitting the material allegations of a petition filed against it in any proceeding under any such law or statute; or shall be dissolved, liquidated, terminated or merged; or shall effect a plan or other arrangement with creditors; or a trustee, receiver, liquidator or custodian shall be appointed for it or for any of its property or assets and shall not be discharged within ninety (90) days after the date of his appointment; or a petition in involuntary bankruptcy or similar proceedings is filed against it and is not dismissed within ninety (90) days after the date of its filing;
- (h) Lender determines that the remaining undisbursed Loan proceeds, together with the proceeds of any Senior Debt, are insufficient to fully pay all of the then-unpaid costs of the Project and the estimated expenses of completion (including the Interest Reserve), and Borrower fails to either (i) deposit with Lender, within three (3) Business

Days following demand, sufficient funds to permit Lender to pay said excess costs as the same become payable or (ii) pay said excess costs directly and deliver to Lender unconditional mechanics' lien waivers therefor (or paid receipts for non-lienable items), at Lender's option;

- (i) A default shall occur with respect to the Senior Debt and shall remain uncured after the expiration of any applicable notice or grace period;
- (j) A default occurs in the performance of Borrower's obligations in any of Section 5.6, 5.7, 5.8, 5.10, 5.13, 5.16, 5.18, 5.19, 5.22, 5.23 or 5.24, hereof;
- (k) The General Contract shall be terminated by either party thereto or either party thereto shall fail to perform its obligations (after any applicable notice and cure period) under the General Contract;
 - (1) Any uncured default by Borrower under the Management Agreement.
- (m) Any failure by Borrower to timely deliver the EB-5 information, which failure continues more than 5 days following notice of such failure from Lender.
- Section 6.2 <u>Rights and Remedies</u>. Upon the occurrence of an Event of Default, unless such Event of Default is subsequently waived in writing by Lender, Lender may exercise any or all of the following rights and remedies, consecutively or simultaneously, and in any order:
 - (a) make one or more Advances of Loan proceeds without liability to make any subsequent Advance;
 - (b) suspend the obligation of Lender to make Advances under this Agreement, without notice to Borrower;
 - (c) declare that the Commitment is terminated whereupon the Commitment shall terminate:
 - (d) declare the entire unpaid principal balance of the Note to be immediately due and payable, together with accrued and unpaid interest on such Advances, without notice to or demand on Borrower;
 - (e) exercise any or all remedies specified herein and in the other Loan Documents, including (without limiting the generality of the foregoing) the right to foreclose the Deed of Trust, and/or any other remedies which it may have therefor at law, in equity or under statute;
 - (f) cure the Event of Default on behalf of Borrower, and, in doing so, enter upon the Project, and expend such sums as it may deem desirable, including reasonable attorneys' fees, all of which shall be deemed to be Advances hereunder, even though

causing the Loan to exceed the face amount of the Note, shall bear interest at the Default Rate provided herein and shall be payable by Borrower on demand; and/or;

(g) Lender may declare an Event of Default under any other Loan Document in accordance with the terms and conditions thereof, and may effectuate any remedies provided for in such agreement.

In addition to the other remedies set forth herein and in the other Loan Documents, Borrower hereby irrevocably authorizes Lender, at any time while an Event of Default continues, to set off any sum due to or incurred by Lender against all accounts, deposits and credits of Borrower with, and any and all claims of Borrower against Lender. Such right shall exist whether or not Lender shall have made any demand hereunder or under any other Loan Document, whether or not said sums, or any part thereof, or deposits and credits held for the account of Borrower is or are matured or unmatured, and regardless of the existence or adequacy of any collateral, or any other security, right or remedy available to Lender. Lender agrees that, as promptly as is reasonably possible after the exercise of any such setoff right, it shall notify Borrower of its exercise of such setoff right; provided, however, that the failure of Lender to provide such notice shall not affect the validity of the exercise of such setoff rights. Nothing in this Agreement shall be deemed a waiver or prohibition of or restriction on Lender to all rights of banker's lien, setoff and counterclaim available pursuant to law.

Section 6.3 Completion of Project by Lender. In addition, in case of the occurrence of an Event of Default specified in Section 6.1(f) hereof, or any Event of Default caused by, or which results in, Borrower's failure, for any reason, to continue with the completion of the construction of the Improvements as required by this Agreement, then Lender may (but shall not be obligated to), in addition to, or in concert with, the other remedies referred to above, take over and complete such construction in accordance with the Plans, with such changes therein as Lender may, in its discretion, deem appropriate, all at the risk, cost and expense of Borrower. Lender may assume or reject any contracts entered into by Borrower in connection with the Project with the exception of timeshare purchase contracts, may enter into additional or different contracts for work, services, labor and materials required, in the judgment of Lender, to complete the Project, may pay, compromise and settle all claims in connection with the construction of the Improvements. All sums, including reasonable attorneys' fees, and charges or fees for supervision and inspection of the construction of the Improvements and for any other necessary or desirable purpose in the discretion of Lender expended by Lender in completing or attempting to complete the construction of the Improvements (whether aggregating more, or less, than the aggregate face amount of the Note), shall be deemed Advances made by Lender to Borrower hereunder, and Borrower shall be liable to Lender, on demand, for the payment of such sums, together with interest on such sums from the date of their expenditure at the rates provided herein. Lender may, in its discretion, at any time abandon work on the Project, after having commenced such work, and may recommence such work at any time, it being understood that nothing in this Section shall impose any obligation on Lender either to complete or not to complete the construction of the Improvements. For the purpose of carrying out the provisions of this Section, Borrower irrevocably appoints Lender its attorney-in-fact, with full power of substitution, to execute and deliver all such documents, to pay and receive such funds, and to take such action as may be necessary, in the judgment of Lender, to complete the Project. This

power of attorney is coupled with an interest and is irrevocable. Lender, however, shall have no obligation to undertake any of the foregoing, and, if Lender does undertake any of the same, it shall have no liability for the adequacy, sufficiency or completion thereof.

ARTICLE VII

INTEREST, FEES AND EXPENSES

Section 7.1 Interest; Fees; and Expenses.

- (a) Included in the Loan are amounts for Interest Reserve. Borrower hereby authorizes Lender to disburse the Interest Reserve to pay interest accrued on the Note and all other expenses and fees directly to the party to whom such payment is owed as set forth in this Article 7. The execution of this Agreement by Borrower shall, and hereby does, constitute an irrevocable authorization to Lender to make such advances provided for in this Article 7, and no further authorization from Borrower shall be necessary to warrant such direct advances, and all such direct advances shall be secured by the Deed of Trust as if made directly by Borrower.
- (b) Lender in its sole discretion may (but shall not be obligated to do so) make such disbursements authorized under this Article 7 notwithstanding that an Event of Default exists under the terms of this Agreement or any other Loan Document. Such disbursements shall constitute an Advance and be added to the principal balance of the Note, and the Lender shall make the applicable Advances to fund any such disbursements. Such disbursements shall bear interest at the Default Rate. The authorization hereby granted is irrevocable, and no further direction or authorization from Borrower is necessary for Lender to make such disbursements. Nothing contained in this Article 7 shall require Lender to disburse Loan proceeds to pay for any of the items set forth in subsection (a) above if the other conditions set forth in this Agreement for Advances are not satisfied.

Section 7.2 Authorization to Make Loan Advances to Cure Borrower's Defaults. If an Event of Default shall occur, Lender (subject to the provisions of this Article 7) may (but shall not be required to) perform any of such covenants and agreements with respect to which Borrower is in Default. Any amounts expended by Lender in so doing and any amounts expended by Lender in connection therewith shall constitute an Advance and be added to the outstanding principal amount of the Loan, and the Lender shall make the applicable Advances to fund any such disbursements. The authorization hereby granted is irrevocable, and no prior notice to or further direction or authorization from Borrower is necessary for Lender to make such disbursements.

ARTICLE VIII

MISCELLANEOUS

Section 8.1 <u>Waiver and Amendment</u>. No failure on the part of Lender or the holder of the Note to exercise and no delay in exercising any power or right hereunder or under any

other Loan Document shall operate as a waiver thereof; nor shall any single or partial exercise of any power or right preclude any other or further exercise thereof or the exercise of any other power or right. The remedies herein and in any other instrument, document or agreement delivered or to be delivered to Lender hereunder or in connection herewith are cumulative and not exclusive of any remedies provided by law. No notice to or demand on either party hereunder not required hereunder or under the Note or any other Loan Document shall in any event entitle such party to any other or further notice or demand in similar or other circumstances or constitute a waiver of the right of Lender, the holder of the Note or Borrower to any other or further action in any circumstances without notice or demand.

No amendment, waiver or consent shall affect the rights or duties of Lender under this Agreement or any other Loan Document unless it is in writing and signed by Lender.

Section 8.2 Expenses and Indemnities.

- (a) Loan Documents. Borrower agrees to pay and reimburse Lender upon demand for all reasonable expenses paid or incurred by Lender (including reasonable fees and expenses of legal counsel) in connection with the collection and enforcement of the Loan Documents, or any of them. Borrower agrees to pay, and save Lender harmless from all liability for, any mortgage registration, mortgage recording, transfer, recording, stamp, or similar tax or other charge due to any governmental entity, which may be payable with respect to the execution or delivery of the Loan Documents. Borrower agrees to indemnify Lender and hold Lender harmless from any loss or expense which may arise or be created by the acceptance of telephonic or other instructions for making the Loan or disbursing the proceeds thereof except for losses or expenses caused by Lender, gross negligence or willful misconduct.
- General Indemnity. In consideration of the Commitment, Borrower further agrees to indemnify and defend Lender and its directors, officers, agents and employees (the "Indemnified Parties") from, and hold each of them harmless against, any and all losses, liabilities, claims, damages, deficiencies, interest, judgments, costs or expenses incurred by them or any of them, including, but without limitation, amounts paid in settlement, court costs, and reasonable fees and disbursements of counsel incurred in connection with any investigation, litigation or other proceeding, arising out of or by reason of any investigation, litigation or other proceeding brought or threatened, arising out of or by reason of their execution of any Loan Document and the transaction contemplated thereby, including, but not limited to, any use effected or proposed to be effected by Borrower of the proceeds of the Loan, but excluding any such losses, liabilities, claims, damages or expenses incurred by reason of the gross negligence or willful misconduct of the relevant Indemnified Party. Any Indemnified Party seeking indemnification under this Section will notify Borrower of any event requiring indemnification within thirty (30) Business Days following such Indemnified Party's receipt of notice of commencement of any action or proceeding, or such Indemnified Party's obtaining knowledge of the occurrence of any other event, giving rise to a claim for indemnification hereunder. Borrower will be entitled (but not obligated) to assume the defense or settlement of any such action or proceeding or to participate in any

negotiations to settle or otherwise resolve any claim using counsel of its choice; provided that:

- (i) Borrower notifies such Indemnified Party in writing that Borrower will indemnify such Indemnified Party from and against the relevant claim;
 - (ii) such counsel is reasonably satisfactory to such Indemnified Party;
- (iii) such claim involves only money damages and does not seek an injunction or other equitable relief;
- (iv) if such Indemnified Party is the Lender, settlement of, or an adverse judgment with respect to, such claim is not, in the good faith judgment of such Indemnified Party, likely to establish a precedential custom or practice materially adverse to the continuing business interests of such Indemnified Party;
- (v) Borrower conducts the defense of such claim actively and diligently;
- (vi) no conflict of interest has arisen which would prevent counsel for Borrower from also representing such Indemnified Party because the defendants in any action include both such Indemnified Party and Borrower; and
- (vii) Borrower will not consent to the entry of any judgment or enter into any settlement with respect to such claim without the prior written consent of such Indemnified Party (not to be withheld unreasonably).

So long as Borrower has assumed the defense of such claim and is conducting such defense in accordance with the foregoing, such Indemnified Party: (y) may retain separate co-counsel at its sole cost and expense and participate in the defense of such claim; and (z) will not consent to the entry of any judgment or enter into any settlement with respect to such claim without the prior written consent of Borrower with respect to such claim (not to be withheld unreasonably).

If Borrower fails to assume such defense or, after doing so, Borrower fails to satisfy any of the above conditions to Borrower's defense, such Indemnified Party (and its counsel) may defend against, and consent to the entry of any judgment or enter into any settlement with respect to, such claim in any manner it may reasonably deem appropriate (and such Indemnified Party need not consult with, or obtain any consent from, Borrower in connection therewith) and Borrower will reimburse such Indemnified Party promptly and periodically for the costs of defending against such claim (including reasonable attorneys' fees and expenses) and Borrower will remain responsible for any loss which such Indemnified Party may suffer resulting from, arising out of, relating to, in the nature of, or caused by such claim to the fullest extent provided for and required by this Agreement.

Section 8.3 <u>Binding Effect; Waivers; Cumulative Rights and Remedies</u>. The provisions of this Agreement shall inure to the benefit of and be binding upon the parties hereto

and their respective heirs, executors, administrators, personal representatives, legal representatives, successors and assigns; provided, however, that neither this Agreement nor the proceeds of the Loan may be assigned by Borrower voluntarily, by operation of law or otherwise, without the prior written consent Lender. No delay on the part of Lender in exercising any right, remedy, power or privilege hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any right, remedy, power or privilege hereunder constitute such a waiver or exhaust the same, all of which shall be continuing. The rights and remedies of Lender specified in this Agreement shall be in addition to, and not exclusive of, any other rights and remedies which Lender would otherwise have at law, in equity or by statute, and all such rights and remedies, together with Lender's rights and remedies under the other Loan Documents, are cumulative and may be exercised individually, concurrently, successively and in any order.

Section 8.4 <u>Incorporation By Reference</u>. Borrower agrees that until this Agreement is terminated by the repayment to Lender of all principal and interest due and owing on the Note and other sums due and owing pursuant to the other Loan Documents, the Note and the other Loan Documents shall be made subject to all the terms, covenants, conditions, obligations, stipulations and agreements contained in this Agreement to the same extent and effect as if fully set forth in and made a part of the Note and the other Loan Documents.

Section 8.5 <u>Survival</u>. All agreements, representations and warranties made in this Agreement shall survive the execution of this Agreement, the making of the Advances by Lender, and the execution of the other Loan Documents, and shall continue until Lender receive payment in full of all of the Obligations and the Indebtedness of Borrower incurred under this Agreement and under the other Loan Documents.

Section 8.6 Governing Law; Waiver of Jury Trial; Jurisdiction. IN ALL RESPECTS, INCLUDING, WITHOUT LIMITING THE GENERALITY OF FOREGOING, MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, THIS AGREEMENT, THE OBLIGATIONS ARISING HEREUNDER, AND ALL QUESTIONS AND ISSUES ARISING HEREUNDER SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEVADA, APPLICABLE TO CONTRACTS MADE AND PERFORMED IN SUCH STATE (WITHOUT REGARD TO PRINCIPLES OF CONFLICT OF LAWS) AND ANY APPLICABLE LAW OF THE UNITED STATES OF AMERICA. TO THE FULLEST EXTENT PERMITTED BY LAW, BORROWER HEREBY UNCONDITIONALLY AND IRREVOCABLY WAIVES ANY CLAIM TO ASSERT THAT THE LAW OF ANY OTHER JURISDICTION GOVERNS THIS AGREEMENT AND THE NOTE, AND THIS AGREEMENT AND THE NOTE SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEVADA.

BORROWER HEREBY WAIVES ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION RELATING TO THE LOAN AND/OR THE LOAN DOCUMENTS. BORROWER, TO THE FULLEST EXTENT PERMITTED BY LAW, HEREBY KNOWINGLY, INTENTIONALLY AND VOLUNTARILY, WITH AND UPON THE ADVICE OF COMPETENT COUNSEL, (A) SUBMITS TO PERSONAL JURISDICTION IN THE STATE

OF NEVADA OVER ANY SUIT, ACTION OR PROCEEDING BY ANY PERSON ARISING FROM OR RELATING TO THIS AGREEMENT, (B) AGREES THAT ANY SUCH ACTION, SUIT OR PROCEEDING MAY BE BROUGHT IN ANY STATE OR FEDERAL COURT OF COMPETENT JURISDICTION IN THE STATE OF NEVADA, (C) SUBMITS TO THE JURISDICTION AND VENUE OF SUCH COURTS AND WAIVES ANY ARGUMENT THAT VENUE IN SUCH FORUMS IS NOT CONVENIENT, AND (D) AGREES THAT IT WILL NOT BRING ANY ACTION, SUIT OR PROCEEDING IN ANY OTHER FORUM. BORROWER FURTHER CONSENTS AND AGREES TO SERVICE OF ANY SUMMONS, COMPLAINT OR OTHER LEGAL PROCESS IN ANY SUCH SUIT, ACTION OR PROCEEDING BY REGISTERED OR CERTIFIED U.S. MAIL, POSTAGE PREPAID, TO BORROWER AT THE ADDRESSES FOR NOTICES DESCRIBED IN THIS AGREEMENT, AND CONSENTS AND AGREES THAT SUCH SERVICE SHALL CONSTITUTE IN EVERY RESPECT VALID AND EFFECTIVE SERVICE (BUT NOTHING HEREIN SHALL AFFECT THE VALIDITY OR EFFECTIVENESS OF PROCESS SERVED IN ANY OTHER MANNER PERMITTED BY LAW).

Section 8.7 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, all of which shall constitute a single Agreement.

Section 8.8 Notices. All notices, demands, requests, consents, approvals or other communications (any of the foregoing, a "Notice") required, permitted or desired to be given hereunder shall be in writing and shall be sent by (a) registered or certified mail, postage prepaid, return receipt requested, (b) Federal Express, UPS or another reputable overnight courier, or (c) by hand by commercial courier service, addressed to the party to be so notified at its address set forth opposite its signature, below, or to such other address as such party may hereafter specify in accordance with the provisions of this Section. Any Notice shall be deemed to have been received: (i) three (3) days after the date such Notice is mailed, (ii) on the date of delivery by hand (or refusal to accept such delivery) if delivered during business hours on a Business Day (otherwise on the next Business Day), and/or (iii) on the next Business Day if sent by an overnight commercial courier. Notices shall be deemed effective if given by counsel to either party, as if given directly by such party. No notice under this Agreement or any other Loan Document shall be ineffective because of the failure to provide a copy of any notice to any party to whom a courtesy copy is to be sent under this Agreement.

Any party may change the address to which any such Notice is to be delivered by furnishing ten (10) days' prior written notice of such change to the other parties in accordance with the provisions of this Section. Notices shall be deemed to have been given on the date as set forth above, even if there is an inability to actually deliver any such Notice because of a changed address of which no Notice was given, or there is a rejection or refusal to accept any Notice offered for delivery.

Section 8.9 No Third-Party Reliance. No third party shall be entitled to rely upon this Agreement or to have any of the benefits of Lender's interest hereunder, unless such third party is an express assignee of all or a portion of Lender's interest hereunder.

Section 8.10 <u>Lender Assignment</u>. Borrower agrees that Lender may assign its interest in the Loan at any time without the prior consent of Borrower, provided that, upon such assignment, the terms of the Loan shall not be modified or amended in any way and provided further, that any such assignment does not violate any terms or conditions of the EB-5 Program.

in se

- Section 8.11 <u>Time of the Essence</u>. Time is of the essence hereof with respect to the dates, terms and conditions of this Agreement.
- Section 8.12 No Oral Modifications. No modification or waiver of any provision of this Agreement shall be effective unless set forth in writing and signed by the parties hereto.
- Section 8.13 <u>Captions</u>. The headings or captions of the Articles and Sections set forth herein are for convenience only, are not a part of this Agreement and are not to be considered in interpreting this Agreement.
- Section 8.14 <u>Borrower-Lender Relationship</u>. The relationship between Borrower and Lender created hereby and by the other Loan Documents shall be that of a borrower and a lender only, and in no event shall Lender be deemed to be a partner of, or a joint venturer with, Borrower.

[END OF AGREEMENT; SIGNATURE PAGES FOLLOW]

IN WITNESS WHEREOF, the parties have executed this Agreement as of the day and year first above written.

BORROWER:

FRONT SIGHT MANAGEMENT, LLC, a Nevada limited liability company

By:_

Name: Ignatius Piazza Title: Manager

Borrower's Address:

1 Front Sight Road Pahrump, NV 89061

With a copy to (which copy shall not constitute notice):

Scott A. Preston, Esq. Preston Arza LLP 8581 Santa Monica Boulevard, #710 West Hollywood, CA 90069

[Signature page 1 of 2 of Construction Loan Agreement]

CALIFORNIA ALL-PURPOSE CERTIFICATE OF ACKNOWLEDGMENT

A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document.

Ctotal of California	
State of California	
County of San Francisco	
On OCT 06,2016 before me, P. Fakeri	(Here insert name and title of the officer)
personally appeared Ignativs Pla	122a ,
the within instrument and acknowledged to me that	ence to be the person(s) whose name(s) is/are subscribed to at he/she/they executed the same in his/her/their authorized on the instrument the person(s), or the entity upon behalf of
	laws of the State of California that the foregoing paragraph
is true and correct. WITNESS my hand and official seal.	P. FAKERI COMM. #2003459 Notary Public - California San Mateo County My Comm. Expires Feb. 6, 2017
Signature of Notary Public	(Notary Seal)
ADDITIONAL OP	TIONAL INFORMATION
DESCRIPTION OF THE ATTACHED DOCUMENT Construction of attached document)	INSTRUCTIONS FOR COMPLETING THIS FORM Any acknowledgment completed in California must contain verbiage exactly as appears above in the notary section or a separate acknowledgment form must be properly completed and attached to that document. The only exception is if a document is to be recorded outside of California. In such instances, any alternative acknowledgment verbiage as may be printed on such a document so long as the verbiage does not require the notary to do something that is illegal for a notary in
(Title or description of attached document continued)	California (i.e. certifying the authorized capacity of the signer). Please check the document carefully for proper notarial wording and attach this form if required.
(Title or description of attached document continued) Number of Pages 51 Document Date 19/6/16 (Additional information)	California (i.e. certifying the authorized capacity of the signer). Please check the document carefully for proper notarial wording and attach this form if required. State and County information must be the State and County where the document signer(s) personally appeared before the notary public for acknowledgment. Date of notarization must be the date that the signer(s) personally appeared which must also be the same date the acknowledgment is completed. The notary public must print his or her name as it appears within his or her
Number of Pages 51 Document Date 19/6/16	California (i.e. certifying the authorized capacity of the signer). Please check the document carefully for proper notarial wording and attach this form if required. State and County information must be the State and County where the document signer(s) personally appeared before the notary public for acknowledgment. Date of notarization must be the date that the signer(s) personally appeared which must also be the same date the acknowledgment is completed.

 ${\tt C.2004-2015\ ProLink\ Signing\ Service, Inc.} + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + All\ Rights\ Reserved\ www. The {\tt ProLink.com} + Nationwide\ Notary\ Service, Inc. + Nationwide\ Nationwide\ Notary\ Service, Inc. + Nationwide\ N$

LENDER:

LAS VEGAS DEVELOPMENT FUND, LLC, a Nevada limited liability company

Зу:		
- J • E	Name:	
	Title:	

Lender's Address:

P.O. Box 3003 916 Southwood Blvd., Suite 1G Incline Village, NV 89450

With a copy to (which copy shall not constitute notice):

EB5 Impact Capital Regional Center LLC 916 Southwood Blvd., Suite 1G Incline Village, NV 89450

And

Michael A. Brand, Esq. 2924 Selwyn Circle Santa Barbara, California 93105

And:

C. Matthew Schulz, Esq. Dentons US LLP 1530 Page Mill Road, Suite 200 Palo Alto, CA 94304-1125

[Signature page 2 of 2 of Construction Loan Agreement]

4847-9752-5523.5

EXHIBIT 4

FIRST AMENDMENT TO LOAN AGREEMENT

This FIRST AMENDMENT TO LOAN AGREEMENT (this "First Amendment") is entered into and effective as of July 1, 2017 (the "First Amendment Effective Date") by and between Las Vegas Development Fund, LLC, a Nevada Limited Liability Company ("Lender") and Front Sight Management, LLC, a Nevada Limited Liability Company, ("Borrower"). Lender and Borrower and their respective permitted successors and assigns are sometimes referred to in this First Amendment individually as a "Party" and collectively as the "Parties".

RECITALS

- A. Lender and Borrower entered into that certain Construction Loan Agreement dated as of October 4, 2016 (the "Original Loan Agreement"). The Original Loan Agreement as amended by this First Amendment is referred to herein as the "Agreement". Pursuant to the Original Loan Agreement, Borrower executed a Promissory Note dated October 6, 2016 (the "Original Note") and a Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated October 6, 2016 (the "Deed of Trust"). Initially capitalized terms not defined herein shall have the respective meanings assigned to such terms in the Original Loan Agreement.
- B. The Parties desire to amend the Original Loan Agreement, the Original Note and the Deed of Trust to modify the rights and obligations of the Parties as further set forth below.

AGREEMENT

NOW THEREFORE, in consideration of the foregoing Recitals, which are hereby incorporated into the operative provisions of this First Amendment by this reference, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties further agree as follows:

1. COMMENCEMENT DATE. The definition of "Commencement Date" in the Original Loan Agreement is hereby deleted and replaced with:

"Commencement Date means October 4, 2016."

2. INTEREST RATE. The definition of Loan Rate in the Original Loan Agreement is amended to read as follows:

"Loan Rate" means the interest rate applicable to the Loan as calculated at an annual rate of 6% during the Initial Term with respect to all Advances made prior to July 1, 2017 and, with respect to such Advances, if extended, at an annual rate of 7% during the Extension Term; and with respect to all Advances made after July 1, 2017 as calculated at an annual rate of 7% during the Initial Term, and, if extended, at an annual rate of 8% during the Extension Term."

And Section 4 of the Note is amended to read as set forth in the Amended and Restated Promissory Note attached hereto as Exhibit A.

3. MAXIMUM LOAN AMOUNT. The maximum Loan amount is hereby reduced from seventy-five million Dollars (\$75,000,000) to fifty million Dollars (\$50,000,000). Accordingly, the reference in Recital A of the Loan Agreement to SEVENTY-FIVE MILLION DOLLARS (\$75,000,000) is hereby amended to FIFTY MILLION DOLLARS (\$50,000,000), and the first sentence of the definition of

JAV I)

"Commitment" in the Loan Agreement is hereby amended to read: "Commitment" means an amount not to exceed Fifty Million Dollars (\$50,000,000)".

Furthermore, the amount shown as the maximum principal amount on the Promissory Note is amended by replacing "\$75,000,000" with "\$50,000,000," and the amount of "Seventy-Five Million and No/100 Dollars (\$75,000,000)" in the first sentence of the Promissory Note is replaced by "Fifty Million and No/100 Dollars (\$50,000,000)" as set forth in the Amended and Restated Promissory Note attached hereto as Exhibit "A".

Additionally, the first sentence of Section 1.1 of the Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing has been amended in said document to read; "Loan, The indebtedness secured by this Deed of Trust is the result of a loan in the original principal amount of up to Fifty Million dollars (\$50,000,000) (the "Loan") provided by Lender to Grantor."

- 4. DATE TO OBTAIN SENIOR DEBT. The date of December 31, 2016, in the last sentence of the definition of Senior Debt, which is the outside date for Borrower to obtain such Senior Debt, is hereby amended to December 31, 2017, provided, however, that Borrower, at its sole election, may extend said date for an additional sixty (60) days from and after said date of December 31, 2017.
- 5. MAINTENANCE OF JOBS REQUIRED FOR EB-5 PURPOSES. Borrower agrees and covenants to continue to employ sufficient full-time employees to meet the jobs creation requirement of the EB-5 Program.
- EB-5 INFORMATION. Borrower has provided Lender with a portion of the information and documentation required pursuant to Section 5.10 of the Original Loan Agreement. The parties acknowledge that Borrower's copies of certain documentation were in an office in or around Santa Rosa, California, which was burned down in a major wildfire in Northern California, Notwithstanding the foregoing, on or before June 30, 2018, Borrower shall provide Lender with copies of major contracts, bank statements. receipts, invoices and cancelled checks or credit card statements or other proof of payment reasonably acceptable to Lender that document that Borrower has invested in the Project at least the amount of money as has been disbursed by Lender to Borrower on or before the First Amendment Effective Date, it being understood that such documentation may evidence investments occurring at any time from and after the date of the Original Loan Agreement up to and including June 30, 2018. Borrower further agrees that, in the event that there is an audit, compliance review or other form of request for such documentation by the USCIS or any successor or affiliated agency, including the U.S. Securities and Exchange Commission or the U.S. Department of Justice, Borrower will, at Borrower's sole cost and expense, promptly reconstruct in its entirety such documentation evidencing the investment of the amount of funds disbursed on or before the First Amendment Effective Date by obtaining copies from third parties. Borrower further agrees that the provisions of the Original Loan Agreement continue to apply with respect to the EB-5 Information (as defined in the Original Loan Agreement) and all provisions of the Original Loan Agreement which require Borrower to provide information and/or documentation to Lender continue to apply and Borrower will comply fully therewith.
- INDEMNIFICATION. Borrower agrees to defend, indemnify and hold Lender harmless from any actual expense, cost, loss or damage, including reasonable attorneys' fees and court costs, paid or incurred by Lender due to (i) Borrower's failure to provide the EB5 documentation for the period from the first disbursement of the Loan proceeds through October 31, 2017, or (ii) Borrower's breach of its obligations contained in Paragraph 6, above.
- 8. AGREEMENT RATIFIED. Except as specifically amended or modified herein, each and every term, covenant, and condition of the Original Loan Agreement, Note and Deed of Trust as amended is

hereby ratified and shall remain in full force and effect. Each and every reference to the "Agreement" in the Original Loan Agreement (including, without limitation, the attachments thereto) shall be deemed to refer to the Original Loan Agreement as amended by this First Amendment.

- 9. GOVERNING LAW. This instrument shall be interpreted and construed in accordance with the laws of the State of Nevada.
- 10. BINDING AGREEMENT. This First Amendment shall be binding upon and inure to the benefit of the Parties hereto and their respective successors and permitted assigns.
- 11. COUNTERPARTS. This First Amendment may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same document.

[Signature page follows]

IN WITNESS WHEREOF, Lender and Borrower have signed this First Amendment as of the First Amendment Effective Date.

BORROWER: FRONT SIGHT MANAGEMENT, LLC, A Nevada Limited Liability Company

By: Name: Ignatius Piazza
Title: Manager

LENDER:

LAS VEGAS DEVELOPMENT FUND, LLC, A Nevada Limited Liability Company

Name: Title:

7

11/14/17

PLEASE SEE ATTACHED
PLEASE SEE ATTACHED
CAUFORNIA ALL-PURPOSE
CAUFORNIA ALL-PURPOSE
ACKNOWLEDGEMENT FORM

CALIFORNIA ALL-PURPOSE ACKNOWLEDGMENT

\$275055555555555555555555555555555555555
es only the identity of the individual who signed the document accuracy, or validity of that document.
Here Insert Name and Title of the Officer
reary PIAZZA 11
dame(s) of Signer(s)
e to be the personial whose name(s) is/are subscribed he/she/they executed the same in his/ber/their sture(s) on the instrument the personial, or the entity a instrument.
I certify under PENALTY OF PERJURY under the laws of the State of California that the foregoing paragraph is true and correct. WITNESS my hand and official seal.
Signature In ARuss ((/
Signature of Notary Public
ONAL leter alteration of the document or form to an unintended document.
next To loss Agreement
Number of Pages: 4
Signer's Name:
□ Corporate Officer – Title(s):
□ Partner – □ Limited □ General □ Individual □ Attorney in Fact
☐ Trustee ☐, Guardian of Conservator
☐ Other:
Signer is Representing:

©2017 National Notary Association

EXHIBIT 5

SECOND AMENDMENT TO LOAN AGREEMENT

This SECOND AMENDMENT TO LOAN AGREEMENT (this "Second Amendment") is entered into and effective as of February 28, 2018 (the "Second Amendment Effective Date") by and between Las Vegas Development Fund, LLC, a Nevada Limited Liability Company ("Lender") and Front Sight Management, LLC, a Nevada Limited Liability Company, ("Borrower"). Lender and Borrower and their respective permitted successors and assigns are sometimes referred to in this Second Amendment individually as a "Party" and collectively as the "Parties".

RECITALS

- A. Lender and Borrower entered into that certain Construction Loan Agreement dated as of October 4, 2016 (the "Original Loan Agreement"). The Original Loan Agreement as amended by this First Amendment is referred to herein as the "Agreement". Pursuant to the Original Loan Agreement, Borrower executed a Promissory Note dated October 6, 2016 (the "Original Note") and a Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated October 6, 2016 (the "Deed of Trust"). Initially capitalized terms not defined herein shall have the respective meanings assigned to such terms in the Original Loan Agreement. The Original Loan Agreement was amended by a First Amendment to Loan Agreement effective as of July 1, 2017 (the "First Amendment") to further extend the date for obtaining the Senior Financing.
- B. Borrower has represented to Lender that further extending the date for obtaining the Senior Debt will benefit the Project by reducing borrowing costs by delaying the Senior Debt until it is strictly necessary to allow construction to proceed at the fastest feasible pace. Borrower has further represented to Lender that construction is currently proceeding at the most expedited pace reasonably possible and that Borrower has received preliminary pricing terms from two lenders for the Senior Debt ("Senior Debt Term Sheets"). The Parties desire to further amend the Original Loan Agreement, as modified by the First Amendment, to modify the rights and obligations of the Parties as further set forth below.

AGREEMENT

NOW THEREFORE, in consideration of the foregoing Recitals, which are hereby incorporated into the operative provisions of this First Amendment by this reference, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties further agree as follows:

- 1. DATE TO OBTAIN SENIOR DEBT. The date of December 31, 2016, in the last sentence of the definition of Senior Debt in the Original Loan Agreement, which was the outside date for Borrower to obtain such Senior Debt, and which date was extended in the First Amendment, is hereby amended to June 30, 2018. Concurrently with the execution of this Second Extension, Borrower shall provide to Lender copies of term sheets, emails and other materials related to the Senior Debt Term Sheets and shall periodically, but no less than monthly, update the same.
- 2. AGREEMENT RATIFIED. Except as specifically amended or modified herein, each and every term, covenant, and condition of the Original Loan Agreement, Note and Deed of Trust as amended is hereby ratified and shall remain in full force and effect. Each and every reference to the "Agreement" in

the Original Loan Agreement (including, without limitation, the attachments thereto) shall be deemed to refer to the Original Loan Agreement as amended by the First Amendment and this Second Amendment.

- 3. GOVERNING LAW. This instrument shall be interpreted and construed in accordance with the substantive laws of the State of Nevada, excluding choice of law principles.
- 4. <u>BINDING AGREEMENT</u>. This Second Amendment shall be binding upon and inure to the benefit of the Parties hereto and their respective successors and permitted assigns.
- 5. <u>COUNTERPARTS</u>. This Second Amendment may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together shall constitute one and the same document.

IN WITNESS WHEREOF, Lender and Borrower have signed this Second Amendment as of the Effective Date.

BORROWER:

FRONT SIGHT MANAGEMENT, LLC, A Nevada Limited Liability Company

Name: Ignatius Piazza

Title: Manager

LENDER:

LAS VEGAS DEVELOPMENT FUND, LLC, A Nevada Limited Liability Company

Бу. ___

Name: Robert W. Dzigola

Title: President & CEO

EXHIBIT 6

APN(s) 045-481-05 and 045-481-06

RECORDING REQUESTED BY and RETURN TO:

Kathryn Holbert, Esq. NV Bar #10084 FARMER CASE & FEDOR 2190 E. Pebble Rd., #205 Las Vegas, NV 89123

DOC #905512

Official Records Nye County NV Deborah Beatty - Recorder 01/18/2019 10:51:43 AM Requested By: E-DOCS SOLUTIONS L Recorded By: MJ RPTT:\$0 Recording Fee: \$285.00 Non Conformity Fee: \$ Page 1 of 5

NOTICE OF BREACH, DEFAULT and ELECTION TO SELL UNDER DEED OF TRUST IMPORTANT NOTICE

NOTICE IS HEREBY GIVEN that: Kathryn Holbert, Esq., is the duly appointed substitute Trustee under that certain Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated October 6, 2016 and recorded on October 13, 2016, as Document No. 860867 of official records in the Office of the Recorder of Nye County, Nevada; ("Deed of Trust"), which was executed by FRONT SIGHT MANAGEMENT, LLC, a Nevada limited liability company, Grantor, as Trustor, to secure certain obligations in favor of LAS VEGAS DEVELOPMENT FUND, LLC, a Nevada limited liability company, together with that certain First Amendment to Construction Deed of Trust, Security Agreement and Fixture Filing dated July 1, 2017 and recorded on January 12, 2018, as Document No. 886510, and any modifications/amendments thereto of the Official Records in the Office of the Recorder of Nye County, State of Nevada ("Deeds of Trust").

Such DEED OF TRUST secures an Amended and Restated Promissory Note for the sum of up to \$50,000,000.00 as well as other material obligations. A breach of the obligations which are secured by such Amended and Restated Promissory Note has occurred and FRONT SIGHT MANAGEMENT, LLC is in default under the terms of the Deeds of Trust as set forth below:

The total amount due is \$345,787.24 which is itemized as \$32,833.33 current interest; \$158,395.80 past due interest; \$138,655.62 legal/attorney fees and costs; and \$15,902.49 in late fees. Additionally, FRONT SIGHT MANAGEMENT, LLC has default regarding various material non-monetary obligations which are set forth in and secured by the Deeds of Trust, including:

- a. Improper use of loan proceeds.
- b. Failure to provide government approved plans for construction.
- c. Material delays in construction.
- d. Material changes to the costs, scope and timing of the construction.
- e. Refusal to comply regarding securing senior debt.
- f. Failure to provide monthly project costs.
- g. Failure to notify lender of the occurrence of events of default.
- Refusal to allow inspection of books and records.
- i. Refusal to allow site inspection by Lender and its representatives.
- j. Failure to provide EB-5 documentation.

NOTICE OF BREACH, DEFAULT and ELECTION TO SELL UNDER DEED OF TRUST
Page 1 of 2

To cure the Default and Reinstate your loan, you must pay all amounts then due at the time of reinstatement, including any additional unpaid amounts that you are obligated to pay by the terms of the Note and the Deed of Trust, such as, but not limited to, advances, taxes, hazard insurance and obligations secured by prior encumbrances, plus Trustee's and/or Attorney's Fees and Costs and Expenses incurred in enforcing the obligation AND cure the above itemized performance obligations.

Pursuant to NRS 104.9604(1)(b) the sale may, at the election of the beneficiary, include personal property.

NOTICE

You may have the right to cure the defaults set forth herein and reinstate the obligations secured by the Deeds of Trust described above. NRS Section 107.080 permits certain defaults to be reinstated without requiring payment of that portion of principal and interest which would not be due had no default occurred (acceleration of principal). Where reinstatement is possible, if the default is not cured within 35 days following the recording and mailing of this Notice, the right of reinstatement shall terminate and the property thereafter may be sold.

To find out the amount you must pay and the other obligations you must fulfill, or to seek to make arrangements to stop the foreclosure, or if your property is in foreclosure for any other reason, contact LAS VEGAS DEVELOPMENT FUND, LLC, c/o Kathryn Holbert, Esq. Farmer Case & Fedor, Las Vegas, NV 89123, 702-579-3900.

That by reason thereof, the present beneficiary under such Deeds of Trust has executed and delivered to said Trustee a written Declaration of Default and Demand for Sale, and has delivered to said Trustee such Deeds of Trust and all documents evidencing obligations secured thereby and has declared and does hereby declare all sums and obligations set forth above which are secured thereby immediately due and has elected to cause the property to be sold to satisfy the obligations secured thereby.

AFFIDAVIT OF AUTHORITY IS ATTACHED HERETO

Dated

Kathryn Holbert, Esq. Successor Trustee

A notary public or other officer completing individual who signed the document, to wh		,	
truthfulness, accuracy, or validity of that do			
STATE OF NEVAOA			349
On January 17, 2019 before	orama 47 a A	Ran-1	a Notary Public
Personally appeared KATHRYN HULBER	wr who prov	ved to me on the basis	, a riotary rubic,
evidence to be the person(s) whose name(s):	is/are subscribed to	the within instrumer	nt and acknowledged
to me that he/she/they executed the same in l			_
his/her/their signature(s) on the instrument the			
acted, executed the instrument.	1	AL	
	Notary Pul	blic Alana Ba	nton
NOTICE OF BREACH, DEFAULT:	and ELECTION TO	SELL UNDER DEED O	F TRUST
	Page 2 of 2	*****	~~~~~~~~~~~~

ALANA BARTON Notary Public State of Nevada No. 18-1915-1 My Appt. Exp. January 10, 2022

AFFIDAVIT OF AUTHORITY TO EXERCISE THE POWER OF SALE

NRS § 107.080(2)(c)

STATE of CALIFORNIA)		
COUNTY of SAN DIEGO))ss.	

The affiant, ROBERT W. DZIUBLA, being first duly sworn upon oath, based on my direct, personal knowledge, or pursuant to personal knowledge that I acquired by a review of the business records, which meet the standards set forth forth in NRS §51.135, of the beneficiary and/or the servicer of the obligation or debt secured by that certain Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated October 6, 2016 and recorded on October 13, 2016, as Document No. 860867 of official records in the Office of the Recorder of Nye County, Nevada; ("Deed of Trust"), which was executed by FRONT SIGHT MANAGEMENT, LLC, a Nevada limited liability company, Grantor, as Trustor, to secure certain obligations in favor of LAS VEGAS DEVELOPMENT FUND, LLC, a Nevada limited liability company, together with that certain First Amendment to Construction Deed of Trust, Security Agreement and Fixture Filing dated July 1, 2017 and recorded on January 12, 2018, as Document No. 886510, and any modifications/amendments thereto of the Official Records in the Office of the Recorder of Nye County, State of Nevada ("Deeds of Trust").

I further attest, under penalty of perjury, that I am the authorized representative of the beneficiary under such Deeds of Trust, which are described in the NOTICE OF BREACH, DEFAULT and ELECTION TO SELL UNDER DEED OF TRUST to which this affidavit is attached.

I further attest, under penalty of perjury, to the following information, as required by NRS § 107.080(2)(c):

1. The full name and business address of the current trustee is:

Kathryn Holbert, Esq. NV Bar No. 10084 Farmer Case & Fedor 2190 E. Pebble Rd., Suite #205 Las Vegas, NV 89123 702-579-3900

2. The full name and business address of the current holder of the Promissory Note which is secured by the Deeds of Trust and the current beneficiary of record of the Deeds of Trust is:

Las Vegas Development Fund, LLC 916 Southwood Blvd., Suite IG Post Office Box 3003 Incline Village, NV 89450

AFFIDAVIT OF AUTHORIZATION
Page 1 of 3

3. The full name and business address of the current servicer of the obligation or debt which is secure by the Deeds of Trust is:

NES Financial Corp. 50 W. San Fernando St., Suite 300 San Jose, CA 95113

- 4. The beneficiary is in actual possession of the Promissory Note which is secured by the Deeds of Trust and is entitled to enforce the debt and/or other obligations which are secured by the Deed of Trust.
- 5. The beneficiary and/or the servicer of the obligations and/or debt which are secured by the Deed of Trust has sent to the obligator/borrower of the obligation and/or debt which are secured by the Deed of Trust a written statement of:
- a. The amount of payment required to make good the monetary deficiency in performance or payment, avoid the exercise of the power of sale and reinstate the terms and conditions of the underlying obligation or debt as existing before the deficiency and/or defaults occurred, as of the date of the statements;
 - b. The amount in default;
 - The principal amount of the obligation or debt secured by the Deed of Trust;
 - d. The amount of accrued interest and late charges,
 - e. A good faith estimate of all fees imposed in connection with the exercise of the power of sale; and
 - f. Contact information for obtaining the most current amounts due and the local or toll free number as required by NRS 107.080(2)(c)(4).
- 6. A local or toll free telephone number that the obligor or borrower of the obligation or debt may call to receive the most current amount due and other items required to cure the obligors defaults under the Deeds of Trust as well as recitation of the information contained in this affidavit is 702-579-3900.
- 7. The following information regarding the recorded instruments that conveyed the interest of the beneficiary is as follows:

Construction Deed of Trust, Security Agreement, Assignment of Leases and Rents and Fixture Filing dated October 6, 2016 and recorded on October 13, 2016, as Document No. 860867 of official records in the Office of the Recorder of Nye County, Nevada; ("Deed of Trust"), which was executed by FRONT SIGHT MANAGEMENT, LLC, a Nevada limited liability company, Grantor, as Trustor, to secure certain obligations in favor of LAS VEGAS DEVELOPMENT FUND, LLC, a

AFFIDAVIT OF AUTHORIZATION

Page 2 of 3

Nevada limited liability company, together with that certain First Amendment to Construction Deed of Trust, Security Agreement and Fixture Filing dated July 1, 2017 and recorded on January 12, 2018, as Document No. 886510, and any modifications/amendments thereto of the Official Records in the Office of the Recorder of Nye County, State of Nevada ("Deeds of Trust").

The beneficiary has and does hereby instruct the Successor Trustee to exercise the power of sale

with respect to the property which is set forth as security under the Deeds of Trust. Robert W. Dziuba, President and CEO of beneficiary LAS VEGAS DEVELOPMENT FUND, LLC A notary public or other officer completing this certificate verifies only the identity of the individual who signed the document, to which this certificate is attached, and not the truthfulness, accuracy, or validity of that document. STATE of CALIFORNIA))ss. COUNTY of SAN DIEGO before me, Amanda Personally appeared Robert W. Dzivbla, who proved to me on the basis of satisfactory evidence to be the person(s) whose name(s) is/are subscribed to the within instrument and acknowledged to me that he/she/they executed the same in his/her/their authorized capacity(ies), and that by his/her/their signature(s) on the instrument the person(s), or entity upon behalf of which the person(s) acted, executed the instrument.

AMANDA ANDERBERG COMM. # 2229114
NOTARY PUBLIC-CALIFORNIA
SAN DIEGO COUNTY
NY COMM. EXP. JAM. 21, 2022

AFFIDAVIT OF AUTHORIZATION
Page 3 of 3

EXHIBIT 7

AMENDED AND RESTATED PROMISSORY NOTE

\$50,000,000.00

Date: July 1, 2017 (the "Effective Date")

FOR VALUE RECEIVED, the undersigned, FRONT SIGHT MANAGEMENT LLC, a Nevada limited liability company ("Borrower"), having an address at I Front Sight Road, Pahrump, NV 89061, promises to pay, as hereinafter provided, to the order of LAS VEGAS DEVELOPMENT FUND LLC, a Nevada limited liability company ("Leuder"), having an address at P.O. Box 3003, 916 Southwood Blvd., Suite 1G, Incline Village NV 89450, without set-off, counterclaim or deduction, the sum of Fifty Million and No/100 Dollars (\$50,000,000.00), or so much thereof as may have been advanced to or made available for the benefit of Borrower pursuant to the Loan Agreement (as such term is hereinafter defined) and remains unpaid from time to time (hereinafter called "Principal Balance"), with interest on the Principal Balance, until paid in full, at the rates per annum hereinafter specified in legal tender for the payment of public and private debts in the United States of America, all in accordance with the terms hereinafter set forth. All interest payable hereunder shall be computed on the basis of a 360-day year, but shall be charged for the actual number of days principal is unpaid.

- I. <u>Payment Location</u>. All payments of principal and interest under this Note shall be made in lawful money of the United States of America by wire transfer in immediately available funds to such account as may be designated by Lender to Borrower in writing.
- 2. <u>Capitalized Terms</u>. Unless the context otherwise indicates, capitalized terms not otherwise defined herein shall have the meanings provided for such terms in that certain Construction Loan Agreement of October 6, 2016, as amended by that certain First Amendment to Construction Loan Agreement of even date herewith by and between Borrower and Lender (as the same may be further amended, modified or supplemented from time to time hereinafter collectively referred to as the "Loan Agreement"), and which terms are incorporated by this reference as if fully set forth herein.
- 3. <u>Identification of Note</u>. This Amended and Restated Promissory Note (this "Note") is the Promissory Note referred to in the Loan Agreement and shall amend, restate and replace in its entirety that certain Promissory Note, made as of October 6, 2016, by Borrower, in favor of Lender, for the purpose of clarifying certain terms and conditions intended to be effective from and after the Effective Date. The Loan Agreement governs the terms of the indebtedness of Borrower to Lender evidenced by this Note and such other indebtedness as more particularly set forth in the Security Documents (defined below).
 - 4. Payments. This Note shall be payable by Borrower to Lender as follows:
 - (a) Interest shall accrue commencing upon the date upon which funds are first released to Borrower from the Loan Escrow, and continuing until such time as Borrower repays such funds to Lender, in whole or in part, as and when permitted in accordance with

4835-0286-1843.2

the Loan Agreement. Borrower shall make current payments of interest on the first (1st) day of each calendar month on that portion of the Principal Balance then outstanding calculated at an annual rate of 6% during the Initial Term with respect to all Advances made prior to July 1, 2017 and, with respect to such Advances, if extended, at an annual rate of 7% during the Extension Term; and, with respect to all Advances made after July 1, 2017 at an annual rate of 7% during the Initial Term, and, with respect to such Advances, if extended, at an annual rate of 8% during the Extension Term.

If any payment date is on a weekend or national holiday, payment shall be made on the next business day.

- (b) The entire unpaid Principal Balance and all interest accrued thereon shall be due and payable in full on the Initial Maturity Date, subject, however, to Borrower's election to extend the Initial Maturity Date in accordance with the terms and conditions set forth in Section 1.6 of the Loan Agreement.
- (c) In the event that the maturity date is extended as set forth in Section 4(b) above, all accrued and unpaid interest pursuant to Section 4(a) above shall be paid to Lender on the Initial Maturity Date.
- (d) Following any Event of Default hereunder or under the Loan Agreement, interest shall accrue at the Default Rate together with, as applicable, any Late Charge in accordance with the Loan Agreement.
- No payment of interest or other consideration made or agreed to be made (e) by Borrower pursuant to this Note or any other instrument referring to or securing this Note shall, at any time, be deemed to have been computed at an interest rate in excess of the maximum rate of interest permissible by law, if any. In the event such payments of interest or other consideration provided for in this Note or any other instrument referring to or securing this Note shall result in payment of an effective rate of interest which, for any period of time, is in excess of the limit of the usury law or any other law applicable to the loan evidenced hereby, all sums in excess of those lawfully collectible as interest for the period in question shall, without further agreement or notice between or by any party or parties hereto, be applied to the Principal Balance immediately upon receipt of such monies by Lender with the same force and effect as though Borrower had specifically designated, and Lender had agreed to accept, such extra payments as a principal payment, without premium or penalty. If the Principal Balance has been fully paid, any such excess amount shall be refunded to Borrower. This provision shall control over every other obligation of Borrower hereunder and under any instrument that secures this Note.
- (f) Except as set forth in Section 4(e) above, all payments made hereunder shall be applied to amounts due in accordance with the Loan Agreement.
- 5. <u>Prepayment.</u> The Principal Balance and accrued interest thereon may be prepaid in full or in part only as provided in the Loan Agreement.
- 6. Security. The payment and performance of this Note and other Obligations are secured by the lien of that certain Deed of Trust, Assignment of Leases and Rents, Security

(and

Agreement and Fixture Filing of even date herewith ("Deed of Trust"), by Borrower for the benefit of Lender encumbering certain real and personal property located in Nye County, Nevada, as more specifically described therein (the "Project"). Advances of the sums evidenced by this Note are to be made pursuant to the Loan Agreement.

Each Boirower, co-maker, endorser, surety and guarantor hereby guaranties payment of this Note, and waives demand for payment, presentment for payment, notice of nonpayment, protest, notice of protest, notice of dishonor, notice of intention to accelerate maturity, notice of acceleration of maturity, notice of intent to foreclose on any collateral securing this Note, all other notices as to this Note, diligence in collection as to each and every payment due hereunder, and all other requirements necessary to charge or hold such person or entity to any obligation hereunder, and agrees that without any notice Lender may take additional security herefor or may release any or all security herefor, or alone or together with any present or future owner or owners of all or any part of the Project or by any other security documents, may from time to time extend, renew, or otherwise modify the date or dates or amount or amounts of payment above recited, or Lender may from time to time release any part or parts of the property and interest subject to the Deed of Trust or any other security documents from the Deed of Trust and/or any other security documents, with or without consideration, and that, in any such case, each Borrower, co-maker, endorser, surety and guarantor shall continue to be bound hereby and to be liable to pay the unpaid balance of the indebtedness evidenced hereby, as so additionally secured, extended, renewed or modified, and notwithstanding any such release, and further agrees to indemnify Lender against and hold Lender harmless from and pay all costs and expenses of collection, including court costs and reasonable attorneys' fees (prior to trial, at trial and on appeal) incurred in collecting the indebtedness evidenced hereby, or in exercising or defending, or obtaining the right to exercise, the rights of Lender hereunder, under the Loan Agreement or under any security document, whether suit be brought or not, and in foreclosure, in bankruptcy, insolvency, arrangement, reorganization and other debtor-relief proceedings, in probate, in other court proceedings, or otherwise, whether or not Lender prevails therein, and all costs and expenses incurred by Lender in protecting or preserving the property and interests which are subject to the Deed of Trust and/or any other security documents.

7. <u>Default</u>. Time is of the essence hereof. The occurrence of an Event of Default under the Loan Agreement shall constitute an Event of Default under this Note. Upon the occurrence of an Event of Default, Lender shall have the rights set forth in Section 6.2 of the Loan Agreement.

Lender shall not by any act, delay, omission or otherwise be deemed to have waived any of its rights or remedies, and no waiver of any kind shall be valid unless in writing and signed by Lender. All rights and remedies of Lender under the terms of this Note, under the terms of the Loan Agreement and/or of any other security document, and under any statutes or rules of law shall be cumulative and may be exercised successively or concurrently by Lender. Borrower agrees that Lender shall be entitled to all the rights of a holder in due course of negotiable instruments. Any provision of this Note which may be unenforceable or invalid under any law shall be ineffective to the extent of such unenforceability or invalidity without affecting the enforceability or validity of any other provision hereof.

- Interest Rate Limitation. It is the intent of Borrower and Lender in the execution of this Note and all other instruments securing this Note that the loan evidenced hereby be exempt from the restrictions of the usury laws of the State of Nevada. In the event that, for any reason, it should be determined that the Nevada usury law is applicable to the Loan, Lender and Borrower stipulate and agree that none of the terms and provisions contained herein or in any of the other Security Documents shall ever be construed to create a contract for the use, forbearance or detention of money requiring payment of interest at a rate in excess of the maximum interest rate permitted to be charged by the laws of the State of Nevada. In such event, if any holder of this Note collects monies which are deemed to constitute interest which would otherwise increase the effective interest rate on this Note to a rate in excess of the maximum rate permitted to be charged by the laws of the State of Nevada, all such sums deemed to constitute interest in excess of such maximum rate will, at the option of Lender, be credited to the payment of the sums due hereunder (without penalty or premium to Borrower) or returned to Borrower. Borrower agrees to pay an effective contracted for rate of interest equal to the rate of interest resulting from all interest payable as provided in this Note, plus any fees, costs and expenses which may be deemed interest under Nevada law.
- 9. Applicable Law: Jury Trial. IN ALL RESPECTS, INCLUDING, WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, MATTERS OF CONSTRUCTION, VALIDITY AND PERFORMANCE, THIS NOTE AND THE OBLIGATIONS ARISING HEREUNDER SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEVADA, APPLICABLE TO CONTRACTS MADE AND PERFORMED IN SUCH STATE (WITHOUT REGARD TO PRINCIPLES OF CONFLICT OF LAWS) AND ANY APPLICABLE LAW OF THE UNITED STATES OF AMERICA. TO THE FULLEST EXTENT PERMITTED BY LAW, BORROWER HEREBY UNCONDITIONALLY AND IRREVOCABLY WAIVES ANY CLAIM TO ASSERT THAT THE LAW OF ANY OTHER JURISDICTION GOVERNS THIS NOTE AND THE SECURITY DOCUMENTS, AND THIS NOTE AND THE SECURITY DOCUMENTS SHALL BE GOVERNED BY AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEVADA.

BORROWER HEREBY WAIVES ANY RIGHT TO A TRIAL BY JURY IN ANY ACTION RELATING TO THE LOAN AND/OR THE SECURITY DOCUMENTS. BORROWER, TO THE FULLEST EXTENT PERMITTED BY LAW, HEREBY KNOWINGLY, INTENTIONALLY AND VOLUNTARILY, WITH AND UPON THE ADVICE OF COMPETENT COUNSEL, (A) SUBMITS TO PERSONAL JURISDICTION IN THE STATE OF NEVADA OVER ANY SUIT, ACTION OR PROCEEDING BY ANY PERSON ARISING FROM OR RELATING TO THIS NOTE, (B) AGREES THAT ANY SUCH ACTION, SUIT OR PROCEEDING MAY BE BROUGHT IN ANY STATE OR FEDERAL COURT OF COMPETENT JURISDICTION IN CLARK COUNTY IN THE STATE OF NEVADA, (C) SUBMITS TO THE JURISDICTION AND VENUE OF SUCH COURTS AND WAIVES ANY ARGUMENT THAT VENUE IN SUCH FORUMS IS NOT CONVENIENT, AND (D) AGREES THAT IT WILL NOT BRING ANY ACTION, SUIT OR PROCEEDING IN ANY OTHER FORUM. BORROWER FURTHER CONSENTS AND AGREES TO SERVICE OF ANY SUMMONS, COMPLAINT OR OTHER LEGAL PROCESS IN ANY SUCH SUIT, ACTION OR PROCEEDING BY REGISTERED OR CERTIFIED U.S. MAIL, POSTAGE PREPAID, TO BORROWER AT THE ADDRESSES FOR NOTICES DESCRIBED IN THIS NOTE, AND CONSENTS AND AGREES THAT SUCH SERVICE SHALL CONSTITUTE IN EVERY

RESPECT VALID AND EFFECTIVE SERVICE (BUT NOTHING HEREIN SHALL AFFECT THE VALIDITY OR EFFECTIVENESS OF PROCESS SERVED IN ANY OTHER MANNER PERMITTED BY LAW).

[SIGNATURE PAGE FOLLOWS]

IN WITNESS WHEREOF, Borrower has caused this Note to be duly executed and delivered effective as of the day and year first above set forth.

BORROWER:

FRONT SIGHT MANAGEMENT LLC,

a Nevada limited liability company

Name: Ignatius Piazza

Title: Manager

PLEASE SEE ATTACHED CALIFORNIA ALL-PURPOSE ACKNOWLEDGEMENT FORM

CALIFORNIA ALL-PURPOSE ACKNOWLEDGMENT

CIVIL CODE § 1189

ATEST PARTIES CONTROL OF THE PARTIES		
A notary public or other officer completing this certificate veri to which this certificate is attached, and not the truthfulness.	fies only the identity of the individual who signed the document, accuracy, or validity of that document.	
State of California		
County of Soroma.		
On Nov. 14, 2017 before me,	John A Russell Notapes Public	
personally appeared LANATIUS ANT	hory PiARZA 11	
personally appeared <u>Ignatius Ant</u>	Vame(s) of Signer(s)	
who proved to me on the basis of satisfactory evidence to the within instrument and acknowledged to me that authorized capacity(les), and that by his/per/their signs upon behalf of which the person(s) acted, executed the	ature(s) on the instrument the person(s), or the entity	
JOHN A. RUSSELL Notary Public - California Sonoma County	l certify under PENALTY OF PERJURY under the laws of the State of California that the foregoing paragraph is true and correct.	
Commission # 2171851 My Comm. Expires Nov 13, 2020	WITNESS my hand and official seal.	
	Signature John Alkass (
Place Notary Seal and/or Stamp Above	Signature of Notary Public	
OPTI	IONAL	
	deter alteration of the document or form to an unintended document.	
Description of Attached Document Title or Type of Document: Amended and	Restated Promission, Note	
Document Date: 11/14/2017.	Number of Pages: 6	
Signer(s) Other Than Named Above:		
Capacity(ies) Claimed by Signer(s)	# [*]	
Signer's Name:	Signer's Name:	
□ Corporate Officer → Title(s):		
□ Partner — □ Limited □ General	□ Partner – □ Limited □ General	
□ Individual · □ Attorney in Fact		
☐ Trustee ☐ Guardian of Conservator		
Other:	Other:	
Signer is Representing:	Signer is Representing:	

©2017 National Notary Association

EXHIBIT 8

LOAN AGREEMENT (Construction Line of Credit)

This Loan Agreement (the "Agreement") is made and entered into as of the day of October, 2017, by and between FRONT SIGHT MANAGEMENT, LLC, a Nevada limited liability company ("Borrower"), and TOP RANK BUILDERS INC., a Nevada corporation ("Top Rank"), MORALES CONSTRUCTION, INC., a Nevada corporation ("Morales"), and ALL AMERICAN CONCRETE & MASONRY INC., a Nevada corporation ("All American"; collectively, Top Rank, Morales and All American may be referred to as "Lender"); provided, that TOP RANK BUILDERS INC. is the duly authorized agent and representative of Lender for purposes of this Agreement.

RECITALS

- A. Borrower has negotiated with and obtained from Lender a construction line of credit in the maximum principal amount of Thirty-Six Million and No/100 United States Dollars (US\$36,000,000.00) for the purpose of financing the construction of that certain project known as the Front Sight Resort and Vacation Club located in Pahrump, Nevada (the "Project").
- B. Lender has agreed to extend said line of credit to Borrower for such purpose upon and subject to all covenants, terms, and conditions hereinafter provided.
- NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

ARTICLE 1. DEFINITIONS AND INTERPRETATIONS

- 1.1 DEFINITIONS. For purposes of this Agreement, the following terms shall have the following meanings:
- 1.1.1 "Actual Expenses" shall mean the necessary and reasonable expenses incurred for the operation, maintenance and administration of Borrower, but shall exclude depreciation, replacement and obsolescence charges or reserves and further excluding amortization of intangibles or other bookkeeping entries of a similar nature.
- 1.1.2 "Advance" or "Advances" shall mean, individually and collectively, an advance of Loan Proceeds pursuant to this Agreement and the Note.
- 1.1.3 "Agreement" shall mean this Loan Agreement, either as originally executed or as it may from time to time be supplemented, modified, extended, renewed, or amended.
- 1.1.4 "All American" shall having the meaning set forth in the Introductory Paragraph of this Agreement.

- 1.1.5 "Assets" shall mean any interest of Borrower in any kind of property, whether real, personal or mixed, tangible or intangible.
- 1.1.6 "Borrower" shall having the meaning set forth in the Introductory Paragraph of this Agreement.
- 1.1.7 "Business Day" shall mean Monday through Friday, excluding any day of the year on which banks are required or authorized to close in the State of Nevada.
- 1.1.8 "Construction Agreements" shall mean those certain construction agreements hereinafter to be entered into from time to time by and between Borrower and one or more of the Construction Contractors with respect to the construction of the Project.
- 1.1.9 "Construction Contractors" shall mean, collectively, Top Rank, Morales and All American, together with their successors and assigns.
- 1.1.10 "Event of Default" shall mean any of the events or occurrences specified in Article 5 hereof, or as otherwise specified in the Loan Documents.
- 1.1.11 "Gross Revenues" shall mean, for any period, all moneys, fees, rates, receipts, rentals, licensing fees, charges, taxes, assessments, issues and income received for, received by or derived from, the Borrower, the operation of the Borrower or its facilities or any other source whatsoever, including, without limitation, grants, contributions, moneys received from the operation of the Borrower's business or the possession of its properties, insurance proceeds or condemnation awards, income from investments, and all rights to receive the same, whether in the form of accounts, accounts receivable, contract rights or other rights and the proceeds of the same whether now owned or held or hereafter coming into being.
- 1.1.12 "Indemnitee(s)" shall have the meaning set forth in Section 7.3 of this Agreement.
- 1.1.13 "Installment Payments" shall have the meaning set forth in Section 3.6 of this Agreement.
- 1.1.14 "Law" shall mean, collectively, all federal, state, and local laws, rules, regulations, ordinances, and codes.
- 1.1.15 "Lender" shall having the meaning set forth in the Introductory Paragraph of this Agreement.
- 1.1.16 "Loan" shall mean the extension of credit by Lender to Borrower in the form of Advances under this Agreement and disbursement of Loan Proceeds pursuant to the provisions of Article 3 below.

- 1.1.17 "Loan Closing" shall mean the date on which the Loan closes, in accordance with Article 3 of this Agreement.
- 1.1.18 "Loan Closing Costs" shall mean any and all fees and costs incurred by Lender in connection with the negotiation and preparation of the Loan Documents, including attorneys' fees, and closing of the Loan as herein provided.
- 1.1.19 "Loan Documents" shall mean, individually and collectively, this Agreement, the Note, resolutions and certifications of Borrower, and such other documents as Lender may require Borrower to give or cause to be given to or for the benefit of Lender as evidence of the Loan.
- 1.1.20 "Loan Proceeds" shall mean all funds advanced by Lender as the Loan to Borrower under this Agreement.
- 1.1.21 "Maturity Date" shall mean the date which is sixty (60) months from and after the date of the first Advance of the Loan, at which time the entire principal balance of the Loan, plus accrued interest thereon, is and shall be due and payable as provided in this Agreement and the Note, subject to acceleration as provided in the Loan Documents.
 - 1.1.22 "Maximum Amount" shall have the meaning set forth in Section 3.1 hereof.
- 1.1.23 "Morales" shall having the meaning set forth in the Introductory Paragraph of this Agreement.
- 1.1.24 "Net Revenues" shall mean the excess of Gross Revenues over Actual Expenses for such period (excluding therefrom any one-time revenues from the sale of any capital asset).
- 1.1.25 "Note" shall mean the Promissory Note of Borrower, in the maximum principal amount of the Loan, payable to the order of Lender, duly executed by Borrower, as required by Lender to evidence the Loan.
- 1.1.26 "Obligations" means the Advances and any and all existing and future indebtedness and liability of every kind, nature and character, direct or indirect, absolute or contingent, joint or several (including all renewals, extensions and modifications thereof and all attorney's fees and expenses incurred by Lender in connection with the collection or enforcement thereof, including but not limited to, the enforcement of this Agreement under provisions of the U. S. Bankruptcy Code whether by motion for relief from stay or otherwise), of the Borrower to the Lender however and whenever created, arising, evidenced or acquired.

- 1.1.27 "Organizational Documents" shall mean the duly filed, certified and/or executed documents or instruments evidencing or confirming the lawful formation and existence of Borrower.
- 1.1.28 "Permitted Liens" mean any liens, charges and encumbrances on title to the real property constituting the Project as of the Closing Date of the Loan, including, but not limited to, that certain Deed of Trust in favor of Las Vegas Development Fund, LLC, to guarantee Borrower's obligations under that certain Construction Loan Agreement by and between Borrower and Las Vegas Development Fund, LLC, together with such other matters of title thereafter approved by Lender in writing. For the avoidance of doubt, any liens, charges and/or encumbrances securing the Senior Debt shall be considered as "Permitted Liens" as and when such liens, charges and/or encumbrances are granted by Borrower in favor of the provider of the Senior Debt.
- 1.1.1 "Person" means any individual, sole proprietorship, general partnership, limited partnership, limited liability partnership, limited liability company, joint venture, trust, unincorporated organization, association, corporation, government, or any Borrower or political division thereof, or any other entity.
 - 1.1.2 "Project" shall having the meaning set forth in Recital A of this Agreement.
- 1.1.3 "Senior Debt" means the additional loan that will be sought by Borrower, and which Borrower will use it best efforts to obtain, from a traditional financial institution specializing in financing projects such as the Project.
- 1.1.4 "Top Rank" shall having the meaning set forth in the Introductory Paragraph of this Agreement.
- 1.2 USE OF DEFINED TERMS. Any defined terms used in the plural shall include the singular, and the masculine gender shall include the feminine and/or neuter, and such terms shall encompass all members of the relevant class.
- 1.3 SCHEDULES AND EXHIBITS. All schedules and exhibits to this Agreement, either as originally existing or as the same may from time to time be supplemented, modified or amended, are incorporated herein by reference.
- 1.4 REFERENCES. Any reference to this Agreement or any other document shall include such document, both as originally executed, and as it may from time to time be amended, supplemented and modified. References herein to Articles, Sections and Exhibits shall be construed as references to this Agreement unless a different document is named.
- 1.5 OTHER TERMS. The term "document" is used in its broadest sense and encompasses agreements, certificates, opinions, consents, instruments and other written material

of every kind. The terms "including" and "include" shall mean "including (include), without limitation."

ARTICLE 2. REPRESENTATIONS AND WARRANTIES OF BORROWER

Borrower hereby represents and warrants to Lender as of the date of this Agreement, the date(s) the Loan Proceeds are disbursed to Borrower, and each and every date during the term of the Loan, or any portion thereof, as the context admits or requires, that:

- 2.1 BORROWER'S CAPACITY. Borrower is a limited liability company, duly organized and validly existing under the laws of the State of Nevada. The Borrower is duly authorized to enter into this Agreement and to carry out its obligations hereunder and the Borrower has duly authorized the execution and delivery of this Agreement by the individual(s) executing this Agreement on behalf of the Borrower.
- 2.2 VALIDITY OF LOAN DOCUMENTS. The Loan Documents are and shall continue to be in all respects valid and binding upon Borrower according to their terms, subject to all Laws, including, without limitation, equitable principles, insolvency Laws, and other matters applying to creditors generally; provided, however, that the implementation of such Laws do not and will not affect the ultimate realization of the Obligations. The execution and delivery by Borrower, and the performance by Borrower, of all its obligations under the Loan Documents have been duly authorized by all necessary action and do not and will not: (a) require any consent or approval not heretofore obtained of any other entity; or (b) violate any provision of any Laws, or of any order, writ, judgment, injunction, decree, determination or award of any court or of any governmental Borrower; or (c) result in a breach of or constitute a default under, cause or permit the acceleration of any obligation owed under, or require any consent under any indenture or loan or credit agreement or any other agreement, lease, or instrument to which Borrower is a party or by which Borrower or any property of Borrower is bound or affected.
- 2.3 BORROWER NOT IN DEFAULT OR VIOLATION. Borrower is not in default under or in violation of any Laws, order, writ, judgment, injunction, decree, determination or award. Borrower is not in default under any obligation, agreement, instrument, loan, or indenture, whether to Lender or otherwise, or any lease. No event has occurred and is continuing, or would result from the making of any Advance, which constitutes an Event of Default, or would constitute an Event of Default but for the requirement that notice be given or time elapse or both.
- 2.4 NO APPROVALS REQUIRED. No authorization, consent, approval, order, license, exemption from, or filing, registration, or qualification with, any governmental Borrower is or will be required to authorize, or is otherwise required in connection with the execution, delivery and the performance by Borrower of all or any of its obligations under the Loan Documents.

- 2.5 PENDING LITIGATION. There are no actions, suits, or proceedings pending, or to the knowledge of Borrower threatened, against or affecting Borrower or involving the validity or enforceability of any of the Loan Documents, at Law or in equity, or before or by any governmental Borrower, except actions, suits, and proceedings that are fully covered by insurance or which, if adversely determined, would not materially impair the ability of Borrower to perform each and every one of its obligations under and by virtue of the Loan Documents; and Borrower is not in default with respect to any order, writ, injunction, decree, or demand of any court or any governmental Borrower.
- 2.6 VIOLATION OF LAWS. There are no violations or notices of violations of any Law relating to Borrower.
- 2.7 SOLVENCY. Borrower is and shall continue to be able to pay its debts as they mature and the realizable value of its Assets is, and at all times that it may have obligations hereunder shall continue to be, sufficient to satisfy any and all obligations hereunder.
- 2.8 PRINCIPAL PLACE OF BUSINESS. The principal place of business of Borrower is, and will continue to be, as set forth under Section 7.12 of this Agreement. In the event that Borrower hereafter intends to move its principal place of business, it shall first give at least thirty (30) days' prior written notice to Lender of its intention so to move, the date that such move is anticipated, and its new address.
- 2.9 PERMITS. Borrower possesses all licenses, approvals, permits, franchises, patents, copyrights, trademarks, and trade names, or rights thereto, that are necessary for the ownership and use of Borrower's properties, and for the business substantially as now conducted and as presently proposed to be conducted by Borrower, and Borrower is not in material violation of any valid rights of others with respect to any of the foregoing.
- 2.10 FULL DISCLOSURE. All information in the loan application, financial statement, certificate, or other document and all information prepared and delivered by Borrower to Lender in obtaining the Loan is correct and complete in all material respects, and there are no omissions therefrom that result in such information being incomplete, incorrect, or misleading in any material adverse respect as of the date thereof.
- 2.11 USE OF PROCEEDS. The proceeds of each Advance will be used by Borrower solely for the purposes specified in this Agreement.

ARTICLE 3. THE LOAN

3.1 THE LOAN. The total amount available for borrowing by Borrower hereunder on a revolving basis is Thirty-Six Million and No/100 United States Dollars (US\$36,000,000.00) (the "Maximum Amount"). The outstanding principal balance of the Loan will not exceed at any time

the sum of Thirty-Six Million and No/100 Dollars (\$36,000,000.00). Funds borrowed and repaid may be re-borrowed, but subject to the following conditions and limitations (in addition to any other conditions or limitations set forth in this Agreement or in the Note): (a) if, at any time, the aggregate amount advanced by Lender to Borrower exceeds the amount of the Maximum Amount, Borrower shall, no later than five (5) days following written notice thereof by Lender, pay down the Loan by such principal amount that exceeds the amount of the Maximum Amount, and (b) each Advance shall be due and payable on the Maturity Date. Advances may be requested as set forth in the Note.

- 3.2 NOTE. The Loan shall be evidenced by the Note. Each payment under the Loan shall be evidenced and recorded upon Lender's loan records, which recordation shall be prima facie evidence of such payment; provided, however, that the failure by Lender to make any such recordation shall not limit or otherwise affect the obligations of Borrower hereunder or under the Note.
- 3.3 INTEREST; PAYMENTS; PREPAYMENT. Principal and interest under the Note shall be due and payable as provided for in the Note. The Note may be prepaid as provided for in the Note.
- 3.4 PURPOSE OF LOAN. Loan Proceeds shall be used by Borrower exclusively for the purpose or purposes set forth in this Agreement, including, without limitation, for the purposes described in Recital A of this Agreement.
- 3.5 CONDITIONS PRECEDENT TO LOAN. In addition to all other conditions of the effectiveness of this Agreement, the Loan Closing shall occur upon, and the obligations of Lender pursuant to this Agreement shall be subject to, the satisfaction of the following conditions, any or all of which may be waived, in whole or in part, by Lender:
- 3.5.1 Borrower, at its sole expense, shall deliver to Lender, on or before the date of any of the Advances, the following, in form and substance satisfactory to Lender, in Lender's sole opinion and judgment:
 - 3.5.1.1 This Agreement;
 - 3.5.1.2 The Note;
- 3.5.1.3 Such resolutions, authorizations or certifications, as applicable, to borrow and/or related documents from Borrower; and
 - 3.5.1.4 True and correct copies of Borrower's Organizational Documents.